ARSHIYA LIMITED

CIN: L93000MH1981PLC024747 Reg Off: 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018 **T:** +91 22 42305500/02 **F:** +91 22 4230 5555

E-mail: info@arshiyalimited.com I www.arshiyalimited.com

NOTICE TO THE SECURED AND UNSECURED CREDITORS

MEETING OF THE SECURED CREDITORS (INCLUDING DEBENTURE HOLDERS) AND UNSECURED CREDITORS OF ARSHIYA LIMITED CONVEY PURSUANT TO THE ORDER DATED 9TH DAY OF DECEMBER, 2019 PASSED BY THE HON'BLE NATIONAL COMPANY LAW TRIBUNAL, MUMBAI BENCH

MEETING OF SECURED CREDITORS (INCLUDING DEBENTURE HOLDERS):

Day	Tuesday
Date	14 th Day of January, 2020
Time	11.00 AM (1100 Hours)
Venue	302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai- 400018, Maharashtra

MEETING OF UNSECURED CREDITORS:

Day	Tuesday
Date	14 th January, 2020
Time	2:00 P.M. (1400 Hours)
Venue	302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai- 400018, Maharashtra

INDEX:

Sr. No	Contents	Page No.
1.	Notice convening meeting of the Secured (including debenture holders) and Unsecured Creditors of Arshiya Limited (' the Company' or 'Demerged Company') pursuant to the Order dated 09th December, 2019 passed by the Hon'ble National Company Law Tribunal, Mumbai Bench ('the Tribunal')	01
2.	Explanatory Statement under Sections 230 (3), 232 (1) and (2) and 102 of the Companies Act, 2013 read with Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 and other applicable provisions of the Companies Act, 2013	06
3.	ANNEXURE A	15
	Composite Scheme of Arrangement between Arshiya Limited And Arshiya Rail Infrastructure Limited and their respective shareholders and creditors ("Scheme") under Sections 230 to 232 read with Section 66 and other applicable provisions of the Companies Act, 2013, as filed before the Tribunal.	
4.	ANNEXURE B-I	29
	Copy of No Observation letter dated 12th Day of July, 2019 from BSE Limited to Arshiya Limited.	
5.	ANNEXURE B-II	32
	Copy of Observation letter dated 15th Day of July, 2019 from National Stock Exchange of India Limited (NSE) to Arshiya Limited.	
6.	ANNEXURE-C	34
	Complaints Report dated 17th Day of November, 2018 and 14th Day of May, 2019 submitted by Arshiya Limited to BSE Limited and the National Stock Exchange of India Limited, respectively.	
7.	ANNEXURE-D	38
	Copy of Compliance Report of Arshiya Limited.	
8.	ANNEXURE-EI	40
	Copy of report adopted by the Board of Directors of the Demerged/ Transferor Company explaining effect of the Scheme on equity shareholders, key managerial personnel, promoters and non-promoter shareholders.	

Sr.	Contents	Page No.
No		40
9.	ANNEXURE-EII Copy of report adopted by the Board of Directors of the Resulting Company explaining effect of the	42
	Scheme on equity shareholders, key managerial personnel, promoters and non-promoter shareholders.	
10.	ANNEXURE-F	44
	Copy of the Share Entitlement Ratio Report obtained from ZADN & Associates Chartered Accountants,	
	dated May 24, 2018.	
11.	ANNEXURE-G	55
	Statutory Auditor's certificate in respect of the accounting treatment proposed in the Scheme.	
12.	ANNEXURE-H	57
	Copy of Fairness Opinion from Merchant Banker.	
13.	ANNEXURE-I-I	62
	Audited Financial Statements of Accounts for the year ended 31st March, 2019 (Standalone &	
	Consolidated) of Demerged Company.	
14.	ANNEXURE-I-II	154
	Audited Financial Statements of Accounts for the year ended 31st March, 2019 of Resulting Company.	
15.	ANNEXURE-J	193
	Order of the NCLT pursuant to which the meeting is to be convened.	
16.	ANNEXURE-K	205
	The applicable information of Arshiya Rail Infrastructure Limited in the format specified for abridged	
	prospectus as provided in Part D of Schedule VIII of the Securities and Exchange Board of India (Issue	
	of Capital and Disclosure Requirements) Regulation, 2018	
17.	ANNEXURE- L	212
	Dissected Financials (List of Assets and Liabilities of Demerged Company which will be transfer to the	
10	Resulting Company).	
18.	ANNEXURE-M	214
	Pre – Scheme and proposed Post-Scheme shareholding pattern of Arshiya Rail Infrastructure Limited	
10	(the Resulting Company).	222
19.	Form of Proxy	223
20.	Attendance Slip	227
21.	Ballot Paper	
22.	Route Map for the Venue of the Meeting	231

FORM NO. CAA. 2

[Pursuant to Section 230 (3) and rule 6 and 7)] BEFORE THE NATIONAL COMPANY LAW TRIBUNAL, MUMBAI BENCH C.A. (CAA)/2926/MB/2019

IN THE MATTER OF THE COMPANIES ACT, 2013; AND IN THE MATTER OF SCHEME OF ARRANGEMENT PURSUANT TO SECTION 230 TO 232 READ WITH SECTION 66 AND SECTION 52 AND OTHER APPLICABLE PROVISIONS OF THE COMPANIES ACT, 2013 AND IN THE MATTER OF COMPOSITE SCHEME OF ARRANGEMENT BETWEEN ARSHIYA LIMITED AND ARSHIYA RAIL INFRASTRUCTURE LIMITED AND THEIR RESPECTIVE SHAREHOLDERS AND CREDITORS.

Arshiya Limited, a Company incorporated under the provisions of Companies Act, 1956 and having its registered office at 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400018 Maharashtra, India.

--First Applicant Company/ Demerged Company

NOTICE GIVEN PURSUANT TO THE ORDER DATED 09TH DECEMBER, 2019 OF THE NATIONAL COMPANY LAW TRIBUNAL, MUMBAI BENCH

То,

The all the Secured Creditors (Including Debenture Holders) and Unsecured Creditors of Arshiya Limited ("Demerged Company")

TAKE NOTICE that by an order dated 09th December, 2019 ("Order"), the Mumbai Bench of the National Company Law Tribunal has directed a meeting to be held of the of secured creditors (Including Debenture Holders) and unsecured creditors of Arshiya Limited ("Demerged Company") for the purpose of considering, and if thought fit, approving with or without modification, the arrangement proposed to be made between the Arshiya Limited ("Demerged Company") and Arshiya Rail Infrastructure Limited ("Resulting Company") and their respective shareholders and creditors ("Scheme") under Sections 230 to 232 read with Section 66 of the Companies Act, 2013.

In pursuance of the Order and as directed therein, further notice is hereby given that a meeting of the Secured Creditors (Including Debenture Holders) of the said company will be held at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai- 400018, Maharashtra on Tuesday, the 14th January, 2020 at 11:00 A.M at which time and place the said Secured Creditors (Including Debenture Holders) are requested to attend.

TAKE FURTHER NOTICE THAT, in pursuance of the said Order and as directed therein further notice is hereby given that a meeting of the Unsecured Creditors of the said company will be held at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai- 400018, Maharashtra on Tuesday, the 14th January, 2020 at 2:00 P.M at which time and place the said Unsecured Creditors are requested to attend.

At the meeting, following resolution will be considered and if thought fit, be passed, with or without modification(s):

"RESOLVED THAT pursuant to the provisions of Sections 230 to 232 read with Section 66 and other applicable provisions, if any, of the Companies Act, 2013, applicable rules and regulations made thereunder (including any statutory modification(s) or re-enactment(s) thereof for the time being in force), the Securities and Exchange Board of India's Circular No. CFD/DIL3/CIR/2017/21 dated March 10, 2017 (as amended from time to time), the Securities and Exchange Board of India (Listing Obligations and Disclosure Requirements) Regulations, 2015 (as may be amended from time to time), No objection letter / observation letters dated 15th Day of July, 2019 and 12th Day of July, 2019 issued by the National Stock Exchange of India Limited and the BSE Limited, respectively, relevant provisions of the Memorandum and Articles of Association of the Company, and subject to sanction/approval(s) of National Company Law Tribunal ("NCLT"), and such other approvals, sanctions and permissions of other regulatory or government bodies /tribunals or institutions as may be applicable, and subject to such conditions and modification(s) as may be prescribed or imposed by NCLT or by any regulatory or other authorities, while granting such consents, approvals and permissions, which may be agreed to by the Board of Directors of the Company/Demerged Company (hereinafter referred to as the "Board"), the Composite Scheme of Arrangement between Arshiya Limited

("Demerged Company") and Arshiya Rail Infrastructure Limited ("Resulting Company")(presently wholly owned subsidiary company of the Demerged Company) and their respective shareholders and creditors ("Scheme"), which provides for the demerger of the "Domestic Business Undertaking" (as defined in the Scheme) and transfer and vesting thereof into the Resulting Company, a copy of which is enclosed with this notice and placed before this meeting and initialled by the Chairperson or the Alternate Chairperson of the meeting, as the case may be, for the purpose of identification, be and is hereby approved.

RESOLVED FURTHER THAT the Board be and is hereby authorized to do all such acts, deeds, matters and things, as it may, in its absolute discretion deem requisite, desirable, appropriate or necessary to give effect to this resolution and effectively implement the Scheme and to accept such modifications, amendments, limitations and/or conditions, if any, which may be required and/or imposed by NCLT while sanctioning the Scheme or by any authorities under law, including passing of such accounting entries and/or making such adjustments in the books of accounts as considered necessary in giving effect to the Scheme, as the Board may deem fit and proper, and to settle any question, difficulty or doubt that may arise in respect of aforesaid without being required to seek any further consent or approval of the equity shareholders of Company or otherwise to the end and intent that they shall be deemed to have given their approval thereto expressly by the authority of this resolution."

TAKE FURTHER NOTICE that in compliance with the provisions of Section 230-232 of the Companies Act, 2013, the Demerged Company has provided the facility of voting by ballot/polling paper at the venue of the meeting to be held on Tuesday, 14th January, 2020.

NCLT has appointed Mr. Ajay Shankarlal Mittal, Managing Director of the Demerged Company as the Chairperson and failing him, Mr. Ashishkumar Bairagra, Independent Director of the Demerged Company as the Alternate Chairperson of the said Meeting, including for any adjournment or adjournments thereof.

Copies of the Scheme and of the Explanatory Statement, under Sections 230, 232 and 102 of the Companies Act, 2013 read with Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016, along with the enclosures as indicated in the Index, can be obtained free of charge at the registered office of the Demerged Company at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai- 400018, Maharashtra India or at the office of its Authorised Representative, Aabid & Co, Company Secretaries, at 302, 3rd Floor, 22 Business Point, Opposite Andheri Subway, Next to DCB Bank, S. V. Road, Andheri (W), Mumbai - 400 058, Maharashtra, India.

The above mentioned Scheme, if approved by the meeting, will be subject to subsequent approval of the NCLT.

A copy of the Explanatory Statement, the said Composite Scheme of Arrangement and other enclosures including the Form of Proxy and Attendance Slip are enclosed and form part of the notice and can be obtained free of charge from the registered office of the Company.

For Arshiya Limited.

-/Sd Ajay S Mittal Chairman appointed for the meeting

Date: 10th December, 2019

Place: Mumbai

Registered Office: 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018

Notes:

- 1. Only Secured Creditors (Including Debenture Holders) or Unsecured Creditors of the Applicant Company may attend and vote (either in person or by proxy) at the Secured Creditors (Including Debenture Holders) or Unsecured Creditors meeting respectively. The representative of body corporate which is a Secured Creditor (Including Debenture Holders) or Unsecured Creditor of the applicant Company may attend and vote at Secured Creditors (Including Debenture Holders) or Unsecured Creditors Meeting provided a certified true copy of the resolution of the Board of Directors or other governing body of the body corporate is deposited at the registered office of the applicant company not less than 48 (forty eight) hours before the time fixed for the aforesaid meetings.
- 2. All alterations made in the form of proxy should be initialed.
- **3.** The Statement under Sections 230(3), 232(1) & (2) and 102 of the Companies Act, 2013 read with Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 is annexed hereto.
- **4.** The notice, Explanatory Statement together with the accompanying documents, is being sent to the Secured and Unsecured Creditors as on 30th September, 2019 in electronic form as well as the physical copy of the same by the permitted mode.
- **5.** The quorum of the meeting of the Secured Creditors and Unsecured Creditors of the Company shall be Five (5). In case the quorum as noted above for the Meeting is not complete at the scheduled time, then the meeting shall be adjourned by half an hour, and thereafter, the person present at the meeting shall be deemed to constitute the quorum.
- 6. In terms of the directions contained in the Order and in compliance with Rule 20 of the Companies (Management and Administration) Rules 2014, the advertisement will be published in 'Active Times' in English language and translation thereof in 'Mumbai Lakshadeep' in Marathi Language, circulated in the State of Maharashtra indicating the day, date, place and time of the meeting and stating that the copies of the Scheme can be obtained free of charge on all working days (except Saturdays, Sundays and public holidays) during 10:00 AM to 5:00 PM from the registered office of the Company.
- 7. NCLT has appointed, Mr. Mohammed Akram, (ACS 22589 C.P. NO. 9411), Practicing Company Secretaries, as the Scrutinizer to scrutinize the voting through e-voting process, through postal ballot and voting at the venue of the Meeting.
- 8. The Scrutinizer, shall on conclusion of the Meeting, unblock the votes in the presence of at least 2 (two) witnesses not in employment of the Company. Thereafter, the Scrutinizer will submit his report to the Chairman of the Meeting, after completion of scrutiny of votes cast by the creditors of the Company through Ballot. The Scrutinizer's decision on the validity of the votes shall be final. The results of the voting on the resolution(s) set out in the Notice will be announced on or before 15th January,2020. The results, together with the Scrutinizer's report, will be displayed on the notice board of the Company at its registered office, on the website of the Company viz. www.arshiyalimited.com, besides being communicated to BSE Limited and National Stock Exchange of India Limited.

FORM NO. CAA. 2

[Pursuant to Section 230 (3) and rule 6 and 7)] BEFORE THE NATIONAL COMPANY LAW TRIBUNAL, MUMBAI BENCH C.A. (CAA)/ 2926/MB/2019 IN THE MATTER OF THE COMPANIES ACT, 2013; AND IN THE MATTER OF SCHEME OF ARRANGEMENT PURSUANT TO SECTION 230 TO 232 READ WITH SECTION 66 AND SECTION 52 AND OTHER APPLICABLE PROVISIONS OF THE COMPANIES ACT, 2013 AND IN THE MATTER OF COMPOSITE SCHEME OF ARRANGEMENT BETWEEN ARSHIYA LIMITED AND ARSHIYA RAIL INFRASTRUCTURE LIMITED AND THEIR RESPECTIVE SHAREHOLDERS AND CREDITORS

Arshiya Limited, a company incorporated under the provisions of Companies Act, 1956 and having its registered office At 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400018 Maharashtra India

----First Applicant Company/ Demerged Company

EXPLANATORY STATEMENT UNDER SECTIONS 230, 232 AND 102 OF THE COMPANIES ACT, 2013 READ WITH RULE 6 OFTHE COMPANIES (COMPROMISES, ARRANGEMENTS AND AMALGAMATIONS) RULES, 2016 ("EXPLANATORY STATEMENT")

1. Pursuant to the order dated 09th December, 2019 ("Order"), passed by the Hon'ble National Company Law Tribunal, Mumbai Bench (the "NCLT") in Company Application Number 2926 of 2019 filed jointly by **Arshiya Limited** ("Demerged Company") and **Arshiya Rail Infrastructure Limited** ("Resulting Company"), a meeting of the Secured Creditors (Including Debenture holders) and Unsecured Creditors of the Resulting Company, is being convened and is to be held at the registered office of the Resulting Company at 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400018 Maharashtra India at 11:00 A.M. and 2:00 P.M.on Tuesday, 14th January, 2020 respectively, for the purpose of considering and if thought fit, approving, with or without modification(s), the proposed Composite Scheme of Arrangement between the joint applicants (that is, the Demerged Company and the Resulting Company) and their respective shareholders and creditors (hereinafter referred to as the "Scheme") under Sections 230 to 232 read with Section 66 and other applicable provisions of the Companies Act, 2013 (the "Act"). A copy of the Scheme setting out details of parties involved in the proposed Scheme, Appointed Date, Effective Date, Share Entitlement Ratio etc., is enclosed as **Annexure A**.

Capitalised terms used herein but not defined shall have the meaning assigned to them in the Scheme, unless otherwise stated.

- 2. In terms of the Order, quorum for the said meeting shall be 5 members. In case the quorum as noted above for the meeting is not complete at the scheduled time, then the meeting shall be adjourned by half an hour and thereafter, the person(s) present at the meeting shall be deemed to constitute the quorum. For the purposes of quorum, valid proxies will also be considered, if the proxy in the prescribed form, duly signed by the persons entitled to attend and vote at the meeting is filed with the registered office of the Demerged Company at least 48 hours before the meeting. Further, in terms of the Order, the NCLT has appointed Mr. Ajay Shankarlal Mittal, Managing Director of the Demerged Company as the Alternate Chairperson and failing him, Mr. Ashishkumar Bairagra, Independent Director of the Demerged Company as the Alternate Chairperson of the said meeting, including for any adjournment or adjournments thereof.
- 3. This Explanatory Statement is being furnished as required under Sections 230(3), 232(1) & (2) and 102 of the Act read with Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016.

A. BACKGROUND OF THE COMPANIES

4. Arshiya Limited, (herein after called the "The Demerged Company" or "AL") is a Listed Public Company, originally incorporated under the provisions of Companies Act, 1956 on 3rd July, 1981 as a Private Company under the name and style of IID Forgings Private Limited. In the Extra Ordinary General Meeting held on 24th day of August, 1982 consent of the Members of the Company has been accorded to change the name of the Company pursuant to conversion from Private Company to Public Company under the name and style of "IID Forgings Limited". Further in the Extra Ordinary General Meeting held on 4th day of May, 2006

consent of the Members of the Company has been accorded to change the name of the Company under the name and style "Arshiya Technologies International Limited". Further the Company on 28th day of September, 2007 has changed the name to "Arshiya International Limited". Further the Company on 5th day of September, 2013 the Company has changed the name under the name and style of "Arshiya Limited".

- 5. The Registered Office, email and website address of the Demerged Company is 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai 400018 Maharashtra India, Email:info@arshiyalimited.comand Website: www. arshiyalimited.com, respectively.
- 6. The main objects for which the Company was formed as set out in the Memorandum of Association are as following:
 - I. To carry on all or any of the business of designing, manufacturing, developing, improving, hiring, repairing, buying, selling and dealing in forgings and castings of ferrous and non-ferrous materials and in any weight for any industry whatsoever, including chilled and malleable castings, special alloy castings. Gunmetal castings, steel castings, gunmetal, copper, brass and aluminium castings and foundry work.
 - II. To carry on the business of providing integrated supply chain and demand chain management services which interalia includes services of air and ocean freight forwarding, cargo consolidation, project logistics air, sea and surface transportation, shipping, chartering of vessels, warehousing, development and providing hubbing facilities at ports and strategic locations development and providing port facilities, container freight stations, inventory and order management services, storage, drumming, consignments at customs, development and providing import and export documentation services, distribution services, analysis and consulting services to enhance supply chain/demand chain logistics, providing information technology services to logistics and value added activities and for the aforesaid purposes to acquire or develop software solutions, acquire on ownership or lease airplanes and ships, trucks, railway wagons and to build or acquire or take on lease tank farms, warehouses, distriparks, container freight stations or develop infrastructure for above services including knowledge Centre for imparting educational training for persons to attain, enhance professional competency in global logistics services.
 - III. To carry on the business of developing, operating and maintaining special economic zones (SEZs)/free trade and warehouse zones (FTWZs), inland container depots (ICDs), industrial parks, logistic parks, warehouses, infrastructure or infrastructure projects; and to act as contractors, builders, town planners, estate developers, engineers, land developers, land consolidators, land scapers, estate agents, immovable property dealers and other allied and/or ancillary activities; and to acquire, build, operate, buy, sell, lease, sub-lease, long lease, leave and license basis, consolidate, exchange, hire or otherwise; lands, buildings, immovable property of any tenure or any interest in the same, SEZs, FTWZs, ICD, warehouses, houses, flats, bungalows, commercial complexes, shopping malls, multiplexes, food courts and other ancillary and/or allied activities, on the land of the company or other land or any immovable property whether belonging to the company or not; and to pull down, rebuild, enlarge, alter any other conveniences and to deal with and improve, in India or abroad either by company or with joint venture or in partnership or on sub-contract basis or otherwise.
 - IV. To carry on the business of any type of transport/ logistic services including but not limited to setting up of rail infrastructure / network within India and abroad including buy, construct, sale ,operates including movement of containers / goods trains using any rail network; and also to acquire, procure, obtain, trade, lease/license or otherwise: container trains, rakes, wagons, boggies; and to create, develop, sale, Purchase, trade or obtain on lease/license basis railway sidings, rail yards, warehouses required for the business of the company and all allied and ancillary services / products related to that.
 - V. To carry on the business developing and maintaining container freight stations, warehousing infrastructure and services, facilities for customs examination or any other regulator/ regulatory authority or department, EDI, empty container yard for storage of shipping containers and other containers or materials; and to carry out repairs and refurbishment of containers, truck, cargo and material handling equipment; and to provide transportation, warehousing, IT & IT infrastructure and services, cold storage and other allied activities; and to provide services of distribution, reverse logistics, forward logistics, supply chain management, value added services, repair and maintenance, manufacture, transportation, consultancy services; and also to provide system/ software solutions, data analytics, acquire, take on lease, hire or otherwise, distribution centres, trucks and material handling equipment as may be necessary to carry on the aforesaid business in India or abroad.
- 7. During the last 5 (five) years, there has been no change in the name, however the registered office of the Company was changed to Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai 400018 registered office 29th September, 2015. Also, the object clause of the Demerged Company was altered by insertion of addition in object clause on 7th November, 2016 which is as stated in point no. 6 above.
- 8. It may be noted that the Demerged Company is a public limited company and its securities are listed on BSE Limited ("BSE") and National Stock Exchange of India Limited ("NSE").

9. The capital structure of the Demerged Company as on March 31, 2018 is as under:

Particulars	Amount (Rs.)
Equity Share Capital	
Authorized capital	
24,75,00,000 Equity Shares of Rs.2 each	49,50,00,000
1,10,00,000 Optionally Convertible Redeemable Preference Shares of Rs. 10 each	11,00,00,000
Issued, subscribed and fully paid-up	
22,82,16,776 Equity Shares of Rs. 2 each fully paid-up	45,64,33,552

There is no change in the Total Authorised Share Capital of the Company. However on 23rd July, 2019 with the approval of shareholders the Authorised capital of the Company is re-classified from Rs. 60,50,00,000 (Rupees Sixty Crores Fifty Lakhs only) comprising of 24,75,00,000 (Twenty-Four Crore Seventy – Five Lakhs only) equity Shares of Rs. 2 (Rupees Two only) each and 1,10,00,000 (One Crore Ten Lakhs only) Zero Percent Optionally Convertible Redeemable Preference Shares of Rs. 10 (Rupees Ten only) each to Rs. 60,50,00,000 (Rupees Sixty Crores Fifty Lakhs only) divided into 28,75,00,000 (Twenty – Eight Crores Seventy Five Lakhs only) Equity Shares of Rs. 2 (Rupees Two) each and 30,00,000 (Thirty Lakhs) Zero Percent Optionally Convertible Redeemable Preference Shares of Rs. 10 (Rupees Ten only) each and 30,00,000 (Thirty Lakhs) Zero Percent Optionally Convertible Redeemable Preference Shares of Rs. 10 (Rupees Ten only) each.

The capital structure of the Demerged Company as on 6th December, 2019 is as under:

Particulars	Amount (Rs.)
Equity Share Capital	
Authorized capital	
28,75,00,000 Equity Shares of Rs.2 each	57,50,00,000
30,00,000 Optionally Convertible Redeemable Preference Shares of Rs. 10 each	3,00,00,000
Issued, subscribed and fully paid-up	
24,60,44,665 Equity Shares of Rs. 2 each fully paid-up	49,20,89,330

10. The names and addresses of the Promoters of the Demerged Company is provided below:

NAMES	ADDRESS	
Ajay S Mittal	Mittal Bhavan 2, 3rd Floor, 62-A, Peddar Road, Mumbai 400026	
Archana A Mittal	Mittal Bhavan 2, 3rd Floor, 62-A, Peddar Road, Mumbai 400026	

11. The list of directors of the Demerged Company as on 06th December, 2019 along with their names and addresses is provided below:

List (List Of Director				
SR No.	Name Of Director	DIN	Designation	Address	
1	Ashishkumar Bairagra	00049591	Independent Director	44, Sadhana Building, B Road, Churchgate Mumbai 400020 Maharashtra India	
2	Tara Sankar Bhattacharya	00157305	Independent Director	8e, B-1, Harbor Heights, N A Sawant Road, Colaba, Mumbai 400005Maharashtra India	
3	Ajay Shankarlal Mittal	00226355	Managing Director	Mittal Bhavan 2, 3rd Floor, 62-A, Peddar Road, Mumbai 400026 Maharashtra India	
4	Rishabh Pankaj Shah	00694160	Independent Director	Saranam Building, Block No. 7, 7th Road, Prabhat Colony, Santacruz (East), Mumbai 400055 Maharashtra India	
5	Archana Ajay Mittal	00703208	Joint Managing Director	Mittal Bhavan 2, 3rd Floor, 62-A, Peddar Road, Mumbai 400026 Maharashtra India	

- 12. **Arshiya Rail Infrastructure Limited**, (hereinafter referred to as "The resulting Company" or "ARIL") is a Public Limited Company originally incorporated under the provisions of the Companies Act, 1956 on 7th April, 2008.
- 13. The Registered Office, email and website address of the Demerged Company is 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai 400018 Maharashtra India, Email:info@arshiyalimited.com and Website: www. arshiyalimited.comrespectively.

- 14. The main objects for which the resulting Company was formed as set out in the Memorandum of Association are as following:
 - I. To carry on the business of setting up of Rail Infrastructure/Network within India and abroad including operations/ movement of Container/Goods Trains using Indian Railway Network and also to acquire, procure, obtain on lease/licence or otherwise Container Trains, Rakes, Wagons, Boggies and Create, Develop or Obtain on lease/licence basis Railway Sidings, Rail Yards, Warehouses required for the business of the Company.
 - II. To carry on the business predominantly in the Northern Region of India, of container freight stations which, inter-alia includes setting up of bonded warehousing infrastructure and services, facilities for customs examination, EDI, empty container yard for storage of shipping containers, repairs and refurbishment of containers, truck, cargo and material handling equipments, transportation, non-bonded warehousing, IT & ITES infrastructure and services, warehousing, cold storage and other cargo related activities.
 - III. To carry on the business of Transport & Handling of Containers/ Goods/ Network within India and abroad including operations/movement of Container/Cargo/Goods Trains using India Railway Network and also to acquire, procure, obtain on lease/licence or otherwise Container Trains, Rakes, Wagons, Boggies and Create, Develop or obtain on lease/ licence basis Railway Sidings, Rail Yards, Warehouses required for the business of the Company.
- 15. There is no change in the name of the Resulting Company, however the registered office was changed on 27th October, 2015 to 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai 400018 and the objects of the Resulting Company was altered on 28th July, 2016 and thereafter on 9th March, 2017 by insertion of additional object clause. The altered object clause of the Company is as stated in point No. 14 above.
- 16. It may be noted that the Resulting Company is a public limited company and its securities are not listed on any stock exchange.
- 17. The capital structure of the Resulting Company as on the 31st March, 2018 is given below:

Particulars	Amount (Rs.)
Equity Share Capital	
Authorized capital	
45,000,000 Equity Shares of Rs.10 each	450,000,000
Issued, subscribed and fully paid-up	
42,384,417 Equity Shares of Rs. 10 each fully paid-up	423,844,170

After March 31, 2018, there has been no change in the authorized, issued, subscribed and paid up share capital of the Resulting Company.

The proposed capital structure of the Resulting Company post the completion of demerger will be as under:

Particulars	Amount (Rs.)
Equity Share Capital	
Authorized capital	
32,52,50,000 Equity Shares of Rs.2 each	65,05,00,000
Issued, subscribed and fully paid-up	
12,30,22,333 Equity Shares of Rs. 2 each fully paid-up	24,60,44,666

18. The name and address of the Promoter(s) of the Resulting Company is provided below:

NAMES	ADDRESS
Arshiya Limited	302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018

19. The list of Directors of the Resulting Company is provided below:

List Of Director				
SR	Name Of Director	DIN	Designation	Address
No.				
1	Ashishkumar Bairagra	00049591	Director	44, Sadhana Building, B Road, Churchgate Mumbai 400020 Maharashtra India
2	Navnit Jugal Kishore Choudhary	00613576	Director	Flat No 114, Aakash Wing, Indraprasth Complex, Satya Nagar, Borivali West, Mumbai 400092 Maharashtra India
3	Ajay Shankarlal Mittal	00226355	Director	Mittal Bhavan 2, 3rd Floor, 62-A, Peddar Road, Mumbai 400026 Maharashtra India
4	Rishabh Pankaj Shah	00694160	Director	Saranam Building, Block No. 7, 7th Road, Prabhat Colony, Santacruz (East), Mumbai 400055 Maharashtra India

B. RELATIONSHIP SUBSISTING BETWEEN PARTIES TO THE SCHEME

20. The Resulting Company is a wholly owned subsidiary of the Demerged Company.

C. BOARD MEETING OF THE DEMERGED COMPANY AND THE RESULTING COMPANY FOR APPROVAL OF THE SCHEME

21. The Scheme has been unanimously approved by the Board of Directors of the Demerged Company vide resolution passed in the meeting held on 24th May, 2018. Out of total 6 (Six) Directors of the Demerged Company, 5 (Five) directors were present in the Meeting, all of whom voted in favour of the resolution.

The details of voting by Directors on the resolution is as under:

NAME OF THE DIRECTORS	DESIGNATION	VOTED IN FAVOR	VOTED AGAINST	ABSTAINED FROM VOTING
Ashishkumar Bairagra	Independent Director	Yes	NA	NA
Tara Sankar Bhattacharya	Independent Director	Yes	NA	NA
Ajay Shankarlal Mittal	Managing Director	Yes	NA	NA
Mukesh Kacker	Independent Director	Yes	NA	NA
Archana Ajay Mittal	Joint -Managing Director	Yes	NA	NA

22. The Scheme has been unanimously approved by the Board of Directors of the Resulting Company vide resolution passed in the meeting held on May 24, 2018. Out of total 4 (Four) Directors as on that date, 3(three) Directors of the Resulting Company were present in the meeting, and voted in favour of the resolution. The details of voting by directors on the resolution is as under:

NAME OF THE DIRECTORS	DESIGNATION	VOTED IN FAVOR	VOTED AGAINST	ABSTAINED FROM VOTING
Ashishkumar Bairagra	Independent Director	Yes	NA	NA
Navnit Jugal Kishore Choudhary	Director	Yes	NA	NA
Ajay Shankarlal Mittal	Director	Yes	NA	NA

D. DETAILS OF THE SCHEME

- 23. The parties to the Scheme are the Demerged Company and the Resulting Company along with their respective shareholders and creditors.
- 24. The "Appointed Date" of the Scheme is April 1, 2019. "Effective Date" means the last of the dates, if applicable, on which the certified or authenticated copy of the order(s) sanctioning the Scheme passed by the National Company Law Tribunal ("NCLT") of Judicature at Mumbai, is filed with the Registrar of Companies, Mumbai. This date in only drawn to capture references to all transactions undertaken between the Appointed Date and date of filing of the NCLT order with Registrar of Companies ("ROC"). Thus, the same is not to be construed as the effective date for the purpose of Section 232(6) of the Act.

25. Share Exchange Ratio

The Resulting Company shall, without any further application or deed, for every 2 (two) fully paid-up equity shares of the Demerged Company, issue and allot to each member of the Demerged Company whose name appears in the register of members of the Demerged Company as on the Record Date or to his/her heirs, executors, administrators or the successors-in-title, as the case may be, 1 (one) fully paid-up equity share of Rs 2 each, of the Resulting Company ('New Equity Shares').

- 26. Rationale for the Scheme and benefits of the Scheme as perceived by the Board of Directors of the Demerged Company are:
 - To integrate / consolidate its DWA, RAIL, PFT & ICD business into ARIL, which would enable Arshiya to focus solely on FTWZ Business ;
 - To provide more flexibility in terms of creating business synergies in the Resulting Company i.e. ARIL, enable cost savings, rationalizing capital requirements and optimizing utilization of valuable resources which will enhance management focus on the different businesses being housed under separate entities, thereby leading to higher operational efficiency;
 - To enhance value for the shareholders and allow focused strategy on expansion/ operation of both the FTWZ and the Domestic Business independently;
 - To achieve and fulfill their objectives more efficiently and economically and the same is also in the interest of all the stakeholders.
- 27. Salient Features of the Scheme:

The Scheme provides for the transfer of the Demerged Undertaking (as defined in the Scheme) to the Resulting Company. There will be no change in the shareholding pattern of the Demerged Company. Upon the Scheme becoming effective and in consideration of the demerger including the transfer and vesting of the Demerged Undertaking in the Resulting Company, the Resulting Company shall, without any further application or deed, for every 2 (two) fully paid-up equity shares of the Demerged Company, issue and allot to each member of the Demerged Company whose name appears in the register of members of the Demerged Company as on the Record Date or to his/her heirs, executors, administrators or the successors-in-title, as the case may be, subject to the provisions of Clause 7.4 of the Scheme, 1 (one) fully paid-up equity share of Rs 2 each, of the Resulting Company ('New Equity Shares').Further, upon issue of the new equity shares to the Shareholders of the Demerged Company on the record date, all existing equity shares held by the existing shareholders of the Resulting Company (i.e. not including the shareholders of the Demerged Company on the record date), shall stand cancelled, without any further act or deed.

The equity shares to be issued pursuant to the Scheme in the Resulting Company will be listed with BSE Ltd. and National Stock Exchange of India Limited.

- 28. Details of capital or debt restructuring, if any: Basis the share exchange ratio mentioned above, the shareholders of the Demerged Company shall be provided with shares of the Resulting Company in the ratio of 2:1 and the shares held by the Demerged Company in the Resulting Company shall be extinguished.
- 29. Amount due to unsecured creditors as on 30th September, 2019:

CREDITORS	NO. OF CREDITORS	AMOUNT DUE
Secured Creditors	17	9,98,83,92,045
Unsecured Creditors	128	12,12,96,050

E. APPROVALS/ SANCTIONS/NO-OBJECTIONS FROM REGULATORY OR GOVERNMENTAL AUTHORITY RECEIVED OR PENDING

- 30. The BSE Limited was appointed as the designated stock exchange by the Demerged Company for the purpose of co-ordinating with the SEBI, pursuant to the SEBI Circular.
- 31. The Demerged Company received "no adverse observation letter" and "no-objection letter" from BSE and NSE respectively, each dated 12th Day of July, 2019 and 15th Day of July, 2019 respectively, conveying their "no adverse observations" and "no-objection" for filing the Scheme with NCLT.

Copies of the aforesaid "no-objection letter" and "no adverse observations letter" of NSE and BSE, respectively, are enclosed as Annexure B (I) and Annexure B (II).

- 32. As required by the SEBI Circular, the Demerged Company had filed the complaint report with BSE and NSE on. The report indicated that the Demerged Company has not received any complaints. A copy of the complaint report submitted by the Demerged Company to BSE and NSE, dated 17th Day of November, 2018 and 14th Day of May, 2019 is enclosed as Annexure C.
- 33. The Demerged Company has also filed the compliance report with BSE and NSE and is enclosed as Annexure D.
- 34. Effect of Scheme on various Parties

A. Directors and Key Managerial Personnel (KMPs)

None of the Directors and key managerial personnel of the Demerged Company and the Resulting Company or their respective relatives have any interest in the Scheme except to the extent the (i) equity shares held by them in the Demerged Company and the Resulting Company directly or as a nominee; and/or (ii) Director(s) are common director(s) in the two companies; and/or (iii) the Director(s), key managerial personnel and their respective relatives are the director(s), partner(s), member(s) and/or beneficiary(ies) of the companies, firms, association of persons, bodies corporates and/or trust, as the case may be, that hold shares in the Demerged Company/ Resulting Company. There will be no adverse effect of the Scheme on the directors and key managerial personnel of the Demerged Company and Resulting Company.

B. Promoters and Non - Promoter equity shareholders of the Demerged Company and the Resulting Company

In compliance with the provisions of Section 232(2)(c) of the Act, the Board of Directors of the Demerged Company and the Resulting Company, in their meetings held on May 24, 2018, have adopted a report, inter alia, explaining the effect of the Scheme on each class of shareholders, key managerial personnel, directors, promoters and non-promoter shareholders. Copy of the reports adopted by the respective Board of Directors of the Demerged Company and the Resulting Company are enclosed as Annexure EI and EII respectively.

C. Depositors

Neither the Demerged Company nor the Resulting Company have accepted any deposits.

D. Creditors and Debenture – Holders /Debenture Trustees

The debts pertaining to the Demerged Undertaking are taken over by the Resulting Company. Any charge, security interest, lien, statutory lien or statutory charge pertaining to any assets of the Demerged Undertaking shall continue to have effect only on the assets of the Demerged Undertaking in the Resulting Company and shall cease to have effect on the assets of the Remaining Business. The debts pertaining to the Remaining Business (as defined in the Scheme) continue to be the debts or liabilities of the Demerged Company. Accordingly, if any creditor has any charge, security interest, lien, statutory lien or statutory charge on any of the assets or properties of Demerged Undertaking of the Demerged Company, such creditor shall continue to enjoy and hold such charge, lien or security interest upon the properties of Demerged Undertaking in the Resulting Company.

The rights of the creditors of the Demerged Company or the Resulting Company shall not be affected by the scheme. There will be no reduction in their claims on account of the Scheme. The creditors will be paid in the ordinary course of the business as and when their dues are payable. There is no likelihood that the creditors would be prejudiced in any manner as a result of the Scheme being sanctioned. The Demerged Company or the Resulting Company has not issued any debentures and accordingly, do not have any debenture holders.

E. Employees

- I. On and from the Effective Date, all permanent employees relating to the Demerged Undertaking, as were employed by the Demerged Company, immediately before such date, shall become the employees of the Resulting Company with the benefit of continuity of service and without any break or interruption in service. It is clarified that the employees of the Demerged Undertaking, who become employees of the Resulting Company by virtue of this Scheme, shall continue to be governed by the same terms of employment as were applicable to them immediately before the demerger. The Resulting Company undertakes to abide by any agreement/settlement, if any, entered into by the Demerged Company with any of its respective employees thereof. The Resulting Company further agrees that for the purpose of payment of any retrenchment compensation, or any other benefits and incentives, if any, such past services with the Demerged Company shall be taken into account.
- II. It is expressly provided that, on the Effective Date, the provident fund, gratuity fund, superannuation fund created or any other special fund existing for the benefit of the employees of the Demerged Company, in relation to the

Demerged Undertaking shall become the funds of the Resulting Company, for all purposes whatsoever in relation to the administration or operation of such fund(s) or in relation to the obligation to make contributions to the said fund(s) in accordance with the provisions thereof as per the terms provided in the respective trust deeds, if any, to the end and intent that all rights, duties, powers and obligations of the Demerged Company, in relation to the Demerged Undertaking in relation to such fund(s) shall become those of the Resulting Company. These funds shall, subject to the necessary approvals and permissions and at the discretion of the Resulting Company, either be continued as separate funds of the Resulting Company for the benefit of the employees of the Demerged Undertaking or be transferred to and merged with other similar funds of the Resulting Company. It is clarified that the services of the employees of the Demerged Company, in relation to the Demerged Undertaking shall be treated as having been continuous for the purpose of the said fund(s); and

- III. With effect from the date of filing of this Scheme with the NCLT and up to and including the Effective Date, the Demerged Company shall not vary or modify the terms and conditions of employment of any of its employees, except with the prior written consent of the Resulting Company.
- 35. There are no investigations or proceedings, pending against the Company under the Companies Act, 2013.
- 36. None of the directors and KMPs or their respective relatives have any material interest in respect of the Scheme.
- 37. A copy of the Scheme has been filed by the Company with the Registrar of Companies, Mumbai.
- 38. The following documents will be open for obtaining extract from or for making or obtaining copies of or inspection by the Equity Shareholders at the Registered Office of the Demerged Company, during 10:00 a.m. to 5:00 p.m., on all working days (except Saturdays, Sundays and Public Holidays) upto the date of the Meeting:
 - a. Draft of the Scheme which is enclosed as Annexure A.
 - b. Copy of the Share Entitlement Ratio Report obtained from ZADN & Associates Chartered Accountants, dated May 24, 2018 which is enclosed as Annexure F.
 - c. Statutory Auditor's certificate in respect of the accounting treatment proposed in the Scheme which is enclosed as Annexure G.
 - d. The Fairness opinion obtained from the merchant banker which is enclosed herewith as Annexure H.
 - e. The audited Accounts of the Demerged Company and the Resulting Company for the Financial Year ended March 31, 2019 which are enclosed as Annexure I I and Annexure I II respectively.
 - f. Copy of the Order of the National Company Law Tribunal pursuant to which the meeting is to be convened which is enclosed as Annexure J.
- 39. The applicable information of Arshiya Rail Infrastructure Limited in the format specified for abridged prospectus as provided in Part D of Schedule VIII of the Securities and Exchange Board of India (Issue of Capital and Disclosure Requirements) Regulation, 2018 which is enclosed as Annexure-K.
- 40. Dissected Financials (List of Assets and Liabilities of Demerged Company which will be transfer to the Resulting Company) is enclosed herewith as Annexure L.
- 41. The Pre Scheme and proposed Post-Scheme share capital structure of the Resulting Company is enclosed as Annexure M.
- 42. Copy of Annual Report of the Demerged & Resulting Companies for the Financial Year 2016-2017, 2017-2018, 2018-2019.
- 43. Contracts or agreements material to the compromise or arrangement N/A
- 44. Any other information or document as the Board believes is necessary N/A.
- 45. It is confirmed that the copy of the Scheme has been filed with the Registrar of Companies, Mumbai by both the Demerged Company and the Resulting Company.
- 46. In view of the information provided hereinabove, and the documents attached alongwith this notice and explanatory statement,

the requirement of Section 232(2) of the Companies Act, 2013 have been complied with.

47. After the Scheme is approved by the equity shareholders and creditors of the Demerged Company, it will be subject to the approval/sanction by NCLT.

For Arshiya Limited.

Sd/-Ajay S Mittal Chairman appointed for the meeting

Date: 10th December, 2019 Place: Mumbai

Registered office: 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400018

SCHEME OF ARRANGEMENT

BETWEEN

ARSHIYA LIMITED

AND

ARSHIYA RAIL INFRASTRUCTURE LIMITED

UNDER SECTIONS 230 TO 232 READ WITH SECTION 66 AND SECTION 52 AND OTHER

APPLICABLE PROVISONS OF THE COMPANIES ACT, 2013

1. INTRODUCTION

- 1.1. Arshiya Limited (hereinafter referred to as "**Demerged Company**" or "Arshiya"), a company incorporated under the Companies Act, 1956 (CIN L93000MH1981PLC024747) and has its registered office at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai 400 018. Arshiya Limited is listed on the BSE and NSE.
- 1.2. Arshiya Rail Infrastructure Limited (hereinafter referred to as "**Resulting Company" or "ARIL**") is a wholly owned subsidiary of Arshiya Limited. ARIL is a company incorporated under the Companies Act, 1956 (CIN U93000MH2008PLC180907), and has its registered office at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai 400 018;
- 1.3. This Scheme of arrangement ("Scheme"), inter alia, provides for demerger of the Demerged Undertaking (hereinafter defined in Part III) of the Demerged Company and transfer and vesting thereof into the Resulting Company and utilization of securities premium of the Demerged and Resulting Company (hereinafter defined under Part II) including consequential or related matters integrally connected therewith.

PREAMBLE

The management of Arshiya Group intends to reorganize its corporate structure spread across various group companies in order to integrate/ consolidate its operations by housing different businesses into two different entities/ separate verticals.

As part of the aforesaid overall Group reorganization, the management had also filed a separate scheme of amalgamation of two of Arshiya's wholly owned subsidiaries, i.e. Arshiya Industrial & Distribution Hub Limited ('**AIDHL**') and Arshiya Transport and Handling Limited ('**ATHL**') into ARIL ('**merger scheme**') with National Company Law Tribunal and is awaiting its approval. The appointed date for the said merger scheme is October 1, 2015. Accordingly, this scheme of arrangement is conditional upon the aforesaid merger scheme becoming effective first. In case the said merger scheme is withdrawn or ARIL ceases to be a wholly owned subsidiary as on the Record Date (defined hereunder), this Scheme will also be withdrawn.

Further, this Scheme is presented under Sections 230 to 232 read with Sections 66 and 52 and other applicable provisions of the Companies Act, 2013 for demerger of Domestic business (as defined hereinafter) of Arshiya Limited (hereinafter referred to as '**Arshiya**' or the '**Demerged Company**') into Arshiya Rail Infrastructure Limited (hereinafter referred to as 'ARIL' or '**Resulting Company**') and utilization of securities premium of both the entities.

BACKGROUND AND RATIONALE FOR THE SCHEME

a. Arshiya, a flagship company of Arshiya Group, is engaged in the business of developing Free Trade Warehousing Zones ('**FTWZ**') and Domestic Warehousing Areas ('**DWA**') as mentioned below to improve logistics infrastructure in India.

FTWZ Business - FTWZ at Panvel offers over 800,000 Sq. ft. of warehousing space with best in class infrastructure which is suitable for clients across industries. The facility is well connected to the National and State Highways and situated only 24 kms from the country's busiest container port and also close to the proposed International Airport in Navi Mumbai.

Domestic Business - Domestic Business comprises of Domestic Warehousing activities being carried out by Arshiya at land admeasuring 43.42 acres of land situated at Khurja (UP) and investments held in the Resulting Company pertaining to Private Freight Terminal ('**PFT**'), Rail Transportation Services ('**RAIL**') businesses being carried in the Resulting Company and Inland Container Depot ('**ICD**'), DWA business being carried in AIDHL.

b. Brief description of the above mentioned business :

PFT: ARIL currently under a PFT license operates Indian Railways traffic business for various customers catering to bulk goods movement and bagged cargo at sidings specifically earmarked for the same.

RAIL: ARIL holds category-I license to run container trains pan-India and is one of the largest Private Container Train Operator (PCTO's) with a rail fleet of 18 rakes and 3,200 owned containers equipped to handle a wide-range of cargo, with a pan India presence. ARIL's service is completely equipped to provide efficient movement of cargo between terminals, hubs and warehouses.

ARIL's infrastructure consists of rail siding, rail terminal equipped with three loop / blast rail lines and three non-ballast rail lines with a capacity to handle up to 20 rakes per day.

ICD: The Khurja ICD is co-located with a state-of-the-art Rail Terminal and FTWZ. ICD-Khurja is the only private ICD in the country to have exclusive connectivity with 6-lane private rail siding offering regular and prompt rail connectivity through owned rakes to all the major gateway ports that service the northern region of India. ICD Khurja is located strategically with multiple road approaches from the major 4/6 lane highways providing a congestion-free movement of cargo and containers.

The Group intends to reorganize its corporate structure and integrate / consolidate its operations by housing the following businesses into two different entities / separate verticals:

1.3.1. FTWZ business in Arshiya

1.3.2. Domestic business (includes DWA, ICD, Rail and PFT business) in ARIL

- 1.4. Though the businesses of all the Group companies complement each other, the Group believes that in the industry, there are other companies who are doing consolidated business of offering rail infrastructure, cargo/container handling services, providing ICD and Domestic warehousing services. The demand in the market for the entities providing consolidated services is higher than the entities providing individual services. In order to earn higher revenue and to cater to the needs of the market, the management intends to consolidate the rail infrastructure, transport handling business, DWA and ICD business. Hence, the management envisages the transfer of Domestic business of Arshiya into ARIL.
- 1.5. Accordingly, the proposed demerger of the Domestic Business (as defined hereinafter) envisaged in this Scheme ('**the Demerger**') would be in order to integrate / consolidate its DWA, RAIL, PFT & ICD business into ARIL, which would enable Arshiya to focus solely on FTWZ Business. This would provide more flexibility in terms of creating business synergies in the Resulting Company i.e. ARIL, enable cost savings, rationalizing capital requirements and optimizing utilization of valuable resources which will enhance management focus on the different businesses being housed under separate entities, thereby leading to higher operational efficiency.

Further, the Scheme would be in the best interests of the shareholders, creditors and employees of Arshiya and ARIL, respectively as it would result in enhanced value for the shareholders and allow focused strategy on expansion/ operation of both the FTWZ and the Domestic Business independently. Pursuant to this Scheme all the shareholders of Arshiya will get shares in ARIL and there would be no change in the economic interest for any of the shareholders of Arshiya pre and post implementation of this Scheme.

- 1.6. Apropos, the Board of Directors of the Demerged Company and the Resulting Company are of the view that the transfer and vesting of the Demerged Undertaking (as defined hereinafter) of the Demerged Company with the Resulting Company will enable both the companies to achieve and fulfill their objectives more efficiently and economically and the same is also in the interest of all the stakeholders.
- 1.7. The Scheme has been approved by the Board of Directors of the Demerged Company and the Resulting Company.
- 1.8. The transfer of the Demerged Undertaking shall be on a going concern basis.

2. PARTS OF THE SCHEME

- 2.1. This Scheme of Arrangement is divided into the following parts:
 - 2.1.1. PART I deals with the definitions and share capital of the Demerged Company (*defined hereinafter*) and the Resulting Company;
 - 2.1.2. PART II deals with the utilization of securities premium of the Demerged and the Resulting Company and the corresponding accounting treatment in connection therewith;
 - 2.1.3. PART III deals with the demerger of the Demerged Undertaking of Demerged Company and its vesting in the Resulting Company;
 - 2.1.4. PART IV deals with the Remaining Business (defined hereinafter) of the Demerged Company (defined hereinafter);
 - 2.1.5. PART V deals with the consideration for the demerger and accounting treatment for the demerger in the books of the Demerged Company and the Resulting Company consequent to the demerger; and
 - 2.1.6. PART VI deals with general terms and conditions applicable to this Scheme.

PART – I

DEFINITIONS AND SHARE CAPITAL

1. **DEFINITIONS**

In this Scheme, unless repugnant to the subject or context or meaning thereof, the following expressions shall have the meanings as set out herein below:

- 1.1. "Act" means the Companies Act, 2013 and rules made thereunder, including any statutory modifications, re-enactments or amendments thereof for the time being in force as the case may be.
- 1.2. "Appointed Date" means 01 April 2019.
- 1.3. **"Demerged Company**" or "**Arshiya**" means Arshiya Limited, a company incorporated under the Companies Act, 1956, and having its registered office at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai 400 018.
- 1.4. **"Domestic Business Undertaking**" or **"Demerged Undertaking**" or **"Domestic Business**" means Arshiya's entire undertaking, business, activities and operations and domestic business activities pertaining to the DWA Business (including Arshiya's investment in its subsidiary, namely ARIL pertaining to PFT, RAIL businesses being carried ARIL). The term Domestic Business Undertaking shall include the following:
 - 1.4.1. All assets (whether movable or immovable, real or personal, corporeal or incorporeal, present, future or contingent, tangible or intangible) wherever situated and of whatever nature, pertaining thereto through which Arshiya carries on the business, activities and operations relating to the Domestic Business.
 - 1.4.2. All the debts, duties, obligations, loans and liabilities, whether present or future, whether secured or unsecured, of the Demerged Company in relation to the Domestic Business as on the Appointed Date comprising of:
 - 1.4.2.1. All the debts, duties, obligations and liabilities, including contingent liabilities which arise out of the activities or operations of the Demerged Company in relation to the Domestic business and all other debts, liabilities, duties and obligations of the Demerged Company relating to the Demerged Undertaking which may accrue or arise after the appointed date but which related up to the date immediately preceding the Appointed Date;
 - 1.4.2.2. Specific loans and borrowings raised, incurred and / or utilized solely for the activities or operation of the Demerged Company in relation to the Domestic Business;

- 1.4.2.3. Liabilities other than those referred in clauses 1.4.2.1 and 1.4.2.2 above and not directly relatable to the Domestic Business, being the amounts of general or multipurpose borrowings of the Demerged Company as stand in the same proportion which the value of the assets transferred under this clause of Domestic Business bears to the total value of the assets of the Demerged Company immediately before the Appointed Date.
- 1.4.3. Without prejudice to the generality of the above, the Domestic Business shall also include in particular:
 - 1.4.3.1. All assets and properties including land, building, plant and machinery, capital work in progress, equipment, furniture and fixture, vehicles, computers, electrical installations and any other fixed asset in relation to the Domestic Business;
 - 1.4.3.2. All current assets, inventory, stock-in-trade, account receivables, loans and advances, prepaid expenses and other assets in relation to the Domestic Business;
 - 1.4.3.3.Cash and cash equivalents, bank balances and bank accounts relating to the Domestic Business including fixed deposits;
 - 1.4.3.4. Security deposits, advances, earnest monies, balances, advance lease rentals or other payments made to or received from the lessors or suppliers or service providers in relation to the Domestic Business and includes deposits and balances with Government, Semi-Government, local and other authorities and bodies, including all tax balances or balances with any tax authority or other statutory body pertaining to the Domestic Business, customers and other persons earnest moneys and/or security deposits paid or received by Arshiya in connection with the Domestic Business;
 - 1.4.3.5. All agreements (including but not limited to agreements with respect to immoveable properties by way of lease, license and business arrangements), rights, contracts, entitlements, permits, licenses, registrations, insurance policies, approvals, consents, engagements, arrangements, subsidies, concessions, exemptions and all other privileges and benefits of every kind, nature and description whatsoever (including but not limited to benefits of tax relief including under the Income-tax Act, 1961 such as credit for advance tax, taxes deducted at source, etc., unutilized deposits or credits, benefits under the VAT/ Sales Tax law / Goods and Services Tax (GST), VAT / sales tax set off, unutilized deposits or credits, benefits of any unutilized MODVAT / CENVAT / Service tax / GST credits, etc.) relating to the Domestic Business;
 - 1.4.3.6. Investments, held by Arshiya in ARIL;
 - 1.4.3.7. All permanent employees of Arshiya excluding those who are engaged in relation to the Remaining Business;
 - 1.4.3.8. All records, files, documents, reports, papers, computer programs, manuals, data catalogues, quotations, sales and advertising materials, list of present and former customers and suppliers, customer credit information, customer pricing information and other records, whether in physical form or electronic form in connection with or relating to the Domestic Business;
 - 1.4.3.9. All intellectual property (if any) including but not limited to technical know-how, assignment of trademarks and other related rights, title and interest vested thereto rights owned or licensed, records, files, papers, data and documents in the name of Arshiya and in relation to the Domestic Business, Brand name and domain name;
 - 1.4.3.10.All pending litigations or proceedings filed by or against the Demerged Company pertaining to the Domestic Business;
 - 1.4.3.11.All loans and cash credit facilities availed of by the Demerged Company for the purposes of the Domestic Business and other liabilities incurred in connection therewith;

Any question that may arise as to whether a specified asset or liability pertains or does not pertain to the Domestic Business or whether it arises out of the activities or operations of the Domestic Business shall be decided by mutual agreement between Arshiya and ARIL. Further, it is clarified that the Optionally Convertible Redeemable Preference Share – I ('OCRPS-I') issued by Arshiya does not relate to Domestic Business and unless converted into equity shares on or before the Record Date, no additional / proportionate consideration shall be payable to its holders pursuant to this Scheme becoming effective. However, any Corporate Action relating to the Outstanding OCRPS-I shall be kept in abeyance till the last date upto which the option for conversion is exercisable.

The expressions which are used in this Scheme and not defined in this Scheme shall, unless repugnant or contrary to the context or meaning hereof, have the same meaning ascribed to them under the Act and other applicable laws, rules, regulations, bye-laws, as the case may be, or any statutory modification or re-enactment thereof from time to time.

- 1.5. "Effective Date" means the last of the dates, if applicable, on which the certified or authenticated copy of the order(s) sanctioning the Scheme passed by the National Company Law Tribunal ("NCLT") of Judicature at Mumbai, is filed with the Registrar of Companies, Mumbai. This date in only drawn to capture references to all transactions undertaken between the Appointed Date and date of filing of the NCLT order with Registrar of Companies ("ROC"). Thus, the same is not to be construed as the effective date for the purpose of Section 232(6) of the Act.
- 1.6. **"Governmental Authority**" or "**Appropriate Authority**" means any applicable central, state or local government, legislative body, regulatory or administrative authority, agency or commission or any court, tribunal, board, bureau, instrumentality, judicial or arbitral body having jurisdiction over the territory of India.
- 1.7. **"National Company Law Tribunal**" or "**NCLT**" or "**Tribunal**" means the competent authority under the provisions of Sections 230 to 232 and other applicable provisions of the Act and specifically refers to the National Company Law Tribunal, Mumbai Bench in respect of Arshiya and ARIL having their registered offices located in Mumbai, Maharashtra.
- 1.8. **"Record Date**" means the date to be fixed jointly by the Board of Directors of Demerged and Resulting Company for the purposes of determining the shareholders of Arshiya to whom shares would be issued in accordance with Clause 7 of this Scheme.
- 1.1. **"Remaining Business**" or "**Remaining Business of Arshiya**" shall mean all undertakings, businesses, activities and operations including assets and liabilities of Arshiya pertaining to the FTWZ business and excludes the Domestic Business.
- "Resulting Company" or "ARIL" means Arshiya Rail Infrastructure Limited a Company incorporated under the Companies Act, 1956 and having its registered office at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai – 400 018.
- 1.3. **"Scheme**" means this Scheme of Arrangement between Arshiya and ARIL, in its present form and / or with any modifications and amendments thereto made under Clause 22 of this scheme as approved or directed by the Tribunal.

2. DATE OF TAKING EFFECT AND OPERATIVE DATE

This Scheme set out herein in its present form or with any modification(s) and amendment(s) made under Clause 23 of this Scheme duly approved or imposed or directed by the Tribunal shall be effective from the Appointed Date but shall be operative from the Effective Date. Therefore, for all regulatory and tax purposes, the Demerger would have been deemed to be effective from the Appointed Date of this Scheme as per Section 232(6) of the Companies Act 2013. Notwithstanding the above, the accounting treatment to be adopted to give effect to the provisions of the Scheme would be in consonance with Indian Accounting Standards 103 ("IND AS 103") and the mere adoption of such accounting treatment will not in any manner effect the vesting of the Demerged Undertaking from the Appointed Date.

3. SHARE CAPITAL

3.1. The share capital of Arshiya Limited as on 30th April 2018 is as under:

Particulars	Amount (Rs.)
Equity Share Capital	
Authorized capital	
247,500,000 Equity Shares of Rs 2 each	495,000,000
11,000,000 Preference Shares of Rs. 10 each	110,000,000
Issued, subscribed and fully paid-up	
229,716,776 Equity Shares of Rs 2 each	459,433,552
OCRPS - I	
57,64,619 OCRPS-I of Rs 10 each	5,76,46,190

Subsequent to 30th April 2018, there has been no change in the authorized, issued and paid up share capital of Arshiya Limited.

3.2. The share capital of the Resulting Company as on 30th April 2018 is as under:

Particulars	Amount (Rs.)
Equity Share Capital	
Authorized capital	
45,000,000 Equity Shares of Rs.10 each	450,000,000
Issued, subscribed and fully paid-up	
42,384,417 Equity Shares of Rs. 10 each fully paid-up	423,844,170

Subsequent to 30th April 2018, there has been no change in the authorized, issued and paid up share capital of Resulting Company.

PART II

UTILISATION OF SECURITIES PREMIUM

4. UTILISATION OF SECURITIES PREMIUM OF DEMERGED COMPANY AND RESULTING COMPANY

- 4.1. It is proposed to write off the aggregate of the balance in Profit and Loss Account as on Appointed Date and the excess of assets over liabilities as provided under clause 8.1.3 ('Arshiya Aggregate Book Losses), against the securities premium account of Arshiya.
- 4.2. It is also proposed to write off the balance in Profit and Loss Account and any excess of consideration over net assets acquired as provided under clause 8.2.4 as on Appointed Date ('ARIL Book Losses'), against the securities premium account of ARIL.
- 4.3. Consequent upon the re-organization / utilization of securities premium, as mentioned in clause 4.1 and 4.2 above, the Arshiya Aggregate Book Losses and ARIL Book Loss as on Appointed Date shall be reduced to NIL.
- 4.4. The utilization of Securities Premium as aforesaid of Arshiya and ARIL respectively, shall be effected as an integral part of and in terms of this Scheme in accordance with the provisions of Section 230(2)(a) and shall constitute sufficient compliance in terms of Section 52 and Section 66 of the Companies Act, 2013 without carrying out separate compliance thereof. Further, the same does not involve either diminution of liability in respect of unpaid share capital or payment to any shareholder of any paid-up share capital. Accordingly, the order of the Tribunal sanctioning the Scheme shall be deemed to be an order under Section 66 of the Act as well as other applicable provisions of Companies Act 2013 confirming the reduction of share capital pursuant to utilization of the securities premium as aforesaid.
- 4.5. The accounting effect of the above sub-clauses shall be directly given in the balance sheet of Arshiya and ARIL respectively.
- 4.6. The utilization of securities premium as aforesaid of Arshiya and ARIL as envisaged in the Scheme shall not affect or impair in any manner the rights and interests of any of the creditors of Arshiya or ARIL, since Arshiya and ARIL shall, post such reduction, continue to be in a position to honor the dues of their respective creditors. Therefore, Arshiya and ARIL seeks liberty of the NCLT for dispensation of words "and reduced" to be added as suffix to its name, as contemplated in Section 66 (2) and 66 (3) of the Act.

PART – III

DEMERGER OF DOMESTIC BUSINESS UNDERTAKING

5. TRANSFER AND VESTING OF THE DOMESTIC BUSINESS UNDERTAKING

On or after the Effective Date and subject to the provisions of this Scheme and with effect from the Appointed Date, the Domestic Business Undertaking (including all the estate, assets, rights, claims, title, interest and authorities including accretions and appurtenances of the Domestic Business Undertaking) pursuant to the provisions of Sections 230 to 232 and with other applicable provisions of the Act shall stand transferred to and vested in or deemed to be transferred to and vested in the Resulting Company, as a going concern without any further act, deed, matter or thing in the following manner:

5.1. Assets

5.1.1. The whole of the Domestic Business Undertaking shall without any further act, deed, matter or thing, stand transferred to and vested in and / or be deemed to be transferred to and vested in the Resulting Company so as to vest in the Resulting Company all rights, title and interest pertaining to the Domestic Business Undertaking;

- 5.1.2. All assets, investments, right, title or interest acquired by the Demerged Company after the Appointed Date but prior to the Effective Date in relation to the Domestic Business Undertaking shall also, without any further act, instrument or deed, be and stand transferred to and vested in and be deemed to have been transferred to and vested in the Resulting Company upon the Effective date pursuant to the provisions of Sections 230 to 232 and with other applicable provisions of the Act; and
- 5.1.3. All the movable assets of the Domestic Business Undertaking and the assets which are otherwise capable of transfer by physical delivery or endorsement and delivery, including cash in hand, shall be so transferred to the Resulting Company and deemed to have been physically handed over by physical delivery or by endorsement and delivery, as the case may be, to the Resulting Company to the end and intent that the property and benefit therein passes to the Resulting Company with effect from the Appointed Date. Such delivery and transfer shall be made on a date mutually agreed upon between the Demerged Company and the Resulting Company. However such date of delivery shall be such date as may be mutually agreed upon by the Demerged Company and the Resulting Company.
- 5.1.4. Pursuant to the Effective Date and with effect from the Appointed Date, all immovable property (including as per Schedule I to this scheme), whether freehold or leasehold, (including but not limited to land, buildings, offices, factories, sites and any other immovable property, including accretions and appurtenances) relating to the Domestic Business Undertaking of the Demerged Company, and any document of title, rights, interest and easements in relation thereto shall stand transferred to and be vested in the Resulting Company, without any act or deed to be done by the Demerged Company and/or any other Appropriate Authority. The Resulting Company shall be entitled to exercise all rights and privileges and be liable to pay all taxes and charges and fulfil all obligations, in relation to or applicable to such immovable properties. The mutation and/or substitution of the title to the immovable properties shall be made and duly recorded in the name of the Resulting Company by the appropriate governmental authorities and third parties pursuant to the sanction of the Scheme by the Tribunal and upon the Effective Date in accordance with the terms hereof without any further act or deed on part of the Demerged Company and/ or the Resulting Company. It is clarified that the Resulting Company shall be entitled to engage in such correspondence and make such representations as may be necessary for the purposes of the aforesaid mutation and/or substitution.

5.2. Contracts

- 5.2.1. All contracts, deeds, bonds, agreements, schemes, arrangements and other instruments of whatsoever nature in relation to the Domestic Business Undertaking to which the Demerged Company is a party or to the benefit of which the Demerged Company may be eligible, and which are subsisting or have effect immediately before the Effective date, shall continue in full force and effect against or in favour of, as the case may be, the Resulting Company in which the **Domestic Business Undertaking** vests by way of demerger hereunder and may be enforced as fully and effectually as if, instead of the Demerged Company, the Resulting Company had been a party or beneficiary or obligee thereto or thereunder; and
- 5.2.2. Without prejudice to the other provisions of this Scheme and notwithstanding the fact that vesting of the Domestic Business Undertaking occurs by virtue of this Scheme itself, the Resulting Company may, at any time after the Effective date in accordance with the provisions hereof, if so required under any law or otherwise, take such actions and execute such deeds (including deeds of adherence), confirmations or other writings or tripartite arrangements with any party to any contract or arrangement to which the Demerged Company is a party or any writings as may be necessary in order to give formal effect to the provisions of this Scheme. The Resulting Company shall, under the provisions of this Scheme, be deemed to be authorized to execute any such writings on behalf of the Demerged Company in relation to the Domestic Business Undertaking and to carry out or perform all such formalities or compliances referred to above on the part of the Demerged Company to be carried out or performed.

5.3. Liabilities

- 5.3.1. All debts, liabilities, contingent liabilities, duties and obligations of every kind, nature and description of the Domestic Business Undertaking shall also, under the provisions of Sections 230 to 232 and all other applicable provisions, if any, of the Act, without any further act or deed, be transferred to or be deemed to be transferred to the Resulting Company, so as to become from the Appointed Date the debts, liabilities, contingent liabilities, duties and obligations of the Resulting Company and it shall not be necessary to obtain the consent of any third party or other person who is a party to any contract or arrangement by virtue of which such debts, liabilities, contingent liabilities, duties and obligations have arisen in order to give effect to the provisions of this sub-clause;
- 5.3.2. Where any of the loans raised and used, liabilities and obligations incurred, duties and obligations of the Demerged Company as on the Appointed Date deemed to be transferred to the Resulting Company, have been discharged by the

Demerged Company after the Appointed Date and prior to the Effective Date, such discharge shall be deemed to have been for and on account of the Resulting Company; and

5.3.3. All loans raised and used and all liabilities and obligations incurred by the Demerged Company for the operations of the Domestic Business Undertaking with prior approval of the Resulting Company after the Appointed Date and prior to the Effective Date, shall, subject to the terms of this Scheme, be deemed to have been raised, used or incurred for and on behalf of the Resulting Company and to the extent they are outstanding on the Effective Date, shall also without any further act or deed be and stand transferred to and be deemed to be transferred to the Resulting Company and shall become the debts, liabilities, duties and obligations of the Resulting Company which shall meet, discharge and satisfy the same.

5.4. Licenses and Permissions

Any statutory licenses, permissions or approvals or consents held by the Demerged Company required to carry on operations of the Domestic Business Undertaking shall stand vested in or transferred to the Resulting Company without any further act or deed, and shall be appropriately mutated by the statutory authorities concerned therewith in favour of the Resulting Company and the benefit of all statutory and regulatory permissions, environmental approvals and consents, registration or other licenses, and consents shall vest in and become available to the Resulting Company as if they were originally obtained by the Resulting Company. In so far as the various incentives, subsidies, rehabilitation schemes, special status and other benefits or privileges enjoyed, granted by any Governmental Authority or by any other person, or availed of by the Demerged Company relating to the Domestic Business Undertaking, are concerned, the same shall vest with and be available to the Resulting Company on the same terms and conditions as applicable to the Demerged Company, as if the same had been allotted and/or granted and/or allowed to the Resulting Company.

PART – IV

REMAINING BUSINESS

6. REMAINING BUSINESS OF THE DEMERGED COMPANY

- 6.1. The Remaining Business of the Demerged Company and all other assets, liabilities, incentives, rights and obligations pertaining thereto shall continue to be vested in and managed by the Demerged Company in the manner as provided below.
- 6.2. All legal and other proceedings including any insurance claims by or against the Demerged Company under any statute, whether pending on the Appointed Date or which may be instituted in future, whether or not in respect of any matter arising before the Effective date and relating to the Remaining Business of the Demerged Company (including those relating to any property, right, power, liability, obligation or duty, of the Demerged Company in respect of the Remaining Business of the Demerged Company) shall be continued and enforced by or against the Demerged Company.
- 6.3. With effect from the Appointed Date:
 - 6.3.1. The Demerged Company shall be deemed to have been carrying on and to be carrying on all business and activities relating to the Remaining Business of the Demerged Company for and on its own behalf;
 - 6.3.2. The Demerged Company may enter into such contracts as the Demerged Company may deem necessary in respect of the Remaining Business;
 - 6.3.3. All profits accruing to the Demerged Company thereon or losses arising or incurred by it relating to the Remaining Business of the Demerged Company shall, for all purposes, be treated as the profits, or losses, as the case may be, of the Demerged Company;
 - 6.3.4. All assets and properties acquired by the Demerged Company in relation to the Remaining Business on and after the Appointed Date shall belong to and continue to remain vested in the Demerged Company; and
 - 6.3.5. All liabilities (including contingent liabilities) loans, debts (whether secured or unsecured) raised or incurred, duties and obligations of every kind, nature and description whatsoever and howsoever arising or accruing in relation to the Remaining Business shall belong to and continue to remain vested in the Demerged Company.

CONSIDERATION AND ACCOUNTING TREATMENT

7. CONSIDERATION

- 7.1. In consideration of the transfer and vesting of the Demerged Undertaking in accordance with the provisions of this Scheme, the paid-up share capital of the Resulting Company shall be increased in the manner set out in this Clause.
- 7.2. Upon the Scheme becoming effective and in consideration of the demerger including the transfer and vesting of the Demerged Undertaking in the Resulting Company, the Resulting Company shall, without any further application or deed, for every 2 (two) fully paid-up equity shares of the Demerged Company, issue and allot to each member of the Demerged Company whose name appears in the register of members of the Demerged Company as on the Record Date or to his/her heirs, executors, administrators or the successors-in-title, as the case may be, subject to the provisions of Clause 7.4 below, 1 (one) fully paid-up equity share of Rs 2 each, of the Resulting Company ('New Equity Shares').
- 7.3. In case of any member's shareholding in the Demerged Company is such that such member becomes entitled to a fraction of 1 (one) equity share of the Resulting Company, the Resulting Company shall not issue fractional share certificate to such member and shall consolidate such fractions and issue the consolidated shares to a trustee nominated by the Board of Directors of the Demerged Company in that behalf, who shall sell such shares and distribute the net sale proceeds (after deduction of expenses incurred) to such members in proportion to their respective fractional entitlements. During consolidation of fractional shares, if the sum of fractional shares is not a whole integer, the Resulting Company shall issue such additional fractional share to the trustee, such that the total shares so issued shall be rounded off to the next whole integer. The issue of fractional share by the Resulting Company to the Trustee, shall form an integral part of the consideration to be paid under the Scheme and that no separate process as may be applicable under the Applicable Law, to that extent, shall be required to be followed by the Resulting Company.
- 7.4. The New Equity Shares issued in terms of clause 7.2 above, shall be listed and / or admitted to trading on the relevant stock exchange/s in India where the equity shares of the Demerged Company are listed and / or admitted to trading.
- 7.5. Upon the Scheme coming to effect, the shares to be allotted to the members of the Demerged Company by the Resulting Company shall be listed and / or admitted to trading on the relevant stock exchange/s in India where the equity shares of the Arshiya are listed and / or admitted to trading as on effective date. Accordingly, the Resulting Company shall take steps for listing simultaneously on all such stock exchange(s) within a reasonable period of the receipt of the final NCLT order sanctioning the Scheme. The Resulting Company shall make necessary applications with the provisions of Applicable Laws, including, as applicable, the provisions of SEBI Circular No. CIR/CFD/CMD/16/2015 dated November 30, 2015, SEBI Circular No. CFD/DIL3/CIR/2017/105 dated September 21, 2017, SEBI Circular No. CFD/DIL3/CIR/2018/2 dated January 03, 2018 and as amended from time to time. The equity shares allotted pursuant to this Scheme shall remain frozen in the depositaries system till relevant directions in relation to listing / trading are provided by the stock exchanges. The Resulting Company shall apply to Securities and Exchange Board of India through Stock Exchange for seeking relaxation under Section 19(2)(b) of Securities Contract (Regulation) Rules, 1957.
- 7.6. The New Equity Shares of the Resulting Company to be issued to the members of the Demerged Company pursuant to Clause 7.2 above shall be subject to the memorandum and articles of association of the Resulting Company and shall rank pari-passu in all respects, including dividend, with the existing equity shares of the Resulting Company.
- 7.7. The issue and allotment of New Equity Shares by the Resulting Company, to the shareholders of Demerged Company as provided in Clause 7.2 is an integral part of the Scheme and shall be deemed to have been carried out as if the procedure laid down under Section 62(1)(c) of the Act and all other relevant Rules, Regulations and Laws for the time being in force were duly complied with.
- 7.8. Upon the scheme becoming effective, the existing equity shares held by the Demerged Company or its nominee in the Resulting Company shall stand cancelled, extinguished and annulled and from the Effective Date. The cancellation, as aforesaid, which amounts to reduction of share capital of the Resulting Company, shall be effected as an integral part of this Scheme itself in accordance with the provisions of Section 66 of the Act and the order of the Tribunal sanctioning the scheme shall be deemed to be also the order under Section 66 of the Act for the purpose of confirming the reduction. The reduction would not involve either a diminution of liability in respect of unpaid share capital or payment of paid-up share capital. Notwithstanding the reduction as aforesaid, the Resulting Company shall not be required to add "and reduced" as suffix to its name and the Resulting Company shall continue in its existing name.

8. ACCOUNTING TREATMENT

The Demerged Company and the Resulting Company shall account for the Scheme in their respective books / financial statements in accordance with applicable Indian Accounting Standards (IND AS) notified under the Companies (Indian Accounting Standards) Rules, 2015 as amended from time to time including as provided herein below:

8.1. ACCOUNTING TREATMENT IN THE BOOKS OF DEMERGED COMPANY

- 8.1.1. Upon the Scheme becoming effective, the value of assets, liabilities, profits/losses or reserves pertaining to the Demerged Undertaking of the Demerged Company as appearing in the books of accounts of the Demerged Company are to be transferred to the Resulting Company in terms of clause 2 of this Scheme and shall be reduced from the respective book value of assets and liabilities of the Demerged Company.
- 8.1.2. Inter-company balances and investments (including Demerged Company's investment in Resulting Company), if any, between Demerged Company and the Resulting Company shall stand cancelled. Further, any receivables or payables, which pertains to the Demerged Undertaking, arising thereon between the Demerged Company and the Resulting Company, inter-corporate loans or balances pertaining to the Demerged Undertaking as arising between the Demerged Company and the Resulting Company and the Resulting Company or vice-versa shall also stand nullified upon the Scheme becoming effective and the Demerged Company shall pass necessary entries in its books of accounts;
- 8.1.3. The excess of assets over liabilities transferred under clause 8.1.1 and after giving effect to clause 8.1.2 above shall be adjusted against Retained Earnings. In case of deficit, the same shall be credited to capital reserve.
- 8.1.4. Notwithstanding the above, the Board of Directors of the Demerged Company are authorized to account for any of these balances in any manner whatsoever, as may be deemed fit, in accordance with accounting principle generally accepted in India, including the Indian Accounting Standards (IND AS) specified under Section 133 of the Companies Act 2013 read with Companies (Indian Accounting Standards) (Amendment) Rules, 2016.

8.2. ACCOUNTING TREATMENT IN THE BOOKS OF THE RESULTING COMPANY

Upon the Scheme coming into effect and with effect from the Effective Date:

- 8.2.1. The Resulting Company shall record all assets and liabilities of the Demerged Undertaking vested in it pursuant to this Scheme, at the respective values thereof, as appearing in the books of account of the Demerged Company with effect from the Effective Date.
- 8.2.2. Any receivables or payables, which pertains to the Demerged Undertaking, arising thereon between the Demerged Company and the Resulting Company, inter-corporate loans or balances pertaining to the Demerged Undertaking as arising between the Demerged Company and the Resulting Company or vice-versa shall also stand nullified upon the Scheme becoming effective and the Resulting Company shall pass necessary entries in its books of accounts;
- 8.2.3. Upon cancellation of the shares held by the Demerged Company in the Resulting Company, the Resulting Company shall debit to its equity share capital account, the aggregate face value of existing equity shares held by the Demerged Company, which stands cancelled hereof. Further, the Resulting Company shall credit the aggregate face value of the new equity shares issued by it to the members of the Demerged Company pursuant to this Scheme to the share capital account in its books of accounts;
- 8.2.4. The surplus / deficit arising on recording of the assets and liabilities as per clause 8.2.1 and 8.2.2 over the shares cancelled and consideration recorded as per clause 8.2.3 above shall be transferred to Capital Reserve / Goodwill in the balance sheet of the Resulting Company;
- 8.2.5. In case of any differences in accounting policies between the Resulting Company and the Demerged Company, the impact of such differences shall be quantified and adjusted in accordance with the applicable accounting principles;
- 8.2.6. To the extent there are any obligations of the Resulting Company towards the Demerged Undertaking, the obligations in respect thereof shall come to an end and corresponding effect shall be given in the books of account of the Resulting Company.

- 8.2.7. All costs and expenses incurred by the Resulting Company in connection with the Scheme and to put it into operation and any other expenses or charges attributable to the implementation of the Scheme shall be debited to the profit & loss account of the Resulting Company. However, costs pertaining to issue of equity shares shall be directly debited to the reserves of the Resulting Company.
- 8.2.8. Notwithstanding the above, the Board of Directors of the Resulting Company are authorized to account for any of these balances in any manner whatsoever, as may be deemed fit, in accordance with accounting principle generally accepted in India, including the Indian Accounting Standards (IND AS) specified under Section 133 of the Companies Act 2013 read with Companies (Indian Accounting Standards) (Amendment) Rules, 2016.

PART VI

GENERAL TERMS & CONDITIONS

9. RECLASSIFICATION OF AUTHORISED SHARE CAPITAL

- 9.1. Upon the Scheme being effective, the Authorised Share Capital of the Resulting Company existing on the Effective Date, without any further application, act, instrument or deed, shall be reclassified from equity shares of face value of Rs 10/- each into corresponding number of equity shares of face value of Rs. 2/- each.
- 9.2. Consequent upon the reclassification of Authorised Share Capital under Clause 9.1 above and upon the Scheme being effective, Clause V of the Memorandum of Association of the Company (relating to the Authorised Share Capital) shall, without any further act, instrument or deed, be and stand altered, modified and amended pursuant to Sections 13, 14, 61 and 230-232 and other applicable provisions of the Act.
- 9.3. It is hereby clarified that for the purposes of clause 9.2 above, the consent of the shareholders to the Scheme shall be deemed to be sufficient for the purposes of effecting the above reclassification in Authorised Share Capital of the Resulting Company, and no further resolution would be required to be separately passed. Upon the Scheme being effective, the Company shall file necessary form for the reclassification of the Authorised Share Capital with the Registrar of Companies, Mumbai. Further, there would be no requirement for any further payment of stamp duty and/or fee (including registration fee) by the Company for the reclassification to the Authorised Share Capital.

10. TAX ASPECTS

It is intended that this Scheme will be in compliance with the conditions relating to "Demerger" as specified under Section 2(19AA) and Section 72A(4) of the Income-tax Act, 1961 such that:

- 10.1. The transfer of the Domestic Business Undertaking will be on a going concern basis with effect from the Appointed Date.
- 10.2. Subject to clause 10.1 above, if any terms or provisions of the Scheme is/are inconsistent with the provisions of Section 2(19AA) and Section 72A(4) of the Income-tax Act, 1961, the provisions of Section 2(19AA) and Section 72A(4) of the Income-tax Act, 1961 shall prevail and the Scheme shall stand modified to the extent necessary to comply with Section 2(19AA) and Section 72A(4) of the Income-tax Act, 1961 as on the Appointed Date; such modification shall not affect other parts of the Scheme. Upon the Effective Date, the Demerged Company and the Resulting Company are expressly permitted to file / revise their income-tax, service tax, GST, value added tax, withholding tax and other statutory returns, notwithstanding that the period for filing / revising such returns may have lapsed. Each of the Demerged Company and the Resulting Company are expressly permitted to amend tax deduction at source and other statutory certificates and shall have the right to claim refunds, advance tax credits, set offs and adjustments relating to their respective incomes / transactions from the Appointed Date.

11. LISTING AGREEMENT AND SEBI COMPLIANCE

- 11.1. Since the Demerged Company is listed company, this scheme is subject to the compliances by the Demerged Company of all requirements under the Listing Regulations and all statutory directives of the Securities Exchange Board of India ('SEBI') through its circulars insofar as they relate to sanction and implementation of the scheme.
- 11.2. The Demerged Company in compliance with Listing Regulations shall apply for the "Observation Letter" to BSE and NSE where its shares are listed.
- 11.3. The Demerged Company shall also comply with the directives of SEBI contained in circular no CFD/DIL3/CIR/2017/21 dated March 10, 2017 (as amended from time to time) issued by SEBI in terms of Regulations 37 of the Listing Regulations.

12. LISTING OF EQUITY SHARES OF RESULTING COMPANY

12.1. Scheme of Demerger is in conformity with the requirements as laid down in sub-rule 19(7) of Securities Contract (Regulation) Rules, 1957 and in terms of the said sub-rule after allotment of New Equity Shares in Resulting Company, Resulting Company shall comply with relevant provisions under SEBI Regulations, as may be applicable in relation to listing of shares allotted, simultaneously on all stock exchanges where the equity shares of Demerged Company are listed.

Resulting Company shall make application to the SEBI in terms of Rule 19(7) of Securities Contract (Regulation) Rules, 1957 for listing of equity shares at all the stock exchanges where the equity shares of Demerged Company are listed on the Appointed Date without complying with the requirements of Rule 19(2)(b) of Securities Contract (Regulation) Rules, 1957.

ARIL shall enter into such arrangements and give such confirmations and / or undertakings as may be necessary in accordance with the applicable laws or regulations for complying with the formalities of the aforesaid stock exchanges. On such formalities being fulfilled, the stock exchanges shall list and / or admit such New Equity Shares also for the purpose of trading. The New Equity Shares allotted by ARIL, pursuant to this Scheme, shall remain frozen in the depositories system till the listing / trading permission is given by the BSE.

13. APPROVAL OF SCHEME BY PUBLIC SHAREHOLDERS

- 13.1. The Approval to this Scheme of Arrangement shall be obtained from the shareholders of the Demerged Company i.e. Arshiya in accordance with SEBI circular no CFD/DIL3/CIR/2017/21 dated March 10, 2017 (as amended from time to time) issued by SEBI in terms of Regulations 37 of the Listing Regulations.
- 13.2. The Scheme shall be acted upon only if the votes cast by the public shareholders in favor of the proposal are more than the number of votes cast by the public shareholders against it.

14. SECURITY

- 14.1. The transfer and vesting of the Demerged Undertaking as aforesaid shall be subject to the existing securities, charges, mortgage and other encumbrances if any, subsisting over or in respect of the property and assets or any part thereof to the extent that such securities, charges, mortgages, encumbrances are created to secure the liabilities forming part of the Demerged Undertaking. It is agreed by and between the Demerged Company and the Resulting Company, that pursuant to the demerger, necessary steps shall be taken in order to effect the change/ modification of charges, if any, in the records of the Registrar of Companies.
- 14.2. It is clarified that unless otherwise determined by the Board of Directors of the Resulting Company, in so far as the assets comprising the Demerged Undertaking are concerned the security or charge relating to loans or borrowings of the Demerged Company, in relation to the Demerged Undertaking, shall without any further act or deed continue to relate to the said assets only after the Appointed Date and the said assets shall not relate to or be available as security in relation to any other borrowings of the Demerged Company;
 - 14.2.1. Similarly, the security or charge relating to loans or borrowings of the Demerged Company, in relation to the Demerged Undertaking, shall continue to relate to the said assets only after the Appointed Date and shall not relate to or be available as security in relation to any other borrowings of the Resulting Company and vice-versa;
- 14.3. The other assets of the Demerged Company shall not relate to or be available as security in relation to the said borrowings of the Demerged Company, in relation to the Demerged Undertaking; and
- 14.4. The Demerged Company may enter into such alternate arrangements with the lenders pursuant to the release of security as per the provisions mentioned herein.

15. EMPLOYEES

- 15.1. On and from the Effective Date, all permanent employees relating to the Demerged Undertaking, as were employed by the Demerged Company, immediately before such date, shall become the employees of the Resulting Company with the benefit of continuity of service and without any break or interruption in service. It is clarified that the employees of the Demerged Undertaking, who become employees of the Resulting Company by virtue of this Scheme, shall continue to be governed by the same terms of employment as were applicable to them immediately before the demerger. The Resulting Company undertakes to abide by any agreement/settlement, if any, entered into by the Demerged Company with any of its respective employees thereof. The Resulting Company further agrees that for the purpose of payment of any retrenchment compensation, or any other benefits and incentives, if any, such past services with the Demerged Company shall be taken into account.
- 15.2. It is expressly provided that, on the Effective Date, the provident fund, gratuity fund, superannuation fund created or any other special fund existing for the benefit of the employees of the Demerged Company, in relation to the Demerged Undertaking shall become the funds of the Resulting Company, for all purposes whatsoever in relation to the administration or operation of such fund(s) or in relation to the obligation to make contributions to the said fund(s) in accordance with the provisions

thereof as per the terms provided in the respective trust deeds, if any, to the end and intent that all rights, duties, powers and obligations of the Demerged Company, in relation to the Demerged Undertaking in relation to such fund(s) shall become those of the Resulting Company. These funds shall, subject to the necessary approvals and permissions and at the discretion of the Resulting Company, either be continued as separate funds of the Resulting Company for the benefit of the employees of the Demerged Undertaking or be transferred to and merged with other similar funds of the Resulting Company. It is clarified that the services of the employees of the Demerged Company, in relation to the Demerged Undertaking shall be treated as having been continuous for the purpose of the said fund(s); and

15.3. With effect from the date of filing of this Scheme with the NCLT and up to and including the Effective Date, the Demerged Company shall not vary or modify the terms and conditions of employment of any of its employees, except with the prior written consent of the Resulting Company.

16. BUSINESS AND PROPERTY IN TRUST

- 16.1. During the period between the Appointed Date and up to and including the Effective Date:
- 16.2. The Demerged Company shall be deemed to have been carrying on all the business and activities relating to the Demerged Undertaking and shall be deemed to hold and stand possessed of the entire business and undertakings in relation to the Demerged Undertaking for and on account of and in trust, on behalf of the Resulting Company.
- 16.3. All the income or profits accruing or arising to the Demerged Company and all costs, charges, expenses or losses incurred by the Demerged Company, in relation to the Demerged Undertaking shall for all purposes of this demerger be treated as the income, profits, costs, charges, expenses and losses of the Resulting Company, as the case may be.
- 16.4. Any of the rights, powers, authorities, privileges, attached, related or pertaining to the Demerged Undertaking exercised by the Demerged Company shall be deemed to have been exercised by the Demerged Company for and on behalf of, and in trust for and as an agent of the Resulting Company; and
- 16.5. The Demerged Company shall carry on the business pertaining to the Domestic Business Undertaking with reasonable diligence and business prudence and shall not alter or diversify business within the Domestic Business Undertaking nor venture into any new business (except for Remaining Business), nor alienate, charge, mortgage, encumber or otherwise deal with the assets or any part thereof except in the ordinary course of business without the prior written consent of the Resulting Company or pursuant to any pre-existing obligation undertaken prior to the date of acceptance of this Scheme by the respective boards of directors of the Demerged Company and the Resulting Company.
- 16.6. The Demerged Company shall not utilize the profits or income in relation to the Demerged Undertaking for the purpose of declaring or paying any dividend in respect of the period falling on and after the Appointed Date, without the prior written consent of the Resulting Company.
- 16.7. The Resulting Company shall be entitled, pending the sanction of the Scheme, to apply to the Governmental Authorities or other appropriate forums as may be required under any applicable law, for such consents, approvals and sanctions which the Resulting Company may require.

17. LEGAL PROCEEDINGS

17.1. All legal, administrative and other proceedings, of whatsoever nature pending in any court or before any authority, judicial, quasi-judicial or administrative or any adjudicating authority and/or arising after the Appointed Date and relating to the Demerged Undertaking, or its respective properties, assets, debts, liabilities, duties and obligations shall be continued and/ or enforced until the Effective date by or against the Demerged Company; and from the Effective Date, shall be continued and enforced by or against the Resulting Company in the same manner and to the same extent as would or might have been continued and enforced by or against the Demerged Company, had the Scheme not been made. On and from the Effective Date, the Resulting Company shall have the right to initiate, defend, compromise or otherwise deal with any legal proceedings relating to the Demerged Undertaking, in the same manner and to the same extent as would or might have been initiated by the Demerged Company as the case may be, had the Scheme not been made; and if any suit, appeal or other proceedings relating to the Demerged Undertaking, of whatever nature by or against the Demerged Company be pending, the same shall not abate or be discontinued or in any way be prejudicially affected by reason of the demerger of the Demerged Undertaking or by anything contained in this Scheme but the proceedings may be continued, prosecuted and enforced by or against the Resulting Company in the same extent as it would or might have been continued, prosecuted and enforced by or against the Demerged Company as if this Scheme had not been made.

18. SAVING OF CONCLUDED TRANSACTIONS

The transfer of properties and liabilities and the continuance of proceedings by or against the Resulting Company, as envisaged under this Scheme, shall not affect any transaction or proceedings already concluded by the Demerged Company, in relation to the Demerged Undertaking on or after the Appointed Date till the Effective Date, to the end and intent that the Resulting Company accepts and adopts all acts, deeds and things done and executed by the Demerged Company, in relation to the Demerged Undertaking, as done and executed on behalf of itself.

19. VALIDITY OF EXISTING RESOLUTIONS, ADJUSTMENTS ETC.

All resolutions passed by the Demerged Company so far as they relate to or to be done or caused to be done in relation to the Demerged Undertaking, shall be deemed to have authorized any Director of the Resulting Company or such other person(s) as authorized by any two Directors of the Resulting Company to do all acts, deeds, things as may be necessary to give effect to these Resolutions.

20. DECLARATION OF DIVIDEND

For the avoidance of doubt, it is hereby declared that nothing in the Scheme shall prevent the Resulting Company from declaring and paying dividends, whether interim or final, to its equity shareholders.

21. APPLICATION TO THE TRIBUNAL

- 21.1. The Demerged Company shall and the Resulting Company, if required, shall make applications/ petitions to the NCLT for sanction of this Scheme, under Sections 230 to 232 read with Sections 66 and 52 and other applicable provisions of the Act; and
- 21.2. Any dispute arising out of this Scheme shall be subject to the jurisdiction of the NCLT, Mumbai Bench.

22. MODIFICATION OR AMENDMENTS TO THE SCHEME

The Demerged Company and the Resulting Company (acting through their Board of Directors, Committee thereof or any director or any other person authorized by the Board of Directors, Committee thereof to this effect) may assent to any modifications or amendments to this Scheme or to any conditions or limitations that the Tribunal may deem fit to direct or impose or which may otherwise be considered necessary, desirable or appropriate by the respective Demerged Company and Resulting Company, including pursuant to the orders of the NCLT and/or any other authorities as they may deem fit to direct or impose or which may otherwise be considered necessary or desirable for settling any question or doubt or difficulty that may arise for implementing and/or carrying out the Scheme. The Demerged Company and the Resulting Company (acting through their Board of Directors, Committee thereof or any director or any other person authorized by the Board of Directors, Committee thereof to this effect) shall be authorized to take such steps and do all acts, deeds and things as may be necessary, desirable or proper to give effect to this Scheme and to resolve any doubts, difficulties or questions whether by reason of any orders of the Tribunal or of any directions given by any other appropriate authorities or for any reason otherwise arising out of this Scheme and/or any matters concerning or connected herewith.

If any part of the Scheme is held invalid or is ruled illegal by the Tribunal or becomes unenforceable for any reason, whatsoever whether under present or future laws, then it is the intention of the Companies that such part in the opinion of the board of any companies, shall be severable from the remainder of the Scheme and the remaining part of this Scheme shall not be affected thereby, unless the deletion of such part, in the opinion of Board of either of the companies, shall cause this Scheme to become materially adverse to either of the companies in which case companies shall attempt to bring about a modification in this Scheme, which will best preserve the benefits and obligations of this Scheme for companies, including but not limited to such part.

23. EFFECT OF NON-RECEIPT OF APPROVALS

- 23.1. In case the Scheme is not approved by the Tribunal or any of the approvals or conditions enumerated in the Scheme have not been obtained or complied with, or for any other reason, if this Scheme cannot be implemented, then the board of directors of the Demerged Company and the Resulting Company shall mutually waive such conditions as they consider appropriate to give effect, as far as possible, to this Scheme and failing such mutual agreement, the Scheme shall become null and void and in such event no rights or liabilities whatsoever shall accrue to or be incurred by either the Resulting Company or the Demerged Company and each party shall bear their respective costs, charges and expenses in connection with this Scheme unless otherwise mutually agreed upon.
- 23.2. If any part of this Scheme hereof is invalid, held illegal by Tribunal, or unenforceable under any present or future laws, then it is the intention of the parties that such part shall be severable from the remainder of the Scheme, and the Scheme shall not be affected thereby, unless the deletion of such part shall cause this Scheme to become materially adverse to any party, in which case the parties shall attempt to bring about a modification in the Scheme, as will best preserve for the parties the benefits and obligations of the Scheme, including but not limited to such part.

24. COSTS, CHARGES AND EXPENSES

Save and except as provided otherwise, all costs, charges, expenses, taxes including duties, levies in connection with the Scheme and its implementation thereof, and matters incidental thereto, shall be borne by the Demerged Company and the Resulting Company respectively.

Annexure BI

BSE - INTERNAL



July 12, 2019

DCS/AMAL/JR/R37/1530/2019-20

The Company Secretary, Arshiya Limited 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai, Maharashtra, 400018

Dear Sir,

Sub: Observation letter regarding the Draft Scheme of Arrangement by Arshiya Limited

We are in receipt of Draft Scheme of Arrangement by Arshiya Limited filed as required under SEBI Circular No. CFD/DIL3/CIR/2017/21 dated March 10, 2017; SEBI vide its letter dated July 12, 2019 has inter alia given the following comment(s) on the draft scheme of arrangement:

- "Company shall ensure that additional information, if any, submitted by the Company, after filing the scheme with the stock exchange, from the date of receipt of this letter is displayed on the websites of the listed company."
- "Company shall duly comply with various provisions of the Circulars."
- "Company is advised that the observations of SEBI/Stock Exchanges shall be incorporated in the petition to be filed before National Company Law Tribunal (NCLT) and the company is obliged to bring the observations to the notice of NCLT."
- "It is to be noted that the petitions are filed by the company before NCLT after processing and communication of comments/observations on draft scheme by SEBI/stock exchange. Hence, the company is not required to send notice for representation as mandated under section 230(5) of Companies Act, 2013 to SEBI again for its comments / observations / representations."

Accordingly, based on aforesaid comment offered by SEBI, the company is hereby advised:

- To provide additional information, if any, (as stated above) along with various documents to the Exchange for further dissemination on Exchange website.
- To ensure that additional information, if any, (as stated aforesaid) along with various documents are disseminated on their (company) website.
- To duly comply with various provisions of the circulars.

In light of the above, we hereby advise that we have no adverse observations with limited reference to those matters having a bearing on listing/de-listing/continuous listing requirements within the provisions of Listing Agreement, so as to enable the company to file the scheme with Hon'ble NCLT. Further, where applicable in the explanatory statement of the notice to be sent by the company to the shareholders, while seeking approval of the scheme, it shall disclose Information about unlisted companies involved in the format prescribed for abridged prospectus as specified in the circular dated March 10, 2017.

However, the listing of equity shares of Arshiya Rail Infrastructure Limited shall be subject to SEBI granting relaxation under Rule 19(2)(b) of the Securities Contract (Regulation) Rules, 1957 and compliance with the requirements of SEBI circular. No. CFD/DIL3/CIR/2017/21 dated March 10, 2017. Further, Arshiya Rail Infrastructure Limited shall comply with SEBI Act, Rules, Regulations, directions of the SEBI and any other statutory authority and Rules, Byelaws, and Regulations of the Exchange.



BSE Limited (Formerly Bombay Stock Exchange Ltd.) Registered Office : Flog_25, PJ Towers, Dalal Street, Mumbai 400 001 India T; +91 22 22/21 22/4735 [E-corp.comm0besindia.com /www.bseindia.com Corporate Identity Number: 1.67120/H42005PLC155188



The Company shall fulfill the Exchange's criteria for listing the securities of such company and also comply with other applicable statutory requirements. However, the listing of shares of Arshiya Rail Infrastructure Limited is at the discretion of the Exchange. In addition to the above, the listing of Arshiya Rail Infrastructure Limited pursuant to the Scheme of Arrangement shall be subject to SEBI approval and the Company satisfying the following conditions:

- To submit the Information Memorandum containing all the information about Arshiya Rail Infrastructure Limited in line with the disclosure requirements applicable for public issues with BSE, for making the same available to the public through the website of the Exchange. Further, the company is also advised to make the same available to the public through its website.
- To publish an advertisement in the newspapers containing all the information Arshiya Rail Infrastructure Limited in line with the details required as per the aforesaid SEBI circular no. CFD/DIL3/CIR/2017/21 dated March 10, 2017. The advertisement should draw a specific reference to the aforesaid Information Memorandum available on the website of the company as well as BSE.
- To disclose all the material information about Arshiya Rall Infrastructure Limited on a continuous basis so as to make the same public, in addition to the requirements if any, specified in Listing Agreement for disclosures about the subsidiaries.
- 4. The following provisions shall be incorporated in the scheme:
 - i. The shares allotted pursuant to the Scheme shall remain frozen in the depository system till listing/trading permission is given by the designated stock exchange."
 - "There shall be no change in the shareholding pattern of Arshiya Rail Infrastructure Limited between the record date and the listing which may affect the status of this approval."

Further you are also advised to bring the contents of this letter to the notice of your shareholders, all relevant authorities as deemed fit, and also in your application for approval of the scheme of Arrangement.

Kindly note that as required under Regulation 37(3) of SEBI (Listing Obligations and Disclosure Requirements) Regulations, 2015, the validity of this Observation Letter shall be Six Months from the date of this Letter, within which the scheme shall be submitted to the NCLT.

The Exchange reserves its right to withdraw its 'No adverse observation' at any stage if the information submitted to the Exchange is found to be incomplete / incorrect / misleading / false or for any contravention of Rules, Bye-laws and Regulations of the Exchange, Listing Agreement, Guidelines/Regulations issued by statutory authorities.

Please note that the aforesaid observations does not preclude the Company from complying with any other requirements.

Further, it may be noted that with reference to Section 230 (5) of the Companies Act, 2013 (Act), read with Rule 8 of Companies (Compromises, Arrangements and Amalgamations) Rules 2016 (Company Rules) and Section 66 of the Act read with Rule 3 of the Company Rules wherein pursuant to an Order passed by the Hon'ble National Company Law Tribunal, a Notice of the proposed scheme of compromise or arrangement filed under sections 230-232 or Section 66 of the Companies Act 2013 as the case may be is required to be served upon the Exchange seeking representations or objections if any.

In this regard, with a view to have a better transparency in processing the aforesaid notices served upon the Exchange, the Exchange has <u>already introduced an online system of serving such</u> <u>Notice along with the relevant documents of the proposed schemes through the BSE Listing</u> <u>Centre.</u>



BSE - INTERNAL

BSE - INTERNAL



Any service of notice under Section 230 (5) or Section 66 of the Companies Act 2013 seeking Exchange's representations or objections if any, <u>would be accepted and processed through the Listing Centre only and no physical filings would be accepted</u>. You may please refer to circular dated February 26, 2019 issued to the company.

Yours faithfully, Nifinkumar Pujari

Senior Manager







National Stock Exchange Of India Limited

Ref: NSE/LIST/18881_I

July 15, 2019

The Company Secretary Arshiya Limited 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018

Kind Attn.: Ms. Savita Dalal

Dear Madam,

Sub: Observation Letter for Scheme of Arrangement between Arshiya Limited and Arshiya Rail Infrastructure Limited

We are in receipt of Scheme of Arrangement between Arshiya Limited ("Demerged Company" or "Arshiya") and Arshiya Rail Infrastructure Limited ("Resulting Company" or "ARIL"). The Scheme of Arrangement provides for Demerger of Domestic Business Undertaking which includes Domestic Warehousing activities (Demerged Undertaking) from Arshiya Limited ("Demerged Company" or "Arshiya") into Arshiya Rail Infrastructure Limited ("Resulting Company" or "ARIL").

Based on our letter reference no Ref: NSE/LIST/18881submitted to SEBI and pursuant to SEBI Circular No. CFD/DIL3/CIR/2017/21 dated March 10, 2017 ('Circular'), SEBI vide letter dated July 12, 2019, has given following comments:

- a. The Company shall ensure that additional information, if any submitted by the Company, after filing the Scheme with the Stock Exchange and from the date of the receipt of this letter is displayed on the website of the listed company.
- b. The Company shall duly comply with various provisions of the Circular.
- c. The Company is advised that the observations of SEBI/Stock Exchanges shall be incorporated in the petition to be filed before National Company Law Tribunal (NCLT) and the company is obliged to bring the observations to the notice of NCLT.
- d. It is to be noted that the petitions are filed by the company before NCLT after processing and communication of comments/observations on draft scheme by SEBI/ stock exchange. Hence, the company is not required to send notice for representation as mandated under section 230(5) of Companies Act, 2013 to SEBI again for its comments/observations/ representations.

It is to be noted that the petitions are filed by the company before NCLT after processing and communication of comments/observations on draft scheme by SEBI/ stock exchange. Hence, the company is not required to send notice for representation as mandated under section 230(5) of Companies Act, 2013 to National Stock Exchange of India Limited again for its comments/observations/ representations.

This Document is Digitally Signed

NSE

Signer: Rajendra P Bhosale Date: Mon, Jul 15, 2019 19:00:11 IST Location: NSE

National Stock Exchange of India Limited | Exchange Plaza, C-1, Block G, Bandra Kurla Complex, Bandra (E), Mumbai – 400 051, India +91 22 26598100 | www.nseindia.com | CIN U67120MH1992PLC069769



Continuation Sheet

Further, where applicable in the explanatory statement of the notice to be sent by the company to the shareholders, while seeking approval of the Scheme, it shall disclose information about unlisted companies involved in the format prescribed for abridged prospectus as specified in the circular dated March 10, 2017.

Based on the draft scheme and other documents submitted by the Company, including undertaking given in terms of Regulation 11 of SEBI (LODR) Regulations, 2015, we hereby convey our "No-objection" in terms of Regulation 94 of SEBI (LODR) Regulations, 2015, so as to enable the Company to file the draft scheme with NCLT.

However, the Exchange reserves its rights to raise objections at any stage if the information submitted to the Exchange is found to be incomplete/ incorrect/ misleading/ false or for any contravention of Rules, Bye-laws and Regulations of the Exchange, Listing Regulations, Guidelines / Regulations issued by statutory authorities.

The validity of this "Observation Letter" shall be six months from July 15, 2019, within which the scheme shall be submitted to NCLT.

Yours faithfully, For **National Stock Exchange of India Limited**

Rajendra Bhosale Manager

P.S. Checklist for all the Further Issues is available on website of the exchange at the following URL <u>http://www.nseindia.com/corporates/content/further_issues.htm</u>

This Document is Digitally Signed



Signer: Rajendra P Bhosale Date: Mon, Jul 15, 2019 19:00:11 IST Location: NSE



Date: 17th November, 2018

To, The General Manager, Department of Corporate Services, BSE Limited, P.J. Towers, Dalal Street, Mumbai – 400 001. Scrip Code: 506074

Ref: Case No. 85922

Sub: Application under Regulation 37 of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015 for the proposed scheme of Arrangement between Arshiya Limited (hereinafter referred to as "Demerged Company" or "Arshiya"), and Arshiya Rail Infrastructure Limited (hereinafter referred to as "Resulting Company" or "ARIL").

This is in continuation to our application under Regulation 37 of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015, read with SEBI (Circular No. CFD/DIL3/CIR/2017/21 dated 10th March, 2017 (earlier SEBI Circular No. CIR/CFD/CMD/16/2015 dated 30th November, 2015) ("Circular) and subsequent information/documents sought by you from time to time.

In terms of the provision of the above said circular and as mentioned in our application dated 10th October, 2018 we hereby submit the Complaints Report for the period commencing from 26th October, 2018 to 16th November, 2018 stating details of complaints/comments on the Scheme of Arrangement (Annexed hereto as an Annexure – I).

The Complaints Report will also be uploaded on the website of the Company i.e. www.arshiyalimited.com.

Request you to kindly take the same on your record.

For Arshiva Limited

Savita Dalal Company Secretary & Compliance Officer

Encl. as above

Lim, a Mumbai

Arshiya Limited

Regd. Off.: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018. India. T: +91 22 4230 5500/502 I F: +91 22 4230 5555 I Email: info@arshiyalimited.com I www.arshiyalimited.com CIN : L93000MH1981PLC024747



ANNEXURE - I

Complaints Report

Part A

Particulars	Number 0	
Number of complaints received directly		
Number of complaints forwarded by Stock Exchange	0	
Total Number of complaints/comments received (1+2)	0	
Number of complaints resolved	Not Applicable	
Number of complaints pending	Not Applicable	
	Number of complaints received directly Number of complaints forwarded by Stock Exchange Total Number of complaints/comments received (1+2) Number of complaints resolved	

Part B

Sr. No.	Name of complainant	Date of complaint	Status (Resolved/Pending)
1.	Not Applicable	Not Applicable	Not Applicable

ya Lim

Mumbai

Ars

For Arshiya Limited 0

Savita Dalal Company Secretary & Compliance Officer

Date: 17th November, 2018



Date: 14th May, 2019

To, Manager - Listing Compliance National Stock Exchange of India Limited, 'Exchange Plaza'. C-1, Block G, Bandra Kurla Complex, Bandra (E), Mumbai - 400 051 Scrip Code: ARSHIYA

Ref: Application No. 18881

Sub: Application under Regulation 37 of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015 for the proposed scheme of Arrangement between Arshiya Limited (hereinafter referred to as "Demerged Company" or "Arshiya"), and Arshiya Rail Infrastructure Limited (hereinafter referred to as "Resulting Company" or "ARIL").

This is in continuation to our application under Regulation 37 of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015, read with SEBI (Circular No. CFD/DIL3/CIR/2017/21 dated 10th March, 2017 (earlier SEBI Circular No. CIR/CFD/CMD/16/2015 dated 30th November, 2015) ("Circular) and subsequent information/documents sought by you from time to time.

In terms of the provision of the above said circular and as mentioned in our application dated 10^{th} October, 2018 we hereby submit the Complaints Report for the period commencing from 22^{nd} April, 2019 to 14^{th} May, 2019 stating details of complaints/comments on the Scheme of Arrangement (Annexed hereto as an Annexure – J).

The Complaints Report will also be uploaded on the website of the Company i.e. http://www.arshiyalimited.com/shareholding-information.html.

Request you to kindly take the same on your record.

For Arshiya Limited MUNARIA Savita Dalal **Company Secretary & Compliance Officer**

Arshiya Limited

Regd. Off.: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018. India. T: +91 22 4230 5500/502 | F: +91 22 4230 5555 | Email: info@arshiyalimited.com | www.arshiyalimited.com CIN : L93000MH1981PLC024747


ANNEXURE - J

Complaints Report

Part - A

Sr. No.	Particulars	Number
1.	Number of complaints received directly	0
2.	Number of complaints forwarded by Stock Exchange	0
3.	Total Number of complaints/comments received (1+2)	0
4.	Number of complaints resolved	Not Applicable
5.	Number of complaints pending	Not Applicable

Part - B

Sr. No.	Name of complainant	Date of Complaint	Status (Resolved/Pending)
1.	Not Applicable	Not Applicable	Not Applicable

For Arshiya Limited



Savita Dalal Company Secretary & Compliance Officer

Date: 14th May, 2019

Arshiya Limited

Regd. Off.: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018. India. T: +91 22 4230 5500/502 | F: +91 22 4230 5555 | Email: info@arshiyalimited.com | www.arshiyalimited.com CIN : L93000MH1981PLC024747



Date: 08th October, 2018

To, The General Manager, Department of Corporate Services, BSE Limited, P.J. Towers, Dalal Street, Mumbai – 400 001.

COMPLIANCE REPORT

It is hereby certified that the draft scheme of arrangement involving Arshiya Limited and Arshiya Rail Infrastructure Limited does not, in any way violate, override or limit the provisions of securities laws or requirements of the Stock Exchange(s) and the same is in compliance with the applicable provisions of SEBI (Listing Obligations and Disclosure Requirements) Regulations, 2015 and this circular, including the following:

SI.	Reference	Particulars
1	Regulations 17 to 27 of LODR Regulations	Corporate governance requirements
2	Regulation 11 of LODR Regulations	Compliance with securities laws
Requi	rements of this circular	3
(a)	Para (I)(A)(2)	Submission of documents to Stock Exchanges
(b)	Para (I)(A)(2)	Conditions for schemes of arrangement involving unlisted entities
(c)	Para (I)(A)(4) (a)	Submission of Valuation Report*
(d)	Para (I)(A)(5)	Auditors certificate regarding compliance with Accounting Standards
(e)	Para (I)(A)(9)	Provision of approval of public shareholders through e-voting

*Valuation Report is not required since there is no change in the shareholding pattern of the Demerged Company / resultant company.

Savia Dalal **Company Secretary**

Lin su Mumbai 4

Ajay S Mittal Managing Director

Certified that the transactions / accounting treatment provided in the draft scheme of arrangement involving Arshiya Limited and Arshiya Rail Infrastructure Limited are in compliance with all the Accounting Standards applicable to a listed entity.

S. Maheshwari

Chief Financial Officer

Arshiya Limited

Ajay S Mittal Managing Director

Regd. Off.: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018. India. T: +91 22 4230 5500/502 I F: +91 22 4230 5555 I Email: info@arshiyalimited.com I www.arshiyalimited.com CIN : L93000MH1981PLC024747

Arshiya

ANNEXURE IV

Date: 23rd April, 2019

To, Manager - Listing Compliance National Stock Exchange of India Limited 'Exchange Plaza'. C-1, Block G, Bandra Kurla Complex, Bandra (E), Mumbai - 400 051

COMPLIANCE REPORT

It is hereby certified that the draft scheme of arrangement involving Arshiya Limited and Arshiya Rail Infrastructure Limited does not, in any way violate, override or limit the provisions of securities laws or requirements of the Stock Exchange(s) and the same is in compliance with the applicable provisions of SEBI (Listing Obligations and Disclosure Requirements) Regulations, 2015 and this circular, including the following:

SI.	Reference	Particulars
1	Regulations 17 to 27 of LODR Regulations	Corporate governance requirements
2	Regulation 11 of LODR Regulations	Compliance with securities laws
Requi	rements of this circular	
(a)	Para (I)(A)(2)	Submission of documents to Stock Exchanges
(b)	Para (I)(A)(2)	Conditions for schemes of arrangement involving unlisted entities
(c)	Para (I)(A)(4) (a)	Submission of Valuation Report*
(d)	Para (I)(A)(5)	Auditors certificate regarding compliance with Accounting Standards
(e)	Para (I)(A)(9)	Provision of approval of public shareholders through e-voting

*Valuation Report is not required since there is no change in the shareholding pattern of the Demerged Company / resultant company.

Savia Dalal **Company Secretary**



Ajay S Mittal Managing Director

Certified that the transactions / accounting treatment provided in the draft scheme of arrangement involving Arshiya Limited and Arshiya Rail Infrastructure Limited are in compliance with all the Accounting Standards applicable to a listed entity.

S. Maheshwari Chief Financial Officer

Arshiya Limited

Ajay S Mittal Managing Director

Regd. Off.: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018. India. T: +91 22 4230 5500/502 I F: +91 22 4230 5555 I Email: info@arshiyalimited.com I www.arshiyalimited.com CIN : L93000MH1981PLC024747

REPORT ADOPTED BY THE BOARD OF DIRECTORS OF ARSHIYA LIMITED AT ITS MEETING HELD ON THURSDAY THE 24th DAY OF MAY, 2018 EXPLAINING THE EFFECT OF SCHEME ON EQUITY SHAREHOLDERS, KEY MANAGERIAL PERSONNEL, PROMOTERS AND NON-PROMOTER SHAREHOLDER

Background:

- 1. The proposed Scheme of arrangement is between Arshiya Limited ("**Demerged company**") and Arshiya Rail Infrastructure Limited ("**Resulting company**") and their respective shareholders and creditors ("**Scheme**") had been approved by the board of directors of the Demerged Company ("**Board**") vide the resolution passed at its meeting on May 24, 2018.
- 2. The provision of Section 230(2)(c) of the Companies Act, 2013 requires the Board to adopt a report explaining effect of the Scheme on each class of Shareholders, Key Managerial Personnel, Promoters and Non-Promoter Shareholders, laying out in particular share exchange ratio and specifying and special valuation difficulties. The said report adopted by the Board is required to be circulated to the shareholders and creditors along with the notice convening the meeting of members/creditors.
- 3. This report of the Board is accordingly prepared in accordance with the requirements specified under Section 230(3) (c) of the Companies Act, 2013.
- 4. The following documents were considered by the Board for the purpose of issuing this report:
 - a. The draft of the Scheme
 - b. Report by ZADN & Associates, Chartered Accountants in respect of share entitlement ratio.
 - c. Fairness opinion dated May 28, 2018 provided by the Marchant Banker in respect of the fairness of the share entitlement ratio.
 - d. Report of the Audit Committee of the Demerged Company, recommending the draft scheme to the Board for approval.
 - e. Certificate dated 24th day of May, 2018 issued by Chaturvedi & Shah, Chartered Accountants the statutory auditors of the company as required under section 232(3) of the companies Act, 2013 certifying that the accounting treatment in the draft scheme is in accordance with the accounting standards and applicable law.

Report:

- 1. The Scheme provides for the transfer of the Demerged Undertaking (as defined in the scheme) to the Resulting Company. There will be no change in the shareholding pattern of the Demerged Company.
- 2. Upon the Scheme becoming effective and in consideration of the demerger including the transfer and vesting of the Demerged Undertaking in the Resulting Company, the Resulting Company shall, without any further application or deed, for every 2 (two) fully paid-up equity shares of the Demerged Company, issue and allot to each member of the Demerged Company whose name appears in the register of members of the Demerged Company as on the Record Date or to his/her heirs, executors, administrators or the successors-in-title, as the case may be, 1 (one) fully paid-up equity share of Rs 2 each, of the Resulting Company ('New Equity Shares').

Effect of the Scheme on the Equity Shareholders (the only class of Shareholders) and promoters, and Non-Promoter Shareholders of the Demerged Company:

- 1. Basis the share exchange ration mentioned above, the Resulting Company shall issue shares to the shareholders of the Demerged Company in the ratio of 2:1 and the shares held by the Demerged Company in the Resulting Company shall be extinguished.
- 2. Upon the Scheme coming to effect, the shares to be allotted to the members of the Demerged Company by the Resulting Company shall be listed and / or admitted to trading on the relevant stock exchange/s in India where the equity shares of the Arshiya are listed and / or admitted to trading as on effective date. Accordingly, the Resulting Company shall take steps for listing simultaneously on all such stock exchange(s) within a reasonable period of the receipt of the final NCLT order sanctioning the Scheme.
- 3. Pursuant to the Scheme, all shareholders of the Demerged Company as on the Record Date (as defined in the scheme) will receive equity shares in the Resulting Company and subsequently, such shareholders of the Demerged Company will hold equity shares in both, the Demerged Company and Resulting Company, it will give such shareholders of the Demerged Company the ability to continue to remain invested in both or either of the Companies, giving them greater flexibility in managing and/or dealing with their investments.

4. The Scheme is beneficial to the respective shareholders, creditors, employees and all stakeholders of the Demerged Company and Resulting Company. The Scheme is expected to contribute in furthering and fulfilling the objectives of both the companies and in the growth and development of their respective businesses.

Basis the aforesaid, we understand that there will be no adverse effect of the scheme on the equity shareholders, directors, promoters and non-promoter shareholders of the Demerged Company.

Effect of the Scheme on the Directors and Key Managerial personnel of the Demerged Company:

5. None of the Directors and Key Managerial Personnel of the Demerged Company and their respective relatives have any interest in the scheme except to the extent the (i) equity shares held by them in the Demerged Company and the resulting Company directly or as nominee; and /or (ii) Director(s) are common director(s) in the two companies; and/or (iii) the Director(s), key managerial personnel and their respective relatives are the director(s), partner(s), members(s) and/or beneficiary(ies) of the companies, firms, association of persons, body corporates and/or trust, as the case may be, that hold shares into the Demerged Company/ Resulting Company. There will be no adverse effect of the scheme on the directors and key managerial personnel of the Demerged Company

For Arshiya Limited.

-/Sd/-Ajay S Mittal Chairman appointed for the meeting

Date: 10th December, 2019 Place: Mumbai

REPORT ADOPTED BY THE BOARD OF DIRECTORS OF ARSHIYA RAIL INFRASTRUCTURE LIMITED AT ITS MEETING HELD ON THURSDAY THE 24th DAY OF MAY, 2018 EXPLAINING THE EFFECT OF SCHEME ON EQUITY SHAREHOLDERS, KEY MANAGERIAL PERSONNEL, PROMOTERS AND NON-PROMOTER SHAREHOLDER

Background:

- 1. The proposed Scheme of arrangement is between Arshiya Limited ("**Demerged company**") and Arshiya Rail Infrastructure Limited ("**Resulting company**") and their respective shareholders and creditors ("**Scheme**") had been approved by the board of directors of the Demerged Company ("**Board**") vide the resolution passed at its meeting on May 24, 2018.
- 2. The provision of Section 230(2)(c) of the Companies Act, 2013 requires the Board to adopt a report explaining effect of the Scheme on each class of Shareholders, Key Managerial Personnel, Promoters and Non-Promoter Shareholders, laying out in particular share exchange ratio and specifying and special valuation difficulties. The said report adopted by the Board is required to be circulated to the shareholders and creditors along with the notice convening the meeting of members/creditors.
- 3. This report of the Board is accordingly prepared in accordance with the requirements specified under Section 230(3) (c) of the Companies Act, 2013.
- 4. The following documents were considered by the Board for the purpose of issuing this report:
 - a. The draft of the Scheme
 - b. Report by ZADN & Associates, Chartered Accountants in respect of share entitlement ratio.
 - c. Fairness opinion dated May 28, 2018 provided by the Marchant Banker in respect of the fairness of the share entitlement ratio.
 - d. Report of the Audit Committee of the Resulting Company, recommending the draft scheme to the Board for approval.
 - e. Certificate dated 24th day of May, 2018 issued by Chaturvedi & Shah, Chartered Accountants the statutory auditors of the company as required under section 232(3) of the companies Act, 2013 certifying that the accounting treatment in the draft scheme is in accordance with the accounting standards and applicable law.

Report:

- 1. The Scheme provides for the transfer of the Demerged Undertaking (as defined in the scheme) to the Resulting Company. There will be no change in the shareholding pattern of the Demerged Company.
- 2. Upon the Scheme becoming effective and in consideration of the demerger including the transfer and vesting of the Demerged Undertaking in the Resulting Company, the Resulting Company shall, without any further application or deed, for every 2 (two) fully paid-up equity shares of the Demerged Company, issue and allot to each member of the Demerged Company whose name appears in the register of members of the Demerged Company as on the Record Date or to his/her heirs, executors, administrators or the successors-in-title, as the case may be, 1 (one) fully paid-up equity share of Rs 2 each, of the Resulting Company ('New Equity Shares').

Effect of the Scheme on the Equity Shareholders (the only class of Shareholders) and promoters, and Non-Promoter Shareholders of the Resulting Company:

- 1. Basis the share exchange ration mentioned above, the Resulting Company shall issue shares to the shareholders of the Demerged Company in the ratio of 2:1 and the shares held by the Demerged Company in the Resulting Company shall be extinguished.
- 2. Upon the Scheme coming to effect, the shares to be allotted to the members of the Demerged Company by the Resulting Company shall be listed and / or admitted to trading on the relevant stock exchange/s in India where the equity shares of the Arshiya are listed and / or admitted to trading as on effective date. Accordingly, the Resulting Company shall take steps for listing simultaneously on all such stock exchange(s) within a reasonable period of the receipt of the final NCLT order sanctioning the Scheme.
- 3. Pursuant to the Scheme, all shareholders of the Demerged Company as on the Record Date (as defined in the scheme) will receive equity shares in the Resulting Company and subsequently, such shareholders of the Demerged Company will hold equity shares in both, the Demerged Company and Resulting Company, it will give such shareholders of the Demerged Company the ability to continue to remain invested in both or either of the Companies, giving them greater flexibility in managing and/or dealing with their investments.

4. The Scheme is beneficial to the respective shareholders, creditors, employees and all stakeholders of the Demerged Company and Resulting Company. The Scheme is expected to contribute in furthering and fulfilling the objectives of both the companies and in the growth and development of their respective businesses.

Basis the aforesaid, we understand that there will be no adverse effect of the scheme on the equity shareholders, directors, promoters and non-promoter shareholders of the Demerged Company.

Effect of the Scheme on the Directors and Key Managerial personnel of the Resulting Company:

5. None of the Directors and Key Managerial Personnel of the Resulting Company and their respective relatives have any interest in the scheme except to the extent the (i) equity shares held by them in the Demerged Company and the resulting Company directly or as nominee; and /or (ii) Director(s) are common director(s) in the two companies; and/or (iii) the Director(s), key managerial personnel and their respective relatives are the director(s), partner(s), members(s) and/or beneficiary(ies) of the companies, firms, association of persons, body corporates and/or trust, as the case may be, that hold shares into the Demerged Company/ Resulting Company. There will be no adverse effect of the scheme on the directors and key managerial personnel of the Resulting Company.

For Arshiya Rail Infrastructure Limited.

-/Sd Ajay S Mittal Chairman appointed for the meeting

Date: 10th December, 2019 Place: Mumbai

ZADN & Associates

4th Floor, Shreeniwas House Hazarimal Somani Marg Next to MTNL Exchange Building Fort, Mumbai 400 001, India Tel: +91 22 4973 5451 / 52 / 53

May 24, 2018

To,

The Board of Directors Arshiya Limited (Demerged Company) 302. Ceejay House, Level - 3 Shiv Sagar Estate, F- Block Dr. Annie Besant Road Worli, Mumbai – 400018, India. To,

The Board of Directors Arshiya Rail Infrastructure Limited (Resulting Company) 302, Ceejay House, Level - 3 Shiv Sagar Estate, F- Block Dr. Annie Besant Road Worli, Mumbai – 400018, India.

Dear Sirs,

Sub: Report on share entitlement ratio for the proposed demerger of Domestic Business Undertaking ('Demerged Undertaking') of Arshiya Limited ('Demerged Company') into Arshiya Rail Infrastructure Limited ('Resulting Company').

We, ZADN & Associates, Chartered Accountants ('We' or 'ZADN' or 'Our' or 'Us'), refer to the Engagement Letter dated May 8, 2018 and related discussions we had with you, wherein the management of Arshiya Limited ('Demerged Company' or 'Arshiya') and Arshiya Rail Infrastructure Limited ('Resulting Company' or 'ARIL') (collectively hereinafter referred to as the 'Companies') requested our report on the share entitlement ratio of equity shares of the Resulting Company to be issued to the equity shareholders of the Demerged Company in connection with the proposed demerger of Domestic Business Undertaking ('Demerged Undertaking') of Arshiya Limited on the Record Date as more elaborately defined in the proposed Scheme of Arrangement between Arshiya Limited and Arshiya Rail Infrastructure Limited, under sections 230 to 232 read with section 66 and section 52 and other applicable provisions of the Companies Act, 2013 hereinafter referred to as 'the Proposed Scheme' or 'the Scheme of Arrangement'.



1. Background:

1.1 Arshiya Limited ('Arshiya' or the 'Demerged Company')

- 1.1.1 Arshiya Limited is a listed public limited company incorporated on July 3, 1981 under the Companies Act, 1956 bearing CIN L93000MH1981PLC024747 and its registered office is situated at 302, Level 3, Ceejay House, F Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai 400 018. Equity shares of Arshiya are listed on BSE Limited ('BSE') and National Stock Exchange of India Limited ('NSE') in India.
 - 1.1.2 Arshiya, is a flagship company of Arshiya Group having interests in the business of developing Free Trade Warehousing Zones ('FTWZs') and Domestic Warehousing Areas ('DWAs') to improve logistics infrastructure in India.
 - 1.1.3 Board of Directors of the Company as on May 12, 2018 are as under:

Name of the Director	Begin Date
Mr. Ashishkumar Bairagra	August 7, 2006
Mr. Ajay Shankarlal Mittal	October 1, 2011
Mr. Shah Pankaj Rishabh	December 31, 2005
Ms. Archana Ajay Mittal	October 25, 2005
Mr. Mukesh Kacker	October 28, 2009
Mr. Savita Kodain	July 10, 2015
Mr. Santosh Mohanlal Maheshwari	February 8, 2017

1.1.4 The Authorized, Issued, Subscribed and Paid-up Share Capital of Arshiya as on May 12, 2018 is as under:

Particulars	Amount in INR
Authorised Share Capital:	
247,500,000 Equity Shares of Rs. 2/- each	495,000,000
11,000,000 Preference Shares of Rs. 10/- each	110,000,000
Total	605,000,000
Issued, Subscribed and Paid-up Share Capital	
229,716,776 Equity Shares of Rs. 2/- each	459,433,552
5,764,619 Preference Shares of Rs. 10/- each	57,646,190
Total	517,079,742

We are informed that there are no subsequent changes in the capital structure and directors on board.

1.1.5 As stated in the audited financial statements of Arshiya for the year ended March 31, 2017, Arshiya is in financial stress and had exited its corporate debt restructuring during FY16 pursuant to which the lending banks have assigned their financial assets pertaining to Arshiya and its other wholly owned

Strictly Private and Confidential



1

subsidiaries companies including ARIL to Edelweiss Asset Reconstruction Company Limited ('EARC').

- 1.1.6 Accordingly, EARC has become the secured lender and all the rights, title and interest of lending banks have vested in EARC. Arshiya has entered into Restructuring Agreement with EARC on March 31, 2017 for restructuring its bank dues wherein the debt is segregated between Restructured Rupee Loan which will carry interest of 10% p.a. and the restructured loans are payable over 7 years tenure and balance loan is to be converted by issuance of equity shares and Optionally Convertible Redeemable Preference Shares ('OCRPS') which may be converted into equity shares at the option of the holder at any time within 18 months from the date of issue of OCRPS.
- 1.1.7 As represented by the management, below table describes the number of OCRPS issued by Arshiya to EARC and their as on date conversion status:

Particulars	No of OCRPS issued	Maximum No of Equity shares to be alloted upon conversion of OCRPS	Allotment of equity shares pursuant to converiosn of OCRPS	No. of OCRPS outstanding as on date	Balance allotment of equity shares
Series I - OCRPS	6,423,329	15,485,554	1,588,038	5,764,619	13,897,516
Series II - OCRPS	1,310,000	2,382,392	2,382,392		-0
Series III - OCRPS	870,000	4,764,785	4,764,785	*	-
Series IV - OCRPS	2,140,000	4,764,785	4,764,785	e i	
Total	10,743,329	27,397,516	13,500,000	5,764,619	13,897,516

As per clause 1.4.3 of the Proposed Scheme, it is clarified that the OCRPS issued by Arshiya does not relate to Domestic Business and accordingly, no additional / proportionate consideration shall be payable to its holders upon the Proposed Scheme becoming effective. However, any Corporate Action relating to the outstanding Series 1 – OCRPS shall be kept in abeyance till last date upto which the option for conversion is exercisable. Further, in case if there is change in any parameters or facts or any assumptions stated in this report, the report will undergo a change to the extent required.

In reference to above, management informed that if the said Series 1 - OCRPS is converted into equity shares on or before the Record Date, they will receive the equity shares of ARIL as per share entitlement report, otherwise, said Series 1 - OCRPS shall continue to be in the books of Arshiya not affecting the identical equity shareholding being created pursuant to Proposed Scheme. Further, we understand from the management that there are no on-going discussions with any other lender(s) for issuance of equity shares / convertible instrument and conversion of convertible instruments other than outstanding 5,764,619 Series 1 - OCRPS held by EARC.

1.2 Arshiya Rail Infrastructure Limited ('ARIL' or the 'Resulting Company')

1.2.1 ARIL is a unlisted public company incorporated on April 7, 2008 under the Companies Act, 1956 bearing CIN - U93000MH2008PLC180907 and its

Strictly Private and Confidential



2

registered office is situated at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai – 400 018.

- 1.2.2 ARIL is currently engaged in the business of providing Private Freight Terminal (PFT) and Rail Transportation Services (Rail). ARIL is also engaged in container train business and holding Category – I license which allows ARIL to operate on Indian Rail network on pan India basis both domestic and exim traffic.
- 1.2.3 Board of Directors of the Company as on May 12, 2018 is as under:

Name of the Director	Begin Date
Mr. Ashishkumar Bairagra	January 16, 2010
Mr. Navnit Jugal Kishore Choudhary	January 15, 2013
Mr. Shah Pankaj Rishabh	March 27, 2015

1.2.4 The Authorized, Issued, Subscribed and Paid-up Share Capital of ARIL as on May 12, 2018 is as under:

Particulars	Amount in INR
Authorised Share Capital:	
45,000,000 Equity Shares of Re.10/- each	450,000,000
Total	450,000,000
Issued, Subscribed and Paid-up Share Capit	tal:
42,384,417 Equity Shares of Re. 10/- each	423,844,170
Total	423,844,170

We are informed that there are no subsequent changes in the capital structure and directors on board. As per clause 9 of the Proposed Scheme, pursuant to and upon the Proposed Scheme becoming effective, it is proposed that Authorised Share Capital of ARIL, without any further application, act, instrument or deed, shall be reclassified to 22,50,00,000 Equity Shares of Rs. 2/- each.

- 1.2.5 As on date, ARIL is a wholly owned subsidiary of Arshiya Limited as the entire paid up equity share capital is held by Arshiya and its nominees.
- 1.2.6 Management of ARIL informed that currently there are no OCRPS holders in ARIL and there are no on-going discussions with any lenders for issuance of equity shares / convertible instruments.
- 1.2.7 The management of Arshiya Group has informed that as part of the overall group reorganization, the management has filed a Scheme of Amalgamation of Arshiya Industrial & Distribution Hub Limited ('AIDHL') and Arshiya Transport and Handling Limited ('ATHL') with Arshiya Rail Infrastructure Limited ('ARIL' or 'Resulting Company') with National Company Law Tribunal at Mumbai and is

Strictly Private and Confidential



awaiting their approval. The appointed date for the proposed merger is October 1, 2015 as stated in the Scheme of Amalgamation.

- 1.2.8 AIDHL and ATHL are 100% subsidiaries of Arshiya and hence fellow subsidiaries of ARIL.
- 1.2.9 Further, based on the earlier Scheme of Amalgamation as adopted by the board of the companies, for every one equity share of AIDHL and ATHL, one equity share of ARIL is to be issued. Accordingly, ARIL would issue 14,992,995 equity shares to the shareholders of AIDHL (i.e. Arshiya) and 50,000 equity shares to the shareholders of ATHL (i.e. Arshiya).
- 1.2.10 Further AIDHL has issued 1,20,000 Zero percent Optionally Convertible Redeemable Preference Shares to Bank of Baroda (BOB) against conversion of loan amounting to INR 1,20,000,000 which was approved by Special Resolution passed on January 17, 2018. The management has informed us that as per the terms of the issuance of OCRPS, Bank of Baroda shall be entitled to equity shares of AIDHL on conversion of the said OCRPS, if the Company fail to make the redemption before the end of sixth year. However, management has represented that the said OCRPS shall be converted into equity shares before the scheme of merger becoming effective. Also as represented to us, Arshiya (Holding Company) has taken a decision in consensus with BOB that Arshiya shall buy these OCRPS and convert the same into equity shares or AIDHL shall redeem the same before the scheme of merger becoming effective.
- 1.2.11 As per the Preamble paragraph of the Proposed Scheme, the Proposed Scheme is also conditional upon the aforesaid merger scheme being effective first and ARIL to continue to be wholly owned subsidiary company of Arshiya as on the Record Date.
- 1.2.12 Please note that if the above representation by the management of Arshiya and ARIL about purchase of OCRPS from BOB and conversion or redemption of the same before the Effective Date of the merger and the Proposed Scheme is not affected then in that scenario, our recommendation of the share entitlement ratio will need to be revisited and might undergo a change, which we are unable to comment upon and it is not part of our current engagement. Further, we understand from the management that there are no on-going discussions with any other lender(s) for issuance of equity or convertible instruments by AIDHL and ATHL.

2. Objective of this Report:

2.1 As described hereinbefore in this report, the Board of Directors of Arshiya proposes to demerge its Domestic Business Undertaking and transfer into ARIL.

Strictly Private and Confidential



4

- 2.2 As defined in the Proposed Scheme, 'Domestic Business Undertaking' means domestic business activities pertaining to development of Domestic Warehousing Areas ('DWAs') including investments held in the Resulting Company.
- 2.3 As informed by the management, the aforesaid proposed demerger is in line with Arshiya Group's intentions to reorganize its corporate structure and businesses which is spread across various group companies (wholly owned subsidiaries) in order to integrate / consolidate its operations by housing similar businesses into respective entities as stated below:
 - FTWZs business in Arshiya or into an independent entity wholly owned by Arshiya.
 - ii. DWAs, RAIL, PFT and Inland Container Depot business in ARIL.
- 2.4 In reference to above, management has informed that the integration / reorganization of corporate structure of the group is to be achieved by way of transfer of Domestic Business Undertaking through the Proposed Scheme with April 1, 2019 as the Appointed Date.
- 2.5 We understand that pursuant to the Proposed Scheme, all the equity shareholders of Arshiya as on the Record Date will be issued equity shares of ARIL as per share entitlement ratio as consideration for the proposed transfer and vesting of Domestic Business Undertaking into ARIL.
- 2.6 The equity shares of Arshiya are listed on BSE and NSE in India. Accordingly, upon the Proposed Scheme coming to effect, the equity shares to be allotted to all the equity shareholders of Arshiya on the Record Date by ARIL will also be listed and / or admitted to trading on BSE and NSE in India, subject to applicable compliances.
- 2.7 Upon the Proposed Scheme becoming effective, all the equity shares held by Arshiya or its nominee in ARIL shall stand cancelled, extinguished and annulled.
- 2.8 In connection with the Proposed Scheme, we have been requested to provide a report on the share entitlement ratio of equity shares of ARIL to the shareholders of Arshiya as proposed by the management.
- 2.9 We understand that consequent to the Proposed Scheme and issuance of equity shares to all the equity shareholders of Arshiya as contemplated in the Proposed Scheme, there will be no impact on the economic and beneficial interest of the equity shareholders of Arshiya.
- 2.10 We understand that this equity share entitlement ratio report will be used for the abovementioned purpose only and to the extent mandatorily required under applicable laws of India.
- 2.11 For the purpose of our report, we have relied on the information, explanations and representations provided to us by the management of Arshiya and ARIL. We have not

Strictly Private and Confidential



carried out any independent tests to establish the accuracy of such information, explanations and representations. In addition, we have listed the scope of work in the course of our assignment, noting any limitations on our assignment. This report is subject to the attached limiting conditions and terms for this assignment, as may be described in this report or in our engagement letter.

3. Sources of Information:

For the purpose of this share entitlement ratio exercise, we have relied on the following information provided to us by the management of the Demerged Company and Resulting Company and information available in the public domain:

- 3.1 Draft Scheme of Arrangement between Arshiya and ARIL proposed to be adopted at the Board meetings of both the companies;
- 3.2 Scheme of Amalgamation of Arshiya Industrial & Distribution Hub Limited ('AIDHL') and Arshiya Transport and Handling Limited ('ATHL') with Arshiya Rail Infrastructure Limited ('ARIL');
- 3.3 Audited financial statements of the Demerged and Resulting Company for the year ended March 31, 2017;
- 3.4 Current Shareholding pattern of the Demerged Companies and Resulting Company;
- 3.5 Desired capital structure and equity shares of ARIL proposed to be issued to the shareholders of Arshiya on demerger of the Domestic Business Undertaking into ARIL;
- 3.6 Representation with regards to dealing with OCRPS issued to Bank of Baroda by AIDHL as detailed in paragraph 1.2.10 of this report;
- 3.7 Other relevant data and information provided to us either in written or oral form or in form of soft copy and discussions with the representatives of management of Arshiya Group:
- 3.8 Representation letter dated May 23, 2018, and
- 3.9 Relevant information in public domain.



6

Strictly Private and Confidential

4. Exclusions and Limitations:

- 4.1 This report is subject to the limitations detailed herein. As such the report is to be read in totality, and not in parts, in conjunction with the relevant documents and representations referred to herein.
- 4.2 Our work was not designed to verify the accuracy or reliability of the information provided to us and nothing in this report should be taken to imply that we have conducted procedures, audits or investigations in an attempt to verify or confirm any of the information supplied to us.
- 4.3 In addition, we do not take any obligation or responsibility for any changes in the information used by us to arrive at our conclusion as set out herein which may occur subsequent to the completion of scope of work and the date of the report.
- 4.4 We further assume that the managements of both all the Companies have brought to our attention any and all factors having an impact on the determination of the share entitlement ratio.
- 4.5 We have been given to understand by the managements of the Companies that they have not omitted any relevant and material factors. Accordingly, we do not express any opinion or offer any form of assurance regarding its accuracy and completeness. We assume no responsibility for any errors in the above information furnished by the Companies and their impact on the present exercise.
- 4.6 Whilst all reasonable care has been taken to ensure that the factual statements in the report are accurate, neither ourselves, nor any of our Partners, Officers or Employees shall, in any way be liable or responsible either directly or indirectly for the contents stated herein. Accordingly, we make no representation or warranty, express or implied, in respect of the completeness, authenticity or accuracy of such factual statements. We expressly disclaim all liabilities, which may arise based upon the information used in this report. We are not liable to any third party in relation to the issue of this report.
- 4.7 We have also relied on information available in public domain; however the accuracy and timelines of the same has not been independently verified by us.
- 4.8 Any person / party, intending to provide finance / invest in the shares / business of the company or for any other reason whatsoever, shall do so after seeking their own professional advice and after carrying out their own due diligence procedures to ensure that they are making an informed decision.
- 4.9 This exercise of commenting on the share entitlement ratio for the proposed demerger is not and must not be interpreted as our opinion or estimate of the absolute market value or independent fair value of the equity shares or any class or number of equity shares pre or post demerger.



Strictly Private and Confidential

7

ZADN & Associates

- 4.10 Our scope of work is limited to expression of our view on the proposed share entitlement ratio and its impact on the economic interest of the shareholders of the Companies. Our report is not, nor should it be construed as, our opining or certifying the compliance of the proposed demerger of the Demerged Undertaking with the provisions of any law or any legal implications or issues arising from such Proposed Scheme.
- 4.11 Our recommendation should be considered to be in the nature of non-binding advice.
- 4.12 This report has been prepared exclusively for the use of Arshiya and ARIL and solely for the purpose of recommending a share entitlement ratio for the Proposed Scheme. Hence, this report should not be provided or used for any purpose, whether in whole or in part without our prior written consent, to any other person and for any other purpose except that as is mentioned earlier in this report. However, we understand that you may share this report with your advisors supporting the proposed transaction as well as statutory authorities. Please note that we do not accept any responsibility to your advisors or any third party with regard to this report.

We would like to draw attention to important representations by the management as detailed in paragraph 1.1.7 and paragraph 1.2.10 of this report. The share entitlement ratio mentioned in this report is on the basis of those representations. While we take no responsibility for implementing the actions or non-implementing the actions by the managements in the representations, we would want to bring to the attention that non-fulfillment of the representations could have an impact on the share entitlement ratio. We are unable to comment on the quantum (whether substantial or not) of impact due to non-fulfillment of the actions committed in those representations.

5. BASIS OF DETERMINATION OF SHARE ENTITLEMENT RATIO:

Our evaluation towards the share entitlement ratio considers the following aspects:

- 5.1 The estimate of net assets, working capital requirements and capital base required for the Domestic Business Undertaking / ARIL as independently assessed by the management of Arshiya and ARIL;
- 5.2 Based on the above assessment the management has proposed the number of equity shares to be issued such that ARIL is appropriately capitalized;
- 5.3 As on date, ARIL is a wholly owned subsidiary ('WOS') of Arshiya and shall continue to be WOS of Arshiya till the effective date / Record Date of the Proposed Scheme.
- 5.4 As detailed earlier, the management is also in the process of merging AIDHL and ATHL with ARIL, all WOS's of Arshiya.



8

Strictly Private and Confidential

ZADN & Associates

- 5.5 Pursuant to clauses in the Proposed Scheme with regards to dealing with Series 1 -OCRPS as detailed in paragraph 1.1.7 of this report and representations by the management in paragraph 1.2.10 of this report would be implemented prior to the Record Date, the equity shares issued in ARIL to the shareholders of Arshiya would be in the same proportion in which they hold equity shares in Arshiya on the Record Date and such allotment would be value neutral to the equity shareholders of Arshiya.
- 5.6 As per Part V clause 7.8 of the Proposed Scheme, upon the scheme becoming effective, all the equity shares held by the Demerged Company or its nominee in the Resulting Company shall stand cancelled, extinguished and annulled. As such, the equity shareholding pattern of ARIL post implementation of the representations and the Proposed Scheme, would be identical to the equity shareholding pattern of Arshiya.
- 5.7 Accordingly as stated in clause 1.5 of the Proposed Scheme, all the equity shareholders of Arshiya as on the Record Date would be entitled to equity shares of ARIL in the same proportion in which they currently own directly in Arshiya resulting into identical shareholders to Arshiya and ARIL. Accordingly, even after the Proposed Scheme coming into effect, the economic and beneficial interest will remain within the existing shareholders of Arshiya in the same proportion as they hold in Arshiya as on the Record Date.
- 5.8 Thus, share entitlement ratio would not have any impact on the ultimate value of the equity shareholders of Arshiya and the proposed demerger of Domestic Business Undertaking of Arshiya into ARIL will be value-neutral to Arshiya's equity shareholders. The background and rationale for the scheme has been provided in the Proposed Scheme.
 - 5.9 Accordingly, the determination of share entitlement ratio of the companies is at best an internal arrangement between the Demerged Company and Resulting Company and its shareholders and a detailed valuation of the Companies to determine the share entitlement ratio would not be relevant in the present case.
 - 5.10 The share entitlement ratio and the number of equity shares to be allotted pursuant to demerger, is of no material relevance since there will be no loss of economic interest in the hands of equity shareholders of Arshiya. Accordingly, for the purpose of recommending a share entitlement ratio we are not attempting to arrive at the absolute values of equity shares of each company.
 - 5.11 Accordingly, in our opinion based on the representations and information provided by the management, as there will be no change in the shareholding pattern of Arshiya pursuant to the Scheme, no valuation report is required as per Circular CFD/DIL3/CIR/2017/21 dated March 10, 2017 issued by SEBI, as may be amended from time to time.



Strictly Private and Confidential

9

- 5.12 Further, we have been informed by the management of ARIL that upon the effective date, the Authorized Equity Share Capital of Arshiya would be increased by the requisite amount, if required, so that issued paid up share capital post the allotment pursuant to the Scheme is lower than the authorized capital.
- 5.13 Considering the above, the share entitlement ratio for demerger of Domestic Business Undertaking of Arshiya into ARIL, as suggested by the managements of both the Companies and as stated under we believe that the share entitlement ratio is fair and equitable considering that all the shareholders of Arshiya upon demerger will become the shareholders of ARIL and ultimate beneficial owners of Arshiya and ARIL will be held by the same shareholders in same proportion. Accordingly,

"1 (One) fully paid equity share of Face Value INR 2 (Rupees Two) each of ARIL is to be issued for every 2 (Two) fully paid equity shares of Face Value INR 2 (Rupees Two) each held in Arshiya."



M. No.100666 Place: Mumbai

Annexure G

CHATURVEDI Chartered Accountants

To, The Board of Directors, Arshiya Limited 302, Level 3, Ceejay House, F-Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai – 400 018

Independent Auditor's Certificate in respect of proposed accounting treatment as mentioned in Scheme of Amalgamation Between Arshiya Limited ('Demerged Company' or 'Arshiya') AND Arshiya Rail Infrastructure Limited ('Resulting Company' or 'ARIL')AND their respective shareholders (hereinafter referred to as 'the Scheme').

- 1. This Certificate is issued in accordance with the terms of our engagement with Arshiya Limited ('the Company').
- The accounting treatment proposed by the Company is specified in Clause 8 of the Scheme. For ease of reference, the said Clause 8 of the Scheme, duly authenticated on behalf of the Company, are reproduced in Annexure to this Certificate ('the Statement') and initialled by us only for the purposes of identification.

Management's Responsibility

3. The responsibility for preparation of the Scheme and its compliance with relevant laws and regulations, including the applicable Indian Accounting Standards read with the rules made thereunder and other Generally Accepted Accounting principles, is that of the Board of Directors of the Companies involved. This responsibility includes the design, implementation and maintenance of internal control relevant to the preparation and presentation of the Scheme and applying an appropriate basis of preparation; and making estimates that are reasonable in the circumstances.

Auditor's Responsibility

4. Our responsibility is only to examine and report whether the accounting treatment proposed by the Company as specified in Clause 8 of the Scheme as reproduced in the Statement is in conformity with the Indian Accounting Standards specified in Section 133 or any other provisions of the Companies Act, 2013 and in line with the Generally Accepted Accounting Principles in India. Nothing contained in this Certificate, nor anything said or done in the course of, or in connection with the services that are subject to this Certificate, will extend any duty of care that we may have in our capacity of the statutory auditors of any financial statements of the Company.

Head Office: 714-715, Tulsiani Chambers, 212, Nariman Point, Mumbai - 400 021, India. Tel.: +91 22 3021 8500 • Fax :+91 22 3021 8595 Other Offices: 44 - 46, 4th Floor, "C" Wing, Mittal Court, Nariman Point, Mumbai - 400 021, India. Tel.: +91 22 4510 9700 • Fax : +91 22 45109722. URL : www.cas.ind.in

Branch : Bengaluru



CHATURVEDI

Chartered Accountants

6. We have complied with the relevant applicable requirements of the Standard on Quality Control (SQC) 1, Quality Control for Firms that Perform Audits and Reviews of Historical Financial Information, and Other Assurance and Related Services Engagements. Further our examination did not extend to any other parts and aspects of a legal and proprietary nature in the aforesaid Scheme.

Opinion

7. Based on our examination and according to the information and explanations provided to us by the Management of the Company, we are of the opinion that the accounting treatment proposed by the Company as specified in Clause 8 of the Scheme and reproduced in the Statement, is in compliance with SEBI (Listing Obligations and Disclosure Requirements) Regulations, 2015 and circulars issued there under and all the applicable Indian Accounting Standards notified by the Central Government under the Companies Act 2013 and other Generally Accepted Accounting Principles in India.

Restriction on Use

8. The Certificate is provided to the Board of Directors of the Company in connection with the scheme and for the purpose of onward submission to the National Company Law Tribunal and other regulatory authorities including Securities and Exchange Board of India and Stock Exchange(s) and should not be used by any other person/ authority or for any other purpose. Accordingly, we do not accept or assume any liability or any duty of care for any other purpose or to any other person to whom this Certificate is shown or into whose hands it may come without our prior consent in writing.

For Chaturvedi & Shah

Chartered Accountants Firm Registration No. 101720W

CITAL FIL

Vijay Napawaliya Partner Membership No.: 109859

Place: Mumbai Date: May 24, 2018 Encl.: as above



Chartered Capital And Investment Ltd.

418-C, "215 ATRIUM", Andheri Kurla Road, Andheri (East), Mumbai-400 093. Tel.: 91-22-6692 4111 / 6222 • Website : www.charteredcapital.net

Date: May 24, 2018

To,

The Board of Directors Arshiya Limited 302, Ceejay House, Level - 3 Shiv Sagar Estate, F- Block Dr. Annie Besant Road Worli, Mumbai – 400018, India To, The Board of Directors Arshiya Rail Infrastructure Limited 302, Ceejay House, Level - 3 Shiv Sagar Estate, F- Block Dr. Annie Besant Road Worli, Mumbai – 400018, India

Dear Sirs,

Subject: Fairness Opinion on Share Entitlement Ratio Report for the proposed demerger of Domestic Business Undertaking ('Demerged Undertaking') of Arshiya Limited ('Arshiya' or 'Demerged Company') into Arshiya Rail Infrastructure Limited ('ARIL' or 'Resulting Company').

I. BACKGROUND

We M/s. Chartered Capital and Investment Limited ('Chartered Capital' or 'CCIL'), SEBI registered Merchant Banker, having SEBI Permanent Registration No. INM000004018 have been appointed by the Arshiya Limited ('Arshiya' or 'Demerged Company') and Arshiya Rail Infrastructure Limited ('ARIL' or 'Resulting Company') to provide a fairness opinion on the Share Entitlement Ratio Report by M/s. ZADN & Associates, Chartered Accountants having their office at 4th Floor, Shreeniwas House, Hazarimal Somani Marg, Next to MTNL Exchange Building, Fort, Mumbai-400001, who were the appointed for issuing report on the Share Entitlement Ratio of equity shares of the Resulting Company to be issued to the shareholders of the Demerged Company in connection with the proposed demerger of Domestic Business Undertaking ('Demerged Undertaking') of Arshiya Limited in the proposed Scheme of Arrangement between Arshiya Limited and Arshiya Rail Infrastructure Limited, under sections 230 to 232 read with section 66 and section 52 and other applicable provisions of the Companies Act, 2013 hereinafter referred to as 'the Proposed Scheme' or 'the Scheme' or 'the Scheme of Arrangement'.



Regd. Office : 711, Mahakant, Opp. V. S. Hospital, Ellisbridge, Ahmedabad - 380 006. CIN NO: L45201GJ1986PLC008577



The fairness opinion has been issued for the proposed demerger of Domestic Business Undertaking of Arshiya Limited into Arshiya Rail Infrastructure Limited pursuant to the requirement and in compliance with SEBI (Listing Obligations and Disclosure Requirement) Regulation, 2015 read with SEBI Circular No. CIR/CFD/CMD/16/2015 dated November 30, 2015, SEBI Circular No. CFD/DIL3/CIR/2017/21 dated March 10, 2017, SEBI Circular No. CFD/DIL3/CIR/2017/26 dated March 23, 2017, SEBI Circular No. CFD/DIL3/CIR/2017/105 dated September 21, 2017 and SEBI Circular No. CFD/DIL3/CIR/2018/2 dated January 03, 2018.

II. COMPANIES THAT ARE PARTY TO THE PROPOSED SCHEME

- 2.1 Arshiya Limited is a listed public limited company incorporated on July 3, 1981 under the Companies Act, 1956 bearing CIN L93000MH1981PLC024747 and its registered office is situated at 302, Level 3, Ceejay House, F Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai 400 018. Equity shares of Arshiya are listed on BSE Limited ('BSE') and National Stock Exchange of India Limited ('NSE') in India.
- 2.2 Arshiya, is a flagship company of Arshiya Group having interests in the business of developing Free Trade Warehousing Zones ('FTWZs') and Domestic Warehousing Areas ('DWAs') to improve logistics infrastructure in India.
- 2.3 Arshiya Rail Infrastructure Limited ("ARIL") is a unlisted public company incorporated on April 7, 2008 under the Companies Act, 1956 bearing CIN -U93000MH2008PLC180907 and its registered office is situated at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai – 400 018.
- 2.4 ARIL is currently engaged in the business of providing Private Freight Terminal (PFT) and Rail Transportation Services (Rail). ARIL is also engaged in container train business and holding Category – I license which allows ARIL to operate on Indian Rail network on pan India basis both domestic and exim traffic.
 - 2.5 As on date, ARIL is a wholly owned subsidiary of Arshiya Limited as the entire paid up equity share capital is held by Arshiya and its nominees.
 - 2.6 The management of Arshiya Group has informed that as part of the overall group reorganization, the management has filed a Scheme of Amalgamation of Arshiya Industrial & Distribution Hub Limited ('AIDHL') and Arshiya Transport and Handling Limited ('ATHL') with Arshiya Rail Infrastructure Limited ('ARIL' or 'Resulting Company') with National Company Law





Tribunal at Mumbai and is awaiting their approval. The appointed date for the proposed merger is October 1, 2015 as stated in the Scheme of Amalgamation.

- 2.7 AIDHL and ATHL are 100% subsidiaries of Arshiya and hence fellow subsidiaries of ARIL.
- 2.8 As per the Preamble paragraph of the Proposed Scheme, the Proposed Scheme is also conditional upon the aforesaid merger scheme as mentioned in point 2.6 above being effective first. In case the scheme as mentioned in point 2.6 above is withdrawn or ARIL ceases to be a wholly owned subsidiary of Arshiya as on the Record date, the proposed scheme will also be withdrawn.

III. <u>INFORMATION RECEIVED AND RELIED UPON BY US FOR THE</u> FAIRNESS OPINION

- 3.1 We, CCIL, have prepared the **Fairness Opinion** on the basis of the following information received from the management of the Companies:
 - a) Draft Scheme of Arrangement between Arshiya and ARIL proposed to be adopted at the Board meetings of both the companies;
 - b) Perused the Report on Share Entitlement Ratio given by M/s. ZADN & Associates, Chartered Accountants dated May 24, 2018.
 - c) Scheme of Amalgamation of Arshiya Industrial & Distribution Hub Limited ('AIDHL') and Arshiya Transport and Handling Limited ('ATHL') with Arshiya Rail Infrastructure Limited ('ARIL');
 - d) Audited financial statements of the Demerged and Resulting Company for the year ended March 31, 2017;
 - e) Current Shareholding pattern of the Demerged Company and Resulting Company;
 - f) Desired capital structure and equity shares of ARIL proposed to be issued to the shareholders of Arshiya on demerger of the Domestic Business Undertaking into ARIL;
 - g) Such other information and explanations that have been provided to us by the management of Arshiya and ARIL;
 - h) Relevant Information in public domain.
- 3.2 This Report is issued on the understanding that Arshiya Limited and Arshiya Rail Infrastructure Limited has drawn our attention to all the relevant





matters, of which they were aware of, concerning the respective company's financial position and businesses, which may have an impact on this Report.

3.3 Our work does not constitute an audit or certification of the historical financial statements of Arshiya Limited and Arshiya Rail Infrastructure Limited, including their working results referred to in this Report. Accordingly, we are unable to and do not express any opinion on the accuracy of any financial information referred to in this Report. We assume no responsibility for any errors in the information submitted by the Managements and their impact on the present exercise.

IV. DISCLAIMER

- 4.1 We have not considered any finding made by other external agencies in carrying out this work.
- 4.2 This report^{*} is intended solely for the limited purpose mentioned earlier and should not be regarded as a recommendation to the investors to invest in Arshiya Limited and Arshiya Rail Infrastructure Limited or deal in any form in the securities of the Companies and should also not be considered as a final equity value of the Companies.
- 4.3 Our report does not, in any way, guarantee that the equity shares of Companies will continue to remain at the price on which the valuation of the shares takes place.
- 4.4 Our report is for the benefit of and confidential use by the Companies. This report is not meant for save and except as specified above, under any Indian or foreign law, statute, act guideline or similar instruction. The Companies are prohibited from using this report other than for its sole limited purpose and not to make a copy of this report available to any party other than those required by statute for carrying out the limited purpose of this report.
- 4.5 We have not made an appraisal or independent valuation of any of the assets or liabilities of any of the Companies and have not conducted an audit or due diligence or reviewed/ validated the financial data except what is provided for in the Valuation Report and financial data provided to us by the Company or Valuer.
- 4.6 This Certificate may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same.
- 4.7 During the course of our work, we have relied upon assumptions made by management of the companies. These assumptions require the exercise of judgement and are subject to uncertainties.





4.8 In no circumstances whatsoever, will Chartered Capital and Investment Limited, its Directors and employees, accept any responsibility of liability towards any third party for consequences arising out of the use of this report.

V. OPINION ON SHARE ENTITLEMENT RATIO REPORT

The proposed demerger of Domestic Business Undertaking of Arshiya into ARIL will be value neutral to the equity shareholders of Arshiya, there will be no impact on the economic and beneficial interest of the equity shareholders of Arshiya and it would not have any impact on the ultimate value of the equity shareholders of Arshiya. Based on the information, material, data made available to us, including the Share Entitlement Ratio Report, to best of our knowledge and belief, Share Entitlement Ratio suggested by M/s. ZADN & Associates, Chartered Accountants under the Scheme, i.e.

"1 (One) fully paid equity share of Face Value INR 2 (Rupees Two) each of ARIL is to be issued for every 2 (Two) fully paid equity shares of Face Value INR 2 (Rupees Two) each held in Arshiya is Fair."

Thanking you,

Yours faithfully,

For and on behalf of, For Chartered Capital and Investment Limited

attan

Amitkumar Gattani Assistant Vice President



Arshiya Limited Balance Sheet as at 31st March, 2019

and the second se	1 1		(Rs. in Lakh
Particulars	Notes	As at 31st March, 2019	As at 31st March, 2018
ASSETS			
Non-Current Assets			
(a) Property, Plant and Equipment	5	73,858.36	94,138.63
(b) Capital Work-in-Progress		76.02	54,155.00
(c) Intangible Assets	6	845.86	1,235.9
(d) Intangible Assets Under Development		60.00	1,235.9
(e) Financial Assets		80.00	
(i) Investments	7	1,34,680.02	1 00 040 0
(i) Loans	8	1,34,080.02	1,32,018.0
	9		1,731.4
(iii) Trade Receivables		6,061.50	0.070.0
(f) Other Non-Current Assets	10	3,320.34	2,273.8
		2,20,634.24	2,31,397.90
Current assets	and the second s	Server and	
(a) Inventories	11	16,505.97	
(b) Financial Assets	10.00	and the second se	
(i) Trade Receivables	12	814.64	764.6
(ii) Cash and Cash Equivalents	13	5.86	135.6
(iii) Bank Balances Other than (ii) above	14	15.17	0.04
(iv) Loans	15	30,327.14	33,279.9
(v) Other Financial Assets	16	4,082,95	1,848.7
(c) Other Current Assets	17	2,134.00	2,319.6
At the Charles of the		53,885.73	38,348.6
Total Assets	L E	2,74,519.97	2,69,746.5
EQUITY AND LIABILITIES	1 [
Equity			
(a) Equity Share capital	18	4,872.29	4,564,34
(b) Other Equity	19	1,66,643.28	1,60,350,3
(b) other Equity	10	1,71,515.57	1,64,914.6
Liabilities		1,71,515,57	1,04,914.0
Non Current Liabilities	1 1		
	+ - +		
(a) Financial Liabilities	25	50,000,000	
(i) Borrowings	20	60,267.20	68,839.8
(ii) Other Financial Liabilities	21	1,612.72	2,191,6
(b) Provisions	22	118.93	151.0
		61,998.85	71,182.4
Current Liabilities			
(a) Financial Liabilities		Sa. 24	
(i) Borrowings	23	9,261.16	10,488.8
(ii) Trade Payables	24		
Micro and Small Enterprises	1.1	37.88	0.8
Others	1.22	584.91	581.6
(iii) Other Financial Liabilities	25	29,679,14	21,506.0
(b) Other Current Liabilities	26	1.434.11	1,062.4
(c) Provisions	27	8.35	9.5
		41,005.55	33,649.4
T . 1 (T . 1) (T) (T . 1) (T		274 640 07	2 60 746 5
Total Equity and Liabilities	a more thanks	2,74,519.97	2,69,746.5

1 to 67

For and on behalf of the Board of Directors of Arshiya Limited

Vijay Napawaliya Partner Membership Number: 109859

Notes to the financial statements

Firm Registration Number 101720W/W100355

RVEDI & SA

As per our report of even date For Chaturvedi & Shah LLP

Chartered Accountants

Place: Mumbai Date: 27th May, 2019 h-ttd

Ajay S Mittal Chairman and Managing Director DIN: 00226355

Ashishkumar Bairagra Independent Director

DIN: 00049591 Sawta Dalal Company Secretary

S. Maheshwari Chief Financial Officer

har

Joint Managing Director

Archana A Mittal

DIN: 00703208

12 Sodaw

Dinesh Kumar Sodani VP: Accounts & Finance

Arshiya Limited

Statement of Profit and Loss for the year ended 31st March, 2019

	En la	
Notes	Year Ended 31st March, 2019	Year Ended 31st March, 2018
28	13,139.98	8,542.02
29	2,192.48	1,020.09
	15,332.46	9,562.11
1. 2. 1		
30	2,583.34	
31	87.63	320.61
32	1,720.28	1,456.61
33	11,236.53	13,761.94
34	1,482.22	2,091.67
35	1,038.20	1,332.46
-	18,148.20	18,963.29
	(2,815.74)	(9,401.18)
36	700.75	(13,296.84)
	(3,516.49)	3,895.66
59		
1		
	-	
	(3,516.49)	3,895.66
n		
	8.11	(2.69)
	8.11	(2.69)
	(3,508.38)	3,892.97
58		2.13
	28 29 30 31 32 33 34 35 36 59	31st March, 2019 28 13,139.98 29 2,192.48 30 2,583.34 31 87.63 32 1,720.28 33 11,236.53 34 1,482.22 35 1,038.20 18,148.20 (2,815.74) 36 700.75 (3,516.49) 59 59 - (3,516.49) - 59 - (3,516.49) - 59 - (3,516.33) - (3,516.33) -

Notes to the financial statements As per our report of even date

For Chaturvedi & Shah LLP

Firm Registration Number 101720W/W100355

MUMEA

DAT

Chartered Accountants

1 to 67

For and on behalf of the Board of Directors of Arshiya Limited

Ajay S Mittal Chairman and Managing Director DIN: 00226355

Ashishkumar Bairagra Independent Director DIN: 00049591

Savita Dalal Company Secretary

Archana A Mittal Joint Managing Director DIN: 00703208

S. Maheshwari **Chief Financial Officer**

sodau

Dinesh Kumar Sodani VP: Accounts & Finance

Vijay Napawaliya

Partner Membership Number: 109859

Place: Mumbal Date: 27th May, 2019

Arshiya Limited Statement of changes in Equity for the year ended 31st March, 2019

the second and a second manage function of the second seco	
Particulars	Rs. in Lakh
Equity Shares of Rs. 2 each issued, subscribed and paid up	
As at 31st March, 2017	3,123.59
Issue of Equity Shares	1,440.75
As at 31st March, 2018	4,564.34
Issue of Equity Shares	307.95
As at 31st March 2010	A 872 29

B. Other Equity (Refer Note No. 19)

(Rs. in Lakh)

and the second sec			Equity Component of	and the second	100 Mar 100			
Particulars	Share application money pending allotment	Money received against share warrants	0% Optionally Convertible Preference shares (OCRPS)	Amalgamation Reserve	Securities Premium Account	General Reserve	Retained Eamings	Total
Balances as on 31st March, 2017	18,766.71		88,620.84	124.80	79,617.43	940.18	(68,332.91)	1,19,737.05
Profit/(loss) for the year Other comprehensive income	, .			1		10	3,895.66 (2.69)	3,895.66 (2.69)
Total comprehensive income for the year	•	•			•		3,892.97	3,892.97
On issue of equity shares	(18,766.71)		(41,068.97)	4	95,278,86	eit.	ł	35,443,18
Money received against share warrants		860.25		5				860.25
Transaction costs on issue of equity shares				3	(37.62)		1	(37.62)
Fair value of financial liabilities	•	*	4	4	4	1	64.05	64.05
Conditional lease rent	•	•	*	4		9	390.41	390.41
Balances as at 31st March, 2018		860,25	47,551.87	124.80	1,74,858.67	940.18	(63,985.48)	1,60,350.30
Profit/(loss) for the year	4	1		ť		15	(3,516.49)	(3,516.49)
Other comprehensive income		9	3	-	-1		8.11	8.11
Total comprehensive income for the year		•				•	(3,508.38)	(3,508.38)
Money received against share warrants	-3	15.00		ţ	r		ł	15.00
On issue of equity shares		(875.25)	(47,551.87)	•	58,213.49	-		9,786.37
Balances as at 31st March, 2019				124.80	2,33,072.16	940.18	(67,493.86)	1,66,643.28

For and on behalf of the Board of Directors of Arshiya Limited

\$ 5 3 Partner Membership Number: 109859 A MUMBA For Chaturvedi & Shah LLP Chartered Accountants Firm Registration Number 101720W/W100355

Seper aline

Ajay S Mittal Chairman and Managing Director DIN: 00226355 , S. Maheshwari Chief Financial Officer

* SINVING

TERED NO

Place: Mumbai Date: 27th May, 2019

Anlandute Archana A Mittal Joint Managing Director DIN: 00703208

3

Savita Dalal Company Secretary

Ashishkumar Bairagra Independent Director DIN: 00049591

Sedau

Arshiya Limited Cash Flow Statement for the year ended 31st March, 2019

Particulars		and strange which a subscription of the second	(Rs. in Lakh Year Ended
		3161 March 2019	st March, 2018
Cash flow from operating activities		1.000	
Profit/(Loss) before tax		(3,516.49)	3,895.6
Adjustments for		and the second se	
Sundry balances written back (net)		(206.27)	(175.8
Driscarding/written off of Property, plant and equipment and Intangible assets		166.34	
Gain on monetization of Property, Plant and Equipment (Refer Note No. 55)		-	(15,633.2
Profit on disposal of Property, plant and equipment (net)		(0.33)	
Bad debts		3.16	
Allowance for doubtful debts		8.68	7.3
Settlement of claims		700.75	(2,001.7
Loss on sale of investment in subsidiary (Refer Note No. 54.2)			4,338.1
Depreciation and amortization expense		1,482.22	2,091.6
Finance costs		11,236.53	13,761.9
Unwinding interest income on loan to subsidiaries		(420.02)	(185.5
Interest income on fixed deposits		(0.19)	
Liability Component of Compound Financial Instruments (OCRPS)		(653.17)	(639.3
Financial guarantees income		(898.96) 12.81	(639.3
Foreign exchange difference (net)	-	7,915.06	5,443.8
Operating profit before working capital changes		7,915.00	5,445.0
Adjustments for		2,583.34	
Change in Inventories		(9,356.47)	(2,183.1
(Increase) in financial and other assets		(9,356.47) (844.68)	(11,630.3
(Decrease) in financial and other liabilities		297.25	(8,369.6
Cash generated from operations		(118.27)	(705.1
Direct taxes paid (net of refunds) Net cash flow from operating activities	(A)	178.98	(9,074.7
Net cash now noin operating activities	~	110.00	10,014.1
Cash flow from investing activities			
Purchase of property, plant and equipments		(49.78)	(89.7
Purchase of intangible assets			(1,120.0
Purchase of Capital work in progress and Intangible assets under development		(45.37)	
Proceeds from sale of property, plant and equipment	1	2.38	60.0
Proceeds from monetization of property, plant and equipments			43,400.0
Capital advance		(1,020.41)	(88.2
Investment made in equity shares		(23.00)	(155.5
Sale of Investment in subsidiaries			330.8
Loans given to subsidiaries (net)		1,892.10	(33,278.2
Interest income on fixed deposits		0.19	
Net cash flow from investing activities	(B)	756.11	9,059.1
Cash flow from financing activities			
Cash flow from financing activities			15.268.3
Issue of Equity shares (including Security Premium)		15.00	860.2
Money received against share warrants		2,611.82	3,200.0
Proceeds from non-current borrowings			(18,071.0
Repayment of non-current borrowings		(1,248.03)	
Short-term borrowings (Net)		(1,227.65)	5,543.6
Unpaid Dividend transfer to IEPF A/c		(0.04)	
Interest paid		(1,200.89)	(6,727.
Net cash flow from financing activities	(C)	(1,049.79)	74.
Net Increase/(Decrease) in cash and cash equivalents (A+B+C)		(114.70)	58.
Cash and cash equivalents at the beginning of the year		135.73	77.2
Cash and cash equivalents at the end of the year (Refer Note No. 13 and 14)		21.03	135.7





Arshiya Limited Cash Flow Statement for the year ended 31st March, 2019

Particulars	Long term Borrowings	Short term Borrowings
As at 1st April, 2018	84,007.14	10,488.81
Add: Transaction cost	305.77	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Less: Conversion of Liability Component of Compound Financial Instruments (OCRPS) into Equity	(10,342.61)	in the second
Add: Non cash items	845.00	
Add/Less: Cash flow (net)	1,363,79	(1,227.65)
As at 31st March, 2019	76,179,09	9,261.16

Notes:

1. Bracket indicates cash outflow.

2. The above cash flow statement has been prepared under the "Indirect Method" as set out in IND AS 7 on Statement of Cash Flow

3. Classification of Land from Property, plant and equipment to invetories has been considered as non cash items (Refer note no.44)

Notes to the financial statements

Jangualin.

Firm Registration Number 101720W/W100355

JEDI & SA

MUMBA

ARTERED AC

CHA

As per our report of even date For Chaturvedi & Shah LLP

Chartered Accountants

Vijay Napawaliya

1 to 67

For and on behalf of the Board of Directors of Arshiya Limited

Ajay S Mittal Chairman and Managing Director DIN: 00226355

Ashishkumar Bairagra Independent Director DIN: 00049591

Savita Dalal Company Secretary

Archana A Mittal Joint Managing Director DIN: 00703208

S. Maheshwari Chief Financial Officer

Dinesh Kumar Sodani VP: Accounts & Finance

Partner Membership Number: 109859

Place: Mumbai Date: 27th May, 2019

Arshiya Limited Notes to the financial statements for the year ended 31st March, 2019

1 Corporate Information

Arshiya Limited (the Company) is a unified supply chain and integrated logistics infrastructure solution provider and is engaged in the business of Free Trade and Warehousing Zone (FTWZ), Domestic Warehousing Zone and value added services along with development, operations and maintenance of FTWZ.

These statements comprises of financial statements of Arshiya limited (CIN : L93000MH1981PLC024747) for the year ended 31st March, 2019. The Company is a public company domiciled in India and is incorporated on 3rd July, 1981 under the provisions of the Companies Act applicable in India. The registered office of the company is located at 302, Level 3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400 018.

FTWZ's are developed under the provisions of Special Economic Zone Act, 2005 and the Special Economic Zone Rules, 2006.

The Company's equity shares are listed on the Bombay Stock Exchange (BSE) and National Stock Exchange (NSE) of India.

The Financial statements of the Company for the year ended 31st March, 2019 were approved and adopted by the Board of Directors in their meeting held on 27th May, 2019.

2 Basis of preparation of financial statements

The financial statements are prepared in accordance with Indian Accounting Standards ("Ind AS") notified by the Ministry of Corporate Affairs ("MCA") pursuant to the Section 133 of the Companies Act, 2013 ("the Act") read with of the Companies (Indian Accounting Standards) Rules 2015, (as amended) and other relevant provisions of the Act.

The financial statements are prepared on a historical cost convention basis, except for certain financial assets and liabilities measured at fair value.

The financial statements are presented in Indian Rupees (Rs.), which is the Company's functional and presentation currency and all values are rounded to the nearest lakh as per the requirement of schedule III, unless when otherwise indicated.

3 Significant Accounting Policies

3.1 Property, Plant and Equipment

Property, plant and equipment are carried at cost less accumulated depreciation and accumulated impairment losses, if any. Cost includes purchase price, borrowing cost and any cost directly attributable to the bringing the assets to its working condition for its intended use.

Depreciation on the property, plant and equipment is provided using straight line method over the useful life of assets as specified in schedule II to the Companies Act, 2013. Leasehold improvements are amortised over the period of lease. Depreciation on property, plant and equipment which are added / disposed off during the year, is provided on pro-rata basis with reference to the date of addition / deletion. Freehold land is not depreciated and under the previous GAAP land was revalued.

The assets' residual values, useful lives and method of depreciation are reviewed at each financial year end and are adjusted prospectively, if appropriate.

Capital work-in-progress includes cost of property, plant and equipment under installation / under development as at the balance sheet date.

Property, plant and equipment are eliminated from financial statement, either on disposal or when retired from active use. Profits / losses arising in the case of retirement / disposal of property, plant and equipment are recognised in the statement of profit and loss in the year of occurrence.

The Company has opted to continue with the carrying values of all of its property. Plant and Equipment as recognised in the previous GAAP financial statements as deemed cost at the transition date i.e. 1st April, 2016.

3.2 Intangible Assets

Intangible assets are carried at cost less accumulated amortisation and accumulated impairment losses, if any. Cost includes expenditure that is directly attributable to the acquisition of the intangible assets.

Identifiable intangible assets are recognised when it is probable that future economic benefits attributed to the asset will flow to the Company and the cost of the asset can be reliably measured.

Cost of Enterprise Resource Planning (ERP) software including expenditure on implementation is to be amortised over a period of ten years based on management's estimate of useful life over which economic benefits will be derived from its use.

Computer softwares are capitalised at the amounts paid to acquire the respective license for use and are amortised over the period of SHUPPe to seven years. The assets' useful lives are reviewed at each financial year end.

Arshiya Limited

Notes to the financial statements for the year ended 31st March, 2019

Trademark are amortised over the period of ten (10) years.

Gains or losses arising from derecognition of an intangible asset are measured as the difference between the net disposal proceeds and the carrying amount of the asset and are recognised in the statement of profit and loss when the asset is derecognised.

Intangible assets under development includes cost of computer software under installation / under development as at the balance sheet date.

The Company has opted to continue with the carrying values of all of its Intangible assets as recognised in the previous GAAP financial statements as deemed cost at the transaction date i.e. 1st April, 2016.

3.3 Leases

A lease is classified at the inception date as a finance lease or an operating lease. A lease that transfers substantially all the risks and rewards incidental to ownership to the Company is classified as a finance lease.

The Company as a lessee

(a) Finance lease

Assets acquired under finance lease are capitalized and the corresponding lease liability is recognised at lower of the fair value of the leased assets and the present value of minimum lease payments at the inception of the lease. Initial costs directly attributable to lease are recognised with the asset under lease.

(b) Operating lease

Lease of assets under which all risks and rewards of ownership are effectively retained by the lessor are classified as operating lease. Lease payments under operating lease are recognised as expenses on accrual basis in accordance with the respective lease agreements.

The Company as a lessor

(a) Finance lease

When assets are leased out under a finance lease, the present value of the minimum lease payments is recognised as a receivable. The difference between the gross receivable and the present value of the receivable is recognised as unearned finance income. Lease income is recognised over the term of the lease using the net investment method before tax, which reflects a constant periodic rate of return.

The lessor derecognises the leased assets and recognises the difference between the carrying amount of the leased assets and the finance lease receivable in the statement of Profit and Loss when recognising the finance lease receivable. This gain or loss is presented in the statement of Profit and Loss in the same line item as that in which the lessor presents gains or losses from sale of similar assets.

(b) Operating lease

Rental income from operating leases is recognised in the statement of profit and loss on a straight line basis over the term of the relevant lease unless another systematic basis is more representative of the time pattern in which economic benefits from the leased assets is diminished.

Initial indirect costs incurred in negotiating and arranging as operating lease are added to carrying value of the leased asset and recognised on a straight line basis over the lease term.

3.4 Inventories

Inventories are measured at lower of cost and net realisable value. Inventory comprises of cost of land and incidental cost thereto.

3.5 Cash and cash equivalents

Cash and cash equivalent in the balance sheet comprise cash at banks, cash on hand and short-term deposits with an original maturity of three months or less, which are subject to an insignificant risk of changes in value.

For the purpose of the statement of cash flows, cash and cash equivalents consist of cash on hand and short-lerm deposits, as defined above, net of outstanding bank overdrafts as which are considered an integral part of the Company's cash management.

3.6 Impairment of assets

An asset is considered as impaired when at the date of Balance Sheet, there are indications of impairment and the carrying amount of the asset, or where applicable, the cash generating unit to which the asset belongs, exceeds its recoverable amount (i.e. the higher of the net asset selling price and value in use) The carrying amount is reduced to the recoverable amount and the reduction is recognized as an impairment loss in the statement of profit and loss. The impairment loss recognized in the prior accounting period is reversed if there has been a change in the estimate of recoverable amount. Post impairment, depreciation is provided on the revised carrying value of the impaired asset over its remaining useful life.



J ss V

Arshiya Limited Notes to the financial statements for the year ended 31st March, 2019

3.7 Financial instruments - initial recognition, subsequent measurement and impairment

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity.

Financial assets -Initial recognition and measurement

All financial assets are initially recognized at fair value. Transaction costs that are directly attributable to the acquisition or issue of financial assets, which are not at fair value through profit or loss, are adjusted to the fair value on initial recognition. Financial assets are classified, at initial recognition, as financial assets measured at fair value or as financial assets measured at amortised cost.

Financial assets - Subsequent measurement:

For the purpose of subsequent measurement, financial assets are classified in two broad categories:-

- a) Financial assets at fair value
- b) Financial assets at amortised cost

Where assets are measured at fair value, gains and losses are either recognised entirely in the statement of profit and loss (i.e. fair value through profit or loss), or recognised in other comprehensive income (i.e. fair value through other comprehensive income).

A financial asset that meets the following two conditions is measured at amortised cost (net of any write down for impairment) unless the asset is designated at fair value through profit or loss under the fair value option.

- (a) Business model test: The objective of the Company's business model is to hold the financial asset to collect the contractual cash flow.
- (b) Cash flow characteristics test: The contractual terms of the financial asset give rise on specified dates to cash flow that are solely payments of principal and interest on the principal amount outstanding.

A financial asset that meets the following two conditions is measured at fair value through other comprehensive income unless the asset is designated at fair value through profit or loss under the fair value option.

- (a) Business model test: The financial asset is held within a business model whose objective is achieved by both collecting contractual cash flow and selling financial assets.
- (b) Cash flow characteristics test: The contractual terms of the financial asset give rise on specified dates to cash flow that are solely payments of principal and interest on the principal amount outstanding.

Financial assets - Equity Investment in subsidiaries

Investments in subsidiaries are recognised at cost as per Ind AS 27 separate financial statements.

Transition to Ind AS

Upon first-time adoption of Ind AS, the Company has elected to continue with the carrying value of all of its investments in subsidiaries as at April 1, 2016 measured as per the previous GAAP and use that carrying value as the deemed cost of investments in subsidiaries.

In respect of interest free loans given to subsidiaries, the difference between the loan amount and its fair value is treated as further investment by the Company in the respective subsidiaries. Where financial guarantees in relation to loans of subsidiaries are provided for no compensation, the fair values are added to investment by the Company in respective subsidiaries.

The Company has accounted for its equity investment in subsidiaries at cost.

Financial assets - Derecognition

A financial assets (or, where applicable, a part of a financial asset or part of a group of similar financial assets) is primarily derecognised (i.e. removed form the Company's statement of financial position) when

- (a) The rights to receive cash flows from the asset have expired, or
- (b) The Company has transferred its rights to receive cash flow from the asset.

Financial liabilities - Initial recognition and measurement:

The financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

Effective interest method

The effective interest method is a method of calculating the amortised cost of a financial instrument and of allocating interest income of expenses over the relevant period. The effective interest rate is the rate that exactly discounts future cash receipts or payments through the expected life of financial instruments, or where appropriate, a shorter period.

7 V h

Arshiya Limited Notes to the financial statements for the year ended 31st March, 2019

Financial liabilities - Subsequent measurement

Financial liabilities are subsequently carried at amortized cost using the effective interest method. For trade and other payables maturing within one year from the balance sheet date, the carrying amounts are approximate at their fair value due to the short maturity of these instruments.

Financial Liabilities - Financial guarantee contracts

Financial guarantee contracts issued by the Company are those contracts that require a payment to be made to reimburse the holder for a loss it incurs because the specified debtor fails to make a payment when due in accordance with the terms of a debt instrument. Financial guarantee contracts are recognised initially as a liability at fair value, adjusted for transaction costs that are directly attributable to the issuance of the guarantee. Subsequently, the liability is measured at the higher of the amount of loss allowance determined and the amount recognised less cumulative amortisation.

Financial Liabilities - Derecognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another, from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit and loss.

Compound Instruments

An issued financial instrument that comprises of both the liability and equity components are accounted as compound financial instruments. The fair value of the liability component is separated from the compound instrument and the residual value is recognised as equity component of other financial instrument. The liability component is subsequently measured at amortised cost, whereas the equity component is not remeasured after initial recognition. The transaction costs related to compound instruments are allocated to the liability components in the proportion to the allocation of gross proceeds. Transaction costs related to equity component is recognised directly in equity and the cost related to liability component is included in the carrying amount of the liability component and amortised using effective interest method.

3.8 Provisions, Contingent Liabilities, Contingent Assets and Commitments:

Provisions are recognised when the Company has a present obligation (legal or constructive) as a result of a past event. It is probable that an outflow of resources embodying economic benefits will be required to settle the obligation and a reliable estimate can be made of the amount of the obligation. If the effect of the time value of money is material, provisions are discounted using equivalent period government securities interest rate. Unwinding of the discount is recognised in the statement of profit and loss as a finance cost. Provisions are reviewed at each balance sheet date and are adjusted to reflect the current best estimate.

Contingent liabilities are disclosed when there is a possible obligation arising from past events, the existence of which will be confirmed only by the occurrence or non-occurrence of one or more uncertain future events not wholly within the control of the Company or a present obligation that arises from past events where it is either not probable that an outflow of resources will be required to settle or a reliable estimate of the amount cannot be made. Information on contingent liability is disclosed in the Notes to the Financial Statements. Contingent assets are not recognised. However, when the realisation of income is virtually certain, then the related asset is no longer a contingent asset, but it is recognised as an asset.

3.9 Dividend Distribution

Annual dividend distribution to the shareholders is recognised as a liability in the period in which the dividends are approved by the shareholders. Any interim dividend paid is recognised on approval by Board of Directors. Dividend payable and corresponding tax on dividend distribution is recognised directly in other equity.

3.10 Revenue recognition

Revenue is recognized upon transfer of control of goods or rendering of services to customers in an amount that reflects the consideration which the Company expects to receive in exchange for those goods or services.

Generally, control is transfer upon shipment of goods to the customer or when the goods is made available to the customer, provided transfer of title to the customer occurs and the Company has not retained any significant risks of ownership or future obligations with respect to the goods shipped.

Revenue is measured at the amount of consideration which the company expects to be entitled to in exchange for transferring distinct goods or services to a customer as specified in the contract, excluding amounts collected on behalf of third parties (for example taxes and duties collected on behalf of the government). Consideration is generally due upon satisfaction of performance obligations and a receivable is recognized when the it becomes unconditional.





Arshiya Limited

Notes to the financial statements for the year ended 31st March, 2019

Revenue are recognized as the related services are performed and revenue from the end of the last invoicing to the reporting date is recognized as unbilled revenue.

Revenues in excess of invoicing are classified as contract assets (which we refer as unbilled revenue) while invoicing in excess of revenues are classified as contract liabilities (which we refer to as unearned revenues).

Trade Receivables

A receivable represents the Company's right to an amount of consideration that is unconditional.

Contract liabilities

A contract liability is the obligation to transfer of services to a customer for which the Company has received consideration (or an amount of consideration is due) from the customer. If a customer pays consideration before the Company transfers services to the customer, a contract liability is recognised when the payment is made Contract liabilities are recognised as revenue when the Company performs under the contract.

- (i) Income from allotment of warehousing spaces and open yard area for use are recognised for the period the material is lying in area as per agreed terms.
- (ii) Revenue from valued services and other activities is recognised when related services are performed as per the contractual terms.
- (iii) Income from Business Conducting Fees shall be recognised as per contractual terms.
- (iv) Revenue from lease of land is recognised as per contract terms agreed between the parties.
- (v) Interest income from a financial asset is recognised when it is probable that the economic benefits will flow to the Company and the amount of income can be measured reliably. Interest income is accrued on a time basis, by reference to the principal outstanding and at the effective interest rate applicable, which is the rate that exactly discounts estimated future cash receipts through the expected life of the financial asset to that asset's net carrying amount on initial recognition.
- (vi) Export benefits under Foreign Trade Policy are recognised when utilized.
- (vii) Dividend income is recognised when the Company's right to receive the payment is established, which is generally when shareholders approve the payment of dividend.

3.11 Foreign currency reinstatement and translation:

Transactions in foreign currencies are initially recorded by the Company at rates prevailing at the date of the transaction. Subsequently monetary items are translated at closing exchange rates as on balance sheet date and the resulting exchange difference recognised in statement of profit and loss. Differences arising on settlement of monetary items are also recognised in statement of profit and loss.

Non-monetary items that are measured in terms of historical cost in a foreign currency are translated using the exchange rates at the dates of the transaction. Non-monetary items carried at fair value that are denominated in foreign currencies are translated at the exchange rates prevailing at the date when the fair value was determined. The gain or loss arising on translation of non-monetary items measured at fair value is treated in line with the recognition of the gain or loss on the change in fair value of the item (i.e., translation differences on items whose fair value gain or loss is recognised in OCI or profit or loss are also recognised in OCI or profit or loss are also recognised in OCI or profit or loss.

Foreign exchange differences regarded as an adjustment to borrowing costs are presented in the statement of profit and loss, within finance costs. All other finance gains / losses are presented in the statement of profit and loss on a net basis.

3.12 Employee benefits

Short term employee benefits are recognized as an expense in the statement of profit and loss of the year in which the related services are rendered.

(a) Defined Contribution Plan

Contribution to Provident Fund etc., a defined contribution plan, is made in accordance with the statute, and is recognised as an expense in the year in which employees have rendered services.

(b) Defined Benefit Plan

Leave encashment being a defined benefit plan is accounted for using the projected unit credit method, on the basis of actuarial valuations carried out by third party actuaries at each Balance Sheet date. Actuarial gains and losses arising from experience adjustments and changes in actuarial assumptions are charged or credited to other comprehensive income in the year in which they arise. Other costs are accounted in statement of profit and loss.



to som

Arshiya Limited Notes to the financial statements for the year ended 31st March, 2019

The cost of providing gratuity, a defined benefit plans, is determined using the Projected Unit Credit Method, on the basis of actuarial valuations carried out by third party actuaries at each Balance Sheet date. Actuarial gains and losses arising from experience adjustments and changes in actuarial assumptions are charged or credited to other comprehensive income in the period in which they arise. Other costs are accounted in statement of profit and loss.

Remeasurements of defined benefit plan in respect of post employment and other long term benefits are charged to the other comprehensive income in the year in which they occur. Remeasurements are not reclassified to statement of profit and loss in subsequent periods.

3.13 Taxes on Income

Income tax expense represents the sum of current tax (including MAT and income tax for earlier years) and deferred tax. Tax is recognised in the statement of profit and loss, except to the extent that it relates to items recognised directly in equity or other comprehensive income, in such cases the tax is also recognised directly in equity or in other comprehensive income. Any subsequent change in direct tax on items initially recognised in equity or other comprehensive income is also recognised in equity or other comprehensive income.

Current tax provision is computed for income calculated after considering allowances and exemptions under the provisions of the applicable Income Tax Laws. Current tax assets and current tax liabilities are off set, and presented as net

Deferred tax is recognised on differences between the carrying amounts of assets and liabilities in the Balance sheet and the corresponding tax bases used in the computation of taxable profit. Deferred tax liabilities are generally recognised for all taxable temporary differences, and deferred tax assets are generally recognised for all deductible temporary differences, carry forward tax losses and allowances to the extent that it is probable that future taxable profits will be available against which those deductible temporary differences, carry forward tax losses and allowances can be utilized. Deferred tax assets and liabilities are measured at the applicable tax rates. The carrying amount of deferred tax assets is reviewed at each balance sheet date and reduced to the extent that it is no longer probable that sufficient taxable profits will be available against which the temporary differences can be utilized.

Credit of MAT is recognised as an asset only when and to the extent there is convincing evidence that the Company will pay normal income tax during the specified period, i.e., the period for which MAT credit is allowed to be carried forward. In the year in which the MAT credit becomes eligible to be recognised as an asset, the said asset is created by way of a credit to the statement of profit and loss and shown as MAT credit entitlement. The Company reviews the same at each balance sheet date and writes down the carrying amount of MAT credit entitlement to the extent there is no longer convincing evidence to the effect that the Company will pay normal income tax during the specified period.

3.14 Borrowing costs

Borrowing costs specifically relating to the acquisition or construction of qualifying assets that necessarily takes a substantial period of time to get ready for its intended use are capitalized (net of income on temporarily deployment of funds) as part of the cost of such assets. Borrowing costs consist of interest and other costs that the Company incurs in connection with the borrowing of funds. For general borrowing used for the purpose of obtaining a qualifying asset, the amount of borrowing costs eligible for capitalization is determined by applying a capitalization rate to the expenditures on that asset.

The capitalization rate is the weighted average of the borrowing costs applicable to the borrowings of the Company that are outstanding during the period, other than borrowings made specifically for the purpose of obtaining a qualifying asset. The amount of borrowing costs capitalized during a period does not exceed the amount of borrowing cost incurred during that period. All other borrowing costs are expensed in the period in which they occur.

3.15 Earnings per Share

Basic earnings per share is computed using the net profit/(loss) for the year attributable to the shareholders' and weighted average number of equity shares outstanding during the year

Diluted earnings per share is computed using the net profit/(loss) for the year attributable to the shareholders' and weighted average number of equity and potential equity shares outstanding during the year including share options, convertible preference shares and debentures, except where the result would be anti-dilutive. Potential equity shares that are converted during the year are included in the calculation of diluted earnings per share, from the beginning of the year or date of issuance of such potential equity shares, to the date of conversion.

3.16 Current and non-current classification

The Company presents assets and liabilities in statement of financial position based on current/non-current classification. The Company has presented non-current assets and current assets, non-current liabilities and current liabilities in accordance with Schedule III, Division II of Companies Act, 2013 notified by MCA.

7 x 1 how

An asset is classified as current when it is:

(a) Expected to be realised or intended to be sold or consumed in normal operating cycle.

(b) Held primarily for the purpose of trading.

c) Expected to be realised within twelve months after the reporting period, or


(d) Cash or cash equivalent unless restricted from being exchanged or used to settle a liability for at least twelve months after the reporting period.

All other assets are classified as non-current.

A liability is classified as current when it is:

- (a) Expected to be settled in normal operating cycle,
- (b) Held primarily for the purpose of trading,
- (c) Due to be settled within twelve months after the reporting period, or
- (d) There is no unconditional right to defer the settlement of the liability for at least twelve months after the reporting period.

All other liabilities are classified as non-current.

The operating cycle is the time between the acquisition of assets for processing and their realisation in cash or cash equivalents. Deferred tax assets and liabilities are classified as non-current assets and liabilities. The Company has identified twelve months as its normal operating cycle.

3.17 Fair value measurement

The Company measures financial instruments at fair value at each balance sheet date.

Fair value is the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The fair value measurement is based on the presumption that the transaction to sell the asset or transfer the liability takes place either:

- (a) In the principal market for the asset or liability, or
- (b) In the absence of a principal market, in the most advantageous market for the asset or liability.

A fair value measurement of a non-financial asset takes into account a market participant's ability to generate economic benefits by using the asset in its highest and best use or by selling it to another market participant that would use the asset in its highest and best use.

The Company uses valuation techniques that are appropriate in the circumstances and for which sufficient data are available to measure fair value, maximizing the use of relevant observable inputs and minimizing the use of unobservable inputs.

All assets and liabilities for which fair value is measured or disclosed in the financial statements are categorized within the fair value hierarchy.

3.18 Off-setting financial Instrument

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable rights to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable rights must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the Company or counterparty.

3.19 Segment Reporting - Identification of Segments

An operating segment is a component of the Company that engages in business activities from which it may earn revenues and incur expenses, whose operating results are regularly reviewed by the company's chief operating decision maker to make decisions for which discrete financial information is available. Based on the management approach as defined in Ind AS 108, the chief operating decision maker evaluates the Company's performance and allocates resources based on an analysis of various performance indicators by business segments and geographic segments.

3.20 Contributed Equity

Equity Shares are classified as equity, incremental costs directly attributable to the issue of new shares or options are shown in equity as deduction, net of tax from the proceeds.

3.21 Cash flow statement

Cash flows are reported using the indirect method, whereby profit before tax is adjusted for the effects of transactions of non-cash nature and any deferrals or accruals of past or future cash receipts or payments. The cash flows from operating, investing and financing activites of the Company are segregated based on the available information.

3.22 Business combinations

Business combinations involving entities that are controlled by the Company are accounted for using the pooling of interests method as follows:



A a sum

- (i) The assets and liabilities of the combining entities are reflected at their carrying amounts.
- (ii) No adjustments are made to reflect fair values, or recognise any new assets or liabilities.
- (iii) Adjustments are only made to harmonise accounting policies.
- (iv) The financial information in the financial statements in respect of prior periods is restated as if the business combination had occurred from the beginning of the preceding period in the financial statements, irrespective of the actual date of the combination. However, where the business combination had occurred after that date, the prior period information is restated only from that date.
- (v) The balance of the Retained Earnings appearing in the financial statements of the transferor is aggregated with the corresponding balance appearing in the financial statements of the transferee or is adjusted against General Reserve.
- (vi) The identities of the reserves are preserved and the reserves of the transferor become the reserves of the transferee.
- (vii) The difference, if any, between the amounts recorded as share capital issued plus any additional consideration in the form of cash or other assets and the amount of share capital of the transferror is transferred to Capital Reserve and is presented separately from Other Capital Reserves.

4 Significant accounting judgements, estimates and assumptions

The preparation of the financial statements requires management to make judgements, estimates and assumptions that affect the reported amounts of revenues, expenses, assets and liabilities, and the accompanying disclosures, and the disclosure of contingent liabilities. Uncertainty about these assumptions and estimates could result in outcomes that require a material adjustment to the carrying amount of assets or liabilities affected in future periods. The key assumptions concerning the future and other key sources of estimation uncertainty at the reporting date, that have a significant risk of causing a material adjustment to the carrying amounts of assets and liabilities within the next financial year, are described below.

The Company based on its assumptions and estimates on parameters available when the financial statements were prepared. However, existing circumstances and assumptions about future developments may change due to market changes or circumstances arising that are beyond the control of the Company. Such changes are reflected in the assumptions when they occur.

4.1 Property, plant and equipment and Intangible Assets

Management reviews the estimated useful lives and residual values of the assets annually in order to determine the amount of depreciation to be recorded during any reporting period. The useful lives and residual values as per schedule II of the Companies Act, 2013 or are based on the Company's historical experience with similar assets and taking into account anticipated technological changes, whichever is more appropriate.

4.2 Income Tax

The Company reviews at each balance sheet date the carrying amount of deferred tax assets. The factors used in estimates may differ from actual outcome which could lead to an adjustment to the amounts reported in the financial statements.

4.3 Contingencies

Management has estimated the possible outflow of resources at the end of each annual reporting financial year, if any, in respect of contingencies/claim/litigations against the Company as it is not possible to predict the outcome of pending matters with accuracy.

4.4 Impairment of financial assets

The impairment provisions for financial assets are based on assumptions about risk of default and expected cash loss. The Company uses judgement in making these assumptions and selecting the inputs to the impairment calculation, based on Company's past history, existing market conditions as well as forward looking estimates at the end of each reporting period.

4.5 Impairment of non-financial assets

The Company assesses at each reporting date whether there is an indication that an asset may be impaired. If any indication exists, or when annual impairment testing for an asset is required, the Company estimates the asset's recoverable amount. An asset's recoverable amount is the higher of an asset is or Cash Generating Units (CGU) fair value less costs of disposal and its value in use. It is determined for an individual asset, unless the asset does not generate cash inflows that are largely independent to those from other assets or groups of assets. Where the carrying amount of an asset or CGU exceeds its recoverable amount, the asset is considered impaired and is written down to its recoverable amount.

In assessing value in use, the estimated future cash flows are discounted to their present value using a pre-tax discount rate that reflects current market assessments of the time value of money and the risks specific to the asset. In determining fair value less cost of disposal, recent market transactions are taken into account. If no such transactions can be identified, an appropriate valuation model is used. These calculations are corroborated by valuation multiples or other available fair value indicators.



4.6 Defined benefits plans

The Cost of the defined benefit plan and the present value of such obligation are determined using actuarial valuations. An actuarial valuation involves making various assumptions that may differ from actual developments in the future. These include the determination of the discount rate, future salary increases, mortality rates and attrition rate. Due to the complexities involved in the valuation and its long-term nature, a defined benefit obligation is highly sensitive to changes in these assumptions. All assumptions are reviewed at each reporting date.

4.7 Recoverability of trade receivable

Judgements are required in assessing the recoverability of overdue trade receivables and determining whether a provision against those receivables is required. Factors considered include the credit rating of the counterparty, the amount and timing of anticipated future payments and any possible actions that can be taken to mitigate the risk of non-payment.

4.8 Provisions

Provisions and liabilities are recognized in the period when it becomes probable that there will be a future outflow of funds resulting from past operations or events and the amount of cash outflow can be reliably estimated. The timing of recognition and quantification of the liability require the application of judgement to existing facts and circumstances, which can be subject to change. Since the cash outflows can take place many years in the future, the carrying amounts of provisions and liabilities are reviewed regularly and adjusted to take account of changing facts and circumstances.

4.9 Fair value measurement of financial instruments

When the fair value of financial assets and financial liabilities recorded in the balance sheet cannot be measured based on quoted prices in active markets, their fair value is measured using valuation techniques including the Discounted Cash Flow (DCF) model. The inputs to these models are taken from observable markets where possible, but where this is not feasible, a degree of judgement is required in establishing fair values. Judgements include considerations of inputs such as liquidity risk, credit risk and volatility. Changes in assumptions about these factors could affect the reported fair value of financial instruments.

4.10 Recent accounting pronouncements:

Standards Issued But Not Effective

On 30th March, 2019, the Ministry of Corporate Affairs (MCA) has notified IND AS 116 – Leases and certain amendment to existing IND AS. These amendments shall be applicable to the Company from 1st April, 2019.

(a) Issue of IND AS 116 - Lease

Ind AS 116 will supersede the current standard on leases i.e. IND AS 17- Leases. As per IND AS 116, the lessor will have to bring to books all the non-cancellable portion of leasing arrangement.

(b) Amendment to existing standards

The MCA has also carried out amendments of the following accounting standards

- (i) IND AS 101- First time adoption of Indian Accounting Standards
- (ii) IND AS 103 Business Combinations
- (iii) IND AS 109 Financial Instruments
- (iv) IND AS 111 Joint Arrangements
- (v) IND AS 12 Income Taxes
- (vi) IND AS 19 Employee Benefits
- (vii) IND AS 23 Borrowing Costs
- (viii) Ind AS 28 Investment in Associates and Joint Ventures

Application of above standards are not expected to have any significant impact on the Company's financial statements.





	\$
	onto
	mote
	ial of
-imited	oucu.
_	tho fi
iya	404 -
Arshiya	Notes to the financial statements fo

Notes to the financial statements for the year ended 31st March, 2019

1	
т	
1	
1	
н	
н	
н	
н	1.
н	F
н	100
н	ž
н	E
н	C
н	- 2
н	13
I	
н	m
1	-
I	70
I	č
1	
L	
I	Ħ
I	5
1	
1	~
1	-
	1
1	2
1	T
1	0
1	0
1	õ
1	ž
3	n
1	-
1	10
4	-

	and the second se								-
Particulars	Freehold Land	Buildings	Plant and Equipments	Furniture and Fixtures	Vehicles	Equipments	Computers	Leasehold Improvements	Total
Gross Carrying Value (at deemed cost)		00000		05 001	00.01		10.02	10 00	1 76 690 64
As at 31st March, 2017	11,439.24	42,099.06	3,411.84	130.05	40.20	14.180.1	10.01	10.00	40'000'07'I
Additions	59.02		10			30.76	1	P	89.78
Disposals	(9.337.96)	(17.202.94)	(1.231.14)	(532.44)	4	(704.65)	•	100	(29,009.13)
As at 31st March, 2018	68,160.30	24,896.12	2,180.70	204.15	48.96	1,017.58	70.01	33.37	96,611.19
Additions	ŕ	3	21.00	8.46	13.96	9.01	17.12		69.55
Disposals/Discarded	ĩ	â	(1.35)	(116.17)	(15.28)	(38.17)	(43.00)	(33.37)	(247.34)
Transfer to Inventories	(19,089,31)	ł			1		1	×	(19,089.31)
As at 31st March, 2019	49,070.99	24,896.12	2,200.35	96.44	47.64	988.42	44.13	•	77,344.09
Accumulated Depreciation									
As at 31st March, 2017	F	1,045.60	321.52	140.97	16.40	452.08	23.27	18.23	2.018.07
Depreciation for the year	1	999.65	304.05	124.84	13.09	320.41	8.67	4.90	1,775.61
Disposals	,1	(539.43)	(207.43)	(180.24)	•	(394.02)	4		(1,321.12)
As at 31st March, 2018	,	1,505.82	418.14	85.57	29.49	378.47	31.94	23.13	2,472.56
Depreciation for the year	4	752.48	209.88	24.38	7,97	153.60	1.18	9	1,149.49
Disposals/Discarded			(0.31)	(60.07)	(11.70)	(19.18)	(21.93)	(23.13)	(136.32)
As at 31st March, 2019	1	2,258.30	627.71	49.88	25.76	512.89	11.19		3,485.73
Net Carrying value as at 31st March, 2019	49,070.99	22,637.82	1,572.64	46.56	21.88	475.53	32.94		73,858.36
Net Carrying value as at 31st March. 2018	68.160.30	23.390.30	1.762.56	118.58	19.47	639.11	38.07	10.24	94,138.63

Notes:

1) Freehold Land includes Rs. 9,735.11 Lakh situated at Nagpur, which is under possession of a lender as per the Order of Hon'ble High Court of Bombay dated 9th May, 2013.

2) Freehold Land measuring 42,59 Acres amounting to Rs. 7,499.35 Lakh is used for Domestic warehousing purpose located at Khurja, Bulandshahr, Uttar Pradesh,

3) During the year, Freehold Land measuring 45.52 Acres amounting to Rs. 19,089.31 Lakh are converted into inventories.

4) Gross carrying value includes cost of vehicles taken on finance lease Rs. 13.96 Lakh.



0

6. Intangible Assets			
H SH SH	1		(Rs. in Lakh)
Particulars	Trade Mark	Software	Total
Gross Carrying Value (at deemed cost)			
As at 31st March, 2017	0.49	588.14	588.63
Additions/other adjustments		1,120.00	1,120.00
As at 31st March, 2018	0.49	1,708.14	1,708.63
Additions	-		4
Disposals/Discarded	1.221	(116.94)	(116.94
As at 31st March, 2019	0.49	1,591.20	1,591.69
Accumulated Amortisation			
As at 31st March, 2017	0.20	156.41	156.61
Amortisation for the year	0.17	315.89	316.06
Deductions			
As at 31st March, 2018	0.37	472.30	472.67
Amortisation for the year		332.73	332.73
Deductions		(59.57)	(59.57
As at 31st March, 2019	0.37	745.46	745.83
Net Carrying value as at 31st March, 2019	0.12	845.74	845.86
Net Carrying value as at 31st March, 2018	0.12	1,235.84	1,235.96



Q 5 Aam

Particulars	As 31st Ma		As 31st Ma	
	No. of Shares	Rs. in Lakh	No. of Shares	Rs. in Lakh
Non-Current Financial Assets				
7. Investments				
(Unquoted Investments carried at Cost) (i) Investments in Equity Instruments of Subsidiaries				
Arshiya Industrial & Distribution Hub Limited (the face value of Rs. 10 each) @	1,72,37,152	44,499 72	1,72,37,152	44 499.73
Arshiya Northern FTWZ Limited (the face value of Rs 10 each) @	1.08,68,677	44,625,29	1,08,68,677	44,625.2
Arshiya Rail Infrastructure Limited (the face value of Rs. 10 each) @	4,23,84,417 50,000	38,369.21 5,00 2.00	4,23,84,417	38,369.21
Arshiya Transport and Handling Limited (the face value of Rs 10 each)			50,000 1,01,158	5.00
Arshiya Technologies (India) Private Limited (the face value of Rs. 10 each)	1,01,158			
Arshiya Lifestyle Limited (the face value of Rs. 10 each)	14.85,000	14.85	14,85,000	14.8
Arshiya Logistics Services Limited (Formerly known as Laxmipati Balaji Exim Trading Ltd.) (the face value Rs. 10 each)	16,00,000	155,50	16,00,000	155.5
Laxmipati Balaji Supply Chain Management Private Limited (the face value of Rs. 10 each)	50,000	5.00	8	
Anomalous Infra Private Limited (the face value of Rs. 10 each)	1,10,000	11,00	-	-
Arshiya Northern Projects Private Limited (the face value Rs. 10 each)	50,000	50,000 5.00 10,000 1.00	• •	1
Arshiya Infrastructure Developers Private Limited (the face value Rs. 10 each)	10,000			
Unrivalled Infrastructure Private Limited (the face value Rs. 10 each)	10,000	1.00	~	-
Total (i)		1,27,694.57	1	1,27,671.5
(All the above equity shares are fully paid up)	1			
(ii) Deemed Equity Investments	1.000			
Investments at amortised cost				
Arshiya Industrial & Distribution Hub Limited		1,116.48		1,112.7
Arshiya Northern FTWZ Limited		696.96		696.9
Arshiya Rail Infrastructure Limited		1,795.82 302.40		1,735.8
Arshiya Transport and Handling Limited Arshiya Lifestyle Limited		1,981.96		498.4
Anomalous Infra Private Limited		1,091.83		
Total (ii)		6,985.45		4,346.4
Total (i+ii)		1,34,680.02		1,32,018.0

@ As per debt covenents of the following Subsidiaries is required to pledge 100% of the shareholding in favor lenders however the Comapny has pledged following number equity shares only:

i) 31st March, 2019 - 70,59,038 (31st March 2018 - 79,46,624) equity shares in Arshiya Northern FTWZ Limited.
 ii) 31st March, 2019 - 51,05,769 (31st March, 2018 - 1,35,86,659) equity shares in Arshiya Industrial & Distribution Hub Limited and
 iii) 31st March, 2019 - 1,54,78,500 (31st March, 2018 - 3,87,32,491) equity shares in Arshiya Rail Infrastructure Limited



K Aer &

	As at	(Rs. in Lakh) As at
Particulars	31st Mar, 2019	31st Mar 2018
Non- Current Financial Assets		
8. Loans	11 1 I I I I I I I I I I I I I I I I I	1 1 1 1 1
Unsecured, considered good unless otherwise stated		
Loans to Subsidiaries (Refer Note No. 56)	1,732.14	1,731.4
	1,732.14	1,731.4
Total	1,732.14	1,731.4
9. Trade Receivables		
Unsecured	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
Considered good - Unsecured (Receivables from Related party (Refer Note No. 56))	6,061.50	
Tetal	6,061.50	-
Total	0,001.30	
10. Other Non- Current Assets		
Capital Advances	1	
Considered good	1,020.41	90.6
Considered doubtful	1,395.00	1,395.0
Less: Provision for impairment	2,415.41 (1,395.00)	1,485.6
	1,020.41	90.6
Security Deposits	59.39	60.8
TDS Receivables/Taxes paid	2,240.54	2,122.2
Total	3,320.34	2,273.8
Current Assets		
a provident		
11. Inventories Land	16,505.97	
card	Taloagiet.	
Total	16,505.97	
anona anona atoma	11	
Current Financial Assets		
12. Trade Receivables		1000
Considered good - Unsecured (Receivables from Related party (Refer Note No. 56))	814.64	764.6
Trade Receivables which have significant increase in credit risk	25.37	16.6
	840.01	781.3
Less: Provision for expected credit losses	(25.37)	(16.0
		· · · · · · · ·
Total	814.64	764.6
13. Cash and Cash Equivalents		
13. Casil and Casil Equivalents		
Balances with banks:		
- in current accounts	5 77	130
Cash on hand	0.09	4.1
Total	5.86	135.6
14. Other Bank Balances		
Deposits with bank to the extent held as margin money	15.17	
Balance with bank in Unpaid dividend account		0.0
2.53.	15.17	0,0
Total	15.17	0.
15. Loans		
15. Loans Unsecured, considered good unless otherwise stated	6	in the second
	30,327.14	33,279.5





.

Particulars		As at 31st Mar 2019	As at 31st Mar 2018
16. Other Financial Assets			
Unbilled Revenue		2,500.00	390.41
Margin money with Lender*		170.00	170.00
Other recoverables		1,412.95	1,288.30
	Total	4,082.95	1,848.7
* To be adjusted at time of final settlement		1 1 1 1	
17. Other Current Assets			1.1.2
Advances to Suppliers		44.42	115.4
Other Advances		93.26	93.2
Prepaid expenses		60.36	54.73
Balances with Statutory, Government Authorities (Refer Note No 51) Cash seized by Income Tax (Refer Note No 48)		1,935.96	1,956.1 100.0
	Total	2,134.00	2,319.6

		(Rs. in Lakh)
Particulars	As at 31st Mar, 2019	As at 31st Mar 2018
18. Equity Share Capital		
Authorised (i) 24,75,00,000 (31st March, 2018 - 24,75,00,000) Equity Shares of Rs. 2 each (ii) 1,10,00,000 (31st March, 2018 - 1,10,00,000) 0% Optionally Convertible Redeemable Preference Shares (OCRPS) of Rs. 10 each	4,950.00 e 1,100.00	4,950.00 1,100.00
Total	6,050.00	6,050.00
Equity Share Capital - issued, subscribed and fully paid 24,36,14,292 (31st March, 2018 - 22,82,16,776) Equity shares of Rs. 2 each	4,872.29	4,564.34
Total	4,872.29	4,564.34

(a) Terms and rights

(i) Terms and rights attached to equity shares The Company has one class of equity share having a par value of Rs. 2 per share. Each holder of equity share is entitled to one vote per share. The shareholders who held shares on the record date are entitled to dividend as may be proposed by the Board of Directors and is subject to approval of the Shareholders at the ensuing General Meeting.

In the event of liquidation of the Company, the holders of Equity Shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in the proportion to the number of equity shares held by the shareholders.

(iii) Terms and rights attached to 0% Optionally Convertible Redeemable Preference Shares (OCRPS)

The Company has 0% optionally convertible redeemable preference shares having a par value of Rs. 10 per share. Each holder of OCRPS has right / entitled to convert into equity shares within 18 months from the date of issue or redemption on or after 20 years as per terms.

(b) Reconciliation of equity shares and optionally convertible preference shares

m (1) (1) (1) (1)	As at 31st Ma	rch 2019	As at 31st Marc	sh 2018
Particulars	Number of Shares	Rs. in Lakh	Number of Shares	Rs. in Lakh
Balance as at the beginning of the year	22,82,16,776	4,564.34	15,61,79,472	3,123.59
Add: Issued during the year	1,53,97,516	307.95	7,20,37,304	1,440.75
Balance as at the end of the year	24,36,14,292	4,872.29	22,82,16,776	4,564.34

(iii) Reconciliation of optionally convertible redeemable preference shares outstanding as at the beginning and end of the year

Pauli sula m	As at 31st March 2019		As at 31st March 2018	
Particulars	Number of Shares	Rs. in Lakh	Number of Shares	Rs. in Lakh
Balance as at the beginning of the year	57,64,619	576.46		-
Add: Issued during the year			1,19,13,329	1,191.33
Less: Converted into equity shares during the year	(57,64,619)	(576.46)	(61,48,710)	(614.87
Balance as at the end of the year			57,64,619	576.46



	aggregate shares in the Company

	As at 31st M	arch 2019	As at 31st Ma	rch 2018
Name of the shareholder	Number of equity shares	Percentage (%) shareholding	Number of equity shares	Percentage (%) shareholding
Archana A Mittal	8,85,59,788	36.35%	8,85,59,288	38.80%
Ajay S Mittal	3,85,61,437	15.83%	3,70,60,937	16.24%
Edelweiss Asset Reconstruction Company Limited (through various trusts)	5,95,59,820	24.45%	4,56,62,304	20.01%

(d) In the Previous year ended 31st March, 2018 the Company had allotted to the Promoter Directors 1,00,00,000 equity shares and 1,00,00,000 share warrants of Rs. 2/- each at a premium of Rs.58.35 per share on preferential basis pursuant to the Restructuring Agreement dated 31st March, 2017 and in terms of special resolution passed on 29th April, 2017 as per applicable provisions of Companies Act, 1956/Companies Act, 2013 and SEBI (Issue of Capital and Disclosure Requirements) Regulation. 85,00,000 share warrants out of 1,00,00,000 share warrants have been converted into Equity shares on 8th November, 2017.

During the year, the Company has allotted 15,00,000 Equity Shares of face value of Rs.2 each to the Promoter upon conversion of equal number of warrants.

(Rs. in L		
Particulars	As at 31st Mar, 2019	As at 31st Mar 2018
19. Other Equity		
(I) Share Application money pending allotment		0.00
Balances as at the beginning of the year		18,766.71
Less: On issue of Equity Shares		(18,766.71)
Balances as at the end of the year	*	-
(ii) Money Received against share warrants	1 Figure 1	1
Balances as at the beginning of the year	860.25	
Add; received during the year	15.00	860.25
Less: On issue of Equity Shares	(875.25)	+
Balances as at the end of the year		860.25
(iii) Equity Component of 0% Optionally Convertible Preference shares (OCRPS)		1 - 1
Balances as at the beginning of the year	47,551.87	88,620.84
Less: On issue of Equity Shares	(47,551.87)	(41,068.97)
Balances as at the end of the year	-	47,551.87
(iv) Amalgamation Reserve		
Balances as at the beginning and end of the year	124.80	124.80
Reserve and Surplus		
(v) Securities Premium Account	and an inclusion	
Balances as at the beginning of the year	1,74,858.67	79,617,43
Add: On issue of Equity Shares	58,213.49	95,278.86
Less: Share issue expenses/ Transaction cost		(37.62)
Balances as at the end of the year	2,33,072.16	1,74,858.67
(vi) General Reserve		
Balances as at the beginning and end of the year	940.18	940.18
(vii) Deficit in the Statement of Profit and Loss	1 1 1 1 1	1.0.3
Balances as at the beginning of the year	(63,985.47)	
Add. Profit/(Loss) for the year	(3,516.49)	
Add: Other comprehensive income/(loss)	8 11	(2.69
Add: Fair Value of Financial instruments		64.05
Add: Conditional Lease rent	-	390.41
Balances as at the end of the year	(67,493.85	(63,985.47)
Total (i to vii)	1,66,643.28	1,60,350.30

Nature and purpose of Reserve and Surplus:

(a) Securities Premium Account:

Securities premium account is created to record premium received on issue of equity shares. The reserve is utilized in accordance with the provision of the Companies Act, 2013.

(b) General Reserve:

General Reserve is used for time to time to transfer of profits from Retained Earnings for appropriation purposes. As the general reserve is created by a transfer from one component of equity to another and is not an item of other comprehensive income, items included in the General Reserve will not be reclassified subsequently to statement of profit and loss.



Notes to the financial statements for the year ended 31st March, 2019

(c) Amalgamation Reserve:

Amalgamation reserve is created on account of scheme of amalgamation of erstwhile BDP (India) Private Limited with the Company approved by the Hon'ble High Court of Judicature at Bombay in earlier years.

(d) Retained Earning:

Retained Earnings are the profit/(loss) of the Company earned till date net of appropriations.

			(Rs. in Lakh
Particulars		As at 31st Mar, 2019	As at 31st Mar 2018
Non-Current Liabilities			
20. Borrowings		the second se	a second second
Secured			
(a) Term Loans		- Sec. in	
From Banks		3,571.43	ALC: No.
From Other Parties		56,686.25	58,497.26
(b) Vehicles Loan from bank		9.52	
Liability Component of Compound Financial Instruments (OCRPS)			10,342.61
	Total	60,267.20	68,839.87

The details of security, terms of repayment and interest on non-current borrowings (which includes current maturities) obtained by the Company are given below:

20.1 Rupee Term loan from Banks

(1) Rupee loan of Rs. 3, 193.29 Lakh (31st March, 2018 - Rs. 3, 189.79 Lakh):

(a) Securities provided

(i) Second charge on movable and immovable Panvel assets of the Company except for the excluded properties under Lease Agreement dated 3rd February, 2018

(ii) Second charge on present and future receivables including the amount deposited to the EARC TRA account of the Company . (iii) The above loans are secured by personal guarantees of two Promoter Directors of the Company.

(b) Terms of Interest rate (i) Rate of Interest is @ 14.50% p.a.

(c) Terms of Repayment:-

Rupee term loan is repayable in Bullet payment at the end of the tenure of loan i.e. 36 months.

(d) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 6.71 Lakh (31st March. 2018 - Rs. 10.21 Lakh).

(2) Rupee loan of Rs. 474.30 Lakh:

(a) Securities provided (i) Second charge on movable and immovable Panvel assets of the Company except for the excluded properties under Lease Agreement dated 3rd February, 2018

(ii) Second charge on present and future receivables including the amount deposited to the EARC TRA account of the Company. (iii) The above loans are secured by personal guarantees of two Promoter Directors of the Company.

(b) Terms of Interest rate

Rate of Interest is @ 14.50% p.a.

(c) Terms of Repayment:-

Rupee term loan is repayable in 78 equal monthly installment commencing from the date of first disbursement i e, 31st August, 2018

(d) The Company has been in default for the repayment of principal amount of Rs. 19.17 Lakh

20.2 Rupee Term loans from Other Parties

(1) Rupee term loan of Rs. 59,513.25 Lakh (31st March, 2018 - Rs. 59,359.23 Lakh):

(a) Security provided:

(i) First charge on all the present and future movable and immovable property, plant and equipment including intangible assets, assignment of rights and benefits but excluding project assets for Khurja FTWZ project, Khurja Distiripark Project, Nagpur project and Rail project on pari passu basis. (ii) Second charge on current assets of the Company but excluding current assets for Khurja FTWZ project, Khurja Distripark Project, Nagpur project

De la Laber

1

and Rail project on pari passu basis. (iii) first pari passu charge by way of hypothecation on the Panvel Receivables both existing and future of whatsoever nature.

- (iv) The above loans are secured by personal guarantees of two Promoter Directors of the Company
- (v) The loans are secured by pledged of shares held by the two Promoter Directors of the Company.

(b) Terms of Interest rate

Rate of Interest is @ 10% p.a. compounded quarterly.

(c) Terms of Repayment:-

	(KS. IN Lakn)
Year	Loan from Others
FY 2017-18	5,671.09
FY 2021-22	14,001.46
FY 2022-23	40,404.50
 Total	60.077.05

VEDI & SH MUMBA ED ACK

Notes to the financial statements for the year ended 31st March, 2019

(d) The Company has been in default for the repayment of principal amount of Rs. 5,671.09 Lakh. (31st March, 2018 - Rs. 5,671.09 Lakh).

(e) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 563.80 Lakh (31st March, 2018 - Rs. 717.82

(2) Rupee term loan of Rs. 2,495.44 Lakh (31st March, 2018 - Rs. 2,672.34 Lakh):

(a) Securities provided

(i) Second charge by way of equitable mortgage/hypothecation on the entire immovable and movable property, plant and equipment of the Company on pari-passu basis.

(ii) The above loans are secured by personal guarantees of two Promoter Directors of the Company.

(b) Terms of Repayment:-

Rupee term loan is repayable in 13 structured quarterly installments commencing from 31st January, 2018

(c) The Company has been in default for the repayment of principal amount of Rs. 670.00 Lakh. (31st March, 2018 - Rs. 428 Lakh)

(d) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 199.56 Lakh (31st March, 2018 - Rs. 405.66

(3) Rupee term loan of Rs. 2,018.65 Lakh:

(a) Securities provided

(i) The above loan are secured by charge on residual cashflow of the Company.

(ii) The above loans are secured by the immovable property held by one Promoter Director of the Company on pari passu basis.

(iii) The above loans are secured by personal guarantees of two Promoter Directors of the Company. (iv) The above loans are secured by pledged of shares held by the one Promoter Director of the Company.

(b) Terms of Interest rate

Rate of Interest is @ 18% p.a.compounded half yearly.

(c) Terms of Repayment:-

Rupee term loan is repayable in bullet payment at the end of the tenure of loan i.e. 18 months.

(d) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 81.35 Lakh.

20.3 Vehicle loans from Bank

Vehicle loans are secured by way of hypothecation of vehicles. Rate of interest is @ 8,55% p.a. and repayment lenure in monthly instalment up to October 2023 and January 2024 respectively.

			(Rs. in Lakh)
Particulars		As at 31st Mar, 2019	As at 31st Mar 2018
21. Other Financial Liabilities Interest accrued but not due on borrowings		221 96	
Financial Liabilities at amortised cost Financial guarantees obligations		1,390.76	2,191,60
	Total	1,612.72	2,191.60
22. Provisions			
Provision for employee benefits (Refer Note No.41) Gratuity Leave encashment		80.60 38.33	101 32 49.70
	Total	118.93	151.02

		T	(Rs. in Lakh)
Particulars		As at 31st Mar, 2019	As at 31st Mar 2018
Current Financial Liabilities			
23. Borrowings		1	
Secured		00.555	
(a) Loan from Other Parties		9,024.05	8,474.05
Unsecured		1.00	1.000
(a) Loans from Promoter Directors		160.11	1,937.76
(b) Inter Corporate Deposits		77,00	77.00
	Total	9,261.16	10,488.81



X Low 6

Notes to the financial statements for the year ended 31st March, 2019

(23.1) Loan from Other Parties

(1) Loan of Rs. 8,474.04 Lakh (31st March, 2018 - Rs. 8,474.04 Lakh)

- (i) Securities provided
 - First Ranking charges on all present and future cash flows, all assets and movable collateral available to the existing lenders of the Company as per the Deed of Hypothecation
 - The above loans are secured by personal guarantees of two Promoter Directors of the Company.

(ii) Terms of interest; @ 18% p.a.

(iii) The Company has been in default for the repayment of principal amount of Rs. 8,474.04 Lakh (31st March, 2018 - Rs. Nil).

(2) Loan of Rs. 550.00 Lakh (i) Securities provided

- Exclusive charges on cash flows of Domestic warehousing building.
- The above loans are secured by mortgage over lands admeasuring 7,130 Sq. mt. of the Company and wholly owned subsidiaries company.
- The above loans are secured by personal guarantees of one Promoter Director of the Company.
 The above loans are secured by corporate guarantees of the two subsidiary Companies.

(ii) Terms of interest: @ 11% p.a.

(23.2) Loans from promoter directors are interest free and repayable on demand.

(23.3) Unsecured Loan from Inter Corporate Deposits:

Intercorporate Deposit of Rs. 77 Lakh (31st March, 2018 - 77 Lakh) is interest free and repayable on demand.

			(Rs. in Lakh)
Particulars		As at 31st Mar, 2019	As at 31st Mar 2018
24. Trade Payables Micro and Small Enterprises (Refer Note No. 40)		37.88	0.89
Others		584.91	581.66
	Total	622.79	582.55

Particulars	As at 31st Mar, 2019	As at 31st Mar 2018
25. Other Financial Liabilities		
Financial Liabilities at amortised cost		
Current maturities of long term debts from banks	1,569.00	1,491.67
Current maturities of long term debts from other parties	14,341.09	13,675.60
Current maturities of vehicle loan	1.80	
Interest accrued and due on borrowings	6,866.45	1,407.85
Interest accrued but not due on borrowings	4,674.69	1,892,26
Interest payable on delayed payments to MSMED creditors (Refer Note No. 40)	2.56	0.22
Unclaimed dividends		0.04
Deposit from Unitholders	401.30	385,68
Financial Guarantee Obligations	575.05	606.11
Payable for capital goods	265.92	1,137.29
Dues to employees	301 03	267.81
Payable for expenses	680.25	641.50
Total	29.679.14	21,506.03

25.1 Term loans from Bank - Rs. 1,472.84 Lakh (31st March, 2018 - Rs. 1,491.67 Lakh):

(i) Securities provided

- Second charge on movable and immovable property, plant and equipments of the Company, present and future on pari-passu

(ii) The above loan is secured by personal guarantees of two Promoter Directors of the Company.

(iii) Terms of Interest rate:

Rate of interest is @ 12% p.a.

(iv) Terms of Repayment & Default:

The bank has been recalled loan of Rs. 1,472.84 Lakh (31st March, 2018 - Rs 1,491.67 Lakh) and interest (including penal interest) of Rs. 178.98 Lakh (31st March, 2018 - Rs. 32 15 Lakh).



V Jam

Notes to the financial statements for the year ended 31st March, 2019

25.2 Term loans from Other Parties:

(1) Loan of RS. 5,000,000 Lakh (31st March, 2018 - Rs. 5,000.00 Lakh) (Refer Note No. 38) Secured by first and exclusive charge on land situated at Village Butibori at Nagpur, Maharashtra. The said loan carries interest @ 20% p.a.

(2) Loan of Rs. 2,000.00 Lakh (31st March, 2018 - Rs. 1,951 52 Lakh) (Refer Note No. 39)
(i) 'Secured by first and exclusive charge on land situated at Khurja, Bulandshahr, Uttar Pradesh.
(ii) The Company has been in default for the repayment of principal amount of Rs. 2,000.00 Lakh. (31st March, 2018 - Rs. 975.00 Lakh)

(iii) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. Nil (31st March, 2018 - Rs. 23.48 Lakh).

25.3 Details of default in payment of Interest on secured loans as on 31st March, 2019 are as follows:

(Rs. in			(Rs. in Lakh)
Year	Banks	Others	Total
FY 2017-18	32.15	1,303.42	1,335,57
FY 2018-19	239.77	5,291.11	5,530.88
Total	271.92	6,594.53	6,866.45

			(Rs. in Lakh)
Particulars		As at 31st Mar, 2019	As at 31st Mar 2018
26. Other Current Liabilities			1
Trade advances received		1.1.1	94.37
Other Payables		1.38	
Statutory Liabilities		1.	
Statutory dues (Refer note below)		774.86	356.34
Interest on delayed payment of statutory dues		657.87	611.77
	Total	1,434.11	1,062,48

Notes:

1) Statutory dues included Tax deducted at sources (TDS), Goods and Service tax (GST), Provident Fund (PF), Profession Tax (PT) and Employee State Insurance Corporation (ESIC)

2) Based on recent Supreme court judgement on structure of component for calculation of Provident Fund dated February 28, 2019 there are various interpretive issues including its applicability thus prospective provision w.e.f. March 01, 2019 been considered of Rs. 0.40 lakhs.

Particulars	As at 31st Mar, 2019	(Rs. in Lakh) As at 31st Mar 2018
27. Provision Provision for employee benefits (Refer Note No. 41) Leave encashment	8,35	9.57
Total	8.35	9.57



	Year Ended 31st Mar 2019	Year Ended 31st Mar 2018
	7 167 87	6,899.36 0.46
	2,500.00	-
	3,351.11	972.91 649.29
	121.00	20.00
Total	13,139.98	8,542.02
	0.19 7.40	- 1.38
d cost		
	420.02	185.51
	898.96	639.31 15.22
	206.27	175.85
	6.14	2.82
	653.17 0.33	1
Total	2,192.48	1,020.09
		_
	2,583.34	4
Total	2,583.34	-
	87.63	320 61
Total	87.63	320.61
	1,625.32	1,394.29
	44.92	30.90
Total	1,720.28	1,456.61
	10,810.67	13,259 46
	9.50	226.36
	88.09	214.39
	335.21	61.51
	11,236.53	13,761.94
Total		
Total		-
Total	1,149.49 332.73	
	rotal Total	3,351.11 121:00 Total 13,139.98 0.19 7.40 898.96 206.27 6.14 653.17 0.33 Total 2,192.48 2.583.34 Total 2,583.34 Total 87.63 Total 87.63 Total 1,625.32 50.04 44.92 Total 1,720.28 10,810.67 2.56



Particulars	Year Ended 31st Mar 2019	Year Ended 31st Mar 2018
35. Other Expenses		
Electricity charges	111.89	230.71
Rent	29.06	13.58
Repairs and maintenance:		
- Building	83.13	104.28
- Plant and Machinery	63.36	84.53
- Others	35.42	53.55
Insurance	11.30	9.06
Rates and taxes	12.65	4.94
Communication expenses	20,75	22.72
Travelling and conveyance expenses	53,17	95.39
Vehicle expenses	36.41	42.28
Printing and stationery	27.77	24.24
Legal and professional fees	130.90	146.45
Security charges	52.27	246.90
Auditor's remuneration:		
- Audit Fees	54.00	54.00
- Limited Review Fees	15.00	18.25
- Certification fees	2.20	1.75
Advertisement and Sales Business Promotion expenses	28.66	78.23
Allowance for doubtful debts	8.68	7.33
Bad debts	3.16	
Foreign exchange differences (net)	12.81	
Director sitting fees	2.65	5.75
Miscellaneous expenses	76.62	88.53
Discarding/written off of Property, plant and equipment and Intangible assets	166.34	
Total	1,038.20	1,332.46
36. Exceptional Items		
Settlement of claims	700.75	(2,001.74
Gain on monetization of property, plant and equipment (Refer Note No. 55)		(15,633.29
Loss on sale of subsidiary (Refer Note No 54.2)		4,338.19
Total	700.75	(13,296.84



A sam

Notes to the financial statements for the year ended 31st March, 2019

- 37 Contingent Liabilities and Commitments
- 37.1 Contingent Liabilities (to the extent not provided for in respect of):

		and the second se	(Rs. in Lakh)
S. No.	Particulars	As at 31st March, 2019	As at 31st March, 2018
(i)	Disputed Income Tax demands	18,515.52	11,087.78
(ii)	Disputed Sales Tax demands		20,51
(iii)	Disputed Service Tax demands	62.68	
(iv)	Disputed Local Body Tax demands	160.33	160.33
(V)	Claims against the Company not acknowledged as debts	2,268.60	2,321.79
(vi)	Import Continuity / Transshipment Bond	10,000.00	10,000.00
(vii)	Bond cum legal under taking	5,196.00	5,196.00
(viii)	Bank Guarantees for Maharashtra Pollution Control Board	15.00	
(ix)	Corporate Guarantees given	18,500.00	

37.2 Capital commitments

Estimated amount of contracts remaining to be executed on capital and other accounts and not provided for (net of advances paid) are Rs. 5,782.08 Lakhs (31st March, 2018 - Rs. Nil)

38 A Public Financial Institution (PFI) agreed to settle their outstanding loan constituting principle and interest of Rs. 16,700 Lakh. Settlement terms and conditions involves payment of Rs. 5,000 Lakh which is secured by land at Nagpur and for balance amount of Rs. 11,700 Lakh, allotment of Optionally Convertible Redeemable Preference Shares - V (OCRPS - V), convertible upto 15,50,000 equity shares at the option of the PFI Considering the same, necessary effect has been given in the books of accounts during the previous year. As per shareholder approval in the EOGM dated 29th January 2018, the company has approved allotment of 11,70,000 OCRPS - V and the same was converted into 15,50,000 Equity shares on 22nd February, 2018 as per settlement terms agreed. Subsequently in the Honorable High Court of Bombay, the Company has made the representation that post allotment of the equity shares as exercised by the PFI, the total outstanding debt remains at Rs. 5,000 lakhs.

During the year ended 31st March, 2019, the PFI has assigned its debt to the Edelweiss Asset Reconstruction Company (EARC). The Company has provided interest in line with major terms negotiated with EARC, till the finalisation of the restructuring agreement.

39 During the year, the Company has defaulted in payment as per consent terms signed with one of the Non-Banking Financial Company (NBFC). Subsequent to the year end 31st March, 2019, the said NBFC has assigned its debt to Edelweiss Asset Reconstruction Company (EARC). Pursuant to said assignment, EARC has become the lender and entitled to recover total dues along with interest at contractual rates and other charges. The company doesn't expect any additional liabilities / charges and liabilities accounted in the books of account are adequate.

40	Disclosures under Micro,	Small And Medium	Enterprises Development	("MSMED") Act, 2006

Sr. No.	Description	As at 31st March 2019	As at 31st March 2018
a)	Principal amount due and remaining unpaid	38.13	0.89
b)	Interest due thereon remaining unpaid	2.56	0.22
c)	Interest paid by the Company in terms of Section 16 of the MSMED Act, 2016, along with the amount of the payment made to the suppliers beyond the appointed day during the year	-	*
d)	Interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under the MSMED Act, 2006		
e)	Interest accrued and remaining unpaid	2.56	0.22
ŋ	Interest remaining due and payable even in the succeeding years, until such date when the interest dues as above ae actually paid to the micro and small enterprises.	1	

Note: Dues to Micro, Small and Medium enterprises have been determined to the extent such parties have been identified on the basis of information collected by the Company and relied upon by the Auditors

41 Employee Benefits

41.1 Disclosure pursuant to Indian Accounting Standard (IND AS) 19 - Employee Benefits

(a) Defined Contribution Plan:

Contribution to Defined Contribution Plan, recognised as expenses for the years are as under:

Particulars	Year ended 31st March, 2019	Year ended 31st March, 2018
Employer's Contribution to Provident Fund	7.91	6 90
Employer's Contribution to Pension Scheme	17 97	15.69
Employer's Contribution to ESIC	0.47	1,02





Notes to the financial statements for the year ended 31st March, 2019

Brief descriptions of the plans (b)

The Company's defined contribution plans are Provident Fund and Employees State Insurance where the Company has no further obligation beyond making the contributions. The Company's defined benefit plans include gratuity The employees are also entitled to leave encashment as per the Company's policy.

(c)

Leave Encashment:		(Rs. in Lakh)
Particulars	As at 31st March 2019	As at 31st March 2018
Provision recognised in the Balance Sheet		
Current Provision as at the end of the year	8.35	9.57
Non Current Provision as at the end of the year	38.33	49.70
Provision recognised in the Balance Sheet	46.68	59.27

Defined benefit plan - Gratuity: (d)

The employee's Gratuity fund is managed by the Life Insurance Corporation of India. The present value of obligation is determined based on actuarial valuation using the Projected Unit Credit Method, which recognised each period of services as giving rise to additional unit of employee benefit entitlement and measures each unit separately to build up to final obligation.

Particulars	Year ended 31st March 2019	Year ended 31st March 2018
I. Actuarial assumptions		
Mortality Table	Indian Assured lives Mortality (2006-08) Ult	Indian Assured lives Mortality (2006-08) Ult
Discount rate	6,95%	7 40%
Expected return on plan assets	7.40%	10-711-
Salary Escalation Rate	9.00%	
Withdrawal Rate	17.00%	
Retirement Age	58 Years	and a set of the set o
II. Change in Present value of defined benefit obligations		
Provision as at the beginning of the year	116.24	99 57
Interest cost	8 59	6.46
Current service cost	16.17	25.73
Benefits paid	(4.73)	
Acquisition adjustments	(31.54)	10.10
	(8.11)	(7.37)
Actuarial (gain)/loss on obligations Provision as at the end of the year	96.62	116.24
III. Change in Fair value of plan assets		
Fair value of plan assets as at the beginning of the year	14.92	14.92
Expected return on plan assets	1.10	1.03
Actual Enterprise's Contributions	1.10	8 15
Benefits paid	-	(8.15
		(1.03
Actuarial gain/(loss) on plan assets Fair value of plan assets as at the end of the year	16.02	14.92
IV. Actual return on plan assets	1	
Expected return on plan assets	1 10	1.03
Actuarial gain/(loss) on plan assets	1	(1.03
Actual return on plan assets	1.10	-
V. Provision recognised in the Balance Sheet	-	
Provision as at the end of the year	96.62	116 24
Fair value of plan assets as at the end of the year	16.02	14.92
Provision recognised in the Balance Sheet	80.60	101.32
VI. Percentage of each category of plan assets to total fair value of plan assets		51
Insurer managed funds	100%	1009
VII. Amount recognised in the Statement of Profit and Loss		
Current service cost	16 17	25.74
Interest cost	8.59	6.46
Expected return on plan assets	(1,10)	(1.03
Net actuarial (gain)/loss to be on obligation		
Expense recognised in Statement of Profit and Loss	23 66	31.17





Notes to the financial statements for the year ended 31st March, 2019

VIII. Amount recognised in the Other Comprehensive Income (OCI)		
Due to Change in financial assumptions	11.73	(2.81)
Due to Change in demographic assumption	(1 54)	-
Due to Change in experience assumption	(18.30)	(4.56)
Expected return on plan assets		1.03
Total remasurement recognised in OCI	(8.11)	(6.34)
IX. Balance Sheet reconciliation		
Opening net provision	101.32	84.64
Expenses recognised in Profit & Loss	23 66	31.17
Actual Employer Contribution	(4.73)	(8.15)
Acquisition adjustments	(31.54)	÷ +
Total Remeasurement recognised in OCI	(8.11)	(6.34)
Closing net provision	80.60	101.32

Salary escalation assumption has been set in discussions with the enterprise based on their estimates of overall long-term salary growth (e) rates after taking into consideration expected earnings inflation as well as performance and seniority related increases.

41.2

Sensitivity analysis:		(Rs. in Lakh)
Particulars	Changes in assumptions	Effect on Gratuity obligation
For the year ended 31st March, 2018		
Salary growth rate	+0.50%	204.08
	-0.50%	194.55
Discount rate	+0 50%	194,59
	-0.50%	204.08
For the year ended 31st March, 2019		
Salary growth rate	+0.10%	99.08
1	-0.10%	94.27
Discount rate	+0.10%	94.27
	-0.10%	99.08

The above sensitivity analysis is based on a change in an assumption while holding all other assumptions constant. In practice, this is unlikely to occur, and changes in some of the assumptions may be correlated. In presenting the above sensitivity analysis, the present value of defined benefit obligation has been calculated using the projected unit credit method at the end of reporting period, which is the same as that applied in calculating the defined obligation liability recognised in the balance sheet.

These plans typically expose the Company to actuarial risks such as. longevity risk and salary risk.

- Interest risk A decrease in the discount rate will increase the plan provision. (a)
- Longevity risk The present value of the defined benefit plan provision is calculated by reference to the best estimate of the mortality of (b) plan participants both during and after their employment. An increase in the life expectancy of the plan participants. As such, an increase the plan's provision.
- Salary risk The present value of the defined plan provision is calculated by reference to the future salaries of plan participants. As such, (c) as increase in the salary of the plan participants will increase the plan's provision.
- 41.3 The weighted average duration of the defined benefit obligation at the end of the reporting period is 5 years (31st March, 2018 6 years).

42 Preparation of financial statements on " Going Concern" basis

In view of the focused emphasis of the Government on logistics infrastructure sector, the proposed restructuring (Refer note no. 50) and considering the fact that the facilities are yet to achieve operational potential besides the strategic locations of the facilities, the management's future outlook of its business is very promising. Accordingly, the financials have been prepared on going concern basis even though the Company continues to incur losses.

- Certain creditors have initiated legal proceedings against the Company and its Directors, and the Company has defaulted in payment of 43 instalments of consent terms for which the Company is in process of negotiating and finalising the revised consent terms and/or making representations to the respective forum. Majority of the creditors have been settled over the past few years and some of the creditors have also shown interest and faith in the logistics infrastructure sector and are being allotted equity shares of the Company.
- The Company is engaged in the business of development, operations and maintenance of Free Trade and Warehousing Zone (FTWZ) 44 and Domestic Warehousing Zone. During the year, certain portion of land which was classified under Property, Plant and Equipment (PPE) is now transferred to inventories at their carrying amounts for future developments.



/

Out of the above land parcels, during the year, the Company has entered into 2 lease agreements aggregating to 5.50 Acres of land with a wholly owned subsidiary company for development of warehouses at FTWZ. Panvel and recognised the revenue from such long-term lease during the year.

45 Loans from various lenders have been assigned by banks to Edelweiss Assets Reconstruction Company Limited (EARC). EARC had restructured the loan and executed the Restructuring Agreement (RA) dated 31st March, 2017 In accordance with RA. EARC has converted part debt into restructured debt, balance assigned loan is to be converted into 3,21,62,304 equity shares and 64,23,329 zero percent optionally convertible redeemable preference shares (OCRPS – Series I) of face value of Rs.10 each at a price of Rs.1,000 each (including premium of Rs.990) of the Company, as per extant SEBI rules and regulations. The EARC has availed the right of conversion of OCRPS into equity.

During the year ended 31st March, 2019:-

- (i) In aggregate 1,38,97,516 equity shares of 2 each on conversion of OCRPS Series I have been allotted to EARC.
- Pursuant to RA, the Company has allotted 15,00,000 Equity Shares of face value of Rs.2 each to the Promoter upon conversion of equal number of warrants.
- 46 The Company has defaulted in agreed repayment schedule of Restructuring Agreement (RA). As per debt covenant, the Company is required to adhere to repayment schedule and any short payment gives Edelweiss Asset Reconstruction Company (EARC) the right to convert whole of the outstanding amount of restructured rupee loan and/or part of the default amount into fully paid up equity shares of the Company. No such notice of conversion in writing has been given by EARC and the Company continues to disclose the amount as non-current and current borrowings as per repayment schedule, in the Balance Sheet.

47 Corporate Guarantees

The Company has issued a corporate guarantee of Rs. 27,724.43 Lakh to the lenders of Arshiya Northern FTWZ limited (ANFTWZ), a subsidiary Company. This guarantee has been invoked by the lenders since ANFTWZ had defaulted in servicing its borrowings towards principal and interest. The Company carried out fair valuation of this corporate guarantee through an independent chartered accountant firm and as per their report the value of security created in favour of the lender is higher than the total liability towards borrowing. Accordingly, no provision is required towards the guarantee so invoked.

48 Cash Seized by Income Tax

The amount of Rs. 100 Lakh cash seized by the Income Tax department at the time of search on 13th June, 2014 has adjusted the said cash seized against demand of the Company and to be specific against Assessment Year 2014-2015. While there is a demand in Assessment Year 2014-2015, the same is contested and the said demand is reflected in Contingent Liability (Refer note no. 37)

- 49 Scheme of arrangement and amalgamation u/s 230 to 232 and other applicable provisions of the Companies Act, 2013 has been filed before the National Company Law Tribunal ('NCLT") between Arshiya Rail Infrastructure Limited, Arshiya Industrial & Distribution Hub Limited and Arshiya Transport & Handling Limited and their respective shareholders. The scheme is conditional on various approval / sanctions and is effective thereafter; accordingly no effect of the said Scheme is given in the financial statements. The meeting of the creditors was held on 6th May, 2019. The Scheme(s) shall be given effect after receipt of necessary approvals.
- 50 The Board of Directors of the Company in their meeting held on 24th May, 2018, has approved a scheme to further reorganize its corporate structure spread across various group companies, in order to integrate/consolidate it's operations by reorganizing different businesses into two entities.

This Scheme is presented under Sections 230 to 232 read with Sections 66 and 52 and other applicable provisions of the Companies Act, 2013 for demerger of "Domestic warehousing business" of the company into Arshiya Rail Infrastructure Limited. This proposed scheme of arrangement is conditional upon approval of an ongoing scheme of group companies i.e. merger of Arshiya Rail Infrastructure Limited, Arshiya Industrial and Distribution Hub Limited and Arshiya Transport & Handling Limited, which is pending with NCLT. No accounting impact and disclosures are considered necessary at this stage pending requisite regulatory approvals.

51 Maharashtra VAT Refund Receivable

As per the Notification dated 16th May, 2013 issued by Government of Maharashtra, MVAT exemption /refund is available to SEZ Developer after 15th October, 2011 (record date). However, the Company has claimed refund of Rs 1,684,55 Lakh in respect of transactions prior to record date, as the Company is of the view that the State Government has exempted it from Local taxes, levies and duties on goods required for authorized operations by a Developer vide GR dated 12th October, 2001 passed by Industries, leavies and Labour Department. Government of Maharashtra. The Company has filed an appeal before High Court of Bombay challenging the Constitutional validity of MVAT on various grounds and has claimed refund of Rs 1,108.80 Lakhs. The Appeal has been admitted, issues are framed and final hearing is pending. Further MVAT refund claim of 575.75 Lakhs is pending with Sales Tax Department as the matter is of similar case. Accordingly, these financial statements reflect a sum of Rs 1,684.55 Lakhs as refund receivable on account of MVAT. In case the refund is not granted, the necessary adjustment entries shall be recorded in the year in which finality is reached.

52 As per Ind-AS 108 "Operating Segment", information has been provided along with the consolidated financial statements of the Group,

VEDI Ч

1.15

Notes to the financial statements for the year ended 31st March, 2019

The Company has Business Conducting and Services Agreement with Arshiya Lifestyle Limited (ALL) (wholly owned subsidiary) in relation to operation of Six Warehouses taken on sub-lease from Ascendas Panvel FTWZ Limited (formerly known as Arshiya Rail Siding and Infrastructure Limited) ("APFL") and operation of Container Yard and Open Yard owned by the Company. The aforesaid Business Conducting and Services Agreement is to be read in the overall context of Lease Deed dated 3rd February, 2018 and other agreements and documents entered into in connection with lease of Six Warehouses by the Company, being owner, to APFL and Sub-Lease of the said Six Warehouses by APFL to ALL and transfer of all rights and obligations under the Existing Unit Holder Agreements entered into by the Company to and in favour of ALL. The Company for the administration and operational expediency entrusted ALL to carry out operating and managing the open yard and the container yard and warehouses whereby ALL agreed to undertake and conduct the business of operating and managing the open yard and the container yard and warehouses and provide other services by utilizing the infrastructure facilities provided by the Company. ALL shall also received all the incomes generated from the warehouses and storage yard, bearing the cost and expenses to operate and maintain the warehouses and storage yard. Pursuant to the aforesaid Business Conducting and Services Agreement, the ALL will pay 99% of excess revenue / Total Income over all the expenses / charges / provisions to the Company as Business Conducting Fees. Accordingly, the Company has recognised as Business Conducting fees Rs. 3,351.11 Lakh during the year ended 3rst March, 2019 (31st March 2018 - Rs. 972.91 Lakh)

54 Investments

- 54.1 The Company's non-current investment in subsidiaries and its non-current / current loans dues from subsidiaries aggregating to Rs. 156,510.60 Lakh. The net worth of aforesaid subsidiaries has either been fully eroded or most of the entities have incurred losses and have accumulated losses. The operations of these subsidiaries are dependent on business plans and various steps taken by the management. The management of Arshiya Group also taken up steps to reorganize its corporate structure spread across various group companies in order to integrate/ consolidate its operations by housing different businesses into two different entities/ separate verticals, through scheme of arrangements. The Company has also obtained valuation report from registered valuer. Based on this and other factors stated, management has considered that no adjustment are required to be made to the carrying value of investment and loans as at 31st March 2019.
- 54.2 The Company has divested its entire investment in a subsidiary company namely Mira Supply Chain Management Private Limited (formerly known as Arshiya Supply Chain Management Private Limited). As a result, the Company has accounted net loss of Rs. 4,338.19 lakhs for the year ended 31st March, 2018 and this loss is grouped under exceptional item.
- 55 The Company, interalia, it's subsidiaries and promoters has executed Lease Deed on 3rd February 2018, in favour of a SPV of Ascendas Property Fund (India) Pte. Limited ("Ascendas" part of the Ascendas-Singbridge Group, Singapore) for grant of leasehold rights of six warehouses at FTWZ Panvel, along with underlying land of those warehouses, identified assets and infrastructure facilities on an initial lease term of 30 (thirty) years. The said transaction is for a total consideration of Rs. 53,400 Lakh (or Rupees Five hundred and thirty four crore), with an upfront lease payment/lump sum rent of Rs. 43,400 Lakh (or Rupees Four hundred and thirty four crore). The balance of Rs. 10,000 Lakh (or Rupees One hundred crore) will be received over four years from transaction closing based on certain performance milestones. The transaction also envisages the terms for construction funding by Ascendas for future growth of the company's business. The Company already possesses the requisite land for the future development.

On transaction closing date of 3rd February 2018, the SPV has acquired long-term leasehold rights from the Company and the same are leased back under an operating lease arrangement pursuant to execution of sub-lease deed dated 3rd February 2018 to Arshiya Lifestyle Limited ("ALL"), a wholly owned subsidiary of the Company, for a sub-lease term of 6 (six) years, renewable as per mutually agreed terms, in consideration of pre-agreed rentals.

Accordingly during the year ended 31st March, 2018 the Company has reduced the value of assets, granted on leasehold rights to Ascendas Panvel FTWZ Limited (formerly known as Arshiya Rail Siding and Infrastructure Limited ("APFL")), from its fixed assets. The gain on grant of leasehold rights to APFL amounting to Rs. 15,633.29 lakh has been credited to profit and loss account of the Company and is disclosed as an exceptional item.

Based on the above, ALL would operate and manage these six warehouses and pay the lease rentals to APFL as defined in the sub-lease agreement. Hence from 3rd February, 2018 onwards all revenue from these assets will be accounted by ALL. However the company will recognise the net revenue in terms of a business conducting agreement entered into between the Company and ALL.



A

56 Related party disclosures, as required by Indian Accounting Standard 24 "Related Party Disclosures" (IND AS-24) as given below:

S. No.	Name of the entity	Country of Incorporation	Proportion of interest (including beneficial interest)/voting power (eith directly/indirectly or through subsidiaries)	
			31st March, 2019	31st March, 2018
	Direct Subsidiaries:			
(i)	Arshiya Rail Infrastructure Limited	India	100%	100%
(11)	Arshiya Northern FTWZ Limited	India	100%	100%
(iii)	Arshiya Industrial & Distribution Hub Limited	India	100%	100%
(iv)	Arshiya Transport and Handling Limited	India	100%	100%
(v)	Arshiya Technologies (India) Private Limited	India	100%	100%
	Arshiya Lifestyle Limited	India	100%	100%
(vii)	Arshiya Logistics Services Limited (Formerly known as Laxmipati Balaji Exim Trading Limited) (w.e.f. 13th Jun, 2017)	India	100%	100%
(viii)	Anomalous Infra Private Limited (w.e.f. 15th October, 2018)	India	100%	Nil
(ix)	Arshiya Northern Projects Private Limited (w.e.f. 25th October, 2018)	India	100%	Nil
(x)	Arshiya Infrastructure Developers Private Limited (w.e.f. 9th January, 2019)	India	100%	Nil
(xi)	Laxmipati Balaji Supply Chain Management Limited (w.e.f 7th May, 2018)	India	100%	Nil
(xii)	Unrivalled Infrastructure Private Limited (w.e.f. 7th January, 2019)	India	100%	Nil
(xiii)	Mira Supply Chain Management Private Limited (formerly known as Arshiya Supply Chain Management Private Limited) (up to 22nd March, 2018)	India	Nil	Nil
	Indirect Subsidiaries:			-
	Held through Arshiya Logistics Services Limited:			
(xiv)	Arshiya 3PL Services Private Limited (w.e.f. 27th August, 2018)	India	India 100%	Nii
-	Held through Arshiya Rail Infrastructure Limited:			
(xv)	Ascendas Panvel FTWZ Limited (formerly known as Arshiya Rail Siding and Infrastructure Limited) (up to 3rd February, 2018)	India	Nil	Nil

(I) Person having significant influence over the Company Mr. Ajay S Mittal - Chairman and Managing Director

Mrs. Archana A Mittal - Joint Managing Director

(II) Key Management Personnel

Mr. Ashish Bairagra - Independent Director Mr. Mukesh Kacker - Independent Director

Mr. Rishabh Shah - Independent Director

Prof. G. Raghuram - Independent Director (till 15th May, 2017)

Mr. Tara Sankar Bhattacharya - Independent Director (w.e.f. 24th May, 2018) Mrs. Savita Dalal - Company Secretary and Compliance Officer

Mr. S. Maheshwari - Chief Financial Officer

(III) Relative of Person having significant influence over the Company Mr. Ananya Mittal - Corporate Strategy Officer (Arshiya Group)

The nature and amount of transactions with the above related parties are as follows

Nature of transactions	Name of the Party	31st March, 2019	31st March, 2018
Revenue from operations	Mira Supply Chain Management Private Limited	4	1,215.97
	Arshiya Logistics Services Limited	4	2,694.77
	Arshiya Lifestyle Limited	3,351.11	972.91
	Anomalous Infra Private Limited	7,001.49	
Income billed to customer on behalf of the Subsidiary Company	Arshiya Lifestyle Limited	1,777.01	257.65
Advance Rent Income	Mira Supply Chain Management Private		112.7



Ox som

Anomalous Infra Private Limited	2.336 62	
Arshiya Rail Infrastructure Limited	63.70	56.88
Arshiya Northern FTWZ Limited	20.06	17.91
	11.60	10.3
Arshiya Transport and Handling Limited	112.42	100.3
Arshiya Lifestyle Limited	212.24	
Arshiva Rail Infrastructure Limited	246.46	240.5
Arshiya Northern FTWZ Limited	126.80	126.8
Arshiya Industrial & Distribution Hub	133.53	178.9
	168,14	0.4
Mira Supply Chain Management Private		89.9
Limited		
Mira Supply Chain Management Private		104.7
Limited		14.63
Arshiya Rail Infrastructure Limited	419.55	418.2
Arshiya Northern FTWZ Limited	41.95	566.7
	101.95	418.2
Arshiya Lifestyle Limited	615.65	
Arshiya Logistics Services Limited	-	(28.71
Arshiya Northern FTWZ Limited	33.76	8.4
Mr. C. Mahashurari	202.68	185.1
Ms. Savita Dalal	16.61	100.
		2.0
Prof. G. Raghuram	-	0.3
Mr. Rishabh Shah	0,40	1.
Mr. T S Bhattacharya	0.70	
Arshiya Rail Infrastructure Limited	788.31	4,831.
Arshiya Northern FTWZ Limited	2,533.34	7,188.
	719.57	14,470.3
Arshiya Technologies (India) Private	0.70	1.3
	5 52	4.
	-	10,706.0
Arshiya Logistics Services Limited	4	3,440.
	Contraction of the second seco	-0
Arshiya Infrastructure Developers Private		
Limited	0.01	
Arshiya Northern Projects Private Limited	0.28	
Laxmipati Balaji Supply Chain	16.46	
	0.01	
		3,751.
		337.
Limited	102.17	514.
	1.00	
Arshiya Transport and Handling Limited	5.15	
Arshiya Lifestyle Limited	3,128.96	725.
		3,361.
Management Limited	6.25	
Mr. Ajay S Mittal	788.46	3,064
Mrs. Archana A Mittal	1,116.14	4,260
Mŕ. Ajay S Mittal	1,153.32	2,594
	Arshiya Northern FTWZ Limited Arshiya Industrial & Distribution Hub Limited Arshiya Transport and Handling Limited Arshiya Transport and Handling Limited Arshiya Infestructure Limited Arshiya Northern FTWZ Limited Arshiya Industrial & Distribution Hub Limited Arshiya Lifestyle Limited Arshiya Lifestyle Limited Mira Supply Chain Management Private Limited Mira Supply Chain Management Private Limited Arshiya Rail Infrastructure Limited Arshiya Northern FTWZ Limited Arshiya Northern FTWZ Limited Arshiya Northern FTWZ Limited Mr. S. Maheshwari Ms. Savita Dalal Mr. Ashishkumar Bairagra Mr. Nukesh Kacker Prof. G. Raghuram Mr. T S Bhattacharya Arshiya Northern FTWZ Limited Arshiya Northern FTWZ Limited Arshiya Northern FTWZ Limited Mr. Ashishkumar Bairagra Mr. Nukesh Kacker Prof. G. Raghuram Mr. T S Bhattacharya Arshiya Rail Infrastructure Limited Arshiya Northern FTWZ Limited <	Arshiya Northern FTWZ Limited 20.06 Arshiya Industrial & Distribution Hub 11.60 Arshiya Transport and Handling Limited 112.42 Arshiya Rail Infrastructure Limited 246.46 Arshiya Infrastructure Limited 126.80 Arshiya Industrial & Distribution Hub 133.53 Arshiya Industrial & Distribution Hub 133.53 Arshiya Industrial & Distribution Hub 133.53 Arshiya Industrial & Distribution Hub 168.14 Mira Supply Chain Management Private - Limited 419.55 Arshiya Industrial & Distribution Hub 101.95 Arshiya Industrial & Distribution Hub 101.95 Limited 615.65 Arshiya Industrial & Distribution Hub 101.95 Limited 615.65 Arshiya Northern FTWZ Limited 33.76 Mr. S. Maheshwari 203.68 Ms. Savita Dalai 16.61 Mr. Ashishkumar Bairagra 1.15 Mr. Rishabh Shah 0.40 Prof. G. Raghuram - Mr. T S Bhatacharya 0.70 Arshiya Rail Infrastructure Limited 788.31 Arshiya R

Nature	Related Party	As at 31st March, 2019	As at 31st March, 2018
Closing Balances			(Rs. in Lak
Corporate guarantees received	Arshiya Northern FTWZ Limited	550.00	
Corporate guarantees received	Arshiya Rail Infrastructure Limited		
Corporate guarantees reduced	Arshiya Lifestyle Limited	4,659.61	1
Corporate guarantees given	Arshiya Lifestyle Limited	241.87	31,316.
Security Deposit repaid/adjusted	Mira Supply Chain Management Private Limited	-	17,226
Security Deposit received	Mira Supply Chain Management Private Limited		11,500
investments sold to	Arshiya Industrial & Distribution Hub Limited		330.
investments purchased from	Mrs. Archana A Mittal	4	0.
Share Warrants converted into Equity	Mr. Ajay S Mittal	860.25	4,959.
	Equity Share	-	583
	Share Warrants Mr. S. Maheshwari		5,820.0
Issue of Equity Shares and Warrants	Mr. Ajay S Mittal Equity Share		5,835.
	Unrivalled Infrastructure Private Limited	1.00	
	Arshiya Northern Projects Private Limited	5.00	-
	Arshiya Infrastructure Developers Private Limited	1.00	
	Laxmipati Balaji Supply Chain Management Limited	5.00	
nvestment in Subsidiaries	Anomalous Infra Private Limited	11.00	
	Arshiya Logistics Services Limited		50.

Nature	Related Party	As at 31st March, 2019	As at 31st March, 2018
	Arshiya Rail Infrastructure Limited	304.38	2,029.26
	Arshiya Northern FTWZ Limited	9,900.62	7,585.45
	Arshiya Industrial & Distribution Hub Limited	15,202.12	14,471.17
	Arshiya Transport and Handling Limited	1,054.68	941.89
	Arshiya Technologies (India) Private Limited	2.40	2.70
	Arshiya Lifestyle Limited	5,584.17	9,980.99
Loans and advances given	Anomalous Infra Private Limited	0.39	
a second the second second	Unrivalled Infrastructure Private Limited	0.01	
	Arshiya Infrastructure Developers Private Limited	0.01	
	Arshiya Northern Projects Private Limited	0.28	
	Laxmipati Balaji Supply Chain Management Limited	10.21	
	Arshiya 3PL Services Private Limited	0.01	
	Arshiya Logistics Services Limited	-	288.87
Trade receivables	Arshiya Lifestyle Limited	553.40	104.81
	Anomalous Infra Private Limited	6,062.99	
Trade Payables	Arshiya Northern FTWZ Limited	-	8.27
a data data data data data data data da	Mr. Ajay S Mittal	104.89	469.75
Loans taken	Mrs. Archana A Mittal	55.22	1,468.01
Share warrants	Mr. Ajay S Mittal	<u>і</u>	860.25
Bernard Contact to Aller	Mr. Ajay S Mittal	1,96,920.00	1,86,370.00
Personal guarantees taken	Mrs. Archana A Mittal	1,96,370.00	1,86,370.00
Equity Shares (excluding share premium)	Mr. S. Maheshwari	20.00	20.00



S X Aom &

	Arshiya Rail Infrastructure Limited	854.69	1,041,15
Financial Guarantee Obligations	Arshiya Industrial & Distribution Hub Limited	536.06	665.90
	Arshiya Northern FTWZ Limited	241.81	368.60
	Arshiya Lifestyle Limited	333.25	498.03
	Arshiya Industrial & Distribution Hub Limited	44,499.72	44,499.72
	Arshiya Northern FTWZ Limited	44,625,29	44,625.29
	Arshiya Rail Infrastructure Limited	38,369.21	38,369.21
	Arshiya Transport and Handling Limited	5.00	5.00
	Arshiya Technologies (India) Private Limited	2.00	2.00
	Arshiya Lifestyle Limited	14.85	14.86
investment in subsidiaries (Refer Note No.7)	Arshiya Logistics Services Limited	155 50	155.50
	Laxmipati Balaji Supply Chain Management Private Limited	5,00	19
	Anomalous Infra Private Limited	11.00	
	Arshiya Northern Projects Private Limited	5.00	
	Arshiya Infrastructure Developers Private	1.00	
	Unrivalled Infrastructure Private Limited	1,00	
	Arshiya Industrial & Distribution Hub Limited	1,116.48	1.112.78
	Arshiva Northern FTWZ Limited	696.96	696.96
Deemed Equity (Refer Note No. 7)	Arshiya Rail Infrastructure Limited	1,795.82	1,735.82
- A COLOR OF A COLOR OF A COLOR OF A	Arshiya Transport and Handling Limited	302.40	302.40
	Arshiya Lifestyle Limited	1,981.96	498.49
	Anomalous Infra Private Limited	1,091,83	
	La constanta da const	00.450.00	
	Arshiya Northern FTWZ Limited	28,450.00	28,450.00
Corporate guarantees/securities issued to	Arshiya Rail Infrastructure Limited Arshiya Industrial & Distribution Hub Limited	51,200.19 29,600.00	48,200.19
	Arshiya Lifestyle Limited	26,130.94	30,548.6
Corporate guarantees/securities received from	Arshiya Rail Infrastructure Limited Arshiya Northern FTWZ Limited	550.00	

57 Loans and Advances in the nature of Loans to Subsidiaries (Pursuant to the SEBI (Listing Obligation and Disclosure Requirement) Regulations, 2015)

(Rs. in Lakh			
Name of the Subsidiary	Year	Amount outstanding as on March 31,	Maximum amount outstanding during the year
Arshiya Rail Infrastructure Limited	2019	304.38	2,128.49
	2018	2,029.26	5,434.21
Arshiya Transport and Handling Limited	2019	1,054.68	1,059,58
tende trent betrever tenden grennen s	2018	941.89	1,054.27
Arshiya Industrial & Distribution Hub Limited	2019	15,202,12	15,202.13
	2018	14,471.17	14,482.75
Arshiya Northern FTWZ Limited	2019	9,900.62	9,900.62
And a descent of the second of the	2018	7,585.45	7,653.69
Arshiva Technologies (India) Private Limited	2019	2.40	3.40
	2018	2.70	2.66
Arshiva Logistics Services Limited	2019		
	2018		853.39
Arshiva Lifestyle Limited	2019	5,584 17	9,980.99
	2018	9,980.99	10,545.51
Anomalous Infra Private Limited	2019	0.39	0.39
	2018		
Unrivalled Infrastructure Private Limited	2019	0.01	0.01
	2018		-
Arshiva Infrastructure Developers Private Limited	2019	0.01	0.01
	2018		
Arshiva Northern Projects Private Limited	2019	0.28	0.28
/ add da representer - 14	2018		and the second second
Laxmipati Balaji Supply Chain Management Limited	2019	10.21	10.21
complete careful and provide the second s	2018		
Arshiva 3PL Services Private Limited	2019	0.01	0.01
And the second se	2018		-
	2019	32,059.28	
Total	2018	35,011.46	

Loans and Advances to Subsidiaries



Notes to the financial statements for the year ended 31st March, 2019

58 Earnings per share:

Particulars	Year ended 31st March 2019	Year ended 31st March 2018
Profit/(Loss) for the year (Rs. in Lakh)	(3,516.49)	3,895.66
Add: Interest adjustment on account of 0% Optionally Convertible Redeemable Preference Shares (OCRPS)		946.32
Total Profit/(Loss) for the year for diluted EPS (Rs. in Lakh)	(3,516,49)	4,841.98
Number of equity shares	and and a	
Weighted average number of equity shares (Number)	23,80,15,279	18,31,20,902
Add: Adjustment on account of Share Warrants	1	5,91,781
Add: Adjustment on account of 0% Optionally Convertible Redeemable Preference Shares (OCRPS)	-	54,82,856
Total Weighted average number of equity shares/shares warrants/OCRPS	23,80,15,279	18,91,95,538
Nominal value per share (Amount in Rs.)	2.00	2.00
Earnings per share – Basic and Diluted (Amount in Rs.)	(1.48)	2.13

0% OCRPS and share warrants had an anti diluting effect on earning per share hence have not been consider for the purpose of computing dilutive earning per share during the previous year.

59 Taxation

- 59.1 In view of loss for the year, no provision for current tax has been made.
- 59.2 The Company has not recognised any deferred tax assets on deductible temporary differences, unused tax losses as it is not probable that the Company will have sufficient future taxable profit which can be available against the available tax losses.

59.3 Unused tax losses for which no deferred tax assets has been recognised

Assessment Year	Business Loss	Unabsorbed Depreciation	Available for utilization till
2014-2015	±.	1,201.54	A.Y. 2022-2023
2015-2016		4,322.75	A Y. 2023-2024
2016-2017	984.75	4,011.34	A Y 2024-2025
2017-2018	45,539.91	3,826.58	A.Y. 2025-2026
2018-2019	13,483.44	559.56	A.Y. 2026-2027
Total	60,008.10	13,921.77	

Assessment Year	Long term Capital Loss	Available for utilization till
2016-2017	1,658.88	A.Y. 2024-2025
Total	1,658.88	

Deferred tax assets as at 31st March, 2019 Rs. 1,283,30 Lakh (31st March, 2018 - Rs. 18,584.87 Lakh) has not been recognised, as there is no convincing evidence that sufficient taxable profits will be available against which the unadjusted tax losses will be utilized by the Company. Details of deferred tax assets are mentioned below:

		(Rs. in Lakh)
Particulars	As at 31st March, 2019	As at 31st March, 2018
Property, plant and equipment	5,919.01	6,460.79
Unabsorbed depreciation	(3,619,66)	(3,557,17)
Expenses allowable on payments under section 43B and 40 (a) (ia)	(3,471.31)	(508.83)
Unabsorbed losses	(16,033.42)	(17,805.69)
Financial Instruments	15,922.08	(3,173.97)
Total Deferred tax assets	(1,283.30)	(18,584.87)

60 During the year, the Company has allocated certain common costs and expenses incurred by it, being the Holding Company, to its subsidiaries aggregating to Rs.563.45 Lakh (31st March, 2018 - Rs. 1,403,22 Lakh.) based on management's estimates of such costs and expenses attributable to them. Hence, Employee benefit expenses (Refer Note No. 32) and certain expenses stated under Other expenses (Refer Note No. 35) are presented as net of allocation of certain common costs and expenses.



×

Notes to the financial statements for the year ended 31st March, 2019

61 Financial Risk Management

The Company's principal financial liabilities comprises of borrowings, trade and other payables and financial guarantees contracts. The main purpose of these financial liabilities is to manage for the Company's and subsidiaries's operations. The Company's financial assets comprises of investment, loans, trade and other receivables, cash and deposits that arises directly from its operations.

The Company's activities expose it to variety of financial risks including credit risk, liquidity risk and market risk. The Company's risks management assessment, management and processes are established to identify and analyze the risks faced by the Company to set up appropriate risks limits and controls, and to monitor such risks and compliances with the same. Risks assessment and management policies and processes are reviewed regularly to reflect changes in market conditions and the Company's activities.

Risk	Exposure arising from	Measurement	Management
Credit risk	Cash and cash equivalents, trade receivables, financial assets measured at amortised cost.		Regular review of credit limits
Liquidity risk	Borrowings and other liabilities	Rolling cash flow forecasts	Availability of borrowing facilities
Market risk – foreign exchange	Recognised financial assets and liabilities not denominated in Indian rupee (INR)		Unhedged
Market risk – interest rate	Long-term borrowings at variable rates	Sensitivity analysis	Unhedged

The Company's risk management is carried out by a corporate finance team under the policies approved by the Board of Directors. The Board provides written principles for overall risk management as well as policies covering specific areas, such as credit risk, interest rate risk.

(a) Credit Risk

The Company is exposed to credit risk, which is risk that counterparty will default on its contractual obligation resulting in a financial loss to the Company. Credit risk arises from cash and cash equivalents as well as credit exposures to trade customers including outstanding receivables.

Trade receivables are typically unsecured and are derived from revenue earned from customers located in India and outside India. Credit risk has always been managed by the Company through continuously monitoring the creditworthiness of customers to which the Company grants credit terms in the normal course of business. Outstanding customer receivables are regularly monitored. Credit risk is high as only few customers' account for majority of the revenue in the year presented. On account of adoption of Ind AS 109, the Company uses expected credit loss model to assess the impairment loss or gain.

(b) Liquidity Risk

Liquidity risk is the risk that the Company may not be able to meet its financial obligations without incurring unacceptable losses. The Company's objective is to, at all times, maintain optimum levels of liquidity to meet its cash and collateral requirements. The Company limits its liquidity risk by ensuring regular monitoring of funds from trade and other receivables. The Company relies on assets light business model through monetization of assets and tie-up of construction funding and operating cash flows to meet its needs for funds.

The table below provides undiscounted cash flows towards financial liabilities into relevant maturity based on the remaining period at the balance sheet to the contractual maturity date.

Particulars	less than 1 year	1 to 5 years	More than 5 year
31st March, 2019			
Financial liabilities		COLUMN A	
Borrowings	25,173.05	61,118 62	-0-
Trade payables	622 79	21	
Creditors for Capital Goods	265.92		1
Financial guarantee obligations	575.05	1,390.76	*
Other financial liabilities	12,926.28	221.96	÷
Te	otal 39,563.09	62,731.34	
31st March, 2018	1		
Financial liabilities		CONTRACT IN	
Borrowings	25,679.56	59,630 96	
OCRPS (Equity and Liability Component)			57,646,19
Trade payables	582.55	8	
Creditors for Capital Goods	1,137.29	A	
Financial guarantee obligations	606.11	2,191.60	
Other financial liabilities	4,595.36		-
Ti	atal 32,600.87	61,822.56	57,646.19

X y som 0

98

Notes to the financial statements for the year ended 31st March, 2019

(c) Market Risk

Market Risk is the risk that the fair value of future cash flow of a financial instruments will fluctuate because of volatility of prices in the financial markets. Market risk can be further segregated as: 1) Foreign currency risk and 2) Interest rate risk.

1 Foreign currency risk

Foreign currency risk is the risk that the fair value or future cash flow or an exposure will fluctuate because of changes in foreign exchange rates. The Company's exposure to the risk of changes in foreign exchange rates relates primarily to the Company's operating activities.

1.1 Foreign currency risk exposure

(i) Details of foreign currency transactions/ balances not hedged by derivative instruments or otherwise are as under-

Particulars	Financial Year Ended	Foreign currency amount	Equivalent amount in INR	
		(Amount in Lakh)	(Rs. in Lakh)	
Trade Receivables		a contact of the second		
1100	31st March, 2019	3.15	218.17	
USD	31st March, 2018	5.32	341.98	
12/201	31st March, 2019	0.02	1.48	
EUR	31st March, 2018	0.05	3.89	
Security Deposit from customers				
USD	31st March, 2019	5.47	378.19	
	31st March, 2018	5.47	351.42	

1.2 Sensitivity

The Sensitivity of profit or loss to changes in the exchange rate arises mainly from foreign currency denominated financial instruments

	Increase/(decrease) in p	profit before tax
Particulars	As at 31st March, 2019	As at 31st March, 2018
FX rate - increase by 1% on closing rate of reporting date	(1.59)	(0.06)
FX rate - (decrease) by 1% on closing rate of reporting date	1.59	0,06

The above amounts have been disclosed based on the accounting policy for exchange differences.

2 Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. Majority of the Company's borrowings is fixed rate borrowings carried at amortised cost, therefore not subject to interest rate risk as defined in IND AS - 107, since neither carrying amount nor the future cash flows will fluctuate because of a change in market interest rates. The Company's interest rate risk arises from long term borrowings with variable rates, which expose the Company to cash flow interest rate risk. The Company's borrowings at the variable rate were mainly denominated in Rupees.

2.1 Interest rate risk exposure

The exposure of the Company's borrowings to interest rate changes at the end of the reporting period are as follows.

		(Rs. in Lakh)
Particulars	As at 31st March, 2019	As at 31st March, 2018
Variable rate borrowing	1,472.84	1,491.67

2.2 Sensitivity of Interest

Profit or loss is sensitive to higher/lower interest expenses from borrowings as a result of changes in interest rates. (Rs. in Lakh)

	Increase/(decrease) in p	rofit before tax
Particulars	31st March, 2019	31st March, 2018
50 bps increase the profit before tax by	(7.36)	(7.46)
50 bps decrease the profit before tax by	7 36	7.46



() am

(De in Lakh)

Notes to the financial statements for the year ended 31st March, 2019

- 62 Fair Value Measurements
- (i)

	Carrying A	mount	Fair	Value
Particulars	As at 31st March, 2019	As at 31st March, 2018	As at 31st March, 2019	As at 31st March, 2018
Financial Assets				
Amortised cost				
Trade Receivables	6,876.14	764.60	6.876.14	764.60
Loans	32,059.28	35,011.46	32,059.28	35,011.46
Cash and Cash Equivalents	5.86	135.69	5.86	135.69
Other Bank Balances	15.17	0.04	15.17	0.04
Other Financial Assets	4,082.95	1,848.71	4,082.95	1,848,71
Total	43,039.40	37,760.50	43,039.40	37,760.50
Financial Liabilities				
Amortised cost				
Borrowings	85,440.25	94,495.95	85,440.25	94,495.95
Trade Payables	622.79	582.55	622.79	582.55
Creditors for Capital Goods	265.92	1,137.29	265.92	1,137.29
Security Deposits	401.30	385.68	401.30	385.68
Financial guarantee obligations	1,965.81	2,797.71	1,965.81	2,797.71
Other financial liabilities	12,746.94	4,209.68	12,746.94	4,209.68
Total	1,01,443.01	1,03,608.86	1,01,443.01	1,03,608.86

(De in Laber)

(ii) Fair Valuation techniques used to determine fair value

The Company maintains procedures to value financial assets or financial liabilities using the best and most relevant data available. The fair values of the financial assets and liabilities are included at the amount that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date.

The following methods and assumptions were used to estimate the fair values:

- (a) The Company assessed that the fair value of cash and cash equivalent, trade receivables, trade payables, and other current financial assets and liabilities approximate their carrying amounts largely due to the short term maturities of these instruments.
- (b) The fair values for loans to subsidiaries, security deposits and other financial liabilities were calculated based on cash flows discounted using a current lending rate. They are classified as level 3 fair values in the Fair value hierarchy due to the inclusion of unobservable inputs including counterparty credit risk.
- (c) The fair values of non current borrowings are based on discounted cash flows using a current borrowing rate. They are classified as level 3 fair values in the fair value hierarchy due to the use of unobservable inputs, including own credit risk.
- (d) Equity Investments in subsidiaries are stated at cost.

(iii) Fair value hierarchy

3

DACEN

This section explains the judgements and estimates made in determining the fair values of the financial instruments that are recognised and measure at fair value. To provide an indication about the reliability of the inputs used in determing fair value, the Company has classified its financial instruments into three levels prescribed under the accounting standard.

- (a) Level 1 Level 1 hierarchy includes financial instruments measured using quoted prices.
- (b) Level 2 The fair value of financial instruments that are not traded in an active market (for example, traded bonds, over the counter derivatives) is determined using valuation techniques which maximize the use of observable market data and rely as little as possible on entity-specific estimates. If all significant inputs required to fair value an instrument are observable, the instrument is included in level 2.
- (c) Level 3 If one or more of the significant inputs are not based on observable market data, the instrument is included in level 3. This is the case for unlisted equity shares, contingent consideration and indemnification assets included in level 3

Notes to the financial statements for the year ended 31st March, 2019

63 Capital Management

For the Company's objective when managing capital is to safeguard the Company's ability to continue going concern in order to provide the return to shareholders and benefit to other stakeholders and to maintain an optional capital structure to reduce the cost of capital, capital includes issued equity capital, share premium and all other equity reserves attributable to the equity holders of the Company

The Company manages its capital structure and makes adjustments in light of changes in economic conditions and the requirements of the financial covenants. To maintain or adjust the capital structure, the Company may adjust the dividend payment to shareholders, return capital to shareholders or issue new shares (if permitted). The Company monitors capital using a gearing ratio, which is total debts divided by total equity.

As stated in Notes to accounts, the Company is also having scheme of arrangements to reorganize the capital structure.

Particulars	As at 31st March, 2019	As at 31st March, 2018
Total Debts	97,203.35	97,796.06
Total Equity	1,71,515.57	1,64,914.64
Total debt to equity ratio (Gearing ratio)	0.57	0.59

Notes:-

(iii)

- (i) Debt is defined as long term and short term borrowings including current maturities of borrowings and interest accrued.
 - Total equity (as shown in balance sheet) includes issued capital and all other equity.

Debt Covenants

Under the terms of Restructuring Agreement (RA), the Company is required to comply with following financial covenants.

Without prior approval of lender, the Company shall not

- (a) Loans, debenture & charge Issue or subscribe to any debentures, shares, raise any loans, deposit from public, issue equity or preference capital, change its capital structure or create any charge on its assets including its cash flow or give any guarantees.
- (b) Dividend on equity shares declare/pay dividend on equity shares unless otherwise approved by the Lender/Business Monitoring Committee in accordance with the provisions of RA.

In order to achieve this overall objective, the capital management, amongst other thing, aims to ensure that it meets financial covenants attached to the interest bearing Loans and borrowings that define capital structure requirements, there have been breaches in the financial covenants of Interest bearing loans and borrowing in the current period and previous period.

The Company has not proposed any dividend in last three year in view of losses incurred.

64 Revenue from contracts with customers (IND AS 115)

(a) The Company disaggregates revenue from contracts with customers by type of services, geography and timing of revenue recognition

Revenue disaggregation by type of services is given note no 28

Revenue disaggregation by geography is as follows:

Geography	Year Ended 31st March, 2019	(Rs. in Lakh) Year Ended 31st March, 2018
India	13,139.98	8,542.02
Outside India		-
Total	13,139.98	8,542.02

Revenue disaggregation by timing of revenue recognition is as follows:

		(RS. In Lakn)	
Geography	Year Ended 31st March, 2019	Year Ended 31st March, 2018	
Services transferred over time	5,972.11	8,541.56	
Consideration on Lease of Land	7,167.87	0.46	
Total	13,139.98	8,542.02	



Reconciliation of Revenue from Operation with contract price.

Particulars	Year Ended 31st March, 2019	(Rs. in Lakh) Year Ended 31st March, 2018
Contract Price	14,916.99	8,799.67
Reduction towards credit note issued to subsidiary	1,777.01	257.65
Revenue from Operations	13,139.98	8,542.02

Transaction Price allocated to remaining performance obligations:

The aggregate amount of the transaction price allocated to the performance obligations that are to be satisfied as of 31st March, 2019 amounts to Rs. 7,109.59 Lakh as per Lease deed. The remaining performance obligation are affected by several factors including Parvel storage revenue, cash flow cover, collections within 90 days or mutually agreed. The management of the Company expects that 35% of the unsatisfied performance obligation will be recognised as revenue during the next reporting period amounting to Rs. 2,500.00 Lakh with balance in future two reporting periods thereafter.

(b) Transitional Provision - IND AS 115 - Revenue from Contracts with Customer

The Ministry of Corporate Affairs (MCA) on 28th March, 2018 notified Ind AS 115 "Revenue from contracts with customers" as part of the Companies (Indian Accounting Standards) Amendment Rules, 2018 and the same is effective for accounting period beginning on or after 1st April, 2018. The Company has applied modified retrospective approach in adopting the new standard and accordingly, the revenue from operations for the year ended 31st March, 2019 is not comparable with the previous year. As a results of change in accounting policies, adjustments to the transition provision has been made in respective item as at 1st April, 2018 with corresponding Impact to equity. Details of changes made in item along with equity have given in below table.

Particulars	Rs. in Lakh
Other Financials assets	1
Unbilled Revenue	390.41
Net Impact on other equity (Increase)	390.41

65 The Company's borrowings have been assigned by bankers to an ARC under CDR package and restructured with NBFC. The ARC and NBFC have charged penal interest/ interest amounting to Rs. 1,303.42 lakh upto the year ended March 31, 2018, which was not accepted by the Company and hence is under negotiation. In light of audit qualifications in previous year as a matter of prudence, the Company has recognised the said interest/penal interest and has accordingly restated the finance cost and other consequential impacts in the year ended March 31, 2018.

Further, during the year ended March 31, 2018 reported figure of finance cost, other Equity and Interest accrued and due on borrowings was Rs. 12,458.52 lakh, Rs. 1,61,263.30 lakh and Rs. 104.43 lakh respectively. Restated figures of finance cost, other equity and Interest accrued and due on Borrowings are Rs 13,761.94 lakh, Rs. 1,59,959.88 lakh and Rs. 1,407.85 lakh respectively. Earning Per Share (EPS) also recalculated based on the restated figures.

66 The Company has sent request letters/ emails to various Parties for confirmations of balances under borrowings, trade receivables and capital advances given to vendors and trade payables etc., to which only few parties have responded. Accordingly, impact of adjustment, if any, will be accounted as and when the same is determinable or accounts are reconciled/settled.

67 Previous year's figures have been regrouped / restated / rearranged wherever necessary.

Signatures to Notes forming part of Financial Statements

For and on behalf of the Board of Directors of For Chaturvedi & Shah LLP Arshiya Limited Chartered Accountants Firm Registration Number 101720W/W100355 NELLEIM Archana A Mittal Ajay S Mittal Vijay Napawaliya Chairman and Managing Director Joint Managing Director Partner DIN: 00703208 DIN: 00226355 Membership Number: 109859 RVED S Maheshwari Ashishkumar Bairagra Chief Financial Officer Independent Director DIN: 00049591 Dinesh Kumar Sodani Savita Dalal Place: Mumbai VP Accounts & Finance Company Secretary Date: 27th May, 2019

Consolidated Balance Sheet as at 31st March, 2019

Particulars	Notes	As at 31st March, 2019	(Rs. In Lakh As at 31st March, 2018
ASSETS			
Non-Current Assets			
(a) Property, Plant and Equipment	7	2,58,156.00	2,82,377.19
(b) Capital Work-in-Progress	1.001	79.62	
(c) Goodwill		19.17	19.17
(d) Intangible Assets	8	4,124,96	5,036,75
(e) Intangible Assets Under Development	ier:	82.21	
(f) Financial Assets	1 1		
(i) Other Financial Assets	9	1,790.02	1,732.58
(g) Other Non-Current Assets	10	6,212.59	4,822.26
(3)	1	2,70,464.57	2,93,987.95
Current assets		al. al. a ner	
(a) Inventories	11	16,505.97	15.66
(b) Financial Assets	1.1	10,000.01	15.00
(i) Trade Receivables	12	4,266,17	2,742.67
(ii) Cash and Cash Equivalents	13	990.56	1,285.84
(iii) Bank Balances Other than (ii) above	14	401.38	498.54
(iv) Loan	15	325.12	450.04
(v) Other Financial Assets	16	8.876.64	12,804.30
(c) Other Current Assets	17	3,950.30	4,289.67
(c) other ourrent Assets	1 "	35,316.14	4,209.07
		35,316.14	21,030.00
Total Assets		3,05,780.71	3,15,624.63
EQUITY AND LIABILITIES Equity (a) Equity Share capital	18	4,872.29	4,564.34
(b) Other Equity	19	48,593.46	66,937.58
		53,465.75	71,501.92
Equity Component of 0% Optionally Convertible Redeemable Preference shares (OCRPS) issued by subsidiary held outside Group		519.09	519.09
Liabilities			
Non Current Liabilities	1 1		
(a) Financial Liabilities		1	
(i) Borrowings	20	1,26,152.68	1,40,059.93
(ii) Other Financial Liabilities	21	612.11	2,300.76
(b) Provisions	22	273.73	203.82
(c) Other Non-Current Liabilities	23	2,335.87	1,852.89
		1,29,374.39	1,44,417.40
Current Liabilities	1 1	and the second second	
(a) Financial Liabilities			
(i) Borrowings	24	12,524.46	13,753.15
(ii) Trade Payables	25		
Micro and small enterprises		84.90	18.11
Others		2,494.36	1,779.68
(iii) Other Financial Liabilities	26	1,02,861.88	79,930.84
(b) Other Current Liabilities	27	4,430.03	3,682.01
(c) Provisions	28	25.85	22.43
A.S. Constanting	100	1,22,421.48	99,186.22
		3,05,780.71	3,15,624.63

1 to 71

Notes to the financial statements As per our report of even date

For Chaturvedi & Shah LLP Chartered Accountants Firm Registration Number 101720W/W100355

MUMBAI

D

Vijay Napawaliya Partner

Membership Number: 109859

õ

*

Place: Mumbal Date: 27th May, 2019 For and on behalf of the Board of Directors of Arshiya Limited

tt

Ajay S Mittal Chairman and Managing Director DIN: 00226355

Ashishkumar Bairagra Independent Director DIN: 00049591

Savita Dalal Company Secretary

MU

Archana A Mittal Joint Managing Director DIN: 00703208

S. Maheshwari Chief Financial Officer and Group President

Dinesh Kumar Sodani VP - Accounts & Finance

	1	Year Ended	(Rs. in Lakh) Year Ended
Particulars	Notes	31st March, 2019	31st March, 2018
			erecinaren, zere
INCOME			
Revenue from operations	29	28,937.38	25,906.69
Other income	30	2,460.09	1,665.19
Total Income (I)		31,397,47	27,571.88
		2.0020010.	
EXPENSES			
Material handling, value optimisation services and other charges		1,047.75	1,211.35
Freight expenses	31	10,954.30	11,668,31
Terminal expenses		357.55	304.26
Other operating expenses		756.02	374.90
Warehouse storage charges		5,484,67	902.54
Employee benefits expense	32	3,806,85	3,634,54
Finance costs	33	27,559.39	31,598,39
Depreciation and amortization expense	34	9,419,56	10,171.76
Other expenses	35	3,929,29	4,047.50
Total Expenses (II)	-	63,315.38	63,913.55
roun Expenses (ii)		03,010.00	00,010.00
Profit/(loss) before exceptional items and tax (I-II)		(31,917.91)	(36,341.67
Exceptional Items (net)	36	(5,167.04)	(39,473.20
		(0,101.04)	(00,470,20
Profit/(loss) before tax		(26,750.87)	3,131.53
	11	and the second second	
Tax expense:	63		
Current tax	1 C L	6,88	
Adjustment of tax relating to earlier periods		0.10	27.42
Profit/(loss) for the year		(26,757.85)	3,104.11
OTHER COMPREHENSIVE INCOME	111		
tems not to be reclassified to profit and loss in subsequent			
periods:		with a start	1.1.1.1
Remeasurement of gains (losses) on defined benefit plans		(28.57)	(9.67
Other Comprehensive income for the year	1	(28.57)	(9.67
outer comprehensive income for the year		(20.57)	(5.67
Total Comprehensive Income for the year		(26,786.42)	3.094.44
			and the second sec
Profit for the year attributable to:		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
Equity holders of the parent		(26,757,85)	3,104.11
Non-controlling interests			
	1	(26,757.85)	3,104.11
Other comprehensive income for the year attributable to:	1		
Equity holders of the parent		(28,57)	(9.67
Non-controlling interests			
		(28.57)	(9.67
Total comprehensive income for the year attributable to:			
Equity holders of the parent		(26,786.42)	3,094.44
Non-controlling interests	-		
	+	(26,786.42)	3,094.44
Earning per share /face value of Ps. 2 each)	62		
Earning per share (face value of Rs. 2 each)	62	(11.24)	1.70
Basic and Diluted			

Arshiya Limited Ildated Ctat

As per our report of even date

For Chaturvedi & Shah LLP Chartered Accountants

Firm Registration Number 101720W/W100355

CEA

EDI & SHAA

JME.

CHARTERED P

Vijay Napawaliya Partner Membership Number: 109859

Place: Mumbai Date: 27th May, 2019

For and on behalf of the Board of Directors of Arshiya Limited

-11 Ajay S Mittal Chairman and Managing Director DIN: 00226355

icoo

Ashishkumar Bairagra Independent Director DIN: 00049591

Savita Dalal Company Secretary

Archana A Mittal Joint Managing Director DIN: 00703208

S. Maheshwari Chief Financial Officer and Group President

Deodam: Dinesh Kumar Sodani

VP - Accounts & Finance

Arshiya Limited Consolidated Statement of changes in Equity for the year ended 31st March, 2019

Equity Share Capital (Refer Note No. 18) 4

Particulars	Rs. In Lakh
Equity Shares of Rs. 2 each issued, subscribed and paid up As at 31st March, 2017	3,123.59
As at 31st March, 2018	4,564.34
Issue of Equity Shares	307.95
As at 31st March. 2019	4,872.29

B. Other Equity (Refer Note No. 19)

(Rs. in Lakh)

Nume application incomes pending notesy pending administration allorment Nume Securities administration allorment Earthy Securities Component of 0X Arrengamental Rearrin Rearrins Arrengamental Securities Rearrins Securities Securities Rearrins Cannot Rearrins 1 18.766.71 - 8.62.04 1.58 124.80 79.6174.43 940.18 1 18.766.71 - - - - - - - - 1 18.766.71 - 8.62.04 1.58 1.24.80 79.6174.43 940.18 1 18.766.71 - - - - - - - - 1 18.766.71 - - - - - - - - - - - 1 1 - - - - - - - - - - - 1 18.766.71 -	Partcuars	「こう」のこれであるとうである	「「「「「「「「」」」」		The second se	A REAL PROPERTY AND ADDRESS OF AD	A NAME OF ALL OF	The second se	And a	二日日の一日の一日の一日の一日の一日の一日の一日の日の日の日の日の日の日の日の日
18,766.71 18,620.84 1.58 124.80 73,617.43 940.18 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 1 1 1 1 1 the year 1 1 <th></th> <th>Share application money pending alloment</th> <th>Money received against share warrants</th> <th>Equity Component of 0% Optionally Convertible Preference shares (OCRPS)</th> <th>Capital Reserve</th> <th>Amalgametion Reserve</th> <th>Securities Premium Account</th> <th>General Reserve</th> <th>Retained</th> <th>B</th>		Share application money pending alloment	Money received against share warrants	Equity Component of 0% Optionally Convertible Preference shares (OCRPS)	Capital Reserve	Amalgametion Reserve	Securities Premium Account	General Reserve	Retained	B
the year (18,766,71) (41,068,97) - <th< td=""><td>Balances as on 31st March, 2017</td><td>18,766.71</td><td></td><td>88,620.84</td><td>1.58</td><td>124.80</td><td>79,617.43</td><td>940.18</td><td>(1,59,750.24)</td><td>28,321.30</td></th<>	Balances as on 31st March, 2017	18,766.71		88,620.84	1.58	124.80	79,617.43	940.18	(1,59,750.24)	28,321.30
the year (18,766,71) (18,766,71) (41,068,97) - 95,278,86 - <t< td=""><td>Profit/(loss) for the year Other comprehensive income</td><td></td><td>Ť.)</td><td></td><td>4</td><td></td><td>1.9</td><td>3.9</td><td>3,104.11 (9.67)</td><td>3,104.11 (9.67)</td></t<>	Profit/(loss) for the year Other comprehensive income		Ť.)		4		1.9	3.9	3,104.11 (9.67)	3,104.11 (9.67)
arrants (18,766.71) 5 (41,088.97) - 95,278.88 - ufty shares - - - - - - 95,278.88 - - ufty shares - - - - - - - 137,622) - <td>Total comprehensive income for the year</td> <td>*</td> <td></td> <td>•</td> <td>•</td> <td>•</td> <td></td> <td></td> <td>3,094.44</td> <td>3,094.44</td>	Total comprehensive income for the year	*		•	•	•			3,094.44	3,094.44
analis 860.25 - + - (37.62) - uity shares - - - - - (37.62) - - uity shares - <	On issue of equity shares	(18,766.71)		(41,068.97)	4	à	95,278.86	ı	ł	35,443.18
ufly shares if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.62) if (37.52) if (47.551.87) if (37.51) if (37.52) if (37.52) if (37.52) if (37.51) if (37.51) if (37.51)	Money received against share warrants		860.25			•		1	4	860.25
The second se	Transaction costs on issue of equity shares	4			ą.	¥	(37.62)	*	- Contraction	(37.62)
	Others (net) Conditional Lease rent		1	,	1	ŕ	*	8	(1,134.38) 390.41	(1, 134.38) 390.41
the year interfs 	Balances as at 31st March, 2018		860.25	47,551.87	1.58	124.80	1,74,858.67	940.18	(1,57,399.77)	66,937.58
the year	Profit/(loss) for the year Other commethensive income	• •		* 1	2.1	• •	1.	14	(26,757,85) (28,57)	(26,757.85)
arants - 15.00 - 15.00 - 58,213.49 - 58,213.49 - 58,213.49 - 58,213.49	Total comprehensive income for the year		•	1.5				1	(26,786.42)	(26,786.42)
- (875.25) (47.551.87) - 58.213.49 -	Money received against share warrants	•	15.00			•	•		•	15.00
	On issue of equity shares		(875.25)	(47,551.87)	-	•	58,213.49	÷		9,786.37
	Others (net)		51	4	§.	1	,	•	(1,359.07)	(1,359.07)
+ + + + + + + + + + + + + + + + + + +	Balances as at 31st March, 2019				1.58	124.80	2,33,072.16	940.18	(1,85,545.26)	48,593.46

Notes to the financial statements As per our report of even date

For Chaturvedi & Shah LLP Chantered Accountants Firm Registration Number 101720W/W100355

A MUMBAI Maper char

For and on behalf of the Board of Directors of Arshiya Limited

TAT 8

Ajay S Mittal Chairman and Managing Director DIN: 00226355

LP*S

Membership Number: 1098

Vijay Napawaliya Partner Dinesh Kumar Sodani VP - Accounts & Finance ١

manuh

Archana A Mittal Joint Managing Director DIN: 00703208

Ashishkumar Bairagra Independent Director DIN: 00049591 2

S. Maheshwari Chief Financial Officer and Group President

Savita Dalal Company Secretar

Place: Mumbai Date: 27th May, 2019

FRED ACU

Arshiya Limited Consolidated Cash Flow Statement for the year ended 31st March, 2019

Particulars	Year Ended	(Rs. in Lakh Year Ended
	31st March, 2019	31st March, 2018
Cash flow from operating activities	11 mar 1 mar 1	
Profit/(Loss) before tax	(26,750.87)	3,131.5
Adjustments for:	1	17107 100
Bad debts	45.81	101.54
Sundry balances written back (net)	(931.35)	(452.89
Discarding/written off of Property, plant and equipment and Intangible assets	166.34	
(Gain)/loss on disposal of Property, plant and equipment	(0.03)	515.64
(Gain) on monetization of Property, plant and equipment (Refer Note No. 59)		(15,633.29
Provision for doubtful debts/Expected credit loss	112.65	(52.12
Excess provision written back	1000	(463.14
Reconciliation of loan accounts (net)		(562.39
Settlement of claims	(5,167.04)	(19,478.4)
Depreciation and amortization expense	9,419.56	10,171.76
Interest expense	27,559.39	31,598.39
Government grant income	(365.49)	(365.49
Financial guarantee income	(227.88)	(2.57
Financial assets carried at amortised cost Gain on derecognised of Liability Component	(193.06)	
Interest income	(653.17)	151.00
Dividend income	(39.72)	(54.23
Foreign exchange differences (net)	43.21	(0.60
Operating profit before working capital changes	3,018.35	(105.30 8,348.37
Adjustments for :	5,010.55	0,340.37
Change in inventories	45.00	
Derease/(Increase) in financial and other assets	15.66 1,058.89	0.07
Decrease in financial and other liabilities	(2,152.32)	(16,518.65 (5,280.32
Cash generated from operations	1,940.58	(13,450.53
Direct taxes paid	(545.51)	(13,430.33
Net cash flow from operating activities (A)		(13,966.53
Cash flow from investing activities	200	
Purchase of property, plant and equipment	(770.00)	14 500 07
Purchase of intangible assets	(779.90)	(4,589.07
A set of the		(2,239.00
Purchase of Capital work in progress and Intangible assets under development	(71.18)	
Proceeds from sale of property, plant and equipment	2.51	95.23
Proceeds from monetization of property, plant and equipments		43,400.00
Capital advances	(1,089.41)	(49.23
Dividend income		0.60
Interest received	39.72	54.23
Net cash flow from investing activities (B)	(1,898.26)	36,672.76
Cash flow from financing activities		
Issue of Equity shares (including security premium)	1	15,268.39
Money received against share warrants	15.00	860.25
Proceeds from non-current borrowings	5,571.82	3,200.00
Repayment of non-current borrowings	(2,550.04)	(36,864.72
Short-term borrowings (net)	(1,228.69)	6,755.45
Unpaid Dividend transfer to IEPF A/c	(0.04)	0,100.40
Interest paid	(1,697.30)	(10,993.44
Net cash flow from financing activities (C)		(21,774.07
Net (decrease)/increase in cash and cash equivalents (A + B + C)	(392.44)	932.16
Cash and cash equivalents as at the beginning of the year	1,784.38	852.22
over and even edentations as at the beginning of the Jobi	1,704.30	032.22





Arshiya Limited Consolidated Cash Flow Statement for the year ended 31st March, 2019

Change in liabilities arises from financing activities

Change in liabilities arises from financing activities		(Rs. in Lakh)	
Particulars	Long term Borrowings	Short term Borrowings	
As at 1st April, 2018	1,76,462.68	13,753,15	
Less: Transaction cost	714.67		
Less: Conversion of Liability Component of Compound Financial instruments (OCRPS) into Equity	(10,342.61)	4	
Add: Non cash items	(73.69)		
Add/Less: Cash flow	3.021.78	(1.228.69)	
As at 31st March, 2019	1,69,782.83	12,524.46	

Notes:

1. Bracket indicates cash outflow.

2. The above cash flow statement has been prepared under the "Indirect Method" as set out in IND AS 7 on Statement of Cash Flow.

Notes to the financial statements

1 to 71

As per our report of even date

For Chaturvedi & Shah LLP Chartered Accountants Firm Registration Number 101720W/W100355

Japanaline

Vijay Napawaliya Partner Membership Number: 109859



Place: Mumbai Date: 27th May, 2019 For and on behalf of the Board of Directors of Arshiya Limited

Ajay S Mittal Chairman and Managing Director DIN: 00226355

Ashishkumar Bairagra Independent Director DIN: 00049591

14

Savita Dalal Company Secretary Anhana littal

Joint Managing Director DIN: 00703208 2

S. Maheshwari Chief Financial Officer and Group President

Sodaw

Dinesh Kumar Sodani VP - Accounts & Finance

Notes to the Consolidated financial statements for the year ended 31st March, 2019

1 Corporate Information

Arshiya Limited (hereinafter referred to "the Parent Company" or "the Company") together with its subsidiaries (collectively referred to as "Group") is a flagship Company of Arshiya Group. It is pioneering Unified Supply Chain and integrated logistics infrastructure solution provider Group headquartered in India. The group businesses comprises of Free Trade and Warehousing Zone (FTWZ), Rail & Rail Infrastructure, Industrial and Distribution hubs, Indian Container Depot (ICD), Domestic Warehousing, Forwarding, Transport & Handling, Supply Chain technology and Management solutions.

These statements comprises of Consolidated Financial Statements ("CFS") of Arshiya limited (CIN: L93000MH1931PI.C024747) and its subsidiaries for the year ended 31st March, 2019. The Company is a public company domiciled in India and is incorporated on 3rd July, 1981 under the provisions of the Companies Act applicable in India. The registered office of the company is located at 302, Level 3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400 018.

The Parent Company's equity shares are listed on the Bombay Stock Exchange (BSE) and National Stock Exchange (NSE) of India.

The Consolidated Financial Statements for the year ended 31st March, 2019 were approved and adopted by board of directors in their meeting held on 27th May, 2019.

2 Basis of preparation of consolidated financial statement

The consolidated financial statements have been prepared in accordance with Indian Accounting Standards ("Ind AS") notified by the Ministry of Corporate Affairs ("MCA") pursuant to the Section 133 of the Companies Act, 2013 ("the Act") read with of the Companies (Indian Accounting Standards) Rules 2015, (as amended) and other relevant provisions of the Act.

The consolidated financial statements have been prepared on a historical cost basis, except for certain financial assets and liabilities, which are measured at fair value / amortized cost.

The consolidated financial statements are presented in Indian Rupees (Rs.) which is the Group's functional and presentation currency and all values are rounded to the nearest lakh as per the requirement of schedule III, unless when otherwise indicated.

3 Basis for Consolidation

The consolidated financial statements comprise of the financial statements of the Parent Company and its subsidiaries as at 31st March, 2019. Control is achieved when the Group is exposed, or has rights, to variable returns from its involvement with the investee and has the ability to affect those returns through its power over the investee.

Generally, there is a presumption that a majority of voting rights result in control. To support this presumption and when the Group has less than a majority of the voting or similar rights of an investee, the Group considers all relevant facts and circumstances in assessing whether it has power over an investee, including the contractual arrangement with the other vote holders of the investee, rights arising from other contractual arrangements, the Group's voting rights and potential voting rights and the size of the Group's holding of voting rights relative to the size and dispersion of the holdings of the other voting rights holders.

The Group re-assesses whether or not it controls an investee if facts and circumstances indicate that there are changes to one or more of the three elements of control. Consolidation of a subsidiary begins when the Group obtains control over the subsidiary and ceases when the Group loses control of the subsidiary. Assets, liabilities, income and expenses of a subsidiary acquired during the year are included in the consolidated financial statements from the date the Group obtains control and assets, liabilities, income and expenses of a subsidiary disposed off during the year are included in the consolidated financial statements till the date the Group ceases to control the subsidiary.

The CFS includes the Financial Statements of the Parent Company and the subsidiaries (as listed in the table below). Subsidiaries are consolidated from the date on which effective control is acquired and are excluded from the date of transfer/disposal.




S. No.	Name of the antity	Country of Incorporation	(including interest)/ votin directly/indire	g power (either
	Direct Subsidiaries:	C	31st March, 2019	31st March, 2018
(i)	Arshiya Rail Infrastructure Limited (ARIL)	Indía	100%	100%
(ii)	Arshiya Northern FTWZ Limited (ANFL)	India	100%	100%
(iii)	Arshiya Industrial & Distribution Hub Limited (AIDHL)	India	100%	100%
	Arshiya Lifestyle Limited (ALL)	India	100%	100%
(v)	Arshiya Logistics Services Limited (formerly known as Laxmipati Balaji Exim Trading Limited) (ALSL) (w.e.f. 13th June, 2017)	India	100%	100%
(vi)	Arshiya Transport and Handling Limited (ATHL)	India	100%	100%
(vii)	Arshiya Technologies (India) Private Limited (ATIPL)	India	100%	100%
(viii)	Laxmipati Balaji Supply Chain Management Private Limited (LBSCM) (w.e.f 7th May, 2018)	India	100%	Nil
(ix)	Anomalous Infra Private Limited (AIPL) (w.e.f. 15th October, 2018)	India	100%	Nil
(x)	Arshiya Infrastructure Developers Private Limited (AIDPL) (w.e.f. 9th January, 2019)	India	100%	Nil
(xi)	Unrivalled Infrastructure Private Limited (UIPL) (w.e.f. 7th January, 2019)	India	100%	Nil
(xii)	Arshiya Northern Projects Private Limited (ANPPL) (w.e.f. 25th October, 2018)	India	100%	Nil
(xiii)	Mira Supply Chain Management Private Limited (formerly known as Arshiya Supply Chain Management Private Limited) (ASCM) (up to 21st March, 2018)	India	Nil	Nil
1	Indirect Subsidiaries:		10	
	Held through Arshiya Logistics Services Limited:			
(xiv)	Arshiya 3PL Services Private Limited (w.e.f. 27th August, 2018)	India	100%	Nil
	Held through Arshiya Rail Infrastructure Limited:			
(xv)	Ascendas Panvel FTWZ Limited (formerly known as Arshiya Rail Siding and Infrastructure Limited) (APFL) (up to 3rd February, 2018)	India	Nil	Nil

4 Consolidation procedure:

(a) Combine like items of assets, liabilities, equity, income, expenses and cash flows of the parent with those of its subsidiaries. For this purpose, income and expenses of the subsidiary are based on the amounts of the assets and liabilities recognised in the consolidated financial statements at the acquisition date.

- (b) Offset (eliminate) the carrying amount of the parent's investment in each subsidiary and the parent's portion of equity of each subsidiary. The difference between the cost of investment in the subsidiaries and the Parent's share of net assets at the time of acquisition of control in the subsidiaries is recognised in the consolidated financial statement as goodwill. However, resultant gain (bargain purchase) is recognized in other comprehensive income on the acquisition date and accumulated to capital reserve in equity.
- (c) Intra-Group balances and transactions, and any unrealized income and expenses arising from intra Group transactions, are eliminated in preparing the consolidated financial statements.
- (d) Consolidated statement of Profit and Loss and each component of OCI are attributed to the equity holders of the parent of the Group and to the non-controlling interests, even if this results in the non-controlling interests having a deficit balance.
- (e) For the acquisitions of additional interests in subsidiaries, where there is no change in the control, the Group recognises a reduction to the non-controlling interest of the respective subsidiary with the difference between this figure and the cash paid, inclusive of transaction fees, being recognised in equity. In addition, upon dilution of controlling interests, the difference between the cash received from sale or listing of the subsidiary shares and the increase to non-controlling interest is also recognised in equity. If the Group loses control over a subsidiary, it derecognises the related assets (including goodwill), liabilities, non-controlling interest and other components of equity, while any resultant gain or loss is recognised in consolidated statement of profit and loss. The results of subsidiaries acquired or disposed off during the year are included in the consolidated statement of Profit and Loss from the effective date of acquisition or up to the effective date of disposal, as appropriate.
- (f) Consolidated financial statements are prepared using uniform accounting policies for like transactions and other events in similar circumstances. If an entity of the group uses accounting policies other than those adopted in the consolidated financial statements for like transactions and events in similar circumstances, appropriate adjustments are made to that Group member's financial statements in preparing the consolidated financial statements to ensure conformity with the Group's accounting policies.
- (g) Consolidated financial statements of all entities used for the purpose of consolidation are drawn up to same reporting date as that of the Parent Company, i.e., year ended on 31st March. When the end of the reporting period of the parent is different from that of a subsidiary, if any, the subsidiary prepares, for consolidation purposes, additional financial information as of the same date as the consolidated financial statements of the parent to enable the parent to consolidate the financial information of the subsidiary, unless it is impracticable to do so.



Notes to the Consolidated financial statements for the year ended 31st March, 2019

(h) In the case of foreign subsidiaries, being non-integral foreign operations, revenue items are consolidated at the average exchange rates prevailing during the year. All assets and liabilities are converted at rates prevailing at the end of the year. Components of equity are translated at closing rate. Any gain / (Loss) on exchange difference arising on consolidation is recognized in the Foreign Currency Translation Reserve (FCTR) through OCI.

The financial statements are presented in Indian Rupees (Rs.), which is the Group's functional and presentation currency and all values are rounded to the nearest lakh, except when otherwise indicated.

5 Significant Accounting Policies

5.1 Property, Plant and Equipment

Property, plant and equipment are carried at cost less accumulated depreciation and accumulated impairment losses, if any. Cost includes purchase price, borrowing cost and any cost directly attributable to the bringing the assets to its working condition for its intended use.

Depreciation on the property, plant and equipment is provided using straight line method over the useful life of assets as specified in schedule II to the Companies Act, 2013. Leasehold improvements are amortized over the period of lease. Depreciation on property, plant and equipment which are added / disposed off during the year, is provided on pro-rata basis with reference to the date of addition / deletion. Freehold land is not depreciated and under the previous GAAP land was revalued.

The assets' residual values, useful lives and method of depreciation are reviewed at each financial year end and are adjusted prospectively, if appropriate.

Capital work-in-progress includes cost of property, plant and equipment under installation / under development as at the balance sheet date.

Property, plant and equipment are eliminated from consolidated financial statement, either on disposal or when retired from active use. Profits / losses arising in the case of retirement / disposal of property, plant and equipment are recognised in the consolidated statement of profit and loss in the year of occurrence.

The Group has opted to continue with the carrying values of all of its property, Plant and Equipment as recognised in the Previous GAAP financial statements as deemed cost at the transition date i.e. 1st April, 2016.

5.2 Intangible Assets

Intangible assets are carried at cost less accumulated amortization and accumulated impairment losses, if any. Cost includes expenditure that is directly attributable to the acquisition of the intangible assets.

Identifiable intangible assets are recognised when it is probable that future economic benefits attributed to the asset will flow to the Group and the cost of the asset can be reliably measured.

Railways License fees is amortized over a period of twenty years being the license period as per agreement.

Cost of Enterprise Resource Planning (ERP) software including expenditure on implementation is to be amortized over a period of ten years based on management's estimate of useful life over which economic benefits will be derived from its use.

Computer softwares are capitalized at the amounts paid to acquire the respective license for use and are amortized over the period of three to seven years. The assets' useful lives are reviewed at each financial year end.

Trademark are amortised over the period of ten (10) years.

Gains or losses arising from derecognition of an intangible asset are measured as the difference between the net disposal proceeds and the carrying amount of the asset and are recognised in the consolidated statement of profit and loss when the asset is derecognised.

Intangible assets under development includes cost of computer software under installation / under development as at the balance sheet date.

The Group has opted to continue with the carrying values of all of its Intangible assets as recognised in the Previous GAAP financial statements as deemed cost at the transaction date i.e. 1st April, 2016.

5.3 Leases

A lease is classified at the inception date as a finance lease cr an operating lease. A lease that transfers substantially all the risks and rewards incidental to ownership to the Group is classified as a finance lease.



0

Notes to the Consolidated financial statements for the year ended 31st March, 2019

The Group as a lessee

(a) Finance lease

Assets acquired under finance lease are capitalized and the corresponding lease liability is recognised at lower of the fair value of the leased assets and the present value of minimum lease payments at the inception of the lease. Initial costs directly attributable to lease are recognised with the asset under lease.

(b) Operating lease

Lease of assets under which all risks and rewards of ownership are effectively retained by the lessor are classified as operating lease. Lease payments under operating lease are recognised as expenses on accrual basis in accordance with the respective lease agreements.

The Group as a lessor

(a) Finance lease

When assets are leased out under a finance lease, the present value of the minimum lease payments is recognised as a receivable. The difference between the gross receivable and the present value of the receivable is recognised as unearned finance income. Lease income is recognised over the term of the lease using the net investment method before tax, which reflects a constant periodic rate of return.

The lessor derecognises the leased assets and recognises the difference between the carrying amount of the leased assets and the finance lease receivable in the consolidated statement of Profit and Loss when recognising the finance lease receivable. This gain or loss is presented in the consolidated statement of Profit and Loss in the same line item as that in which the lessor presents gains or losses from sale of similar assets.

(b) Operating lease

Rental income from operating leases is recognised in the consolidated statement of profit and loss on a straight line basis over the term of the relevant lease unless another systematic basis is more representative of the time pattern in which economic benefits from the leased assets is diminished.

Initial indirect costs incurred in negotiating and arranging as operating lease are added to carrying value of the leased asset and recognised on a straight line basis over the lease term.

5.4 Inventories

Inventories are valued at the lower of cost and net realizable value. The cost of inventories comprises of cost of land and incidental cost thereto, cost of purchase, cost of conversion and other costs incurred in bringing the inventories to their respective present location and condition. Cost is computed on the First in first out basis.

5.5 Cash and cash equivalents

Cash and cash equivalent in the balance sheet comprise cash at banks, cash on hand and short-term deposits with an original maturity of three months or less, which are subject to an insignificant risk of changes in value.

For the purpose of the statement of cash flows, cash and cash equivalents consist of cash on hand and short-term deposits, as defined above, net of outstanding bank overdrafts as which are considered an integral part of the Group's cash management.

5.6 Impairment of assets

An asset is considered as impaired when at the date of Balance Sheet, there are indications of impairment and the carrying amount of the asset, or where applicable, the cash generating unit to which the asset belongs, exceeds its recoverable amount (i.e. the higher of the net asset selling price and value in use). The carrying amount is reduced to the recoverable amount and the reduction is recognized as an impairment loss in the consolidated statement of profit and loss. The impairment loss recognized in the prior accounting period is reversed if there has been a change in the ostimate of recoverable amount. Post impairment, depreciation is provided on the revised carrying value of the impaired asset over its remaining useful life.

5.7 Financial instruments - initial recognition, subsequent measurement and impairment

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity.

(i) Financial assets -Initial recognition and measurement

All financial assets are initially recognized at fair value. Transaction costs that are directly attributable to the acquisition or issue of financial assets, which are not at fair value through profit or loss, are adjusted to the fair value on initial recognition. Financial assets are classified, at initial recognition, as financial assets measured at fair value or as financial assets measured at amortized cost.

(ii) Financial assets - Subsequent measurement:

For the purpose of subsequent measurement, financial assets are classified in two broad categories:-

- (a) Financial assets at fair value
- (b) Financial assets at amortized cost

Where assets are measured at fair value, gains and losses are either recognised entirely in the consolidated statement of profit and loss (i.e. fair value through profit or loss), or recognised in other comprehensive income (i.e. fair value through other comprehensive income).



Som K

Notes to the Consolidated financial statements for the year ended 31st March, 2019

A financial asset that meets the following two conditions is measured at amortized cost (net of any write down for impairment) unless the asset is designated at fair value through profit or loss under the fair value option.

- (a) Business model test: The objective of the Group's business model is to hold the financial asset to collect the contractual cash flow.
- (b) Cash flow characteristics test: The contractual terms of the financial asset give rise on specified dates to cash flow that are solely payments of principal and interest on the principal amount outstanding.

A financial asset that meets the following two conditions is measured at fair value through other comprehensive income unless the asset is designated at fair value through profit or loss under the fair value option.

- (a) Business model test: The financial asset is held within a business model whose objective is achieved by both collecting contractual cash flow and selling financial assets.
- (b) Cash flow characteristics test: The contractual terms of the financial asset give rise on specified dates to cash flow that are solely payments of principal and interest on the principal amount outstanding.

Financial assets - Derecognition

A financial assets (or, where applicable, a part of a financial asset or part of a group of similar financial assets) is primarily derecognised (i.e. removed form the Group's statement of financial position) when:

- (a) The rights to receive cash flows from the asset have expired, or
- (b) The Group has transferred its rights to receive cash flow from the asset.

(iii) Financial liabilities - Initial recognition and measurement:

The financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

Effective interest method

The effective interest method is a method of calculating the amortized cost of a financial instrument and of allocating interest income or expenses over the relevant period. The effective interest rate is the rate that exactly discounts future cash receipts or payments through the expected life of financial instruments, or where appropriate, a shorter period.

Financial liabilities - Subsequent measurement

Financial liabilities are subsequently carried at amortized cost using the effective interest method. For trade and other payables maturing within one year from the balance sheet date, the carrying amounts are approximate at their fair value due to the short maturity of these instruments.

Financial Liabilities - Financial guarantee contracts

Financial guarantee contracts issued by the Group are those contracts that require a payment to be made to reimburse the holder for a loss it incurs because the specified debtor fails to make a payment when due in accordance with the terms of a debt instrument. Financial guarantee contracts are recognised initially as a liability at fair value, adjusted for transaction costs that are directly attributable to the issuance of the guarantee. Subsequently, the liability is measured at the higher of the amount of loss allowance determined and the amount recognised less cumulative amortization.

Financial Liabilities - Derecognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another, from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the consolidated statement of profit and loss.

Compound Instruments

An issued financial instrument that comprises of both the liability and equity components are accounted as compound financial instruments. The fair value of the liability component is separated from the compound instrument and the residual value is recognised as equity component of other financial instrument. The liability component is subsequently measured at amortised cost, whereas the equity component is not remeasured after initial recognition. The transaction costs related to compound instruments are allocated to the liability and equity component is not remeasured after initial recognition. The transaction costs related to compound instruments are allocated to the liability and equity components in the proportion to the allocation of gross proceeds. Transaction costs related to equity component is recognised directly in equity and the cost related to liability component is included in the carrying amount of the liability component and amortised using effective interest method.

5.8 Provisions, Contingent Liabilities, Contingent Assets and Commitments:

Provisions are recognised when the Group has a present obligation (legal or constructive) as a result of a past event. It is probable that an outflow of resources embodying economic benefits will be required to settle the obligation and a reliable estimate can be made of the amount of the obligation. If the effect of the time value of money is material, provisions are discounted using equivalent period government securities interest rate. Unwinding of the discount is recognised in the consolidated statement of profit and loss as a finance cost. Provisions are reviewed at each balance sheet date and are adjusted to reflect the current best estimate.



1 X Am

Notes to the Consolidated financial statements for the year ended 31st March, 2019

Contingent liabilities are disclosed when there is a possible obligation arising from past events, the existence of which will be confirmed only by the occurrence or non-occurrence of one or more uncertain future events not wholly within the control of the Group or a present obligation that arises from past events where it is either not probable that an outflow of resources will be required to settle or a reliable estimate of the amount cannot be made. Information on contingent liability is disclosed in the Notes to the Consolidated Financial Statements. Contingent assets are not recognised. However, when the realisation of income is virtually certain, then the related asset is no longer a contingent asset, but it is recognised as an asset.

5.9 Dividend Distribution

Annual dividend distribution to the shareholders is recognised as a liability in the period in which the dividends are approved by the shareholders. Any interim dividend paid is recognised on approval by Board of Directors. Dividend payable and corresponding tax on dividend distribution is recognised directly in other equity.

5.10 Revenue recognition

Revenue is recognized upon transfer of control of goods or rendering of services to customers in an amount that reflects the consideration which the Group expects to receive in exchange for those goods or services.

Generally, control is transfer upon shipment of goods to the customer or when the goods is made available to the customer, provided transfer of title to the customer occurs and the Group has not retained any significant risks of ownership or future obligations with respect to the goods shipped.

Revenue is measured at the amount of consideration which the Group expects to be entitled to in exchange for transferring distinct goods or services to a customer as specified in the contract, excluding amounts collected on behalf of third parties (for example taxes and duties collected on behalf of the government). Consideration is generally due upon satisfaction of performance obligations and a receivable is recognized when the it becomes unconditional.

Revenue are recognized as the related services are performed and revenue from the end of the last invoicing to the reporting date is recognized as unbilled revenue.

Revenues in excess of invoicing are classified as contract assets (which we refer as unbilled revenue) while invoicing in excess of revenues are classified as contract liabilities (which we refer to as unearned revenues).

Trade Receivables

A receivable represents the Group's right to an amount of consideration that is unconditional.

Contract liabilities

A contract liability is the obligation to transfer of services to a customer for which the Group has received consideration (or an amount of consideration is due) from the customer. If a customer pays consideration before the Group transfers services to the customer, a contract liability is recognised when the payment is made Contract liabilities are recognised as revenue when the Group performs under the contract.

(a) Free Trade Warehousing Zone (FTWZ)

(i) Income from allotment of warehousing spaces and open yard area for use are recognised for the period the material is lying in area as per agreed terms.

- (ii) Revenue from valued services and other activities is recognised when related services are performed as per the contractual terms.
- (iii) Export benefits under Foreign Trade Policy are recognised when utilized.

(b) Inland Container Depot (ICD)

- Income from Container handling, storage and Rail and Road transportation are recognised on proportionate completion of the movement and delivery of goods to the party/designated place.
- (ii) Income from ground rent is recognised for the period the container is lying in the ICD area.
- (c) Rail Transport Operations
- (i) Revenue from sale of services e.g. rail freight income is recognized as per the terms of contracts with customers based on stage of completion when the outcome of the transactions involving rendering of services can be estimated reliably. For fixed-price contracts, revenue is recognised based on the actual service provided to the end of the reporting period as a proportion of the total services to be provided (percentage of completion method).
- (ii) Measurement of revenue: Estimates of revenues, costs or extent of progress toward completion are revised if circumstances change. Any resulting increases or decreases in estimated revenues or costs are reflected in profit or loss in the period in which the circumstances that give rise to the revision become known by management.
- (iii) Revenue from handling and other ancillary services is recognised at the time of rendering of service which is at the time of loading/unloading of container/cargo.



(d) Domestic Warehousing

Revenue from allotment of warehousing space and open yard area for use is accounted on accrual basis as per agreed terms of contract.

- (e) Interest income from a financial asset is recognised when it is probable that the economic benefits will flow to the Group and the amount of income can be measured reliably. Interest income is accrued on a time basis, by reference to the principal outstanding and at the effective interest rate applicable, which is the rate that exactly discounts estimated future cash receipts through the expected life of the financial asset to that asset's net carrying amount on initial recognition.
- (f) Dividend income is recognised when the right to receive the payment is established, which is generally when shareholders approve the payment of dividend.

5.11 Foreign currency reinstatement and translation:

Transactions in foreign currencies are initially recorded by the Group at rates prevailing at the date of the transaction. Subsequently monetary items are translated at closing exchange rates as on balance sheet date and the resulting exchange difference recognised in the consolidated statement of profit and loss. Differences arising on settlement of monetary items are also recognised in the consolidated statement of profit and loss.

Non-monetary items that are measured in terms of historical cost in a foreign currency are translated using the exchange rates at the dates of the transaction. Non-monetary items carried at fair value that are denominated in foreign currencies are translated at the exchange rates prevailing at the date when the fair value was determined. The gain or loss arising on translation of non-monetary items measured at fair value is treated in line with the racognition of the gain or loss on the change in fair value of the item (i.e., translation differences on items whose fair value gain or loss is recognised in OCI or profit or loss are also recognised in OCI or profit or loss, respectively).

Foreign exchange differences regarded as an adjustment to borrowing costs are presented in the consolidated statement of profit and loss, within finance costs. All other finance gains / losses are presented in the consolidated statement of profit and loss on a net basis.

5.12 Employee benefits

Short term employee benefits are recognized as an expense in the statement of profit and loss of the year in which the related services are rendered.

(a) Defined Contribution Plan

Contribution to Provident Fund, a defined contribution plan, is made in accordance with the statute, and is recognised as an expense in the year in which employees have rendered services

(b) Defined Benefit Plan

Leave encashment being a defined benefit plan is accounted for using the projected unit credit method, on the basis of actuarial valuations carried out by third party actuaries at each Balance Sheet date. Actuarial gains and losses arising from experience adjustments and changes in actuarial assumptions are charged or credited to other comprehensive income in the year in which they arise. Other costs are accounted in the consolidated statement of profit and loss.

The cost of providing gratuity, a defined benefit plans, is determined using the Projected Unit Credit Method, on the basis of actuarial valuations carried out by third party actuaries at each Balance Sheet date. Actuarial gains and losses arising from experience adjustments and changes in actuarial assumptions are charged cr credited to other comprehensive income in the period in which they arise. Other costs are accounted in the consolidated statement of profit and loss.

Remeasurements of defined benefit plan in respect of post employment and other long term benefits are charged to the other comprehensive income in the year in which they occur. Remeasurements are not reclassified to the consolidated statement of profit and loss in subsequent periods.

5.13 Taxes on Income

Income tax expense represents the sum of current tax (including MAT and income tax for earlier years) and deferred tax. Tax is recognised in the consolidated statement of profit and loss, except to the extent that it relates to items recognised directly in equity or other comprehensive income, in such cases the tax is also recognised directly in equity or in other comprehensive income. Any subsequent change in direct tax on items initially recognised in equity or other comprehensive income.

Current tax provision is computed for income calculated after considering allowances and exemptions under the provisions of the applicable Income Tax Laws. Current tax assets and current tax liabilities are off set, and presented as net.

Deferred tax is recognised on differences between the carrying amounts of assets and liabilities in the Balance sheet and the corresponding tax bases used in the computation of taxable profit. Deferred tax liabilities are generally recognised for all taxable temporary differences, and deferred tax assets are generally recognised for all deductible temporary differences, carry forward tax losses and allowances to the extent that it is probable that future taxable profits will be available against which those deductible temporary differences, carry forward tax losses and allowances to the extent that it is probable that future taxable profits will be available against which those deductible temporary differences, carry forward tax losses and allowances can be utilized. Deferred tax assets and liabilities are measured at the applicable tax rates. The carrying amount of deferred tax assets is reviewed at each balance sheet date and reduced to the extent that it is no longer probable that sufficient taxable profits will be available against which the temporary differences can be utilized.

XI



Notes to the Consolidated financial statements for the year ended 31st March, 2019

Credit of MAT is recognised as an asset only when and to the extent there is convincing evidence that the Group will pay normal income tax during the specified period, i.e., the period for which MAT credit is allowed to be carried forward. In the year in which the MAT credit becomes eligible to be recognised as an asset, the said asset is created by way of a credit to the consolidated statement of profit and loss and shown as MAT credit entitlement. The Group reviews the same at each balance sheet date and writes down the carrying amount of MAT credit entitlement to the extent there is no longer convincing evidence to the effect that the Group will pay normal income tax during the specified period.

5.14 Borrowing costs

Borrowing costs specifically relating to the acquisition or construction of qualifying assets that necessarily takes a substantial period of time to get ready for its intended use are capitalized (net of income on temporarily deployment of funds) as part of the cost of such assets. Borrowing costs consist of interest and other costs that the Group incurs in connection with the borrowing of funds. For general borrowing used for the purpose of obtaining a qualifying asset, the amount of borrowing costs eligible for capitalization is determined by applying a capitalization rate to the expenditures on that asset.

The capitalization rate is the weighted average of the borrowing costs applicable to the borrowings of the Group that are outstanding during the period, other than borrowings made specifically for the purpose of obtaining a qualifying asset. The amount of borrowing costs capitalized during a period does not exceed the amount of borrowing cost incurred during that period. All other borrowing costs are expensed in the period in which they occur.

5.15 Earnings per Share

Basic earnings per share is computed using the net profit/loss for the year attributable to the shareholders' and weighted average number of equity shares outstanding during the year.

Diluted earnings per share is computed using the net profit/loss for the year attributable to the shareholders' and weighted average number of equity and potential equity shares outstanding during the year including share options, convertible preference shares and debentures, except where the result would be anti-dilutive. Potential equity shares that are converted during the year are included in the calculation of diluted earnings per share, from the beginning of the year or date of issuance of such potential equity shares, to the date of conversion.

5.16 Current and non-current classification

The Group presents assets and liabilities in statement of financial position based on current/non-current classification. The Group has presented non-current assets and current assets, non-current liabilities and current liabilities in accordance with Schedule III, Division II of Companies Act, 2013 notified by MCA.

An asset is classified as current when it is:

- (a) Expected to be realised or intended to be sold or consumed in normal operating cycle,
- (b) Held primarily for the purpose of trading,
- (c) Expected to be realised within twelve months after the reporting period, or
- (d) Cash or cash equivalent unless restricted from being exchanged or used to settle a liability for at least twelve months after the reporting period.

All other assets are classified as non-current.

A liability is classified as current when it is:

- (a) Expected to be settled in normal operating cycle,
- (b) Held primarily for the purpose of trading,
- (c) Due to be settled within twelve months after the reporting period, or
- (d) There is no unconditional right to defer the settlement of the liability for at least twelve months after the reporting period.

All other liabilities are classified as non-current.

The operating cycle is the time between the acquisition of assets for processing and their realisation in cash or cash equivalents. Deferred tax assets and liabilities are classified as non-current assets and liabilities. The Group has identified twelve months as its normal operating cycle.

5.17 Fair value measurement

The Group measures financial instruments at fair value at each balance sheet date.

Fair value is the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The fair value measurement is based on the presumption that the transaction to sell the asset or transfer the liability takes place either:

- (a) In the principal market for the asset or liability, or
- (b) In the absence of a principal market, in the most advantageous market for the asset or liability.

A fair value measurement of a non-financial asset takes into account a market participant's ability to generate economic benefits by using the asset in its highest and best use or by selling it to another market participant that would use the asset in its highest and best use.

The Group uses valuation techniques that are appropriate in the circumstances and for which sufficient data are available to measure fair value, maximizing the use of relevant observable inputs and minimizing the use of unobservable inputs.





Notes to the Consolidated financial statements for the year ended 31st March, 2019

All assets and liabilities for which fair value is measured or disclosed in the consolidated financial statements are categorized within the fair value hierarchy.

5.18 Off-setting financial Instrument

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable rights to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable rights must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the Group or counterparty.

5.19 Segment Reporting - Identification of Segments

An operating segment is a component of the Group that engages in business activities from which it may earn revenues and incur expenses, whose operating results are regularly reviewed by the Group's chief operating decision maker to make decisions for which discrete financial information is available. Based on the management approach as defined in Ind AS 108, the chief operating decision maker evaluates the Group's performance and allocates resources based on an analysis of various performance indicators by business segments and geographic segments.

5.20 Government Grants

Grants from the government are recognised at their fair value where there is a reasonable assurance that the grant will be received and the Group will comply with all attached conditions. Government grants relating to income are deferred and recognised in the profit or loss over the period necessary to match them with the ccsts that they are intended to compensate and presented within other income.

Government grants relating to the purchase of property, plant and equipment are included in non-current liabilities as deferred income and are credited to profit or loss on a straight-line basis over the expected lives of the related assets and presented within other income.

5.21 Contributed Equity

Equity Shares are classified as equity, incremental costs directly attributable to the issue of new shares or options are shown in equity as deduction, net of tax from the proceeds.

5.22 Cash flow statement

Cash flows are reported using the indirect method, whereby profit before tax is adjusted for the effects of transactions of non-cash nature and any deferrals or accruals of past or future cash receipts or payments. The cash flows from operating, investing and financing activities of the Group are segregated based on the available information.

5.23 Business combinations

Business combinations involving entities that are controlled by the Group are accounted for using the pooling of interests method as follows:

- (i) The assets and liabilities of the combining entities are reflected at their carrying amounts.
- (ii) No adjustments are made to reflect fair values, or recognise any new assets or liabilities.
- (iii) Adjustments are only made to harmonise accounting policies.
- (iv) The financial information in the financial statemen's in respect of prior periods is restated as if the business combination had occurred from the beginning of the preceding period in the financial statements, irrespective of the actual date of the combination. However, where the business combination had occurred after that date, the prior period information is restated only from that date.
- (v) The balance of the Retained Earnings appearing in the financial statements of the transferor is aggregated with the corresponding balance appearing in the financial statements of the transferee or is adjusted against General Reserve.
- (vi) The identities of the reserves are preserved and the reserves of the transferor become the reserves of the transferee.
- (vii) The difference, if any, between the amounts recorded as share capital issued plus any additional consideration in the form of cash or other assets and the amount of share capital of the transferor is transferred to Capital Reserve and is presented separately from Other Capital Reserves.

6 Significant accounting judgements, estimates and assumptions

The preparation of the consolidated financial statements requires management to make judgements, estimates and assumptions that affect the reported amounts of revenues, expenses, assets and liabilities, and the accompanying disclosures, and the disclosure of contingent liabilities. Uncertainty about these assumptions and estimates could result in outcomes that require a material adjustment to the carrying amount of assets or liabilities affected in future periods. The key assumptions concerning the future and other key sources of estimation uncertainty at the reporting date, that have a significant risk of causing a material adjustment to the carrying amounts of assets and liabilities within the next financial year, are described below.

The Group based on its assumptions and estimates on parameters available when the consolidated financial statements were prepared. However, existing circumstances and assumptions about future developments may change due to market changes or circumstances arising that are beyond the control of the Group. Such changes are reflected in the assumptions when they occur.

6.1 Property, plant and equipment and Intangible Assets

Management reviews the estimated useful lives and residual values of the assets annually in order to determine the amount of depreciation to be recorded during any reporting period. The useful lives and residual values as per schedule II of the Companies Act, 2013 or are based on the Group's historical experience with similar assets and taking into account anticipated technological changes, whichever is more appropriate.



A DE

6.2 Income Tax

The Group reviews at each balance sheet date the carrying amount of deferred tax assets. The factors used in estimates may differ from actual outcome which could lead to an adjustment to the amounts reported in the financial statements.

6.3 Contingencies

Management has estimated the possible outflow of resources at the end of each annual reporting financial year, if any, in respect of contingencies/claim/litigations against the Group as it is not possible to predict the outcome of pending matters with accuracy.

6.4 Impairment of financial assets

The impairment provisions for financial assets are based on assumptions about risk of default and expected cash loss. The Group uses judgement in making these assumptions and selecting the inputs to the impairment calculation, based on Group's past history, existing market conditions as well as forward looking estimates at the end of each reporting period.

6.5 Impairment of non-financial assets

The Group assesses at each reporting date whether there is an indication that an asset may be impaired. If any indication exists, or when annual impairment testing for an asset is required, the Group estimates the asset's recoverable amount. An asset's recoverable amount is the higher of an asset's or Cash Generating Units (CGU) fair value less costs of disposal and its value in use. It is determined for an individual asset, unless the asset does not generate cash inflows that are largely independent to those from other assets or groups of assets. Where the carrying amount of an asset or CGU exceeds its recoverable amount, the asset is considered impaired and is written down to its recoverable amount.

In assessing value in use, the estimated future cash flows are discounted to their present value using a pre-tax discount rate that reflects current market assessments of the time value of money and the risks specific to the asset. In determining fair value less cost of disposal, recent market transactions are taken into account. If no such transactions can be identified, an appropriate valuation model is used. These calculations are corroborated by valuation multiples or other available fair value indicators.

6.6 Defined benefits plans

The Cost of the defined benefit plan and the present value of such obligation are determined using actuarial valuations. An actuarial valuation involves making various assumptions that may differ from actual developments in the future. These include the determination of the discount rate, future salary increases, mortality rates and attrition rate. Due to the complexities involved in the valuation and its long-term nature, a defined benefit obligation is highly sensitive to changes in these assumptions. All assumptions are reviewed at each reporting date.

6.7 Recoverability of trade receivable

Judgements are required in assessing the recoverability of overdue trade receivables and determining whether a provision against those receivables is required. Factors considered include the credit rating of the counterparty, the amount and timing of anticipated future payments and any possible actions that can be taken to mitigate the risk of non-payment.

6.8 Provisions

Provisions and liabilities are recognized in the period when it becomes probable that there will be a future outflow of funds resulting from past operations or events and the amount of cash outflow can be reliably estimated. The timing of recognition and quantification of the liability require the application of judgement to existing facts and circumstances, which can be subject to change. Since the cash outflows can take place many years in the future, the carrying amounts of provisions and liabilities are reviewed regularly and adjusted to take account of changing facts and circumstances.

6.9 Fair value measurement of financial instruments

When the fair value of financial assets and financial liabilities recorded in the balance sheet cannot be measured based on quoted prices in active markets, their fair value is measured using valuation techniques including the Discounted Cash Flow (DCF) model. The inputs to these models are taken from observable markets where possible, but where this is not feasible, a degree of judgement is required in establishing fair values. Judgements include considerations of inputs such as liquidity risk, credit risk and volatility. Changes in assumptions about these factors could affect the reported fair value of financial instruments.

6.10 Recent accounting pronouncements:

Standards Issued But Not Effective

On 31st March, 2019, the Ministry of Corporate Affairs (MCA) has notified IND AS 116 – Leases and certain amendment to existing IND AS. These amendments shall be applicable to the Group from 1st April, 2019.

(a) Issue of IND AS 116 - Lease

Ind AS 116 will supersede the current standard on leases i.e. IND AS 17- Leases. As per IND AS 116, the lessor will have to bring to books all the non-cancellable portion of leasing arrangement.



to

(b) Amendment to existing standards

The MCA has also carried out amendments of the following accounting standards

- (i) IND AS 101- First time adoption of Indian Accounting Standards
- (ii) IND AS 103 Business Combinations
- (iii) IND AS 109 Financial Instruments
- (iv) IND AS 111 Joint Arrangements
- (v) IND AS 12 Income Taxes
- (vi) IND AS 19 Employee Benefits (vii) IND AS 23 - Borrowing Costs
- (viii) Ind AS 28 Investment in Associates and Joint Ventures

Application of above standards are not expected to have any significant impact on the Consolidation financial statements.



of tam

										(Rs. in Lakh)
Particulars	Freehold Land	Buildings	Railway Terminal	Plant and Equipments	Furniture and Fixtures	Vehicles	Equipments	Computers	Leasehold Improvements	Total
Gross Carrying Value (at deemed cost) As at 31st March. 2017	1.54.665.44	1.13.376.41	14.283.94	38.595.29	1.852.06	20.79	2.390.81	240.86	33.37	3 25 535 25
Additions			49.60	40.92			37.74			4,589.07
Disposals	(9,337.96)	(17,202.94)	•	(1,588.05)	(871.94)	3	(705.64)	(9.17)	. 1	(29,715.70)
Other Adjustments	A I	x		(8.32)		8.32		1	1	
As at 31st March, 2018	1,49,788.29	96,173.47	14,333.54	37,039.84	980.12	105.39	1,722.91	231.69	33.37	3,00,408.62
Additions	2,583.34	608.20	55.03	62.37	22.11	13.96	19.97	18.58	161.15	3,544.71
Disposals	1	,	4	(1.35)	(116.13)	(23.61)	(38.15)	~	(33.37)	(255.61)
transfer to Inventories	(19.089.31)								4	(19,089.31)
As at 31st March, 2019	1,33,282.32	96,781.67	14,388.57	37,100.86	886.10	95.74	1,704.73	207.27	161.15	2,84,608.41
Accumulated Depreciation								1		
As at 31st March, 2017	.1	3,751.87	1,154.50	3,809.09	304.46	25.54	936.75	85.55	18.23	10,085.99
Depreciation for the year	•	3,696.23	1,157.25	3,791.05	276.55	20.84	432.25	70.80	4.90	9,449.87
Disposals	4	(539.43)		(298.41)	(266.57)		(394.72)	(5.30)	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	(1,504.43)
As at 31st March, 2018		6,908.67	2,311.75	7,301.73	314.44	46.38	974.28	151.05	23.13	18,031.43
Depreciation for the year	a.	3,397.64	1,159.96	3,669.17	137.24	15.11	159.16	22.87	4.00	8,565.15
Disposals		1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		(0.31)	(60.07)	(19.55)	(19.18)	(21.93)	(23.13)	(144.17)
As at 31st March, 2019		10,306.31	3,471.71	10,970.59	391.61	41.94	1,114.26	151.99	4.00	26,452.41
Net Carrying value as at 31st March, 2019	1,33,282.32	86,475.36	10,916.86	26,130.27	494.49	53.80	590.47	55.28	157.15	2,58,156.00
Net Carrying value as at 31st March, 2018	1,49,788.29	89,264.80	12,021.79	29,738.11	665.68	59.01	748.63	80.64	10.24	2.82.377.19

Notes:

1) Freehold Land includes Rs. 9,735.11 Lakh situated at Nagpur, which is under possession of a lender as per the Order of Honble High Court of Bombay dated 9th May, 2013.

2) Freehold Land measuring 45.52 Acres amounting to Rs. 19,089.31 Lakh are converted into inventories.

3) Gross carrying value includes cost of vehicles taken on finance lease Rs.13.96 Lakh.

4) In accordance with the Indian Accounting Standard (IND AS -36) on "Impairment of Assets", the management during the year carried out an exercise of identifying the assets that may have been impaired in accordance with the said IND AS. On the basis of this review carried out by the management, there was no impairment loss on property, plant and equipment during the year ended 31st March, 2019.



				(Rs. in Lakh)
Particulars	Trade Mark	Computer Software	Rail License Fees	Total
Gross Carrying Value (at deemed cost)	1112	A second second	and an and a second	
As at 31st March, 2017	0.49	1,036.94	3,208.33	4,245.76
Additions		2,239.00	-	2,239.00
Deductions	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(235.50)	1. mar 1. 2 al 1.	(235.50
As at 31st March, 2018	0.49	3,040.44	3,208.33	6,249.26
Additions	1	14		
Deductions		(116.94)		(116.94
As at 31st March, 2019	0.49	2,923.50	3,208.33	6,132.32
Accumulated Amortisation	1	1		
As at 31st March, 2017	0.20	448.31	250.00	698.51
Amortisation for the year	0.17	471.88	249.84	721.89
Deductions		(207.89)	2 10 10	(207.89
As at 31st March, 2018	0.37	712.30	499.84	1,212.51
Amortisation for the year		604.57	249.84	854.41
Deductions	-	(59.56)		(59.56
As at 31st March, 2019	0.37	1,257.31	749.68	2,007.36
Net Carrying value as at 31st March, 2019	0.12	1,666.19	2,458.65	4,124.96
Net Carrying value as at 31st March, 2018	0.12	2,328.14	2,708.49	5,036.75



Ja & som

Particulars	1 A 24	As at 31st March, 2019	As at 31st March, 2018
Non- Current Financial Assets		1	
9. Other Financial Assets		-	
Security deposits		1,790.02	1,732.58
Non-Current Assets	Total	1,790.02	1,732.58
10. Other Non- Current Assets Capital Advances			
Considered good		1,964.05	965.29
Considered doubtful		1,395.00	1,395.00
Land Bar and a stranger and		3,359.05	2,360.29
Less: Provision for impairment		(1,395.00) 1,964.05	(1,395.00 965.29
Security deposits		59.39	60.89
Prepaid expenses		1,054.33	1,190.39
TDS receivables/Taxes paid		3,129.68	2,601.82
Service tax paid under protest		5.14	3.87
Current Assets	Total	6,212.59	4,822.26
Current Assets			
11. Inventories (Valued at lower of Cost and Net Realisable value)			
Stores and spares			15.66
Land		16,505.97	-
	Total	16,505.97	15.66
Current Financial Assets			
12. Trade Receivables Secured, Considered good		210.77	2,742.67
Unsecured, Considered good		4.055.40	2,142.01
Trade Receivables which have significant increase in credit risk		183.90	71.25
		4,450.07	2,813.92
Less: Provision for expected credit losses		(183.90)	(71.25
	Total	4,266.17	2,742.67
13. Cash and Cash Equivalents			
Balances with banks:			
- in current accounts #		982.11	1,278.68
Cash on hand		8.45	7.16
	Total	990.56	1,285.84

Cash and cash equivalents as at 31st March, 2019 comprises of restricted bank balances held in escrow account with bank amounting to Rs. 803.81 Lakh (31st March, 2018 - Rs.744.85 Lakh). This account can only be operated with the specific permission / instruction in terms of the Escrow Agreement entered into by the Parent Company, ALL and ALSL with Ascendas Panvel FTWZ Limited (formerly known as Arshiya Rail Siding and Infrastructure Limited (APFL).

			(Rs. in Lakh)
Particulars		As at 31st March, 2019	As at 31st March, 2018
14. Other Bank Balancos			
Deposits with banks to the extent held as margin money		373.29	457.68
Interest accrued on fixed deposit		28.09	40.82
Unpaid dividends			0.04
	Total	401.38	498.54
15. Loan			
Unsecured, considered good unless otherwise stated Loan to other		325.12	+
	Total	325.12	

URVEDI ICCDUN

y Is &

Notes to the Consolidated financial statements for the year ended 31st March, 2019

			(Rs. in Lakh)
Particulars	1 - 200	As at 31st March, 2019	As at 31st March, 2018
16. Other Financial Assets			
Security deposits		3,251.26	9,418.61
Unbilled revenue		2,508.27	390.41
Margin money with Lender*		170.00	170.00
Interest accrued on fixed deposits		5.45	
Other recoverables		2,941.66	2,825.28
	Total	8,876.64	12,804.30
* To be adjusted at time of final settlement			
17. Other Current Assets			
Advances to suppliers		84.85	290.22
Advances to employees		2.87	16.58
Other advances		146.04	30.05
Prepaid expenses		350.00	315.10
TDS receivables/Taxes paid		10.67	
Balances with Statutory, Government authorities (Refer Note No. 52)		3,355.87	3,537.72
Cash seized by Income Tax (Refer Note No 48)			100.00
	Total	3,950.30	4,289.67

		(Rs. in Lakh)
Particulars	As at 31st March, 2019	As at 31st March, 2018
18, Share Capital	1000	
Authorised (i) 24,75,00,000 (31st March, 2018 - 24,75,00,000) Equity Shares of Rs. 2 each (ii) 1,10,00,000 (31st March, 2018 - 1,10,00,000) 0% Optionally Convertible Redeemable Preference Shares (OCRPS) of Rs. 10 each	4,950.00 1,100.00	4,950.00 1,100.00
Total	6,050.00	6,050.00
Equity Share Capital - Issued, subscribed and fully paid 24,36,14,292 (31st March, 2018 - 22,82,16,776) Equity shares of Rs. 2 each	4,872.29	4,564.34
Total	4,872.29	4,564.34

(a) Terms and rights

(i) Terms and rights attached to equity shares

The Parent Company has one class of equity share having a par value of Rs. 2 per share. Each holder of equity share is entitled to one vote per share. The shareholders who held shares on the record date are entitled to dividend as may be proposed by the Board of Directors and is subject to approval of the Shareholders at the ensuing General Meeting.

In the event of liquidation of the Parent Company, the holders of Equity Shares will be entitled to receive remaining assets of the Parent Company, after distribution of all preferential amounts. The distribution will be in the proportion to the number of equity shares held by the shareholders.

(ii) Terms and rights attached to 0% Optionally Convertible Redeemable Preference Shares (OCRPS) The Parent Company has 0% optionally convertible redeemable preference shares having a par value of Rs. 10 per share. Each holder of OCRPS has right / entitled to convert into equity shares within 18 months from the date of issue or redemption on or after 20 years as per terms.

(b) Reconciliation of equity shares and optionally convertible preference shares

Particulars	As at 31st Ma	rch, 2019	As at 31st Mar	ch, 2018
Paroculars	Number of Shares	Rs. in Lakh	Number of Shares	Rs. in Lakh
Balance as at the beginning of the year	22,82,16,776	4,564.34	15,61,79,472	3,123.59
Add: Issued during the year	1,53,97,516	307.95	7,20,37,304	1,440.75
Balance as at the and of the year	24 36 14 292	4 872 29	22 82 16 776	4 564 34

(ii) Reconciliation of optionally convertible redeemable preference shares outstanding as at the beginning and end of the year

Particulars	As at 31st Ma	rch, 2019	As at 31st Marc	ch, 2018
rardculars	Number of Shares	Rs. in Lakh	Number of Shares	Rs. in Lakh
Balance as at the beginning of the year	57,64,619	576.46		
Add: Issued during the year			1,19,13,329	1,191.33
Less: Converted into equity shares during the year	(57,64,619)	(576.46)	(61,48,710)	(614.87)
Balance as at the end of the year			57,64,619	576.46



Notes to the Consolidated financial statements for the year ended 31st March, 2019

(c) Details of equity shares held by the shareholders holding more than 5% of the aggregate shares in the Parent Company

	As at 31st M	arch, 2019	As at 31st Ma	rch, 2018
Name of the shareholder	Number of equity shares	Percentage (%) shareholding	Percentage (%) shareholding	Percentage (%) shareholding
Archana A Mittal	8,85,59,788	36.35%	8,85,59,288	38.80%
Ajay S Mittal	3,85,61,437	15.83%	3,70,60,937	16.24%
Edelweiss Asset Reconstruction Company Limited (through various trusts)	5,95,59,820	24.45%	4,56,62,304	20.01%

(d) In Previous year ended 31st March, 2018 the Parent Company had allotted to the Promoter Directors 1,00,00,000 equity shares and 1,00,00,000 share warrants of Rs. 2 each at a premium of Rs.58.35 per share on preferential basis pursuant to the Restructuring Agreement dated 31st March, 2017 and in terms of special resolution passed on 29th April, 2017 as per applicable provisions of Companies Act, 1956/Companies Act, 2013 and SEBI (Issue of Capital and Disclosure Requirements) Regulation. 85,00,000 share warrants out of 1,00,00,000 share warrants have been converted into Equity shares on 8th November, 2017.

During the year, the Parent Company has allotted 15,00,000 Equity Sharos of face value of Rs.2 each to the Promoter upon conversion of equal number of warrants.

	As at	(Rs. in Lakh As at
Particulars	31st March, 2019	31st March, 2018
19. Other Equity		
(i) Share Application money pending allotment		111 111 1
Balances as at the beginning of the year Less: On issue of Equity Shares		18,766.7
Balances as at the end of the year	-	(18,766.71
Dalances as at the end of the year	2	
(ii) Money Received against share warrants	1000	
Balances as at the beginning of the year	860.25	
Add: received during the year	15.00	860.2
Less: On issue of Equity Shares	(875.25)	1
Balances as at the end of the year	•	860.25
(iii) Equity Component of 0% Optionally Convertible Redeemable Preference shares (OCRPS)		
Balances as at the beginning of the year	47.551.87	88.620.84
Less: On issue of Equity Shares	(47,551.87)	(41,068 97
Balances as at the end of the year	-	47,551.87
(iv) Capital Reserve	1 A 1 A 1	
Balances as at the beginning and end of the year	1.58	1.58
(v) Amalgamation Reserve	1.000	
Balances as at the beginning and end of the year	124.80	124.80
Reserve and Surplus		
(vi) Securities Premium Account		
Balances as at the beginning of the year	1,74,858.67	79,617.43
Add: On issue of Equity Shares	58,213.49	95,278.8
Less: Share issue expenses/ Transaction cost		(37.6)
Balancos as at the end of the year	2,33,072.16	1,74,858.6
(vii) General Reserve		
Balances as at the beginning and end of the year	940.18	940.18
(viii) Deficit in the Statement of Profit and Loss	and the second se	
Balances as at the beginning of the year	(1,57,399.77)	(1,59,750.2
Add/Less: Profit/(Loss) for the year	(26,757.85)	3,104.1
Less: Other comprehensive loss	(28.57)	(9.6
Less: Others (net)	(1,359.07)	(1.134.3)
Add: Conditional Lease rent		390.4
Balances as at the end of the year	(1,85,545.26)	(1,57,399.77
Total (i to viii)	48,593.46	66,937.58

Nature and purpose of Reserve and Surplus:

(a) Securities Premium Account:

Securities premium account is created to record premium received on issue of equity shares. The reserve is utilized in accordance with the provision of the Companies Act, 2013.

(b) General Reserve:

General Reserve is used for time to time to transfer of profits from Retained Earnings for appropriation purposes. As the general reserve is created by a transfer from one component of equity to another and is not an item of other comprehensive income, items included in the General Reserve will not be reclassified subsequently to statement of profit and loss.

EDI&S MUMBAI ACCOUN

U Aar

Notes to the Consolidated financial statements for the year ended 31st March, 2019

(c) Amalgamation Reserve:

Amalgamation reserve is created on account of scheme of amalgamation of erstwhile BDP (India) Private Limited with the Parent Company approved by the Hon'ble High Court of Judicature at Bombay in earlier years.

(d) Retained Earning:

Retained Earnings are the profit/(loss) of the Group earned till date net of appropriations.

	As at	(Rs. in Lakh As at
Particulars	31st March, 2019	31st March, 2017
Non-Current Liabilities		
20. Borrowings		
Secured		
(a) Term Loans		
From Banks (Refer Note No. 20.1)	3,571.43	1
From Other Parties (Refer Note No. 20.2)	1,21,809.11	1,29,036.41
(b) Vehicles Loan from bank	9.52	
Liability Comportent of Compound Financial Instruments (OCRPS)	762.62	11,023.52
Tot	al 1,26,152.68	1,40,059.93

The details of security, terms of repayment and interest on non-current borrowings (which includes current maturities) obtained by the Group are given below:

(20.1) Rupee Term loan from Banks:

20.1.1 Parent Company

(1) Rupee term loan of Rs. 3,193.29 Lakh (31st March, 2018 - Rs. 3,189.79 Lakh):

(a) Details of security

(i) Second charge on movable and immovable Panve! assets of the Parent Company except for the excluded properties under Lease Agreement dated 3rd February, 2018

(ii) Second charge on present and future receivables including the amount deposited to the EARC TRA account of the Parent Company . (iii) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

and the second se

(b) Terms of Interest rate

(i) Rate of Interest is @ 14.50% p.a.

(c) Terms of Repayment:-

Rupee term loan is repayable in Bullet payment at the end of the tenuro of loan i.e. 36 months.

(d) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 6.71 Lakh (31st March, 2018 - Rs.10.21 Lakh).

(2) Rupee term loan of Rs. 474.30 Lakh:

(a) Details of security

(i) Second charge on movable and immovable Panvel assets of the Parent Company except for the excluded properties under Lease Agreement dated 3rd February, 2018

(ii) Second charge on present and future receivables including the amount deposited to the EARC TRA account of the Parent Company.

(iii) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

(b) Terms of Interest rate

Rate of Interest is @ 14.50% p.a.

(c) Terms of Repayment:-

Rupee term loan is repayable in 78 equal monthly installment commencing from the date of first disbursement i.e. 31st August, 2018.

(d) The Parent Company has been in default for the repayment of principal amount of Rs. 19.17 Lakh.

(20.2) Rupee Term loans from Other Parties

20.2.1 Parent Company

(1) Rupee term loan of Rs, 59,513.25 Lakh (31st March, 2018 - Rs. 59,359.23 Lakh):

(a) Security provided:

(i) First charge in all the present and future movable and immovable property, plant and equipment including intangible assets, assignment of rights and benefits of the Parent Company but excluding project assets for Khurja FTWZ project, Khurja Distiripark Project, Nagpur project and Rail project on pari passu basis.

(iii) Second charge on current assets of the Parent Company but excluding current assets for khurja FTWZ project, Khurja Distripark Project, Nagpur project and Rail project on pari passu basis.



(iii) first pari passu charge by way of hypothecation on the Panvel Receivables both existing and future of whatsoever nature.

(iv) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

(v) The loans are secured by pledged of shares held by the two Promoter Directors of the Parent Company.

(b) Terms of Interest rate

Rate of Interest is @ 10% p.a. compounded quarterly.

(c) Terms of Repayment:-

	(Rs. in Lakh)
Year	Loan from Others
FY 2017-18	5,671.09
FY 2021-22	14,001.46
FY 2022-23	40,404.50
Total	60,077.05

(d) The Parent Company has been in default for the repayment of principal amount of Rs. 5,671.09 Lakh. (31st March, 2018 - Rs. 5,671.09 Lakh).

(e) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 563.80 Lakh (31st March, 2017 - Rs. 717.82 Lakh).

(2) Rupee term loan of Rs. 2.495.44 Lakh (31st March, 2018 - Rs. 2.672.34 Lakh)

(a) Securities provided

(i) Second charge by way of equitable mortgage/hypothecation on the entire immovable and movable property, plant and equipment of the Parent Company on pari-passu basis.

(ii) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

(b) Terms of Repayment:-

Rupee term loan is repayable in 13 structured quarterly installments commencing from 31st January, 2018.

(c) The Parent Company has been in default for the repayment of principal amount of Rs. 670 Lakh. (31st March, 2018 - Rs.428 Lakh)

(d) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 199.56 Lakh (31st March, 2018 - Rs. 405.66 Lakh).

(3) Rupee term loan of Rs. 2.018.65 Lakh;

(a) Securities provided

(i) The above loan are secured by charge on residual cashflow of the Parent Company.

- (ii) The above loans are secured by the immovable property held by one Promoter Director of the Parent Company on pari passu basis.
- (iii) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.
- (iv) The above loans are secured by pledged of shares held by the one Promoter Director of the Parent Company.

(b) Terms of Interest rate

Rate of Interest is @ 18% p.a.compounded half yearly.

(c) Terms of Repayment:-

Rupee term loan is repayable in bullet payment at the end of the tenure of loan i.e. 18 months.

(d) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 81.35 Lakh.

20.2.2 ARIL

(1) Rupee term loan of Rs. 33.647.31 Lakh (31st March, 2018 - Rs. 33.502.36 Lakh):

(a) Securities provided

(i) First charge on all movable assets (including rakes, containers, equipment's) and immovable properties of ARIL both present and future on pari passu basis.

10

- (ii) Second charge by way of hypothecation of the entire current assets of ARIL on pari passu basis.
- (iii) Pledge of 100% equity shares of ARIL held by the Parent Company.
- (iv) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.
- (v) The above loans are secured by corporate guarantees of the Parent Company.

(b) Terms of Interest rate:

Rate of Interest is @ 10% p.a.compounded quarterly.



(c) Terms of Repayment:-	(Rs. in Lakh)
Year	Loan from Banks
FY 2019-2020	1,744.64
FY 2020-2021	6,139,19
FY 2021-2022	2,276.51
FY 2022-2023	23,954,16
Total	34,114.50

(d) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 467.19 Lakh (31st March, 2018 - Rs. 612.14 Lakh).

20.2.3 ANFL

(1) Rupee term loan from Other Parties of Rs. 10.627.10 Lakh (31st March. 2017 - Rs. 10.447.22 Lakh): (a) Security provided:

(i) First charge on fixed assets of ANFL both present and future on pari passu basis.

(ii) First Pari Passu charge/assignment/security interest on the ANFL's rights under the project documents, contracts (including guarantees) and all licenses, permits, approvals, consents and insurance policies.

(iii) Assignment of contractor guarantees, liquidated damages, letter of credit, guarantee or performance under any project agreement or contract in favour of ANFL.

(iv) Second charge on current assets of ANFL.

(v) Pledge of 40,52,778 equity shares of ANFL held by the Parent Company.
(vi) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

(vii) The above loans are secured by personal guarantees of the Parent Company.

(vii) The above loans are secured by corporate guarantees of the Parent

(b) Terms of Interest rate:

Rate of Interest is @ 10% p.a. compounded quarterly.

(c) Terms of Repayment:-	(Rs. in Lakh)
Year	Loan from Banks
FY 2019-2020	2,113.15
FY 2020-2021	323.92
FY 2021-2022	3,385.55
FY 2022-2023	5,323.63
Total	11,146,25

(d) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 519.15 Lakh (31st March, 2018 - Rs. 699.03 Lakh).

20.2.4 AIDHL

(1) Rupee term loan from Other Parties of Rs. 26.591.94 Lakh (31st March. 2018 - Rs. 26.589.56 Lakh): (a) Security provided:

(i) First charge on all movable and immovable properties of AIDHL both present and future on pari passu basis.

(ii) First charge by way of hypothecation of the entire current assets of AIDHL on pari passu basis.

(iii) Pledge of 100% equity shares of AIDHL held by the Parent Company.

(iv) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

(v) The above loans are secured by corporate guarantees of the Parent Company.

(b) Terms of Interest rate:

Rate of Interest is @ 10% p.a. compounded quarterly.

ALLEL PLATE LAND

(c) Terms of Repayment:-	(Rs. in Lakh)
Year	Loan from Banks
FY 2019-2020	1,885.69
FY 2020-2021	4,034.74
FY 2021-2022	2,209.30
FY 2022-2023	18,470.27
Total	26,600.00

(d) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. 8.06 Lakh (31st March, 2018 - Rs. 10.44 Lakh).

20.3 Vehicle loans from Bank

Vehicle loans are secured by way of hypothecation of vehicles. Rate of interest is @ 8.55% p.a. and repayment tenure in monthly instalment up to October 2023 and January 2024 respectively.



1 d Jam

Notes to the Consolidated financial statements for the year ended 31st March, 2019

Particulars		As at 31st March, 2019	(Rs. in Lakh) As at 31st March, 2018
21. Other Financial Liabilities			
Financial Liabilities at amortised cost Security deposit from unit holders Other financial liabilities Financial guarantees obligations Interest accrued but not due on borrowings Advance warehouse rent		377.71 221.56 12.44	358.37 1,718.36 224.03
	Total	612.11	2,300.76
22. Provisions			
Provision for employee benefits (Refer Note No.40) Gratuity Leave encashment		171.95 101.78	123.76 80.06
	Total	273.73	203.82
23. Other Non-Current Liabilities			
Lease equalisation reserve Government grants		703.55 1,632.32	129.20 1,723.69
	Total	2,335.87	1,852.89

(Rs.		(Rs. in Lakh)	
Particulars		As at 31st March, 2019	As at 31st March, 2018
Current Financial Liabilities		1.000	
24. Borrowings			
Secured			
(a) Working Capital facility (Cash Credit from banks) (Refer Note No. 24.1)		263 30	263.34
(b) Loan from Other Parties (Refer Note No. 24.2)		12,024.05	11,474.05
Unsecured			
(a) Loans from Promoter Directors (Refer Note No. 24.3)		160.11	1,937.76
(b) Inter Corporate Deposits (Refer Note No. 24.4)		77.00	78.00
	Total	12,524.46	13,753.15

24.1 Working capital facility (Cash Credit) from banks:

24.1.1 ANFL

(1) Rs. 263.30 Lakh (31st March, 2018 - Rs. 263.34 Lakh):

(a) Securities provided : (i) First charge on entire current assets of ANFL both present and future on pari passu basis.

(ii) Second pari passu charge on the assets charged for term loan of ANFL on first pari passu charge to lenders.

(iii) Pledge of 40,52,778 equity shares of ANFL held by the Parent Company.

(iv) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

(v) The above loans are secured by corporate guarantees of the Parent Company.

(b) Terms of interest:

Rate of interest on working capital is @ 14% p.a.

(c) ANFL has been in continuing default for the repayment of principal amount of Rs. 263.30 Lakh since FY 2014-15.

(24.2) Loan from Other Parties:

24.2.1 Parent Company

(1) Loan of Rs. 8,474.04 Lakh (31st March, 2018 - Rs. 8,474.04 Lakh)

(a) Securities provided - First Ranking charges on all present and future cash flows, all assets and movable collateral available to the existing lenders of the Parent Company as per the Deed of Hypothecation.

- The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

(b) Terms of interest: @ 18% p.a.

(c) The Parent Company has been in default for the repayment of principal amount of Rs. 8,474.04 Lakh (31st March, 2018 - Rs. Nil).



Notes to the Consolidated financial statements for the year ended 31st Marcli, 2019

(2) Loan of Rs. 550.00 Lakh (a) Securities provided

- Exclusive charges on cash flows of Domestic warehousing building.
- The above loans are secured by mortgage over lands admeasuring 7,130 Sq. mt. of the Company and wholly owned subsidiaries company.
- The above loans are secured by personal guarantees of one Promoter Director of the Parnet Company.
- The above loans are secured by corporate guarantees of the two subsidiary Companies i.e. Arshiya Rail Infrastructure Limited and Arshiya Northern FTWZ Limited.

(b) Terms of interest: @ 11% p.a.

24.2.2 AIDHL

(1) Loan of Rs. 3,000.00 Lakh (31st March, 2018 - Rs. 3,000.00 Lakh):

- (a) Securities provided
- (i) First Ranking charges on all present and future cash flows, all assets and movable collateral available to the existing lenders of AIDHL as per the Deed of Hypothecation.
- (ii) The above loans are secured by personal guarantees of two Promotor Directors of the Parent Company.
- (iii) The above loans are secured by corporate guarantees of the Parent Company.

(b) Terms of interest:

Rate of interest on said loan is @ 18% p.a.

(c) Terms of Repayment & Default:

Default in repayment of principal of Rupees 3,000.00 Lakh as at 31st March, 2019. The same has been recalled by the lender.

(24.3) Unsecured Loan from Promoter Directors:

(24.3.1) Parent Company

Loans from promoter directors of the Parent Company are interest free and repayable on demand.

(24.4) Unsecured Loan from Inter Corporate Deposits:

24.4.1 Parent Company

Intercorporate Deposit of Rs. 77 Lakh (31st March, 2018 - Rs. 77 Lakh) is interest free and repayable on demand.

24.4.2 ALL

Intercorporate Deposit of Rs. Nil (31st March, 2018 - Rs. 1 Lakh) is interest free and repayable on demand.

Particulars	11.2°%	As at 31st March, 2019	(Rs. in Lakh) As at 31st March, 2018
25. Trade Payablos Micro and small enterprises (Refer Note No.39) Others		84.90 2.494.36	18.11 1,779.68
	Total	2.579.26	1.797.79

(Rs.		
Particulars	As at 31st March, 2019	As at 31st March, 2018
26. Other Financial Liabilities	1-1	
Financial Liabilities at amortised cost		
Current maturities of long term debts from banks	20,583.78	22,527.15
Current maturities of long term debts from other parties	23,044.57	13,875.60
Current maturities of vehicle loan	1.80	-
Interest accrued and due on borrowings	38,978,24	27,947.92
Interest accrued but not due on borrowings	10,411.84	4,569.33
Interest payable on delayed payments to MSMED creditors (Refer Note No.39)	2.56	0.22
Unclaimed dividends		0.04
Deposit from Unitholders	910.82	938.16
Financial guarantees obligations	10.55	-
Payable for capital goods	7,228.05	8,422.54
Dues to employees (including full and final settlement)	710.31	474.12
Payable for expenses	979.36	1,145.00
Other Payables		30.76
Total	1,02,861.88	79,930.84





Notes to the Consolidated financial statements for the year ended 31st March, 2019

26.1 Rupee Term Loan from Banks:

26.1.1 Parent Company

(1) Rupee Term loan - Rs. 1.472.84 Lakh (31st March, 2018 - Rs. 1.491.67 Lakh) (a) Securities provided:

- Second charge on movable and immovable property, plant and equipments of the Parent Company, present and future on pari-passu,

(b) The above loan is secured by personal guarantees of two Promoter Directors of the Parent Company.

(c) Terms of Interest rate:

Rate of interest is @ 12% p.a.

(d) Terms of Repayment & Default:

The bank has been recalled loan of Rs. 1,472.84 Lakh (31st March, 2018 - Rs.1,491.67 Lakh) and interest (including penal interest) of Rs. 178.98 Lakh (31st March, 2018 - Rs. 32.15 Lakh).

26.1.2 ARIL

(1) Rupee Term loan from Banks of Rs. 6,910.60 Lakh (31st March. 2018 - Rs. 8,931.30 Lakh):

(a) Securities provided

(i) First charge on all movable assets (including rakes, containers, equipment's) and immovable properties of ARIL both present and future on pari passu basis.

(Re in Lakh)

(ii) Second charge by way of hypothecation of the entire current assets of ARIL on pari passu basis.

(iii) Pledge of 100% equity shares of the ARIL held by the Parent Company. (iv) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

(v) The above loans are secured by corporate guarantees of the Parent Company.

(b) Terms of Interest rate:

(i) Rate of Interest is @ 10.45% p.a. - 16.25% p.a.

(c) Terms of Pennymont

lo/ raina	or nepaymont.	(No. III Lakii)
	Year	Loan from Banks
	FY 2012-2013	22.50
	FY 2013-2014	75.00
0	FY 2014-2015	277.50
	FY 2015-2016	3,695.60
1.	FY 2018-2019*	2,840.00
	Total	6,910.60

* Refer Note No. 48

(d) Details of default in repayment of principal on secured loans as on 31st March, 2019 are as follows:

	(Rs. in Lakh)
Year	Loan from Banks
FY 2012-2013	22.50
FY 2013-2014	75.00
FY 2014-2015	277.50
FY 2015-2016	3,€95.60
FY 2018-2019	2.840.00
Total	6,910.60

The above loan has been recalled by Banks.

26.1.3 ANFL

(1) Rupee Term loans - Rs. 12.104.18 Lakh (31st March, 2018 - Rs. 12.104.18 Lakh): (a) Securities provided:

(i) First charge on fixed assets of ANFL both present and future on pari passu basis.

(ii) First Pari Passu charge/assignment/security interest on the ANFL's rights under the project documents, contracts (including guarantees) and all licenses, permits, approvals, consents and insurance policies.

(iii) Assignment of contractor guarantees, liquidated damages, letter of credit, guarantee or performance under any project agreement or contract in favour of ANFL.

(iv) Second charge on current assets of ANFL.

(v) Pledge of 40,52,778 equity shares of ANFL held by the Parent Company.

(vi) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.

(vii) The above loans are secured by corporate guarantees of the Parent Company.

(b) Terms of Interest rate: Rate of interest is @ 13% p.a.

c) Terms of Repayment:-	(Rs. in Lakh)
Year	Loan from Banks
FY 2012-2013	604.22
FY 2013-2014	1,410.23
FY 2014-2015	1,680.76
FY 2015-2016	8,408.97
Total	12,104.18





(d) The Banks has been recalled loan of Rs. 12,104.18 Lakh (31st March, 2018 - Rs. 12,104.18 Lakh) and interest of Rs. 14,947.59 Lakh (31st March, 2018 - Rs. 11,638.50 Lakh),

Details of default in repayment of principal on secured loans as on 31st March, 2019 are as follows:

	(Rs. in Lakh)
Year	Loan from Banks
FY 2012-2013	604.22
FY 2013-2014	1,410.23
FY 2014-2015	1,680.76
FY 2015-2016	8,408.97
Total	12,104.18

(26.2) Term loans from Other Parties

26.2.1 Parent Company

(1) Loan of Rs. 5,000.00 Lakh (31st March, 2018 - Rs. 5,000.00 Lakh) (Refer Note No. 46) Secured by first and exclusive charge on land situated at Village Butibori at Nagpur, Maharashtra. The said loan carries interest @ 20% p.a.

(2) Loan of Rs. 2,000.00 Lakh (31st March, 2018 - Rs. 1,951.52 Lakh) (Refor Note No. 45)

(i) 'Secured by first and exclusive charge on land situated at Khurja, Bulandshahr, Uttar Pradesh.

- (ii) The Parent Company has been in default for the repayment of principal amount of Rs. 2,000 Lakh. (31st March, 2017 Rs. 975 Lakh).
- (iii) The amortised cost disclosed above is net off incidental cost of borrowings aggregating of Rs. Nil (31st March, 2018 Rs. 23.48 Lakh).

26.2.2 ARIL

(1) Rupee Term loans - Rs. 2,960.00 Lakh

(a) Securities provided:

(i) First pari passu charge on all present and future cash flows of ARIL.

(ii) First pari passu charge on all movable and immovable assets of ARIL.

(iii) Charge on cash flows and movable assets by deed of hypothecation.

- (iv) The above loans are secured by personal guarantees of two Promoter Directors of the Parent Company.
- (v) The above loans are secured by corporate guarantees of the Parent Company.

(b) Terms of Interest rate:

Rate of interest is @ 20% p.a.

(c) Terms of Repayment:-	(Rs. in Lakh)
Year	Loan from Others
FY 2022-2023	2,960.00

(d) The above loan has been recalled during the year.

26.2.2 ANFL

(1) Loan of Rupees Nil (31st March, 2018 - Rs. 200.00 Lakh):

(a) Securities provided:

(i) The above loan is secured by personal guarantees of two Promoter Directors of the Parent Company.

(ii) Charge on movable property has been registered and on immovable property i.e. land admeasuring 1.88 acres is to be registered.

(b) Terms of Interest rate: Rate of interest is @ 11% p.a.

(26.3) Details of default in payment of interest on secured loans as on 31st March, 2019 are as follows:

INS. IN LAKA				
Total	Others	Banks	Year	
3,038.40	191	3,038.40	FY 2013-2014	
2,706.46		2,706.46	FY 2014-2015	
3,123.67	÷	3,123.67	FY 2015-2016	
3,628.35	· · · ·	3,628 35	FY 2016-2017	
8,144.73	3,737.82	4,406.91	FY 2017-2018	
18,336.63	11,321.62	7,015.01	FY 2018-2019	
38,978.24	15,059.44	23,918.80	Total	





Particulars	As at 31st March, 2019	(Rs. in Lakh) As at 31st March, 2018
27. Other Current Liabilities		
Advance received from Customers	333.50	6
Government Grants	365.49	639.61
Trade advances received	0.0	282.59
Other Advances	1.89	1.74
Statutory dues (Refer note below)	2,120.97	1,397.22
Interest on delayed payment of statutory dues	1,608.18	1,360.85
Total	4,430.03	3,682.01

Notes: (1) Statutory dues included Tax deducted at sources (TDS), Provident Fund (PF), Profession Tax (PT) and Employee State Insurance Corporation (ESIC)

(2) Based on recent Supreme court judgement on structure of component for calculation of Provident Fund dated February 28, 2019 there are various interpretive issues including its applicability thus prospective provision w.e.f. March 01, 2019 been considered of Rs. 1.36 lakhs.

Government Grant		(Rs. in Lakh)
Particulars	As at 31st March, 2019	As at 31st March, 2018
Opening balance Income recognised during the year	2,363.30 (365.49)	2,728.79
Closing balance	1,997.81	2,363.30
Non-current liabilities Current liabilities	1,632.32 365.49	1,723.69 639.61
		(Rs. in Lakh)
Particulars	As at 31st March, 2019	As at 31st March, 2018
28. Provision Provision for employee benefits (Refer Note No. 40) Gratuity Leave encashment	3.77 22.08	6.12 16.31
Total	25.85	22.43



y E Aom

(Rs. in La		
Particulars	Year Ended 31st March, 2019	Year Ended 31st March, 2018
29. Revenue from Operations		*
Sale of services	A Commenter	
Rail freight income	12,074.57	12,374.23
Storage income	8,623,97	8,422.96
Conditional Lease rent	2,500.00	0,422.00
Road freight income	2,506.21	1,906.23
Material handling and other services	2.864.45	815.18
Terminal income	233.63	175.40
Domestic warehousing income	121.00	20.00
Finance lease income	3.00	
Other operating income	10.55	2,192.69
Total	28,937.38	25,906.69
30. Other Income		
Interest income on		
Bank fixed deposits	39.72	54.23
Loans to others	25.07	24.94
Others	8.54	61.13
Dividend income		0.60
Government grants	365,49	365.49
Financial guarantee income	227.88	2.57
Financial assets carried at amortised cost	193.06	1
Foreign exchange difference (net)	e l	105.30
Provision for doubtful debts written back		109.61
Excess provision written back	1. to 1.	463.14
Sundry balance written back	931.35	452.89
Gain on derecognised of Liability Component	653.17	-
Gain on disposal of Property, plant and equipment	0.03	
Miscellaneous income	15.78	25.29
Total	2,460.09	1,665.19
31. Freight Expenses		
Rail freight expenses	9,449.49	10,215.90
Road freight expenses	1,504.81	1,452.41
Total	10,954.30	11,668.31
32. Employee Benefits Expense		
Salaries, wages and bonus	3,568.32	3,435.33
Contribution to provident and other funds	102.07	93.09
Staff welfare expenses	136.46	106.12
Total	3,806.85	3,634.54



A x Aar

Notes to the Consolidated financial statements for	the year ended 31st March, 2019
--	---------------------------------

(Rs. in La		
Particulars	Year Ended 31st March, 2019	Year Ended 31st March, 2018
33. Finance Cost		
Interest expense on borrowings	25,951,79	30,647.03
Interest expense on statutory dues	315.67	707.04
Unwinded interest expense on security deposits	7.93	156.48
Interest expense on others	914.37	3.52
Interest expense on MSMED vendors (Refer Note No. 39)	4.01	0.25
Other borrowing costs	365.62	84.07
Total	27,559.39	31,598.39
4. Depreciation and Amortisation Expense		
Depreciation on property, plant and equipment	8,565.15	9,449.87
Amortisation on intangible assets	854.41	721.89
Total	9,419.56	10,171.76
35. Other Expenses		
Electricity charges	562.86	325.80
Rent	553.88	497.85
Repairs and maintenance:	A CONTRACTOR OF	
Building	160.77	809.46
Plant and Machinery	74.75	84.53
Others	132.88	135.36
Insurance	55.73	60.80
Rates and taxes	69.52	34.98
Communication expenses	73.55	73.24
Travelling and conveyance expenses	438.30	447.72
Vehicle expenses	94.38	102.69
Printing and stationery	49.64	44.20
Legal and professional fees	409.31	344.60
Security charges	415.37	392.09
Advertisement and Business Promotion expenses	73.35	198.08
Auditor's remuneration:	1.	
- Audit Fees	137.69	104.79
- Limited Review Fees	15.00	18.25
- Certification fees	31.95	1.75
Provision for doubtful debts	112.65	57 49
Bad Debts	45.81	101.54
Foreign exchange differences (net)	43.21	10,000
Miscellaneous expenses	209.70	184.66
Director sitting fees	2.65	
Discarding/written off of Property, plant and equipment and Intangible assets	166.34	27.62
Total	3,929.29	4,047.50
6. Exceptional Items		1
		1500.000
Reconciliation of loan accounts (net)	15 107 01	(562.39)
Settlement of claims	(5,167.04)	(19,478.47
Loss on sale of subsidiary (Refer Note No. 57)		(4,314.69
Gain on monetization of property, plant and equipment (Refer	18	(15,633.29
Note No. 59)		
Loss on sale/discard of Property, plant and equipment	1	515.64
Total	(5,167.04)	(39,473.20)



y Pz S V X som

Notes to Consolidated financial statements for the year ended 31st March, 2019

37 Contingent Liabilities and Commitments

37.1 Contingent Liabilities (to the extent not provided for in respect of):

(Rs			(Rs. in Lakh)
S. No.	Particulars	As at 31st March, 2019	As at 31st March, 2018
(a)	Disputed Income Tax Demands	18,515.52	11,087.78
(b)	Disputed Sales Tax demands		20.51
(c)	Disputed Service Tax demand	114.23	51,55
(d)	Disputed Local Body Tax demand	160.33	160.33
(e)	Claims against the group not acknowledged as debts	2,291.99	2,397.12
(f)	Bank Guarantees	116.25	162.00
(g)	Letter of Credit (Letter of Credit given in favour of Railways for availing e-freight facility for haulage payment)	100.00	100.00
(h)	Import Continuity / Transshipment Bond / Custodian cum Carrier Bond	43,901.21	31,910.21
(i)	Corporate Guarantees given	18,500.00	

37.2 Capital commitments

Estimated amount of contracts remaining to be executed on capital and other accounts and not provided for (net of advances paid) are Rs. 6,345.90 Lakh (31st March, 2018 - Rs. 449.60 Lakh)

38 Operating lease commitments

(a) The Arshiya Lifestyle Limited (wholly owned subsidiary) has entered into operating lease arrangements for certain warehouse facilities. The lease is non-cancellable and is for a period of 6 years and may be renewed for a further period of 6 years based on mutual agreement of the parties. The lease agreements provide for an increase in the lease payments by 5 % every year from lease commencement date.

Particulars	As at 31st March 2019	As at 31st March 2018
Future Non-Cancellable minimum lease commitments		
Within one year	4,916.73	4,642.37
Later than one year but not later than five years	21,214.21	21,009.65
Later than five years		4,896.67

(b) The Group has taken office on lease under non-cancellable operating lease expiring at the end of 3 years. The leases have varying terms, escalation clauses and renewal rights. On renewal, the terms of the leases are renegotiated.

(Rs. in L		
Particulars	As at 31st March, 2019	As at 31st March, 2018
Future Non-Cancellable minimum lease commitments		
Within one year	351.92	340.65
Later than one year but not later than five years	699.19	1,051.11
Later than five years	· · · · ·	+

39 Details of dues to Micro, Small And Medium Enterprises as per MSMED Act, 2006

	(Rs. in La		
S. No.	Description	As at 31st March, 2019	As at 31st March, 2018
a)	Principal amount due and remaining unpaid	84.98	18.11
b)	Interest due thereon remaining unpaid	4.01	0.22
c)	Interest paid by the Group in terms of Section 16 of the MSMED Act, 2016, along with the amount of the payment made to the suppliers beyond the appointed day during the year		
d)	Interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under the MSMED Act, 2006	2	-
e)	Interest accrued and remaining unpaid	4.01	0.22
f)	Interest remaining due and payable even in the succeeding years, until such date when the interest dues as above ae actually paid to the micro and small enterprises.	÷.	

Note: Dues to Micro, Small and Medium enterprises have been determined to the extent such parties have been identified on the basis of information collected by the Group and relied upon by the Auditors.



1 ~ Aa

Notes to Consolidated financial statements for the year ended 31st March, 2019

- 40 **Employee Benefits**
- 40.1 Disclosure pursuant to Indian Accounting Standard (IND AS) 19 Employee Benefits
- (a) **Defined Contribution Plan:**
 - Contribution to Defined Contribution Plan, recognised as expenses for the years are as under:

(Rs. in La		
Particulars	Year ended 31st March, 2019	Year ended 31st March, 2018
Employer's Contribution to Provident Fund	21.58	19.63
Employer's Contribution to Pension Scheme	49.02	44.61
Employer's Contribution to ESIC	2.12	3.30

(b) Brief descriptions of the plans

The Group's defined contribution plans are Provident Fund and Employees State Insurance where the Group has no further obligation beyond making the contributions. The Group's defined benefit plans include gratuity. The employees are also entitled to leave encashment as per the Group's policy.

- Encombrant (c)

Particulars	As at 31st March, 2019	As at 31st March, 2018
Provision recognised in the Balance Sheet		
Current Provision as at the end of the year	22.08	16.3*
Non Current Provision as at the end of the year	101.78	80.06
Provision recognised in the Balance Sheet	123.86	96.37

Defined benefit plan - Gratuity: (d)

The employee's Gratuity fund is managed by the Life Insurance Corporation of India. The present value of obligation is determined based on actuarial valuation using the Projected Unit Credit Method, which recognised each period of services as giving rise to additional unit of employee benefit entitlement and measures each unit separately to build up to final obligation.

		(Rs. in Lakh
Particulars	31st March, 2019	31st March, 2018
I. Actuarial assumptions. Mortality Table	Indian Assured lives Mortality (2006-08) Ult	Indian Assured live Mortality (2006-08) Ult
Discount rate	6.95%	
Expected return on plan assets	7.40%	
Salary Escalation Rate	9.00%	
Withdrawal Rate	17.00%	
Retirement Age	58 Years	
II. Change in Present value of defined benefit obligations		
Provision as at the beginning of the year	200.03	171.79
Interest cost	14.87	11.03
Current service cost	42.21	55.01
Benefits paid	(22.25)	32.45
Actuarial (gain)/loss on obligations	28.56	(10.62
Provision as at the end of the year	263.42	259.66
III. Change in Fair value of plan assets		
Fair value of plan assets as at the beginning of the year	81.66	118.68
Expected return on plan assets	6.04	15.67
Actual Enterprise's Contributions	(22.25)	(32.45
Benefits paid	22.25	32.45
Unrecognised asset due to limit in Para 64(b)		5.2
Actuarial gain/(loss) on plan assets		(4.59
Fair value of plan assets as at the end of the year	87.70	129.76
IV. Actual return on plan assets	1	
Expected return on plan assets	6.04	15.67
Actuarial gain/(loss) on plan assets		(4.59
Actual return on plan assets	6.04	11.08
V. Provision recognisod in the Balance Sheet		and the second
Provision as at the end of the year	263.42	259.66
Fair value of plan assets as at the end of the year	87.70	129,78
Provision recognised in the Balance Sheet	175.72	129.88





VI. Percentage of each category of plan assets to total fair value of plan assets Insurer managed funds	100%	1000
	100%	100%
VII. Amount recognised in the Statement of Profit and Loss		
Current service cost	42.21	55.01
Interest cost	11.04	11.03
Expected return on plan assets	(2.20)	(19.22)
Net actuarial (gain)/loss to be on obligation	-	-
Expense recognised in Statement of Profit and Loss	51.04	46.83
VIII. Amount recognised in the Other Comprehensive Income (OCI)		
Due to Change in financial assumptions	32.31	(4.29)
Due to Change in demographic assumption	(4.19)	0.06
Due to Change in experience assumption	0.45	(6.39)
Expected return on plan assets		(4.59)
Total remasurement recognised in OCI	28.57	(9.67)
IX. Balance Sheet reconciliation		-
Opening net provision	118.37	53.11
Expenses recognised in Profit & Loss	51.04	57.91
Actual Employer Contribution	(22.25)	28.53
Net transfer by group companies		
Total Remeasurement recognised in OCI	28.56	(9.67)
Closing net provision	175.72	129.88

(e) Salary escalation assumption has been set in discussions with the enterprise based on their estimates of overall long-term salary growth rates after taking into consideration expected earnings inflation as well as performance and seniority related increases.

	0 1 1 1 1 1 1 1 1 1 1	
Particulars	Changes in assumptions	Effect on Gratuity obligation
For the year ended 31st March, 2018		
Salary growth rate	+0.50%	289.11
	-0.50%	275.58
Discount rate	+0.50%	275.63
The state of the second s	-0.50%	289.11
For the year ended 31st March, 2019		
Salary growth rate	+0.10%	273.60
	-0.10%	253.86
Discount rate	+0.10%	253.79
	-0.10%	273.82

The above sensitivity analysis is based on a change in an assumption while holding all other assumptions constant. In practice, this is unlikely to occur, and changes in some of the assumptions may be correlated. In presenting the above sensitivity analysis, the present value of defined benefit obligation has been calculated using the projected unit credit method at the end of reporting period, which is the same as that applied in calculating the defined obligation liability recognised in the balance sheet.

These plans typically expose the Group to actuarial risks such as: longevity risk and salary risk.

- (a) Interest risk A decrease in the discount rate will increase the plan provision.
- (b) Longevity risk The present value of the defined benefit plan provision is calculated by reference to the best estimate of the mortality of plan participants both during and after their employment. An increase in the life expectancy of the plan participants. As such, an increase the plan's provision.
- (c) Salary risk The present value of the defined plan provision is calculated by reference to the future salaries of plan participants. As such, as increase in the salary of the plan participants will increase the plan's provision.
- 40.3 The weighted average duration of the defined benefit obligation at the end of the reporting period is 5 years (31st March, 2018 6 years).



Aan

Notes to Consolidated financial statements for the year ended 31st March, 2019

41 Preparation of financial statements on " Going Concern" basis

(a) One of the subsidiary Company viz Arshiya Rail Infrastructure Limited has accumulated losses and negative net worth. Some of its lenders have recalled their loans and the subsidiary company is in the process of negotiating the revised payment terms. The subsidiary Company is EBIDTA positive and with the commencement of the two dedicated freight corridors, the said subsidiary company will benefit immensely. Moreover a unique contract entered into with one of the largest global shipping lines has already started to improve the profitability. Also, the proposed merger of another subsidiary company with this subsidiary company, would make their operations not only complementary, but enhance their profitability.

In view of the focussed emphasis of the Government on logistics infrastructure sector, the proposed restructuring and considering the fact that the facilities are yet to achieve full operational potential besides the strategic locations of the facilities, the management's future outlook of its businesses is very promising. Accordingly, the financials have been prepared on going concern basis including based on financial support from the Parent Company.

- (i) Diversified focus from only Steel industry to Cement, Agro and Tiles also so as to have a balance of product mix,
- (ii) Stabilizing of PFT business with Long term contracts and constructing the second line,
- (iii) Standardization of Containers to be able to better utilize the assets,
- (iv) Government focus on Multi-modal logistic and transport services to increase the throughput of the infrastructure already created by the company,
- (v) Government focus on the revamping of the Railway Boards and increasing clarity on regulatory aspects to support resolution of the regulatory issues,
- (vi) Two Dedicated Freight Corridor(s) (DFC) along the Western and Eastern part of India to support increase in the business volume,
- (vii) Government announcement of Jewar Airport and Merut Highway connecting Jewar via Khurja and thereby connecting to Yamuna Expressway(i.e. Delhi to Agra),
- (b) Certain creditors have initiated legal proceedings against the Group and its Directors and the Group has defaulted in payment of instalments of consent terms for which the Group is in process of negotiating and finalising the revised consent terms. Majority of the creditors have been settled over the past few years and some of the creditors have also shown interest and faith not only in the logistics infrastructure sector but also in Arshiya Group and are being allotted equity shares of the Parent Company.
- (c) The Parent Company, ATHL, ANFL and AIDHL has accumulated losses and certain creditors have initiated legal proceeding against this companies and their Directors for recovery of the amounts due. However in certain cases has executed consent terms or is in the process of finalizing consent terms with the creditors.

The Parent Company has given its warehouses on long term lease and received upfront lease payments. The management has also initiated various other steps such as construction and future development within the FTWZ, restructuring the Parent Company and it's subsidiaries business operations. The Parent Company, has given a support letter to extend, for the foreseeable future, any financial support which may be required by those subsidiaries. Considering the strength of locational advantages, future outlook as assessed by the management and business plan, those subsidiaries' management is confident to continue as a going concern. The long term prospects, however, are dependent on various factors and consolidated financial statements of those entities have accordingly been continued to be prepared on going concern basis.

The management is in the process of restructuring its business operations and steps are as under:

- Competitive advantage of the FTWZ with easily accessible to two most important retail market in NCR- Gurgaon and Delhi to increase utilization,
- (ii) Aligning warehouse and distribution center logistics to support companies in alignment with business strategy and provide a competitive edge in Mutli-modal Logistics.
- (iii) ICD operational facility has now entered into long term contract with global shipping majors,
- (iv) The planned long term contract for transportation of Reefer cargo to increase revenue,
- (v) Increasing throughput through collaborative 'Pooling of assets' with other ICD and Private Container Train Operators (PCTO).
- (vi) Increasing interest from various Global customers for integrated solutions including rail transport and FTWZ,
- 42 Loans from various lenders have been assigned by banks to Edelweiss Assets Reconstruction Company Limited (EARC). EARC had restructured the loan and executed the Restructuring Agreement (RA) dated 31st March, 2017. In accordance with RA, EARC has converted part debt into restructured debt, balance assigned loan is to be converted into 3,21,62,304 equity shares and 64,23,329 zero porcent optionally convertible redeemable preference shares (OCRPS Series I) of face value of Rs.10 each at a price of Rs.1,000 each (including premium of Rs.990) of the Company, as per extant SEBI rules and regulations. The EARC has availed the right of ronversion of OCRPS into equity.

During the year ended 31st March, 2019;-

- (i) In aggregate 1,38,97,516 equity shares of 2 each on conversion of OCRPS Series I have been allotted to EARC.
- (ii) Pursuant to RA, the Company has allotted 15,00,000 Equity Shares of face value of Rs.2 each to the Promoter upon conversion of equal number of warrants.



Notes to Consolidated financial statements for the year ended 31st March, 2019

43 The Parent Company is engaged in the business of development, operations and maintenance of Free Trade and Warehousing Zone (FTWZ) and Domestic Warehousing Zone. During the year ended 31st March, 2019, certain portion of Land which was classified under Property, Plant and Equipment (PPE) is now transferred to inventories at their carrying amounts for future developments.

Out of the above land parcels, during the year ended 31st March, 2019 the Parent Company has entered into 2 lease agreements aggregating to 5.50 Acres of land with a wholly owned subsidiary company for development of warehouses at FTWZ, Panvel.

- 44 The Parent Company and its subsidiaries have defaulted in agreed repayment schedule of Restructuring Agreement (RA). As per debt covenant, the Parent Company and its subsidiaries are required to adhere to repayment schedule and any short payment gives Edelweiss Asset Reconstruction Company (EARC) the right to convert whole of the outstanding amount of restructured rupee loan and/or part of the default amount into fully paid up equity shares of the Parent Company. No such notice of conversion in writing has been given by EARC and the Parent Company and its subsidiaries continues to disclose the amount as non-current and current borrowings as per repayment schedule, in the Balance Sheet.
- 45 During the year ended 31st March, 2019, the Parent Company has defaulted in payment as per consent terms signed with one of the Non-Banking Financial Company (NBFC). Subsequent to the year end, the said NBFC has assigned its debt to Edelweiss Asset Reconstruction Company (EARC). Pursuant to the said assignment, EARC become the lender and entitled to recover total dues alongwith interest at contractual rates and other charges. The Parent Company doesn't expect any additional liabilities / charges and liabilities accounted in the books of account are adequate.
- 46 A Public Financial Institution (PFI) agreed to settle their outstanding loan constituting principle and interest of Rs. 16,700 Lakh. Settlement terms and conditions involves payment of Rs. 5,000 Lakh which is secured by land at Nagpur and for balance amount of Rs. 11,700 Lakh, allotment of Optionally Convertible Redeemable Preference Shares V (OCRPS V), convertible upto 15,50,000 equity shares at the option of the PFI. Considering the same, necessary effect has been given in the books of accounts during the previous year. As per shareholder approval in the EOGM dated 29th January 2018, the Parent Company has approved allotment of 11,70,000 OCRPS V and the same was converted into 15,50,000 Equity shares on 22nd February, 2018 as per settlement terms agreed. Subsequently in the Honorable High Court of Bombay, the Parent Company has made the representation that post allotment of the equity shares as exercised by the PFI, the total outstanding debt remains at Rs. 5,000 Lakh.

During the year ended 31st March, 2019, the PFI has assigned its debt to the Edelweiss Asset Reconstruction Company (EARC). The Company has provided interest in line with major terms negotiated with EARC, till the finalisation of the restructuring agreement.

47 A subsidiary Company i.e. Arshiya Rail Infrastructure Limited had entered into one-time settlement (OTS) with a Bank during the year ended 31st March, 2019 and the effect was taken as an exceptional item during the year ended 31st March, 2019. However, the subsidiary Company has defaulted in payment as per the terms of the OTS. As a result, the subsidiary Company needs to reverse the exceptional gain recorded during the year ended 31st March, 2019 and needs to recognise Interest on the entire liability as per the original terms. The subsidiary Company is in discussion with the lender for additional time to repay.

The subsidiary Company has not reversed the gain, nor provided for additional interest. Had the Subsidiary Company reversed the gain and provided for additional interest, exceptional item would have been lower by Rs. 6,604.55 Lakh and finance cost would have been higher by Rs. 3,500.76 Lakh having consequential impact on total comprehensive income for the year ended 31st March. 2019.

48 Cash Seized by Income Tax

The amount of Rs. 100 Lakh cash seized by the Income Tax department at the time of search on 13th June, 2014 has adjusted the said cash seized against demand of the Parent Company and to be specific against Assessment Year 2014-2015. While there is a demand in Assessment Year 2014-2015, the same is contested and the said demand is reflected in Contingent Liability (Refer note no. 37).

- 49 Scheme of arrangement and amalgamation u/s 230 to 232 and other applicable provisions of the Companies Act, 2013 has been filed before the National Company Law Tribunal ('NCLT") between Arshiya Rail Infrastructure Limited, Arshiya Industrial & Distribution Hub Limited and Arshiya Transport & Handling Limited and their respective shareholders. The scheme is conditional on various approval / sanctions and is effective thereafter, accordingly no effect of the said Scheme is given in the consolidated financial statements. The Creditors meeting of the respective companies was held on 6th May 2019. The Scheme(s) shall be given effect after receipt of necessary approvals.
- 50 The Board of Directors of the Parent Company in their meeting held on 24th May, 2018, has approved a scheme to further reorganize its corporate structure spread across various group companies, in order to integrate/consolidate it's operations by reorganizing different businesses into two entities subject to various approvals.



× Aam

Notes to Consolidated financial statements for the year ended 31st March, 2019

This Scheme is presented under Sections 230 to 232 read with Sections 66 and 52 and other applicable provisions of the Companies Act, 2013 for demerger of "Domestic warehousing business" of the Parent Company into Arshiya Rail Infrastructure Limited. This proposed scheme of arrangement is conditional upon approval of an ongoing scheme of group companies i.e. merger of Arshiya Rail Infrastructure Limited, Arshiya Industrial and Distribution Hub Limited and Arshiya Transport & Handling Limited, which is pending with NCLT. No accounting impact and disclosures is considered and necessary at this stage pending requisite regulatory approvals.

51 Cenvat Credit Receivable:

The Group has been legally advised that post merger of the AIDHL with ARIL, the unutilized cenvat credit of the AIDHL can be utilized for discharging the service tax liability of ARIL.

52 Indirect Tax Refund Receivable

- (i) As per the Notification dated 16th May, 2013 issued by Government of Maharashtra, MVAT exemption /refund is available to SEZ Developer after 15th October, 2011 (record date). However, the Parent Company has claimed refund of Rs.1,684.55 Lakh in respect of transactions prior to record date, as the Parent Company is of the view that the State Government has exempted it from Local taxes, levies and duties on goods required for authorized operations by a Developer vide GR dated 12th October, 2001 passed by Industries, Energy and Labour Department, Government of Maharashtra. The Parent Company has filed an appeal before High Court of Bombay challenging the Constitutional validity of MVAT on various grounds and has claimed refund of Rs.1,108.80 Lakhs. The Appeal has been admitted, issues are framed and final hearing is pending. Further MVAT refund claim of 575.75 Lakhs is pending with Sales Tax Department as the matter is of similar case. Accordingly, these financial statements reflect a sum of Rs.1,684.55 Lakhs as refund receivable on account of MVAT. In case the refund is not granted, the necessary adjustment entries shall be recorded in the year in which finality is reached.
- (ii) Refunds receivable in respect of VAT, Service Tax, Local Entry Tax and Service Tax for which appeals are pending with respective Appellate Authorities. The Management is of the view that the refunds claimed as above aggregating to Rs. 355.05 Lakh are considered good for recovery on account of refunds being received by other SEZ developers on similar grounds.

The Management is of the view that the refunds claimed as above are considered good for recovery.

- 53 ARIL has procured certain capital goods under EPCG scheme at concessional rate of duty. On non fulfillment of certain conditions, ARIL may become liable to pay differential custom duty along with interest thereon such procurement. The management is hopeful of completing the expected obligation within the stipulated time.
- 54 Corporation Bank has filed a suit with Debt Recovery Tribunal, New Delhi, towards recovery against Arshiya Rail Infrastructure Limited, the Parent Company as a Corporate Guarantor and two Promoter Directors of the Parent Company as Guarantors, for Rs. 8,012.60 lakhs. The same is pending before the DRT Delhi. The matter is sub-judice.
- 55 During the year ended 31st March, 2018, two lenders of a subsidiary i.e. Arshiya Industrial and Distribution Hub Limited ("AIDHL") have assigned their rights, title, and interest in financial assistance granted by them to Edelweiss Assets Reconstruction Company Limited (EARC). Post assignment of loans, EARC has become a secured lender of AIDHL and right, title and interest of the lenders have vested into EARC.

Pursuant to the assignment of such loans, and in terms of the restructuring package approved by EARC for the loans so assigned, the subsidiary has executed Restructuring Agreement (RA) with EARC, on behalf of EARC Trusts on 13th January, 2018 taking the aggregate amount of assigned loans to Rs 20,998 Lakh.

As a result of this restructuring and assignment of debts of lenders the gain earned amounting to Rs 10.398.92 Lakh has been credited to the Consolidated Statement of Profit and Loss for the year ended on 31st March, 2018. This has been disclosed as part of an exceptional item.

- 56 During the year ended 31st March, 2018 a subsidiary i.e. AIDHL has completed one time settlement (OTS) with a lender in respect of the term loan taken. OTS stipulates payment and allotment of Optionally Convertible Redeemable Preference Shares. AIDHL has made a payment of Rs 3,000 Lakh on 18th January,2018 and issued 1,20,000 OCRPS. Gain of Rs. 7,790.75 Lakh on this OTS has been credited to the Consolidated Statement of Profit & Loss as an exceptional item.
- 57 During the year ended 31st March, 2018 the Parent Company and a subsidiary i.e. Arshiya Northern FTWZ Ltd ("ANFTZ") have divested their entire investment in one of subsidiary namely Mira Supply Chain Management Private Limited (formerly known as Mira Supply Chain Management Private Limited) ("ASCM") on 2nd January 2018 by way of transfer of equity shares to a subsidiary i.e. AIDHL on 22nd March, 2018, AIDHL has divested its entire shareholding in ASCM. Pursuant to above, Rs.4,314.69 Lakh gain is accounted in consolidated financial statement which has been considered as exceptional item during the year ended 31st March 2018.
- 58 As per provision of sub section 1 of section 203 of Companies Act, 2013 (w.e.f. 1st April, 2014), ANFL is required to appoint a Company Secretary. However, ANFL has not complied with the said requirement and is in the process of identifying a suitable candidate for this role.



Notes to Consolidated financial statements for the year ended 31st March, 2019

59 During the year ended 31st March, 2018, on 23 November 2017, the Parent Company, interalia, it's subsidiaries and promoters had executed Share Purchase Agreement of Ascendas Panvel FTWZ Limited [formerly known as Arshiya Rail Siding and Infrastructure Limited ("APFL", i.e. a step-down subsidiary/"SPV")], with Ascendas Property Fund (India) Pte Ltd ('Ascendas') for sale of 100% of it's equity holding, having Rs. 5 Lakh paid up equity capital, to Ascendas. This SPV holds the status of a co-developer.

During the year ended 31st March, 2018, the Parent Company, interalia, it's subsidiaries and promoters have executed a Lease Deed on 3rd February 2018, in favour of a SPV of Ascendas Property Fund (India) Pte. Limited ("Ascendas" - part of the Ascendas-Singbridge Group, Singapore) for grant of leasehold rights of six warehouses at FTWZ Panvel, along with underlying land of those warehouses, identified assets and infrastructure facilities on an initial lease term of 30 (thirty) years. The said transaction is for a total consideration of Rs. 53,400 Lakh (or Rupees Five hundred and thirty four crore), with an upfront lease payment/lump sum rent of Rs. 43,400 Lakh (or Rupees Four hundred and thirty four crore). The balance of Rs. 10,000 Lakh (or Rupees One hundred crore) will be received over four years from transaction closing based on certain performance milestones. The transaction also envisages the terms for construction funding by Ascendas for future growth of the Parent Company's business. The Parent Company already possesses the requisite land for the future development.

On transaction closing date of 3rd February 2018, the SPV has acquired long-term leasehold rights from the Parent Company and the same are leased back under an operating lease arrangement pursuant to execution of sub-lease deed dated 3rd February 2018 to Arshiya Lifestyle Limited ("ALL"), a wholly owned subsidiary of the Parent Company, for a sub-lease term of 6 (six) years, renewable as per mutually agreed terms, in consideration of pre-agreed rentals.

Accordingly during the year ended 31st March, 2018 the Parent Company has reduced the value of assets, granted on leasehold rights to APFL, from its fixed assets. The gain on grant of leasehold rights to APFL amounting to Rs. 15,633.29 lakhs has been credited to the statement of profit and loss and is disclosed as an exceptional item.

Based on the above, ALL would operate and manage these six warehouses and pay the lease rentals to APFL as defined in the sublease agreement. Hence from 3rd February, 2018 onwards all revenue from these assets will be accounted by ALL. However the Parent Company will recognize the net revenue in terms of a business conducting agreement entered into between the Parent company and ALL.

60 Disclosure pursuant to Indian Accounting Standard (IND AS) 108 - Segment Information

60.1 Primary Segment Information

The Group operates in three primary reportable business segments, i.e. "Free Trade and Warehousing Zone" and "Rail Transport Operations/Inland Container Depot" and "Domestic Warehousing Zone" and one geographical segment i.e. India as per Accounting Standard 108 – "Segment Reporting".

60.2 Segment Revenue, results, assets and liabilities

Revenue and results have been identified to a segment on the basis of relationship to operating activities of the segment.

Segment assets and segment liabilities represent assets and liabilities in respective segments. Segment assets include all operating assets used by the operating segment and mainly includes trade receivable, inventories and other receivables. Segment liabilities primarily include borrowings, trade payables and other liabilities. Assets and liabilities which cannot be allocated to any of the segments are shown as a part of unallocable assets and liabilities.

60.3 The chief operational decision maker monitors the operating results of its Business Segments separately for the purpose of making decisions about resource allocation and performance assessment. Segment performance is evaluated based on profit or loss and is measured consistently with profit or loss in the financial statements. Operating segment have been identified on the basis of the nature of services.



0 tem

60.4 Segmental Information as at and for the year ended 31st March, 2019 is as follows:-

Particulars	Year Ended	Year Ended
Segment Revenue	31st March 2019	31st March 2018
FTWZ	14,758.88	12,233.47
Rail Transport Operations/ICD	13,750.16	13,653.22
Domestic Warehousing	428.34	20.00
Less: Inter Segment	420.04	20.00
Total Revenue from Operations	28,937.38	25,906.69
Segment Results Before Tax and Interest		
FTWZ	1,325.17	1,006.72
Rail Transport Operations/ICD	(4,567.90)	(4,289.19)
Domestic Warehousing	(1,106.05)	(1,458.01)
Total Segment Result	(4,348.78)	(4,740.48
Less: Unallocated Expenses net of Income	9.74	2.80
Less: Finance Costs	27,559.39	31,598.39
Less: Exceptional Items (Net)	(5,167.04)	(39,473.20)
Loss before tax	(26,750.87)	3,131.53
Less: Tax Expenses	6.98	27.42
Loss after tax	(26,757.85)	3,104.11
Segment Assets		
FTWZ	1,81,666.82	1,85,856.32
Rail Transport Operations/ICD	73,038.42	77,717.22
Domestic Warehousing	48,982.63	49,852.04
Unallocated	2,092.84	2,199.05
Total	3,05,780.71	3,15,624.63
Segment Liabilities		
FTWZ	11,981.34	13,267.09
Rail Transport Operations/ICD	7,746.04	7,581.09
Domestic Warehousing	120.38	3.60
Unallocated	2,31,948.11	2,22,751.84
Total	2,51,795.87	2,43,603.62
Other Disclosures		
Capital Expenditure		and the second
FTWZ	3,256.98	5,944.51
Rail Transport Operations/ICD	274.28	883.56
Domestic Warehousing	13.45	
Unallocated		
Total	3,544.71	6,828.07
Depreciation and amortisation expenses		
FTWZ	3,094.37	3,694.97
Rail Transport Operations/ICD	4,913.85	4,998.78
Domestic Warehousing	1,411.34	1,478.01
Unallocated		
Total	9,419.56	10,171.76
Non-cash Expenditure		
FTWZ	300.09	85.29
Rail Transport Operations/ICD	24.71	101.36
Domestic Warehousing		
Unallocated		
Total	324.80	186.65



A × have

- 61 Related party disclosures, as required by Indian Accounting Standard 24 "Related Party Disclosures" (IND AS-24) as given below:
- (I) Person having significant influence over the Parent Company Mr. Ajay S Mittal – Chairman and Managing Director Mrs. Archana A Mittal – Joint Managing Director
- (II) Key management personnel

Mr. Ashish Bairagra - Independent Director

- Mr. Mukesh Kacker Independent Director
- Prof. G. Raghuram Independent Director (till 15th May, 2017)
- Mr. Rishabh Shah Independent Director
- Mr. Tara Sankar Bhattacharya Independent Director (w.e.f. 24th May, 2018)
- Ms. Savita Dalal Company Secretary of Arshiya Limited
- Mr. Sanjay Sukhram Lakkhan Company Secretary of Arshiya Industrial & Distribution Hub Limited (till 7th February, 2019)
- Ms. Avani Dipakkumar Lakhani Company Secretary of Arshiya Rail Infrastructure Limited (w.e.f. 26th March, 2019)
- Mr. S. Maheshwari Chief Financial Officer of Arshiya Limited (w.e.f. 8th February, 2017) and Group President
- Mr. Navnit Choudhary Chief Financial Officer of Arshiya Northern FTWZ Limited (w.e.f. 8th February, 2017)

Mr. Dinesh kumar Sodani - Chief Financial Officer of Arshiya Rail Infrastructure Limited (w.e.f. 8th February, 2017 to 12th January, 2018)

Mr. Mukesh Khathuria - Chief Financial Officer of Arshiya Industrial & Distribution Hub Limited (w.e.f. 8th February, 2017 to 31st March, 2018)

Mr. Dinesh Kumar Sodani - Chief Financial Officer of Arshiya Industrial & Distribution Hub Limited (w.e.f. March 26, 2019) Mr. Vinod Jain - Chief Financial Officer of Arshiya Rail Infrastructure Limited (till 25th March, 2019)

- Mr. Amardeep Gupta Chief Financial Officer of Arshiya Rail Infrastructure Limited (w.e.f. 26th March, 2019)
- Mr. Siddarth Kasturia Chief Executive Officer of Arshiya Rail Infrastructure Limited (w.e.f. 26th March, 2019)
- Mr. Amit Gupta Chief Executive Officer of Arshiya Industrial & Distribution Hub Limited (w.e.f. March 26, 2019)
- (III) Relative of Person having significant influence over the Parent Company Mr. Ananya Mittal – Corporate Strategy Officer

(IV) Enterprise owned or significantly influenced by key management personnel or their relatives Rudradev Properties Private Limited

Mega Management Services Private Limited

Welldone Software Consultancy Private Limited

Noval FTWZ Limited (formaly known as Arshiya Central FTWZ Limited)

The nature and amount of transactions with the above related parties are as follows:

Nature of transaction	SNAMO of the Party	Veal Ended (15) March (2019)	Year Ended Stat March, 2018
Remuneration paid to Key Managerial Person and		25.68	25.68
Relative of Person having significant influence over the Group	Mr. S. Maheshwari	203.68	185.15
ine Group	Ms. Savita Dalal	16.61	- <u></u>
	Ms. Avani Dipakkumar Lakhani	0.10	
	Mr. Amardeep Gupta	0.36	+
	Mr. Vinod Jain	20.49	8
	Mr. Siddarth Kasturia	0.95	-
Director sitting fees	Mr. Ashishkumar Bairagra	1.15	2.0
	Mr. Mukesh Kacker	0.40	1.8
	Prof. G. Raghuram	-	0.2
	Mr. Rishabh Shah	0.40	1.7
	Mr. T S Bhattacharya	0.70	
Interest income	Noval FTWZ Limited	0.12	40.00



(Rs in Lakh)

Mr. Ajay S Mittal	788.46	3,064.09
Mrs. Archana A Mittal	1,116.14	4,260.42
Noval FTWZ Limited	325.00	-
Mr. Ajay S Mittal	1.153.32	2,594.62
Mrs. Archana A Mittal	2,528.94	2,860.31
Rudradev Properties Private Limited		1.00
Mega Management Services Private Limited	-	13.32
Welldone Software Consultancy Private Limited	14	0.19
Mrs. Archana A Mittal	25110	0.49
the second se		F.005.00
		5,835.00 5,820.00
		5,620.00
Equity Share	-	583.50
Mr. Ajay S Mittal	860.25	4,959.75
	Mrs. Archana A Mittal Noval FTWZ Limited Mr. Ajay S Mittal Mrs. Archana A Mittal Rudradev Properties Private Limited Mega Management Services Private Limited Welldone Software Consultancy Private Limited Mrs. Archana A Mittal Mrs. Archana A Mittal Mrs. Archana A Mittal Mrs. Archana A Mittal Mrs. Archana S Mittal Equity Share Share Warrants Mr. S. Maheshwari Equity Share	Mrs. Archana A Mittal 1,116.14 Noval FTWZ Limited 325.00 Mr. Ajay S Mittal 1,153.32 Mrs. Archana A Mittal 2,528.94 Rudradev Properties Private Limited - Mega Management Services Private - Limited - Welldone Software Consultancy - Private Limited - Mrs. Archana A Mittal - Mrs. Alpay S Mittal - Equity Share - Share Warrants - Mr. S. Maheshwari - Equity Share -

Closing balances			(Rs. in Lakh)
Nature	Related Party	As at 31st March, 2019	As at 31st March, 2018
Loans and Advances taken	Mr. Ajay S Mittal	1C4.89	469.75
	Mrs. Archana A Mittal	55.22	1,468.01
	Rudradev Properties Private Limited	-	1.00
Loan given	Noval FTWZ Limited	325.12	
Evaligiven	Noval FTVV2 Linited	525.12	
Share warrants	Mr. Ajay S Mittal	-	860.25
Equity Shares (excluding share premium)	Mr. S. Maheshwari	20.00	20.00
Personal guarantees taken	Mr. Ajay S Mittal	3,24,812.00	3,11,657.00
reisonal guarantees taken	Mrs. Archana A Mittal	3,24,262.00	3,11,262.00



y tam

Notes to Consolidated financial statements for the year ended 31st March, 2019

62 Earnings per share:

Particulars	Year ended 31st March 2019	Year ended 31st March 2018
Profit/(Loss) for the year (Rs. in Lakh)	(26,757,85)	3,104,11
Add: Interest adjustment on account of 0% Optionally Convertible Redeemable Preference Shares (OCRPS)		946.32
Total Profit/(Loss) for the year for diluted EPS (Rs. in Lakh)	(26,757.85)	4,050.43
Number of equity shares		
Weighted average number of equity shares (Number)	23,80,15,279	18,31,20,902
Add: Adjustment on account of Share Warrants	A	5,91,781
Add: Adjustment on account of 0% Optionally Convertible Redeemable Preference Shares (OCRPS)	9	54,82,856
Total Weighted average number of equity shares/shares warrants/OCRPS	23,80,15,279	18,91,95,538
Nominal value per share (Amount in Rs.)	2.00	2.00
Earnings per share – Basic and Diluted (Amount in Rs.)	(11.24)	1.70

0% OCRPS and share warrants had an anti diluting effect on earning per share hence have not been consider for the purpose of computing dilutive earning per share during the previous year.

63 Taxation

63.1 The Group has not recognised any deferred tax assets on deductible temporary differences, unused tax losses as it is not probable that the Group will have sufficient future taxable profit which can be available against the available tax losses.

63.2 Unused tax losses for which no deferred tax assets has been recognised

ess L	055	Unabso Deprecia		Availab	e for utilizatior till
	544.34		2,073.04	AY	2020-2021
	3.11		5,419.06	AY	2021-2022
	9,848.88	1	7,342.15	A.Y	2022-2023
	1,144.94	1	8,231.03	A.Y	2023-2024
	2,414.75	1	4,589.27	A.Y	. 2024-2025
8	33,302.00	1	4,097.35	A.Y	2025-2026
1	6,974.67		5.968.48	A.Y	2026-2027
	1,806.52		6,537.27	A.Y	2027-2028
1,2	26,039.21	8	4,257.65		

Assessment Year	sessment Year Long term Capital Loss	
2016-2017	1,658.88	A.Y. 2024-2025
Total	1,658.88	

Deferred tax assets as at 31st March, 2019 Rs.27,429.06 Lakh (31st March, 2018 - Rs. 45,714.08 Lakh) has not been recognised, as there is no convincing evidence that sufficient taxable profits will be available against which the unadjusted tax losses will be utilized by the Group. Details of deferred tax assets are mentioned below:

	(Rs. in Lakh)	
Particulars	As at 31st March, 2019	As at 31st March, 2018
Property, plant and equipment	16,991.77	17,129.21
Unabsorbed depreciation	(21,908.29)	(17,225.95)
Expenses allowable on payments under section 43B and 40 (a) (ia)	(6,087.89)	(12,538.11)
Unabsorbed losses	(33,047.07)	(30,844.25)
Financial Instruments	16,622.42	(2,234.98)
Total Deferred tax assets	(27,429.06)	(45,714.08)

64 Financial Risk Management

The Group's principal financial liabilities comprises of borrowings, trade and other payables and financial guarantees contracts. The main purpose of these financial liabilities is to manage for the Group's operations. The Group's financial assets comprises of trade and other receivables, cash and deposits that arises directly from its operations.

The Group's activities expose it to variety of financial risks including credit risk, liquidity risk and market risk. The Group's risks management assessment, management and processes are established to identify and analyze the risks faced by the Group to set up appropriate risks limits and controls, and to monitor such risks and compliances with the same. Risks assessment and management policies and processes are reviewed regularly to reflect changes in market conditions and the Group's activities.




Arshiya Limited Notes to Consolidated financial statements for the year ended 31st March, 2019

Risk	Exposure arising from	Measurement	Management
Credit risk.	Cash and cash equivalents, trade receivables, financial instruments, financial assets measured at amortised cost.		Regular review of credit limits
Liquidity risk	Borrowings and other liabilities	Rolling cash flow forecasts	Availability of borrowing facilities
Market risk – foreign exchange	Recognised financial assets and liabilities not denominated in Indian rupee (INR)		Unhedged
Market risk – interest rate	Long-term borrowings at variable rates	Sensitivity analysis	Unhedged

The Group's risk management is carried out by a corporate finance team under the policies approved by the Board of Directors. The Board provides written principles for overall risk management as well as policies covering specific areas, such as credit risk, interest rate risk.

(a) Credit Risk

The Group is exposed to credit risk, which is risk that counterparty will default on its contractual obligation resulting in a financial loss to the Group. Credit risk arises from cash and cash equivalents as well as credit exposures to trade customers including outstanding receivables.

Trade receivables are typically unsecured and are derived from revenue earned from customers located in India. Credit risk has always been managed by the Group through continuously monitoring the creditworthiness of customers to which the Group grants credit terms in the normal course of business. Outstanding customer receivables are regularly monitored. Credit risk is high as only few customers' account for majority of the revenue in the year presented. On account of adoption of Ind AS 109, the Group uses expected credit loss model to assess the impairment loss or gain.

(b) Liquidity Risk

Liquidity risk is the risk that the Group may not be able to meet its financial obligations without incurring unacceptable losses. The Group's objective is to, at all times; maintain optimum levels of liquidity to meet its cash and collateral requirements. The Group limits its liquidity risk by ensuring regular monitoring of funds from trade and other receivables. The Group relies on assets light business model through monetization of assets and tie-up of construction funding and operating cash flows to meet its needs for funds.

The table below provides undiscounted cash flows towards financial liabilities into relevant maturity based on the remaining period at the balance sheet to the contractual maturity date.

Contractual maturities of financial liabilities Particulars	less than 1 year	1 to 5 years	(Rs. in Lakh More than 5 year
31st March, 2019	toot statt t feat		more churre your
Financial liabilities		1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	
Borrowings	56,154.61	1,26,473.26	
OCRPS (Equity and Liability Component)		1,200.00	0 4
Trade payables	2,579.26	+	
Creditors for Capital Goods	7,228.05		
Financial guarantee obligations	10.55	-	-
Other financial liabilities	51,993.13	612.11	
То	tal 1,17,965.60	1,28,285.37	
31st March, 2018			
Financial liabilities	1		
Borrowings	50,179.38	1,31,491.71	
OCRPS (Equity and Liability Component)			58,846,19
Trade payables	1,797.79	- 14 C	
Creditors for Capital Goods	8,422.54		
Financial guarantee obligations		224.03	-
Other financial liabilities	35,105.55	2,076.73	*
To	tal 95,505.26	1,33,792.47	58,846.19

(c) Market Risk

Market Risk is the risk that the fair value of future cash flow of a financial instruments will fluctuate because of volatility of prices in the financial markets. Market risk can be further segregated as: 1) Foreign currency risk and 2) Interest rate risk



Arshiya Limited Notes to Consolidated financial statements for the year ended 31st March, 2019

1 Foreign currency risk

Foreign currency risk is the risk that the fair value or future cash flow or an exposure will fluctuate because of changes in foreign exchange rates. The Group's exposure to the risk of changes in foreign exchange rates relates primarily to the Group's operating activities.

1.1 Foreign currency risk exposure

(i) Details of foreign currency transactions/ balances not hedged by derivative instruments or otherwise are as under:

Particulars	Financial Year Ended	Foreign currency amount	Equivalent amount in INR
a capacity		(Amount in Lakh)	(Rs. in Lakh)
Bank balances			
USD	31st March, 2019		÷.
030	31st March, 2018	0.13	8.56
Trade receivables			
USD	31st March, 2019	46.09	3,207.35
030	31st March, 2018	29.62	1,903.03
QAR	31st March, 2019		-
QAR	31st March, 2018	0.04	0.79
EUR	31st March, 2019	0.41	28.39
EUR	31st March, 2018	0.47	36.85
Security Deposit from customers			
USD	31st March, 2019	6.20	428.85
030	31st March, 2018	5.61	360.71
EUR	31st March, 2019	0.02	1.35
EUR	31st March, 2018	0.03	2.72
AED	31st March, 2019	0.35	6.65
AED	31st March, 2018		
Advance from customers			
USD	31st March, 2019	0.35	26.45
050	31st March, 2018	0.53	34.01
FUE	31st March, 2019	0.0003	0.02
EUR	31st March, 2018		

1.2 Sensitivity

The Sensitivity of profit or loss to changes in the exchange rate arises mainly from foreign currency denominated financial instruments.

	Increase/(decrease) in	(Rs. in Lakh) profit before tax
Particulars	31st March, 2019	31st March, 2018
FX rate - increase by 1% on closing rate of reporting date	27.79	15.52
FX rate - (decrease) by 1% on closing rate of reporting date	(27.79)	(15.52)

The above amounts have been disclosed based on the accounting policy for exchange differences.

2 Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial inctrument will fluctuate because of changes in market interest rates. Majority of the Group's borrowings is fixed rate borrowings carried at amortised cost, therefore not subject to interest rate risk as defined in IND AS - 107, since neither carrying amount nor the future cash flows will fluctuate because of a change in market interest rates. The Group's interest rate risk arises from long term borrowings with variable rates, which expose the Group to cash flow interest rate risk. The Group's borrowings at the variable rate were mainly denominated in Rupees.

2.1 Interest rate risk exposure

The exposure of the Group's borrowings to interest rate changes at the end of the reporting period are as follows:

		(Rs. in Lakh)
Particulars	31st March, 2019	31st March, 2018
Variable rate borrowing	20,750.92	22,790.49

2.2 Sensitivity of Interest

Profit or loss is sensitive to higher/lower interest expenses from borrowings as a result of changes in interest rates. (Rs. in Lakh)

	Increase/(decrease) in	profit before tax
Particulars	31st March, 2019	31st March, 2018
50 bps increase the profit before tax by	(103.75)	(113.95)
50 bps decrease the profit before tax by	103.75	113.95
		^



Notes to the Consolidated financial statements for the year ended 31st March, 2019

- 65 Fair Value Measurements
- (i) Financial Instruments by Category

Financial Instruments by Category		Company of the local day is a second s		(Rs. in Lakh)
	Carrying	Amount	Fair V	alue
Particulars	31st March, 2019	31st March, 2018	31st March, 2019	31st March, 2018
Financial Assets				
Amortised cost			1.2.2.2.1	
Trade Receivables	4,266.17	2,742.67	4,266.17	2,742.67
Cash and Cash Equivalents	990.56	1,285.84	990.56	1,285.84
Other Bank Balances	401.38	498.54	401.38	498.54
Loan	325.12		325.12	-
Security Deposits	5,041.28	11,151.19	5,041.28	11,151.19
Other Financial Assets	5,625.38	3,385.69	5.625.38	3,385.69
Total	16,649.89	19,063.93	16,649.89	19,063.93
Financial Liabilities				
Amortised cost				
Borrowings	1,82,307.29	1,90,215.83	1,82,307.29	1,90,215.83
Trade Payables	2,579.26	1,797.79	2,579.26	1,797.79
Creditors for Capital Goods	7,228.05	8,422.54	7,228.05	8,422.54
Security Deposits	1,288.53	1,296.53	1,288.53	1,296.53
Other financial liabilities	51,327.26	36,109.78	51,327.26	36,109.78
Total	2,44,730.39	2,37,842.47	2,44,730.39	2,37,842.47

(ii) Fair Valuation techniquos used to determine fair value

The Group maintains procedures to value financial assets or financial liabilities using the best and most relevant data available. The fair values of the financial assets and liabilities are included at the amount that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date.

The following methods and assumptions were used to estimate the fair values:

- (a) The Group assessed that the fair value of cash and cash equivalent, trade receivables, trade payables, and other current financial assets and liabilities approximate their carrying amounts largely due to the short term maturities of these instruments.
- (b) The fair values for security deposits and other financial liabilities were calculated based on cash flows discounted using a current lending rate. They are classified as leve! 3 fair values in the Fair value hierarchy due to the inclusion of unobservable inputs including counterparty credit risk.
- (c) The fair values of non current borrowings are based on discounted cash flows using a current borrowing rate. They are classified as level 3 fair values in the fair value hierarchy due to the use of unobservable inputs, including own credit risk.
- (ii) Fair value hierarchy

This section explains the judgements and estimates made in determining the fair values of the financial instruments that are recognised and measure at fair value. To provide an indication about the reliability of the inputs used in determing fair value, the Group has classified its financial instruments into three levels prescribed under the accounting standard.

- (a) Level 1 Level 1 hierarchy includes financial instruments measured using quoted prices.
- (b) Level 2 The fair value of financial instruments that are not traded in an active market (for example, traded bonds, over the counter derivatives) is determined using valuation techniques which maximize the use of observable market data and rely as little as possible on entity-specific estimates. If all significant inputs required to fair value an instrument are observable, the instrument is included in level 2.
- (c) Level 3 If one or more of the significant inputs are not based on observable market data, the instrument is included in level 3. This is the case for unlisted equity shares, contingent consideration and indemnification assets included in level 3.





Notes to Consolidated financial statements for the year ended 31st March, 2019

66 Capital Management

For the Group's objective when managing capital is to safeguard the Group's ability to continue going concern in order to provide the return to shareholders and benefit to other stakeholders and to maintain an optional capital structure to reduce the cost of capital, capital includes issued equity capital, share premium and all other equity reserves attributable to the equity holders of the Group.

The Group manages its capital structure and makes adjustments in light of changes in economic conditions and the requirements of the financial covenants. To maintain or adjust the capital structure, the Group may adjust the dividend payment to shareholders, return capital to shareholders or issue new shares (if permitted). The Group monitors capital using a gearing ratio, which is debts divided by total equity.

As stated in Notes to accounts, the Group is also having scheme of arrangements to reorganize the capital structure.

Particulars	As at 31st March, 2019	As at 31st March, 2018
Total Debts	2,31,919.33	2,22,733.08
Total Equity	53,465.75	71,501.92
Total debt to equity ratio (Gearing ratio)	4.34	3.12

Notes:-

- (i) Debt is defined as long term and short term borrowings including current maturities of borrowings and interest accrued.
- (ii) Total equity (as shown in balance sheet) includes issued capital and all other equity.

Debt Covenants

Under the terms of Restructuring Agreement (RA), the Group is required to comply with following financial covenants:

Without prior approval of lender, the Group shall not:

- (a) Loans, debenture & charge Issue or subscribe to any debentures, shares, raise any loans, deposit from public, issue equity or preference capital, change its capital structure or create any charge on its assets including its cash flow or give any guarantees.
- (b) Dividend on equity shares declare/pay dividend on equity shares unless otherwise approved by the Lender/Business Monitoring Committee in accordance with the provisions of RA.

In order to achieve this overall objective, the capital management, amongst other thing, aims to ensure that it meets financial covenants attached to the interest bearing Loans and borrowings that define capital structure requirements, there have been breaches in the financial covenants of Interest bearing loans and borrowing in the current period and previous period. The Parent Company has not proposed any dividend in last three year in view of losses incurred.

67 Revenue from contracts with customers (IND AS 115)

(a) The Group disaggregates revenue from contracts with customers by type of products and services, geography and timing of revenue recognition.

Revenue disaggregation by type of services is given note no. 29.

Revenue disaggregation by geography is as follows:

Geography	Year Ended 31st March, 2019	(Rs. in Lakhs) Year Ended 31st March, 2018
India	28,918.71	25,868.69
Outside India	18.67	38.00
Total	28,937.38	25,906.69

Revenue disaggregation by timing of revenue recognition is as follows:

		(Rs. in Lakhs)
Geography	Year Ended 31st March, 2019	Year Ended 31st March, 2018
Services transferred over time	28,937,38	25,906.69
Total	28,937.38	25,906.69

Transaction Price allocated to remaining performance obligations:

The aggregate amount of the transaction price allocated to the performance obligations that are to be satisfied as of 31st March, 2019 amounts to Rs. 7,109.59 Lakh as per Lease deed. The remaining performance obligation are affected by several factors including Panvel storage revenue, cash flow cover, collections within 90 days or mutually agreed. The management of the Parent Company expects that 35% of the unsatisfied performance obligation will be recognised as revenue during the next reporting period amounting to Rs. 2,500.00 Lakh with balance in future two reporting periods thereafter.



Notes to Consolidated financial statements for the year ended 31st March, 2019

(b) Transitional Provision - IND AS 115 - Revenue from Contracts with Customer

The Ministry of Corporate Affairs (MCA) on 28th March, 2018 notified Ind AS 115 "Revenue from contracts with customers" as part of the Companies (Indian Accounting Standards) Amendment Rules, 2018 and the same is effective for accounting period beginning on or after 1st April, 2018. The Group has applied modified retrospective approach in adopting the new standard and accordingly, the revenue from operations for the year ended 31st March, 2019 is not comparable with the previous year. As a results of change in accounting policies, adjustments to the transition provision has been made in respective item as at 1st April, 2018 with corresponding Impact to equity. Details of changes made in item along with equity have given in below table.

Particulars	Rs. in Lakh
Other Financials assets	
Unbilled Revenue	390.41
Net Impact on other equity (Increase)	390.41

68 The Group's borrowings have been assigned by bankers to an ARC / restructured with banks under CDR Package and restructured with NBFC. Certain lenders had invoked the CDR package. The ARC / CDR Lenders and NBFC have charged penal interest/ additional interest/ interest amounting to Rs. 2,599.23 lakh upto the year ended March 31, 2018, which was not accepted by the Group and hence is under negotiation. In light of audit qualifications in previous year as a matter of prudence, the Group has recognised the said interest/penal interest and has accordingly restated the finance cost and other consequential impacts in the year ended March 31, 2018.

Above amount include additional interest amounting to Rs. 655.89 lakhs which is pertaining to period upto 31st March, 2017. Therefore this is adjusted in retained earning as on 1st April, 2017. Due to this reported figure of opening other Equity was Rs. 28,977.19 lakh and now restated figure of opening other Equity is Rs. 28,321.30 lakhs.

Further, during the year ended March 31, 2018 reported figure of finance cost, other Equity and Interest accrued and due on borrowings was Rs. 29,655.06 lakh, Rs. 69,146.40 lakh and Rs. 25,348.69 lakh respectively. Restated figures of finance cost, other equity and Interest accrued and due on Borrowings are Rs.31,598.39 lakh, Rs. 66,937.58 lakh and Rs.27,947.92 lakh respectively. Earning Per Share (EPS) also recalculated based on the restated figures.



san

Arshiya Limited Notes to the Consolidated financial statements for the year ended 31st March, 2019

69 Information required for Consolidated Financial Statement pursuant to Schedule III of the Companies Act, 2013

				FY 2018-19	18-19		
vi	Name of the subsidiary	Total Equity	quity	Share in Other Comprehensive Income	nsive Income	Share in Total Comprehensive Income	nensive Income
No.		As % of Consolidated Total Equity	Rs. in Lakh	As % of Consolidated Other Comprehensive Income	Rs. in Lakh	As % of Consolidated Total Comprehensive Income	Rs. in Lakh
-	Parent Subsidiaries Indian:						
-	Arshiya Limited	320.80%	1,71,515.57	(28.39%)	8.11	13.10%	(3,508.38)
N	Arshiva Rail Infrastructure Limited	(26.11%)	(13,958,69)	85.37%	(24.39)	15.62%	(4,183.51
0	Arshiya Northern FTWZ Limited	36.33%	19,424.06	25.94%	(7.41)	29.73%	(7,962.93)
4	Arshiva Industrial & Distribution Hub Limited	27.12%	14,498.06	5.95%	(1.70)	23.65%	(6,334.79)
5	Arshiva Transport and Handling Limited	(2.01%)	(1,074.59)		100-00 - 00 - 00 - 00 - 00 - 00 - 00 -	0.43%	(114.85
9	Arshiya Technologies (India) Private Limited	(0.01%)	(3.70)		1	%00.0	(0.82)
2	Arshiya Lifestyle Limited	3.74%	2,001.61	1.23%	(0.35)	(0.10%)	26.51
00	Arshiya Logistics Services Limited (formerly known as Laxmipati Balaji Exim Trading Limited)	(0.14%)	(73.91)	9.91%	(2.83)	0.42%	(111.70)
6	Anomalous Infra Private Limited	1.74%	931.59	· · · · · · · · · · · · · · · · · · ·	i	0.64%	(171.24
10	Arshiya Northern Projects Private Limited	(%00.0)	(1.03)			0.02%	(6.03)
11	Arshiya Infrastructure Developers Private Limited	%00.0	0.74		4	%00.0	(0.26)
12	Laxmipati Balaji Supply Chain Management Private Limited	0.01%	4.53		-	%00'0	(0.47)
3	13 Unrivalied Infrastructure Private Limited	0.00%	0.74		*	0.00%	(0.26)
	Stepdown Subsidiary:						
	Held through Arshiya Logistics Services Limited						
-	Arshiya 3PL Services Private Limited	0.01%	4.03	1		%00'0	(26.0)
	Consolidation Adjustments/Eliminiations*	(261.48%)	(1,39,803.26)		-	16.49%	(4,416.72)
	Total	100.00%	53,465.75	100.00%	(28.57)	100.00%	(26,786.42)





Arshiya Limited Notes to the Consolidated financial statements for the year ended 31st March, 2019

				51-/107 1J	81-11		
0		Total Equity	quity	Share in Other Comprehensive Income	nsive Income	Share in Total Comprehensive Income	ensive Income
No.	Name of the subsidiary	As % of Consolidated Total Equity	Rs. in Lakh	As % of Consolidated Other Comprehensive Income	Rs. in Lakh	As % of Consolidated Total Comprehensive Income	Rs. in Lakh
-	Parent Subsidiaries Indian:						
-	Arshiya Limited	230.64%	1,64,914.64	(27.82%)	(2.69)	125.81%	3,892.97
2	Arshiva Rail Infrastructure Limited	(13.76%)	(9,835,18)	74.35%	7.19	(347.03%)	(10,738.56)
6	Arshiya Northern FTWZ Limited	38.30%	27,386.99	(%62.66)	(9.65)	(301.09%)	(9,316.94)
4	Arshiya Industrial & Distribution Hub Limited	29.14%	20,832.84	1.45%	0.14	33.30%	1,030.57
2	Arshiya Transport and Handling Limited	(1.34%)	(959.74)			(3.34%)	(103.33)
9	Arshiya Technologies (India) Private Limited	(%00.0)	(2.88)			(0.05%)	(1.65)
1	Arshiva Lifestyle Limited	2.51%	1,798.21	ı		(0.72%)	(22.38)
80	Arshiya Logistics Services Limited (formerly known as Laxmipati Balaji Exim Trading Limited)	0.05%	37.79	179.11%	17.32	(3.18%)	(98.54)
თ	Mira Supply Chain Management Private Limited (formerly known as Arshiya Supply Chain Management Private Limited) (up to 21st March, 2018)	1		(27.30%)	(2.64)	(51.47%)	(1,592.76)
	Stepdown Subsidiary:						
	Held through Arshiya Rail Infrastructure Limited						
10	Ascendas Parvel FTWZ Limited (formerly known as Arshiya Rail Siding and Infrastructure Limited) (up to 3rd February, 2018)		•	•	1	0.06%	1.79
	Consolidation Adjustments/Eliminiations*	(185.55%)	(1.32,670.75)	-	Ū	647.72%	20,043.27
	Total	100.00%	71,501.92	100.00%	9.67	100.00%	3,094,44

The above figures for the Parent Company and subsidiaries are presented before intercompany eliminations and consolidation adjustments in order to conform to the respective financial statements of the Parent Company and Subsidiaries.



Arshiya Limited Notes to Consolidated financial statements for the year ended 31st March, 2019

- 70 The Group has sent request letters/ emails to various Parties for confirmations of balances under borrowings, trade receivables and capital advances given to vendors and trade payables etc., to which only few parties have responded. Accordingly, impact of adjustment, if any, will be accounted as and when the same is determinable or accounts are reconciled/settled.
- 71 Previous year's figures have been regrouped / restated / rearranged wherever necessary.

Signatures to Notes forming part of Financial Statements

EDI & SHA

ED A

For Chaturvedi & Shah LLP Chartered Accountants Firm Registration Number 101720W/W100355

CHA

Arshiya Limited

For and on behalf of the Board of Directors of

Vijay Napawaliya Partner Membership Number: 109859 Ajay S Mittal Chairman and Managing Director DIN: 00226355

Ashishkumar Bairagra Independent Director DIN: 00049591

Place: Mumbai Date: 27th May, 2019

Savita Dalal Company Secretary

Archana A Mittal Joint Managing Director DIN: 00703208

S. Maheshwari

Chief Financial Officer and Group President

(1w

Dinesh Kumar Sodani VP - Accounts & Finance

Statement containing the salient features of the financial statements of subsidiaries / associates companies / joint ventures [Pursuant to first proviso to sub-section (3) of section 129 of the companies Act, 2013, read with rule 5 of the companies (Accounts) Rules, 2014 - AOC-1]

No Name of the subsidiary Financial Period Curr anded	Financial Period Cum ended	Currency	Equity	Other Equity	Total assets [9]	Total liabilities excluding equity and other equity)	Investments	Iunover	Profit / (Loss)before taxation	Provision for taxation	Profit (Loss) after faxation	% of Share holding(1")
1 Arshiya Industrial & Distribution Hub 31st March, 2019 1 imaged	31st March, 2019	INR	1.723.72	13,293.43	67,614.82	52,597.67	State of the	1,192.12	(6,333.09)		(60'828'9)	100%
2 Arshiva Northern FTWZ Limited 31st March, 2019	31st March. 2019	A HAR	1.086.87	18,337,19	75,763,88	56,339,82	のないないの	1 460.80	(7,955.52)		(1.955.52)	100%
3 Arshiya Rail Intrastructure Limited	31st March, 2019		4,238.44	(18,197.13)	51,313.60	65,272.29		13,614.83	(4,159.12)		(4,159.12)	100%
 Arahiya Transport And Handling 21st March, 2018 Immedia 	31st March, 2019	INI	5.00	(1,079.59)	20:0	1,074.61			(114.85)		(98:411)	100%
5 Arshiya Technologies (India) Private 31st March, 2019 Limited	31st March, 2019	IN	10.12	(13.82)	0.20	3.90			(0.82)		(0.82)	100%
6 Arshiva Lifestyle Limited - 31st March, 2019	21st March. 2019	A NUMBER	148.50	1,853.11	4,707.56	2,705.95	のないのであった	10,465,30	33.84	6.98	26.86	100%
 Arshiya Logistics Services Limited (formenty known as Laxmipati Balaji Exim Trading Limited) 	31st March, 2019	INR	160.00	(133.91)	1,958.26	2,032.17	5.00	6,795.46	(108.88)		(108.88)	100%
8 Anomalous Infra Private Limited 31st March. 2019	31st March. 2019	RNE	11.00	920,59	7.003.80	6,072,21	でんでいたたい	いいたちょう	(171.24)		(H2178) E	100%
9 Arshiya Northern Projects Private Limited	31st March, 2019		5.00	(6.03)	5.00	6.03			(6.03)	Contraction of the second	(6.03)	100%
C Arshiya Intrastructure Developers 31st March 2019 2-toole 1 miled	31st March, 2019	INR	1.00	(0.26)	1.00	0.26			(0.26)			100%
1 Laxmipati Balaji Supply Chain Management Private Limited	31st March, 2019	INR	5.00	(0.47)	22.32	17.79			(0.47)		(0.47)	100%
12 Unrivated Infrastructure Private 31st March, 2019 [1]	.31st March, 2019	INR	1:00	(0.26)	1.00	0.26			(0.26)		(0.26)	100%
13 Arshiva 3PI Services Private Limited 31st March 2019	31st March 2019	INR	5.00	(10.97)	4.54	0.51	「日本」として		(20:0)	Aler	(16:0)	Nil (2*)

(1*) % of Share Holding including beneficial interest/ voting power (either directly/indirectly or through subsidiaries).
(2*) 100% held through Arshiya Logistics Services Limited (formerly known as Laxmipat Balaji Exim Trading Limited).

Savita Dalal Company Secretary Chief Financial Officer and Group President S. Maheshwari Ashishkumar Bairagra Independent Director DIN: 00049591 1 Archana A Mittal Joint Managing Director DIN: 00703208 inhanonte For and on behalf of the Board of Directors of Arshiya Limited Ajay S Mittal Chairman and Managing Director DIN: 00226355 R Place: Mumbai Date: 27th May, 2019 5 6

ED FIN APT HS

l "sodaw"

Dinesh Kumar Sodani VP - Accounts & Finance

ARSHIYA RAIL INFRASTRUCTURE LIMITED BALANCE SHEET AS AT MARCH 31, 2019

		and the second	(Rupees in lakhs
		00000 H, 2005	The second s
ASSETS			a and a share and a share a share a share
Non-Current Assets	1	1	
(a) Property, Plant and Equipment	5	44 222 27	
(c) Intangible Assets	6	44,322.37	47,856.82
(d) Financial Assets	ľ	2,825.28	3,223.16
(i) Other Financial Assets	7	663.64	
(e) Other Non-Current Assets	8	1,335.26	818.60
		49,146.55	1,227.72
Current assets			
(a) Inventories			
(b) Financial Assets	9	-	15.66
(i) Trade Receivables	10		
(ii) Cash and Cash Equivalents	10	1,323.33	666.59
(III) Bank Balances Other than (II) above	11	120.24	375.35
(iv) Other Financial Assets	12	267.30	386.16
(c) Other Current Assets	13	322.37	348.59
(c) outer content hasets	14	2,167.05	
		2,107.03	2,100.81
TOTAL ASSETS	1	51,313.60	55,227.11
EQUITY AND LIABILITIES			
Equity	1 1		
(a) Equity Share capital	1		
(b) Other Equity	15 16	4,238.44	4,238.44
fall annen auferiet		(18,197.13)	(14,073.62
Liabilities	1	(13,958-69)	(9,835.18)
Non Current Llabilities			¥ .*.
(a) Financial Liabilities			* ,
(i) Borrowings	17	31,902.67	
(b) Provisions	18	120.52	33,538.04
(c) Other Non-Current Liabilities	19	1,632.32	16.42
		33,655.51	1,723,69
Current Liabilities			55,478.15
(a) Financial Llabilities	1		
(i) Borrowings	20	304.38	2,029.26
(ii) Trade Payables	21		2,025.20
(A) Total outstanding dues of Micro and Small Enterprises		15.77	5.24
(B)Total outstanding dues of creditors Other than Micro and			
Small Enterprises		1,307.44	620.88
(iii) Other Financial Liabilities	22	28,103.68	25,206,84
(b) Other Current Liabilities	23	1,874.90	1,918.23
(c) Provisions	24	10.61	3.69
		31,616.78	29,784.14
TOTAL EQUITY & LIABILITIES	ŀ	51,313.60	55,227.11

Notes to the financial statements As per our Report of even date

For Chaturvedi & Shah LLP Chartered Accountants Firm Registration Number 101720W/W100355



Place : Mumbai Date: May 27, 2019 For and on behalf of Board of Directors of Arshiya Rail Infrastructure Limited

<u>`</u>0 - N

1-57

Ajay S Mittal Director DIN : 00226355

Amardeep Gupta

Chief Financial Officer

NFK L.U./BAI ы

Navnit Choudhary Director DIN : 00613576

Avani

Avani Dipakkumar Lakhani Company Secretary

ARSHIYA RAIL INFRASTRUCTURE LIMITED STATEMENT OF PROFIT AND LOSS FOR THE YEAR ENDED MARCH 31, 2019

			(Rupees in lakhs
		, 1994 - Mareland 1994: Balangi - San Salangi - S	an chuitean An Antair an Antair à
ing the second sec		منتشيب بالماني والمعا	مەركىيەت مىلەر يۈكەر كىلىكەر مەركىيە مەركىيەت مىلەر يېزىكەر كىلىكەر مەركىيەت
INCOME			
Revenue from operations	25	13,614.83	13,598.33
Other income	26	641.87	1,459.3
Total Income (I)		14,256.70	15,057.7
EXPENSES			
Cost of operations	27	11,400.27	11,507.3
Employee benefits expenses	28	1,121.91	826.7
Finance costs	29	7,015.57	8,018.9
Depreciation and amortization expenses	30	4,099.64	4,130.6
Other expenses	31	1,253.59	895.4
Total Expenses (II)		24,890.98	25,379.20
tota Expanses (i)		24,030.30	23,373.20
Loss before exceptional items and tax (1-II)	1	(10,634.28)	(10,321.49
Exceptional Items (Net)	32	(6,475.16)	424.2
Loss before tax		(4,159.12)	(10,745.7
Tax expense:	45		
Current tax			
Deferred tax		-	-
Loss for the year		(4,159.12)	(10,745.75
OTHER COMPREHENSIVE INCOME (OCI)			
Item not to be reclassified to profit and loss :			
Remeasurement of gains/ (losses) on defined benefit plans	37	(24.39)	7.19
TOTAL COMPREHENSIVE INCOME FOR THE YEAR		(4,183.51)	(10,738.5)
Earnings per Equity shares (Face Value Rupees 10 each)	33		
Basic/ Diluted earnings per share (in Rupees)	1 1	(9.81)	(25.3)

Notes to the financial statements As per our Report of even date

For Chaturvedi & Shah LLP Chartered Accountants Firm Registration Number 101720W/W100355

Happen alin

Vijay Napawaliya Partner Membership Number. 109859

Place : Mumbai Date: May 27, 2019



· •

For and on behalf of Board of Directors of Arshiya Rail Infrastructure Limited

1-57

Ajay S Mittal Director DIN : 00226355

Supta Amarde Chief Financial Officer

N Charait- 40

Navnit Choudhary Director DIN : 00613576

Avant

Avani Dipakkumar Lakhani Company Secretary



ARSHIYA RAIL INFRASTRUCTURE LIMITED STATEMENT OF CHANGES IN EQUITY FOR THE YEAR ENDED MARCH 31, 2019

A Equity Share Capital (Refer Note 15)

Equicy Share Capital (Relet Note 15)	(Rupees in lakhs)
Participation (Participation of the second sec	Matinal
Equity Shares of Rupees 10 each issued, subscribed and paid up	
As at April 1, 2017	
Equity Shares	4,238.44
Issue of equity share during the year	-
As at March 31, 2018	4,238.44
Equity Shares	4,238.44
Issue of equity share during the year	-
As at March 31, 2019	4,238.44

B Other Equity (Refer Note 16)

				(F	Rupees in lakhs
	ાં સેપ્સુઝ રાજ્યનામાં ગુજરામાં આવ્યું ગુજરામાં આવ્યું આવ્યું સેપ્સામના	2-220000 	ીમોન્ટ ઉપનાસ આપ્યાસ અને આપ્યાસ અનેઓિસ અન્યાં અનેઓાસ્ટ્સ	anger annangen annangen annangen annangen annangen annangen	ас с. (б. с.).
As at April 1, 2017	38,123.31	(43,189.57)	1,564.45	171.37	(3,330.44
Loss for the year	-	(10,745.75)	-	-	(10,745.75
Other comprehensive income	-	7.19		-	7.19
Total comprehensive income for the year		(10,738.56)	-		(10,738.56
Fair valuation of financial guarantees given	-	(4.62)	-	-	(4.62
As at March 31, 2018	38,123.31	(53,932.75)	1,564.45	171.37	(14,073.62
Loss for the year		(4,159.12)	-	-	(4,159.12
Other comprehensive income		(24.39)	-		(24.39
Total comprehensive income for the year	-	(4,183.51)	•	-	(4,183.51
Fair valuation of financial guarantees given	-	-	60.00	-	60.00
As at March 31, 2019	38,123.31	(58,116.26)	1,624.45	171.37	(18,197.13

Notes to the financial statements As per our Report of even date 1-57

For Chaturvedi & Shah LLP

Chartered Accountants Firm Registration Number 101720W/W100355

IRVED

MUM

ACCOUNT

SH



Vijay Napawaliya Partner Membership Number. 109859

Place : Mumbai Date: May 27, 2019 For and on behalf of Board of Directors of

Arshiya Rail Infrastructure Limited

Alay S Mittal

Ajay S Mittal Director DIN : 00226355

Amardeep Gupta Chief Financial Officer

Narrin nous

Navnit Choudhary Director DIN : 00613576

Avon

Avani Dipakkumar Lakhani Company Secretary



ARSHIYA RAIL INFRASTRUCTURE LIMITED Cash Flow Statement for the year ended March 31, 2019

					(Rupees in lakhs)
enas Atrija					
	····	<u> </u>	<u></u>	en oblitacionis dolaren e	and the second
CASH FLOW FROM OPERATING ACTIVITIES					
Loss for the year before tax				(4,159.12)	{10,745.75
-					
Adjustments for :				4,099.54	4 4 7 9 4 4
Depredation and amortization expenses				(21.10)	4,130.61 (29.38
Interest Income Loss on sale/discarded Property, plant and equipme	ant			(21.10)	182.79
Gain on sale of Investment	5414				(5.00
Sundry Balances Written Back (net)				(245.67)	
Finance Expense				7,015.57	8,018.94
Bad Debts Written off				-	101.30
Settlement of Claims				(6,475.16)	302.54
Unwinding of Interest on loan to related party					(476.14
Fair value of financial Instruments				(5.76)	{1.28
Government grant – income				(365.49)	(365.49
Financial Guarantee Income				(3.85)	(0.77
Advance rent				6.13	1.49
OPERATING PROFIT / (LOSS) BEFORE WORKING CA	PITAL CHANGES			(154.81)	499.35
Adjustments for				1	l
Trade & other payables				640.92	(817.42
Inventories				15.66	0.07
Trade & other receivables				(483.73)	1,532.56
CASH GENERATED FROM OPERATIONS				18.04	1,214.56
Direct Tax (Paid)/ Refunds				(113.22)	24.58
NET CASH FLOW FROM OPERATING ACTIVITIES			Total (A)	(95.18)	1,239.14
CASH FLOW FROM INVESTING ACTIVITIES					1
Purchase of Property, Plant and Equipment				(167.78)	(295.31
Purchase of Intangible assets				-	(500.00
Purchase of Investment				(77.69)	-
Proceeds from Sale of Investment				196.55	31.85
Proceeds from Sale of Property, Plant and Equipme	int			0.47	-
Interest Income				21.10	29.38
NET CASH FLOW FROM INVESTING ACTIVITIES			Total (B)	(27.35)	(734.08
CASH FLOW FROM FINANCING ACTIVITIES					
Proceeds from borrowings - Non current				2,960.00	3,220.74
Repayment of borrowings - Non current				(1,060.00)	
Borrowing - current (Net)				(1,788.58)	1,402.08
Interest paid on borrowings				(244.00)	(1,172.85
NET CASH FLOW FROM FINANCING ACTIVITIES			Total (C)	(132.58)	(306.63
Net Increase/(Decrease) in cash and cash equivaler	nts		(A+8+C)	(255.11)	198.43
Cash and cash equivalents at the beginning of the			·····	375.35	176.92
Cash and Cash Equivalents at the beginning of the				120.24	375.35
And and room standing as one addressing as the					
Cash and Cash Equivalents at the end of the year				120.24	375.35
*Note:- Changes in liabilities arising from financing	activities :				
Particulars	March 31, 2018	Cash flow	INDAS Impact	Other non cash	March 31, 2019

Particulars	March 31, 2018	Cash flow	INDAS impact	Other non cash	March 31, 2019
				adjustment	
Long term borrowing (Refer Note no. 17 & 22)	42,469.34	1,900.00	144.94	996.38	43,517.90
Short term borrowing (Refer Note no. 20)	2,029.26	(1,788.58)	63.70		304.38
Shore term worrenning the state					

Notes to the financial statements As per our Report of even date

For Chaturvedi & Shah LLP Chartered Accountants

Firm Registration Number 101720W/W100355

URVED/

MUMB

ACCOUNT

SIL

×

CH

Igness alig C Vijay Napawaliya Partner

Membership Number. 109859

Place : Mumbai Date: May 27, 2019 1-57

Ajay S Mittal

DIN: 00226355

p Gupta

Chief Financial Officer

Director

Ama

For and on behalf of Board of Directors of Arshiya Rail Infrastructure Limited

NFRIS UMBAI Naunitual

Navnit Choudhary Director DIN:00613576

Avant

Avani Dipakkumar Lakhani Company Secretary

1 CORPORATE INFORMATION:

Arshiya Rail Infrastructure Limited (CIN : U93000MH2008PLC180907) is a public company domiciled in India and Is incorporated on April 7, 2008 under the provisions of the Companies Act applicable in India. The registered office of the company is located at 302, Level 3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400 018.

Arshiya Rail Infrastructure Limited (ARIL) is a subsidiary of Arshiya Limited (AL). AL is listed on Bombay Stock Exchange and National Stock Exchange. In April 2008, AL acquired a Category-I license to operate a pan-India rail logistic service, giving rise to ARIL. ARIL is a specialized entity of AL, offering unprecedented rail infrastructure, including private modern rakes, customized containers, Private Freight Terminal (PFT), pan-India network and superior connectivity. The company is engaged in Private Container Train Operator (PCTO) business and is holding Category-I license which allows the company to operate on Indian Rail network on pan India basis both Domestic and Exim Traffic. ARIL's unique offering provide unparalleled efficiencies with capability of large scale evacuation of cargo from Ports, Domestic Distriparks, Free Trade and Warehousing Zones , Inland Container Depot (ICD) and customer Sidings.

The financial statements of the Company for the year ended 31st March, 2019 were approved and adopted by board of directors in their meeting held on 27th May 2019.

2 BASIS OF PREPARATION:

The financial statements of the Company have been prepared in accordance with Indian Accounting Standards (Ind AS) notified under the Companies (Indian Accounting Standards) Rules, 2015 (as amended) and the relevant provisions of the Companies Act, 2013 ("the Act").

The Company prepared its financial statements in accordance the Indian Accounting Standards (IND AS) are notified under the section 133 of the Companies Act 2013, read together with paragraph 7 of the Companies (Accounts) Rules, 2014 (Indian GAAP).

The financial statements have been prepared on a historical cost convention, except for certain financial assets and liabilities, which are measured at fair value/ amortised cost.

The financial statements are presented in Indian Rupees (Rs.), which is the Company's functional and presentation currency and all values are rounded to the nearest lakhs, except when otherwise indicated.

3 SIGNIFICANT ACCOUNTING POLICIES AND CRITICAL ACCOUNTING ESTIMATE AND JUDGEMENT:

3.1 Property, Plant and Equipment:

Property, plant and equipment are carried at cost less accumulated depreciation and accumulated impairment losses, if any. Cost includes purchase price, borrowing cost and any cost directly attributable to the bringing the assets to its working condition for its intended use.

Depreclation on the property, plant and equipment is provided using straight line method over the useful life of assets as specified in schedule II to the Companies Act, 2013. Depreciation on property, plant and equipment which are added / disposed off during the year, is provided on pro-rata basis with reference to the date of addition / deletion. Freehold land is not depreciated and under the previous GAAP land was revalued.

The asset's residual values, useful lives and method of depreciation are reviewed at each financial year end and are adjusted prospectively, if appropriate.

Capital work-in-progress includes cost of property, plant and equipment under installation / under development as at the balance sheet date.

Leasehold Improvements are depreciated over the period of lease.

Property, plant and equipment are eliminated from financial statement, either on disposal or when retired from active use. Profits / losses arising in the case of retirement / disposal of property, plant and equipment are recognised in the statement of profit and loss in the year of occurrence.

The Company has opted to continue with the carrying values of all of its property, plant and equipment as recognised in the indian GAAP financial statements as deemed cost at the transition date i.e. April 1, 2016.

3.2 Intangible Assets :

Intangible assets are carried at cost less accumulated amortisation and accumulated impairment losses, if any. Cost includes expenditure that is directly attributable to the acquisition of the intangible assets.

Identifiable Intangible assets are recognised when it is probable that future economic benefits attributed to the asset will flow to the Company and the cost of the asset can be reliably measured.

Railways License fees is amortised over a period of twenty years being the license period as per agreement.

Computer softwares are capitalised at the amounts pald to acquire the respective license for use and are amortised over the period of five years. The assets' useful lives are reviewed at each financial year end.

Gains or losses arising from derecognition of an intangible asset are measured as the difference between the net disposal proceeds and the carrying amount of the asset and are recognised in the statement of profit and loss when the asset is derecognised.

The Company has opted to continue with the carrying values of all of its intangible assets as recognised in the Indian GAAP financial statements as deemed cost at the transition date i.e. April 1, 2016.

3.3 Leases:

A lease is classified at the inception date as a finance lease or an operating lease. A lease that transfers substantially all the risks and rewards incidental to ownership to the Company is classified as a finance lease.





The Company as a lessee

(a) Finance lease

Assets acquired under finance lease are capitalized and the corresponding lease liability is recognised at lower of the fair value of the leased assets and the present value of minimum lease payments at the inception of the lease. Initial costs directly attributable to lease are recognised with the asset under lease.

(b) Operating lease

Lease of assets under which all risks and rewards of ownership are effectively retained by the lessor are classified as operating lease. Lease payments under operating lease are recognised as expenses on accrual basis in accordance with the respective lease agreements.

The Company as a lessor

(a) Finance lease

When assets are leased out under a finance lease, the present value of the minimum lease payments is recognised as a receivable. The difference between the gross receivable and the present value of the receivable is recognised as unearned finance income. Lease income is recognised over the term of the lease using the net investment method before tax, which reflects a constant periodic rate of return.

The lessor derecognises the leased assets and recognises the difference between the carrying amount of the leased assets and the finance lease receivable in the statement of Profit and Loss when recognising the finance lease receivable.

(b) Operating lease

Rental income from operating leases is recognised in the statement of profit and loss on a straight line basis over the term of the relevant lease unless another systematic basis is more representative of the time pattern in which economic benefits from the leased assets is diminished.

Initial indirect costs incurred in negotiating and arranging as operating lease are added to carrying value of the leased asset and recognised on a straight line basis over the lease term.

3.4 Inventories:

Inventories are valued at the lower of cost and net realizable value. The cost of inventories comprises of cost of purchase, cost of conversion and other costs incurred in bringing the inventories to their respective present location and condition. Cost is computed on the First in first out basis.

3.5 Cash and cash equivalents:

Cash and cash equivalent in the balance sheet comprise cash at banks, cash on hand and short-term deposits with an original maturity of three months or less, which are subject to an insignificant risk of changes in value.

For the purpose of the statement of cash flows, cash and cash equivalents consist of cash and short-term deposits, as defined above, net of outstanding bank overdrafts as they are considered an integral part of the Company's cash management.

3.6 Impairment of assets:

An asset is considered as impaired when at the date of Balance Sheet, there are indications of impairment and the carrying amount of the asset, or where applicable, the cash generating unit to which the asset belongs, exceeds its recoverable amount (i.e. the higher of the net asset selling price and value in use). The carrying amount is reduced to the recoverable amount and the reduction is recognized as an impairment loss in the statement of profit and loss. The impairment loss recognized in the prior accounting period is reversed if there has been a change in the estimate of recoverable amount. Post impairment, depreciation is provided on the revised carrying value of the impaired asset over its remaining useful life.

3.7 Financial Instruments - Initial recognition, subsequent measurement and Impairment:

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity.

I) Financial assets -initial recognition and measurement:

All financial assets are initially recognized at fair value. Transaction costs that are directly attributable to the acquisition or issue of financial assets, which are not at fair value through profit or loss, are adjusted to the fair value on initial recognition. Financial assets are classified, at initial recognition, as financial assets measured at fair value or as financial assets measured at amortised cost.

Financial assets - Subsequent measurement:

For the purpose of subsequent measurement, financial assets are classified in two broad categories:-

- a) Financial assets at fair value
- b) Financial assets at amortlsed cost

BLE processes are measured at fair value, gains and losses are either recognised entirely in the statement of profit and loss (i.e fair value through profit BLE processes are comprehensive income).





A financial asset that meets the following two conditions is measured at amortised cost (net of any write down for impairment) unless the asset is designated at fair value through profit or loss under the fair value option.

a) Business model test: The objective of the Company's business model is to hold the financial asset to collect the contractual cash flow.
 b) Cash flow characteristics test: The contractual terms of the financial asset give rise on specified dates to cash flow that are solely payments of principal and interest on the principal amount outstanding.

A financial asset that meets the following two conditions is measured at fair value through other comprehensive income unless the asset is designated at fair value through profit or loss under the fair value option.

a) Business model test: The financial asset is held within a business model whose objective is achieved by both collecting contractual cash flow and selling financial assets.

b) Cash flow characteristics test: The contractual terms of the financial asset give rise on specified dates to cash flow that are solely payments of principal and interest on the principal amount outstanding.

Ali other financial asset is measured at fair value through profit or loss.

Financial assets - Equity Investment in subsidiarles

The Company has accounted for its equity investment in subsidiaries at cost.

Financial assets - Derecognition

A financial assets (or, where applicable, a part of a financial asset or part of a group of similar financial assets) is primarily derecognised (i.e. removed from the Company's statement of financial position) when:

a) The rights to receive cash flows from the asset have expired, or
 b) The Company has transferred its rights to receive cash flow from the asset.

II) Financial liabilities - Initial recognition and measurement:

The financial llabilities are recognised initially at fair value and, in the case of loans and borrowings, net of directly attributable transaction costs.

Financial liabilities - Subsequent measurement:

Financial liabilities are subsequently carried at amortized cost using the effective interest method. For trade and other payables maturing within one year from the balance sheet date, the carrying amounts are approximate at their fair value due to the short maturity of these instruments.

Financial Liabilities - Financial Guarantee contracts:

Financial guarantee contracts issued by the Company are those contracts that require a payment to be made to reimburse the holder for a loss it incurs because the specified debtor falls to make a payment when due in accordance with the terms of a debt instrument. Financial guarantee contracts are recognised initially as a liability at fair value, adjusted for transaction costs that are directly attributable to the issuance of the guarantee. Subsequently, the liability is measured at the higher of the amount of loss allowance determined and the amount recognised less cumulative amortisation.

Financial Liabilities - Derecognition:

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another, from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit and loss.

Compound Instruments

An issued financial instrument that comprises of both the liability and equity components are accounted as compound financial instruments. The fair value of the liability component is separated from the compound Instrument and the residual value is recognised as equity component of other financial instrument. The liability component is subsequently measured at amortised cost, whereas the equity component is not remeasured after initial recognition. The transaction costs related to compound instruments are allocated to the liability and equity components in the proportion to the allocation of gross proceeds. Transaction costs related to equity component is recognised directly in equity and the cost related to liability component and amortised using effective interest method.





3.8 Provisions, Contingent Liabilities, Contingent Assets and Commitments:

Provisions are recognised when the Company has a present obligation (legal or constructive) as a result of a past event. It is probable that an outflow of resources embodying economic benefits will be required to settle the obligation and a reliable estimate can be made of the amount of the obligation. If the effect of the time value of money is material, provisions are discounted using equivalent period government securities interest rate. Unwinding of the discount is recognised in the statement of profit and loss as a finance cost. Provisions are reviewed at each balance sheet date and are adjusted to reflect the current best estimate.

Contingent liabilities are disclosed when there is a possible obligation arising from past events, the existence of which will be confirmed only by the occurrence or non-occurrence of one or more uncertain future events not wholly within the control of the Company or a present obligation that arises from past events where it is either not probable that an outflow of resources will be required to settle or a reliable estimate of the amount cannot be made. Information on contingent liability is disclosed in the Notes to the Financial Statements. However, when the realisation of income is virtually certain, then the related asset is no longer a contingent asset, but it is recognised as an asset.

3.9 Revenue recognition

Revenue is recognized upon transfer of control of goods (equipment) or rendering of services to customers in an amount that reflects the consideration which the Company expects to receive in exchange for those goods or services.

Income from services is recognized upon completion of services as per the terms of contracts with the customers. Revenue is measured based on the transaction price, which is the consideration, price concessions and incentives, if any, as specified in the contract with the customer. Revenue also excludes taxes collected from customers.

Amounts billed for services in accordance with contractual terms but where revenue is not recognized, have been classified as unearned revenue and disclosed under current liabilities

Contract assets are recognized when there is excess of revenue earned over billings on contracts. Contract assets are classified as unbilled receivables (only act of involcing is pending) when there is unconditional right to receive cash, and only passage of time is required, as per contractual terms. Unearned revenue ("contract liability") is recognized when there is billing in excess of revenues.

Revenue is measured at the amount of consideration which the company expects to be entitled to in exchange for transferring distinct goods or services to a customer as specified in the contract, excluding amounts collected on behalf of third parties (for example taxes and duties collected on behalf of the government). Consideration is generally due upon satisfaction of performance obligations and a receivable is recognized when it becomes unconditional.

Revenue fron sale of services e.g rail freight income recognised as per the terms of contracts with customers based on stage of completion when the outcome of the transactions involving rendering of services can be estimated reliably. For Fixed-price contract, revenue is recognised based on the actual service provided to the end of the reporting period as a proportion of the total services to provided (Percentage of completion method)

Measement of revenue : Estimates of revenues, cost or extent of progress toward completion are revised if circumstances change. Any resulting increases or decreases in estimated revenues or costs are reflected in profit or loss in the period in which the circumstances that give rise to the revision become known by management.

Revenue from handling and other ancillary services is recognised at the time of rendering of service which is at the time of loading/unloading of container/cargo.

Contract balances

Trade receivables

A receivable represents the Company's right to an amount of consideration that is unconditional.

Contract liabilities

A contract liability is the obligation to transfer goods or services to a customer for which the Company has received consideration (or an amount of consideration is due) from the customer. If a customer pays consideration before the Company transfers goods or services to the customer, a contract liability is recognised when the payment is made. Contract liabilities are recognised as revenue when the Company performs under the contract.

Interest Income:

Interest income from a financial asset is recognised when it is probable that the economic benefits will flow to the Company and the amount of income can be measured reliably. Interest income is accrued on a time basis, by reference to the principal outstanding and at the effective interest rate applicable, which is the rate that exactly discounts estimated future cash receipts through the expected life of the financial asset to that asset's net carrying amount on initial recognition.

Dividend Income:

Dividend income is recognised when the right to receive the payment is established.





3.10 Foreign currency reinstatement and translation:

Transactions in foreign currencies are initially recorded by the Company at rates prevailing at the date of the transaction. Subsequently monetary items are translated at closing exchange rates as on balance sheet date and the resulting exchange difference recognised in statement of profit and loss. Differences arising on settlement of monetary items are also recognised in statement of profit and loss.

Non-monetary items that are measured in terms of historical cost in a foreign currency are translated using the exchange rates at the dates of the transaction. Non-monetary items carried at fair value that are denominated in foreign currencies are translated at the exchange rates prevailing at the date when the fair value was determined. The gain or loss arising on translation of non-monetary items measured at fair value is treated in line with the recognition of the gain or loss on the change in fair value of the item (i.e., translation differences on items whose fair value gain or loss is recognised in OCI or profit or loss are also recognised in OCI or profit or loss, respectively).

Foreign exchange differences regarded as an adjustment to borrowing costs are presented in the statement of profit and loss, within finance costs. All other finance gains / losses are presented in the statement of profit and loss on a net basis.

3.11 Employee Benefits:

Short term employee benefits are recognized as an expense in the statement of profit and loss of the year in which the related services are rendered.

(a) Defined Contribution Plan

Contribution to Provident Fund, a defined contribution plan, is made in accordance with the statute, and is recognised as an expense in the year in which employees have rendered services.

(b) Defined Benefit Plan

Leave encashment being a defined benefit plan is accounted for using the projected unit credit method, on the basis of actuarial valuations carried out by third party actuaries at each Balance Sheet date. Actuarial gains and losses arising from experience adjustments and changes in actuarial assumptions are charged or credited to other comprehensive income in the year in which they arise. Other costs are accounted in statement of profit and loss.

The cost of providing gratuity, a defined benefit plans, is determined using the Projected Unit Credit Method, on the basis of actuarial valuations carried out by third party actuaries at each Balance Sheet date. Actuarial gains and losses arising from experience adjustments and changes in actuarial assumptions are charged or credited to other comprehensive income in the period in which they arise. Other costs are accounted in statement of profit and loss.

Remeasurements of defined benefit plan in respect of post employment and other long term benefits are charged to the other comprehensive income in the year in which they occur. Remeasurements are not reclassified to statement of profit and loss in subsequent periods.

3.12 Taxes on income:

Income tax expense represents the sum of current tax (including MAT and income tax for earlier years) and deferred tax. Tax is recognised in the statement of profit and loss, except to the extent that it relates to items recognised directly in equity or other comprehensive income, in such cases the tax is also recognised directly in equity or in other comprehensive income. Any subsequent change in direct tax on items initially recognised in equity or other comprehensive income is also recognised in equity or other comprehensive income.

Current tax provision is computed for income calculated after considering allowances and exemptions under the provisions of the applicable income Tax Laws. Current tax assets and current tax liabilities are off set, and presented as net.

Deferred tax is recognised on differences between the carrying amounts of assets and liabilities in the Balance sheet and the corresponding tax bases used in the computation of taxable profit. Deferred tax liabilities are generally recognised for all taxable temporary differences, and deferred tax assets are generally recognised for all deductible temporary differences, carry forward tax losses and allowances to the extent that it is probable that future taxable profits will be available against which those deductible temporary differences, carry forward tax losses and allowances can be utilised. Deferred tax assets and liabilities are measured at the applicable tax rates. The carrying amount of deferred tax assets is reviewed at each balance sheet date and reduced to the extent that it is no longer probable that sufficient taxable profits will be available against which the temporary differences can be utilised.

Credit of MAT is recognised as an asset only when and to the extent there is convincing evidence that the Company will pay normal income tax during the specified period, i.e., the period for which MAT credit is allowed to be carried forward. In the year in which the MAT credit becomes eligible to be recognised as an asset, the said asset is created by way of a credit to the statement of profit and loss and shown as MAT credit entitlement. The Company reviews the same at each balance sheet date and writes down the carrying amount of MAT credit entitlement to the extent there is no longer convincing evidence to the effect that the Company will pay normal income tax during the specified period.





3.13 Borrowing Costs:

Borrowing costs specifically relating to the acquisition or construction of qualifying assets that necessarily takes a substantial period of time to get ready for its intended use are capitalized (net of income on temporarily deployment of funds) as part of the cost of such assets. Borrowing costs consist of interest and other costs that the Company incurs in connection with the borrowing of funds. For general borrowing used for the purpose of obtaining a qualifying asset, the amount of borrowing costs eligible for capitalization is determined by applying a capitalization rate to the expenditures on that asset. The capitalization rate is the weighted average of the borrowing costs applicable to the borrowings of the Company that are outstanding during the period, other than borrowings made specifically for the purpose of obtaining a qualifying asset. The amount of borrowing costs capitalized during a period does not exceed the amount of borrowing cost incurred during that period. All other borrowing costs are expensed in the period in which they occur.

3.14 Earnings per share:

Basic earnings per share is computed using the net profit for the year attributable to the shareholders' and weighted average number of equity shares outstanding during the year.

Diluted earnings per share is computed using the net profit for the year attributable to the shareholders' and weighted average number of equity and potential equity shares outstanding during the year including share options, convertible preference shares and debentures, except where the result would be anti-dilutive. Potential equity shares that are converted during the year are included in the calculation of diluted earnings per share, from the beginning of the year or date of issuance of such potential equity shares, to the date of conversion.

3.15 Current and non-current classification:

The Company presents assets and liabilities in statement of financial position based on current/non-current classification.

The Company has presented non-current assets and current assets before equity, non-current liabilities and current liabilities in accordance with Schedule III, Division II of Companies Act, 2013.

An asset is classified as current when it is:

a) Expected to be realised or intended to be sold or consumed in normal operating cycle,

b) Held primarily for the purpose of trading,

c) Expected to be realised within twelve months after the reporting period, or

d) Cash or cash equivalent unless restricted from being exchanged or used to settle a liability for at least twelve months after the reporting period. All other assets are classified as non-current.

A llability is classified as current when it is:

a) Expected to be settled in normal operating cycle,

b) Held primarily for the purpose of trading,

c) Due to be settled within twelve months after the reporting period, or

d) There is no unconditional right to defer the settlement of the liability for at least twelve months after the reporting period.

All other liabilities are classified as non-current.

The operating cycle is the time between the acquisition of assets for processing and their realisation in cash or cash equivalents. Deferred tax assets and liabilities are classified as non-current assets and liabilities. The Company has identified twelve months as its normal operating cycle.

3.16 Fair value measurement:

The Company measures financial instruments at fair value at each balance sheet date.

Fair value is the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The fair value measurement is based on the presumption that the transaction to sell the asset or transfer the liability takes place either:

a) In the principal market for the asset or liability, or

b) in the absence of a principal market, in the most advantageous market for the asset or liability.

A fair value measurement of a non-financial asset takes into account a market participant's ability to generate economic benefits by using the asset in Its highest and best use or by selling it to another market participant that would use the asset in its highest and best use.

The Company uses valuation techniques that are appropriate in the circumstances and for which sufficient data are available to measure fair value, maximizing the use of relevant observable inputs and minimizing the use of unobservable inputs.

All assets and liabilities for which fair value is measured or disclosed in the financial statements are categorized within the fair value hierarchy.

3.17 Off-setting financial Instrument:

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable rights to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable rights must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankrueter states of company or counterparty.





3.18 Government Grants

Grants from the government are recognised at their fair value where there is a reasonable assurance that the grant will be received and the company will comply with all attached conditions. Government grants relating to income are deferred and recognised in the profit or loss over the period necessary to match them with the costs that they are intended to compensate and presented within other income. Government grants relating to the purchase of property, plant and equipment are included in non-current liabilities as deferred income and are

credited to profit or loss on a straight-line basis over the expected lives of the related assets and presented within other income.

3.19 Segment Reporting - Identification of Segments

An operating segment is a component of the Company that engages in business activities from which it may earn revenues and incur expenses, whose operating results are regularly reviewed by the company's chief operating decision maker to make decisions for which discrete financial information is available. Based on the management approach as defined in Ind AS 108, the chief operating decision maker evaluates the Company's performance and allocates resources based on an analysis of various performance indicators by business segments and geographic segments.

3.20 Contributed Equity

Equity Shares are classified as equity, incremental costs directly attributable to the issue of new shares or options are shown in equity as deduction, net of tax from the proceeds.

3.21 Cash flow statement

Cash flows are reported using the indirect method, whereby profit before tax is adjusted for the effects of transactions of non-cash nature and any deferrals or accruals of past or future cash receipts or payments. The cash flows from operating, investing and financing activities of the Company are segregated based on the available information.

3.22 Dividend Distribution

Annual dividend distribution to the shareholders is recognised as a liability in the period in which the dividends are approved by the shareholders. Any interim dividend paid is recognised on approval by Board of Directors. Dividend payable and corresponding tax on dividend distribution is recognised directly in other equity.

4 SIGNIFICANT ACCOUNTING JUDGEMENTS, ESTIMATES AND ASSUMPTIONS:

The preparation of the financial statements requires management to make judgements, estimates and assumptions that affect the reported amounts of revenues, expenses, assets and liabilities, and the accompanying disclosures, and the disclosure of contingent liabilities. Uncertainty about these assumptions and estimates could result in outcomes that require a material adjustment to the carrying amount of assets or liabilities affected in future periods. The key assumptions concerning the future and other key sources of estimation uncertainty at the reporting date, that have a significant risk of causing a material adjustment to the carrying amounts of assets and liabilities within the next financial year, are described below. The Company based on its assumptions and estimates on parameters available when the financial statements were prepared. However, existing circumstances and assumptions about future developments may change due to market changes or circumstances arising that are beyond the control of the Company. Such changes are reflected in the assumptions when they occur.

4.1 Property, plant and equipment and Intangible Assets:

Management reviews the estimated useful lives and residual values of the assets annually in order to determine the amount of depreciation to be recorded during any reporting period. The useful lives and residual values as per schedule II of the Companies Act, 2013 or are based on the Company's historical experience with similar assets and taking into account anticipated technological changes, whichever is more appropriate.

4.2 Income Tax:

The Company reviews at each balance sheet date the carrying amount of deferred tax assets. The factors used in estimates may differ from actual outcome which could lead to an adjustment to the amounts reported in the financial statements.

4.3 Contingencies:

Management has estimated the possible outflow of resources at the end of each annual reporting financial year, if any, in respect of contingencies/claim/litigations against the Company as it is not possible to predict the outcome of pending matters with accuracy.

4.4 Impairment of financial assets:

The Impairment provisions for financial assets are based on assumptions about risk of default and expected cash loss. The Company uses judgement in making these assumptions and selecting the inputs to the impairment calculation, based on Company's past history, existing market conditions as well as forward looking estimates at the end of each reporting period.





4.5 Impairment of non-financial assets:

The Company assesses at each reporting date whether there is an indication that an asset may be impaired. If any indication exists, or when annual impairment testing for an asset is required, the Company estimates the asset's recoverable amount. An asset's recoverable amount is the higher of an asset's or Cash Generating Units (CGU) fair value less costs of disposal and its value in use. It is determined for an individual asset, unless the asset does not generate cash inflows that are largely independent to those from other assets or groups of assets. Where the carrying amount of an asset or CGU exceeds its recoverable amount, the asset is considered impaired and is written down to its recoverable amount.

In assessing value in use, the estimated future cash flows are discounted to their present value using a pre-tax discount rate that reflects current market assessments of the time value of money and the risks specific to the asset. In determining fair value less cost of disposal, recent market transactions are taken into account. If no such transactions can be identified, an appropriate valuation model is used. These calculations are corroborated by valuation multiples or other available fair value indicators.

4.6 Defined benefits plans:

The Cost of the defined benefit plan and the present value of such obligation are determined using actuarial valuations. An actuarial valuation involves making various assumptions that may differ from actual developments in the future. These include the determination of the discount rate, future salary increases, mortality rates and attrition rate. Due to the complexities involved in the valuation and its long-term nature, a defined benefit obligation is highly sensitive to changes in these assumptions. All assumptions are reviewed at each reporting date.

4.7 Recoverability of trade receivable:

Judgements are required in assessing the recoverability of overdue trade receivables and determining whether a provision against those receivables is required. Factors considered include the credit rating of the counterparty, the amount and timing of anticipated future payments and any possible actions that can be taken to mitigate the risk of non-payment.

4.8 Provisions:

Provisions are recognized in the period when it becomes probable that there will be a future outflow of funds resulting from past operations or events and the amount of cash outflow can be reliably estimated. The timing of recognition and quantification of the liability require the application of judgement to existing facts and circumstances, which can be subject to change. Since the cash outflows can take place many years in the future, the carrying amounts of provisions and liabilities are reviewed regularly and adjusted to take account of changing facts and circumstances.

4.9 Fair value measurement of financial instruments :

When the fair value of financial assets and financial liabilities recorded in the balance sheet cannot be measured based on quoted prices in active markets, their fair value is measured using valuation techniques including the Discounted Cash Flow (DCF) model. The inputs to these models are taken from observable markets where possible, but where this is not feasible, a degree of judgement is required in establishing fair values. Judgements include considerations of inputs such as liquidity risk, credit risk and volatility. Changes in assumptions about these factors could affect the reported fair value of financial instruments.





				in Graa Millio Millio		- Martine	a share	and the second sec	
Gross Carrying Value As at April 1, 2017	15,138.85	309.52	25,507.44	10.74	35.91	2.21	14,283.94	•	55,288.61
Additions	204.81	,	40.92				49.60		295.33
Disposals		ı	(261.08)	(3.80)		(1.12)			(266.00)
Other Adjustments	•		(8.32)	•	8.32		I		•
As at March 31, 2018	15,343.66	309.52	25,278.96	6.94	44,23	1.09	14,333.54		55,317.94
Additions		•	41.37	,	•	0.72	55.03	70.66	167.78
Disposals	•		•	,	(8.33)	1	I	I	(8.33)
Other Adjustments	ı		,		ſ	•	,	ł	'
As at March 31, 2019	15,343.66	309.52	25,320.33	6.94	35.90	1.81	14,388.57	70.66	55,AT7.39
Accumulated Depreciation									
As at April 1, 2017	ı	63.11	2,546.51	1.95	7.07	0.02	1,154.50		3,773.16
Deprectation for the year	,	53.42	2,547.88	1.95	5.67		1,157.25	,	3,766.17
Deductions	ł		(81.05)	(1.14)	4.00	(0.0Z)			(78.21)
As at March 31, 2018		116.53	5,013.34	2.76	16.74		2,311.75		7,461.12
Depreciation for the year	•	3.11	2,530.42	1.32	5.06	0.12	1,159.97	1.76	3,701.76
Deductions	ı		•		(7.86)				(7.86)
As at March 31, 2019		119.64	7,543.76	4.08	13.94	0.12	3,471.72	1.76	11,155.02
Net Carrying value as at March 31, 2019	15,343.66	189.88	17,776.57	2.86	21.96	1.69	10,916.85	68.90	44,322.37
	46 343 66	107 00	20.765.62	4.18	77 49	1.00	PT_100.01		47,856.82





	11		(Rupees in Jakhs
at the second	Seconda Inderando	的时间的现在分词。	- Gui
Gross Carrying Value		and a second	
As at April 1, 2017	213.31	3,208.33	3,421.6
Additions	500.00		500.0
Disposals	-	-	-
As at March 31, 2018	713.31	3,208.33	3,921.6
Additions			_
Disposals	•	•	-
As at March 31, 2019	713.31	3,208.33	3,921.6
Accumulated Amortisation			
As at April 1, 2017	84.03	250.00	334.0
Amortisation for the year	114.61	249.84	364.4
Deductions		-	-
As at March 31, 2018	198.64	499.84	698.48
Amortisation for the year	148.04	249.84	397.88
Deductions	-	-	
As at March 31, 2019	346.68	749.68	1,096.36
Net Carrying value as at March 31, 2019	366.63	2,458.65	2.825.28
Net Carrying value as at March 31, 2018	514.67	2,708.49	3,223.16



6



7 OTHER NON CURRENT FINANCIAL ASSETS

OTHER NON CURRENT FINANCIAL ASSETS		(Rupees in lakhs)
		an an ann an
Unsecured, considered good unless otherwise stated Financial assets carried at amortised cost		Parks
Security Deposits	57.88	14.21
Financial Guarantee	605.76	804.39
Total	663.64	818.60

OTHER NON CURRENT ASSETS 8

OTHER NON CURRENT ASSETS		(Rupees in lakhs)
and the second	in an Algeoide (1000)	and the second states of the
Unsecured, considered good unless otherwise stated		e and the bolder and defining the second of
Capital Advances	869.64	864.64
Advances other than Capital advances		
- Other Advances - gratulty (Refer Note 37)	-	10.05
- Prepaid Rent	1.89	3.79
- TDS Receivable	458.59	345.37
- Service tax paid under protest	5.14	3.87
Total	1,335.26	1,227.72

INVENTORIES 9

			(Rupees in lakhs)
attanist Prata.		· ···································	Mittan I. and
Stores and Spares		-	15.66
	Total	-	15.66

10 CURRENT ASSETS - TRADE RECEIVABLES

CURRENT ASSE	TS - TRADE RECEIVABLES		(Rupees in lakhs)
: •	ាល់នៅនេស៊ីសេសូមើរ៉ុក្ខ។ 	Ataylı Ataylı (Öğü	Magazie (1965) Magazie (1965)
Trade R	eceivables considered good - Secured		
Trade R	eceivables considered good - Unsecured	1,323.33	666.59
Trade Re	eceivable which have Significant increase in Credit Risk	29.11	4.39
Trade Re	Trade Receivable -credit Impaired	-	-
		1,352.44	670.98
Less: Pro	Less: Provision for expected credit loss	29.11	4.39
		29.11	4.39
	Total	1,323.33	666.59





-

.

11 CURRENT ASSETS - CASH AND CASH EQUIVALENTS

		an it in all at	的现在分词
Balances with banks in current accounts Cash on hand		119.96 0.28	375.29 0.06
	Total	120.24	375.35

12 CURRENT ASSETS - OTHER BANK BALANCES

		fundanes un igunal
A surplik gen	ration And the Ing. Albert	
Deposits with banks to the extent held as margin money Interest Accrued on Fixed Deposit	258.12 9.18	
Tota	267.30	386.16

13 OTHER CURRENT FINANCIAL ASSETS

OTHER CURRENT FINANCIAL ASSETS			(Rupees in lakhs)
14.98 L. 1.98 L. 1.98 L. 1.99 L		(196-194) (196-194)(196-194) (196-194)(196-194)	in ar Arthress and a
Unsecured, considered good unless otherwise stated Financial assets carried at amortised cost Security Deposits Financial Guarantee		73.44 248.93	111.83 236.76
Т	otal	322.37	348.59

14 OTHER CURRENT ASSETS

OTHER CURRENT ASSETS			(Rupees in lakhs)
2.15×0.42 b :			
		an a	Miner I. Alies
Advances other than Capital advances			
- Advances to Related Parties		-	0.23
- Advances to Suppliers		39.84	81.34
- Advances to Employees		2.87	16.58
-Others		16.54	15.55
Others			
- Other receivable		35.66	35.66
- Prepaid expenses		1.54	2.02
- Balance with Government Authority		37.36	157.08
	Total	133.81	308.4





(Rupees in lakhs)

(Ruppes in lakbs)

		(Rupees in lakh
na stan de 1919 - Julio - Juli	en en vog elfe som statione	tan Marting Ing Arries
i) Authorised Share Capital Equity Shares		
4,36,50,000 (As at March 31, 2018: 4,36,50,000) Equity shares of Rupees 10 each	4,365.00	4,365.0
Preference Shares		
13,50,000 (As at March 31, 2018: 13,50,000) Preference Shares of Rupees 10 each	135.00	135.0
Total	4,500.00	4,500.0
ii) Issued, Subscribed & Fully Paid up		
4,23,84,417 (As at March 31, 2018: 4,23,84,417) Equity Shares of Rupees 10 each	4,238.44	4,238.4
Total	4,238.44	4,238.4

ii) Reconciliation of number of Equity Shares outstanding at the beginning and at the end of the year:

The States	ំ ស្រុកអង្ករង្គម្លាំក្នុង ស្រុកអង្ករង្គម្លាំងក្នុង	ning and a single state And a grant of the state of the state
Equity Share Capital Equity shares of Rupees 10 each issued, subscribed and fu	bliq yili	
At April 1, 2017	4,23,84,417	4,238.44
issued during the year	-	-
At March 31, 2018	4,23,84,417	4,238.44
issued during the year	-	-
At March 31, 2019	4,23,84,417	4,238.44

Terms/rights attached to equity shares

The company has only one class of equity shares having par value of Rupees 10 per share. Each holder of equity shares is entitled to one vote per share. The company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject to the approval of the shareholders in the ensuing Annual General Meeting.

In the event of liquidation of the company, the holders of equity shares will be entitled to receive remaining assets of the company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the shareholders.

iii. Details of shareholders holding more than 5% shares in the company

		·····	i sala ziniko
्रियमेरेव ु	COD-DEBUG	dunder	Solution -
			the second second second second
4,23,83,817		4,23,83,817	
l	100.00	1	100.00
600		600	
	4,23,83,817	4,23,83,817 100.00	4,23,83,817 4,23,83,817 100.00





		(Rupees in lakhs)
6 	en e	na ki Manaki paga naki po
Securities Premium Account Retained Earnings Equity Component of Guarantee given by Parent Company Equity Component of Ioan from Parent Company	38,123.31 (58,116.26) 1,624.45 171.37	38,123.31 (53,932.75) 1,564.45 171.37
Total	(18,197.13)	(14,073.62)

(a) Securities Premium Account

(a) Securities Premium Account		fuelses ut lowest
This due the second	eren ga Attente also attente -	્યક ના કરવા આ ગામના સાહેલ્ટ
Opening balance Add: On issue of Equity shares	38,123.31	38,123.31
Closing Balance	38,123.31	38,123.31

(Runees in Jakhs)

(b) Retained Earnings		(Rupees in lakhs)
Stephine Base	A CARLES AND A CAR	later to Attended to generated attended
Opening balance Add/(Less):	(53,932.75)	(43,189.57)
Net Profit/(Loss) for the year	(4,159.12)	(10,745.75)
Other comprehensive income	(24.39)	7.19
Equity component of Guarantee given	-	(4.62)
Closing balance	(58,116.26)	(53,932.75)

c) Equity Component of Guarantee given by Parent Company		(Rupees in lakhs)
novit esti	na a Manada na dinisi y	fat and States and some states
Equity Component of Guarantee given by Parent Company	4.554.45	4 5 6 4 15
Opening balance Add/(Less): Transaction during the year	1,564.45 60.00	1,564.45
Closing balance	1,624.45	1,564.45

(d) Nature & purpose of Reserves

Securities Premium Account:

Securities premium reserve represents the amount received in excess of the face value of the equity shares. The utilisation of the securities premium reserve is governed by the Provision of Companies Act, 2013.

Retained Earnings :

Retained Earnings are the profits/losses of the Company earned till date net of appropriations.

Equity Component of Guarantee given by Parent Company:

The fair value of financial guarantees given to the lenders of the Company by the Parent Company is recognised as a deemed equity component.

Equity Component of Ioan from Parent Company:

The difference between the fair value of Interest free loans on the date of Issue and the transition price is recognised as a a RVEDI & Annual equity component by the Parent Company.

computation of the fair value benefit, the Company has estimated the fair value of the financial library to the date of the financial library to the date of the facts and circumstances relevant to the company.

17 NON CURRENT BORROWINGS

Hort contain bornto mitos				(Rupees in lakins)
÷.			ेक्षर द्वा देवस्वाहेन्द्र द्वार, जोगवेद्रः	terre de la companya de la companya Companya de la companya de la company
Secured -Term Loans From Others (r	efer note 17.1 below	v)	31,902.67	33,538.04
		Total	31,902.67	33,538.04

17.1 Rupee term loan from other parties :-

- Rupee term loans (including current maturity) of Rupees 33,647.31 lakhs (March 31, 2018 : Rupees 33,538.04 lakhs) are secured by

(1) Details of Security

i. First charge on all movable assets (including rakes, containers, equipment's) and immovable properties of the company both present and future on pari passu basis.

ii. Second charge by way of Hypothecation of the entire current assets of the company on pari passu basis.

iii. Pledge of 100% equity shares of the Company held by Promoters.

iv. Personal guarantees from Promoters of Holding Company.

v. Corporate Guarantee from Holding Company.

(2) Terms of Interest rate:

- on Term Loans from others 10% p.a. compounded quarterly,

(3) Terms of repayment :-

		(Rupees in lakhs)
	and the second state of the second	
9 - est <u>- e ben</u>	2019-2020	1,744.63
	2020-2021	6,139.19
1	2021-2022	2,276.52
	2022-2023	23,954.16
	Total	34,114.50

(4) The amortised cost disclosed above is net off of incidental cost of borrowings aggregating to Rupees 467.19 lakhs for March 31, 2019 and Rupees 612.13 lakhs for March 31, 2018.





(December 1) better

(Rupper In Jakks)

18 NON CURRENT LIABILITIES - PROVISIONS

8	NON CURRENT LIABILITIES - PROVISIONS			(Rupees in lakhs)
	ν νη φράτιβο το το τ		Alexian States	Company and the States
	Provision for employee benefits Gratuity (Refer Note 37) Leave encashment (Refer Note 37)		72.03	-
		Total	48.49	16.42

OTHER NON CURRENT LIABILITIES 19

OTHER NON CURRENT LIABILITIES		(Rupees in lakhs)
$(1)^{(1)} (r_{A}(t_{A}$		- A.76
na n	Bertial Style Barts	BEACH OD . SERVER !
Government Grants	1,632.32	1,723.69
Total	1,632.32	1,723.69

20	CURRENT BORROWIN	IGS			(Rupees in lakhs)
		1273)nêley(6160		 (N. 1741) (M. 2008)	Aleria Alerian Section 6
	Unsecured Loans from He	olding Company (refer no	te 20.1 below & Note 41)	304.38	2,029.26
			Total	304.38	2,029.26

20.1 Interest free loan upto 1 year and repayable on demand.





21 CURRENT LIABILITIES- TRADE PAYABLES

CURRENT LIABILITIES- TRADE PAYABLES		(Rupees in lakhs)
equeration of the second se	en an Den se se server	ing and the second of the second s
Total outstanding dues of Micro and Small Enterprises (Refer note 39 & 52) Total outstanding dues of creditors Other than Micro and Small Enterprises (Refer Note 52)	15.77 1,307.44	5.24 620.88
Total	1,323.21	626.12

OTHER CURRENT FINANCIAL LIABILITIES 22

IER CURRENT FINANCIAL LIABILITIES		(Rupees in lakhs)
ni yanta Tana ang ang ang ang ang ang ang ang ang	an dan serien dari kan serien Serien dari kan serien dari kan s	and a second second Second second
Financial Liabilities at amortised cost		
Current maturities of long term borrowings		
Term Loans from banks (Refer Note 22.1)*	6,910.60	8,931.30
Term Loans from others (Refer Note 17.1 & 22.2)	4,704.64	-
Interest accrued and due on borrowings**	11,710.49	13,183.84
Interest accrued but not due on borrowings	3,280.15	1,370.35
Security Deposit	41.44	57.64
Financial Guarantee		3.85
Others		
Project Creditors(Refer Note 35 & 52)	959.28	1,054.01
Employee's Dues ***	216.98	103.59
Payable for Expenses	280.10	502.26
Total	28,103.68	25,206.84

* Include Loan aggregating to Rupees 6,910.60 lakhs (March 31, 2018: Rupees 8,931.30 lakhs) recalled by banks.

**Include Interest accrued and due on Term Loans aggregating to Rupees 8,289.98 lakhs (March 31, 2018: Rupees 11,893.70 lakhs) recalled by banks.

**Include interest accrued and due on Term Loans aggregating to Rupees 390.64 lakhs recalled by others.

*** include Full and Final settlement of Rupees 25.28 lakhs (March31, 2018 Rupees 14.04 lakhs)

22.1 Rupee term loan from Banks :-

- Rupee term loans of Rupees 6910.60 lakhs (March 31, 2018 : Rupees 8931.30 lakhs) are secured by (1) Details of Security

i, First charge on all movable assets (including rakes, containers, equipment's) and immovable properties of the company both present and future on pari passu basis.

ii. Second charge by way of Hypothecation of the entire current assets of the company on pari passu basis.

iii, Pledge of 100% equity shares of the Company held by Promoters.

iv. Personal guarantees from Promoters of Holding Company.

v. Corporate Guarantee from Holding Company/ Promoter.

(2) Terms of Interest rate:

- on Term Loans from Banks from 10.45% p.a - 16.25% p.a.

(3) Terms of repayment :-

		(Rupees in lakhs)
	· History B. Style	Raffe - app. 1 af 2.
		- Refer 2-1908
	2012-2013	22.50
	2013-2014	75.00
	2014-2015	277.50
	2015-2016	3,695.60
AVE A	2018-2019*	2,840.00
TURVEDTO	Totai	6,910.60
* MURVE Pra . * Reference no.	48	



CCOUN

(4) Amount and period of default in repayment of borrowings

(4) Miloune and bei				(Rupees in lakhs)
<u>, , ,</u>	and the second	·	(24-9+4)s.	Anderski (Standar) 1. julij: Standar (Standar) 1. julij: Standar (Standar)
Current maturity	of Rupee Term loans		22.50	2012-13
			75.00	2013-14
			277.50	2014-15
			3,695.60	2015-16
			2,840.00	2018-19
Total			6,910.60	

22.2 Rupee term loan from Others :-

- Rupee term loans of Rupees 2,960.00 lakhs are secured by (1) Details of Security

I. First pari passu charge on all present and future cash flows of the Company

ii.First parl passu charge on all movable assets and immovable assets of the company.

ill. Charge on cash flows and movable assets by deed of Hypothecation.

Iv. Personal guarantees from Promoters of Holding Company.

v. Corporate Guarantee from Holding Company.

(2) Terms of Interest rate:

- on Term Loans from others @ 20% p.a payable quarterly

(3) Terms of repayment :-

		(Rupees in lakhs)
	 	a star
са. 1911 г.	1116年1月4日4月4日(1 ³³ 2)年末。	. Astrony (Artokie).
	 2022-2023*	2,960.00
	Total	2,960.00

* During the year the loan has been recalled by others.

(4) Amount and period of default in repayment of borrowings

		(Rupees in lakhs)
······································	અંગ્રેશ કર્ _ય	Ann an an thir and
Current maturity of Rupee Term loans	2,960.00	2018-19
Total	2,960.00	





** Amount and period of default in payment of interest on borrowings**

			(Rupees in lakhs)
			21§2
	an ta sa An Santa	×	1933 q. ¹ .
Interest accrued & due on borrowing	2013-14	965.05	
	2014-15	876.51	-
	2015-16	1,035.94 1,287.23	-
	2017-18	1,512.65	1,378.00
	2018-19	2,612.60	2,042.51
Total		8,289.98	3,420.51

OTHER CURRENT LIABILITIES 23

OTHER CURRENT LIABILITIES		(Rupees in lakhs)
™ sztákottás	Acta Alberto Alberto	and the second s
Advance received from Customers	107.02	151.12
Other Advances	0.51	1.74
Government Grants	365.49	639.61
Others		
Statutory Liabilities*	811.31	698.99
Interest on Delayed payment of Statutory dues	\$90.57	426.77
Tot	al 1,874.90	1,918.23

* Statutory liabilities include TDS, Goods & Service Tax, Service tax, PF, ESIC payable, Employee professional tax

OVERNMENT GRANTS		(Rupees in lakhs)
$\sim (\hat{z}_{i+1}^{*}\hat{H})^{-1}$	્યે. છે. ગુણુ તે અને કર્યો - ગ્રીધિક્રમ	Norm Maria (Maria)
Opening balance Released to statement of profit and loss	2,363.30 (365.49)	2,728.79 (365.49)
Closing balance	1,997.81	2,363.30
Current	365.49	639.61
Non Current	1,632.32	1,723.69

24 CURRENT LIABILITIES - PROVISIONS

4	CURRENT LIABILITIES - PROVISIONS		(Rupees in lakhs)
	ataráitjetb),or	(1967) MARTER 2010	Arraite Arraite and Arraite
	Provision for employee benefits Leave encashment (Refer Note 37)	10.61	3.69
	Total	10.61	3.69





25 REVENUE FROM OPERATIONS

REVENUE FROM OPERATIONS			(Rupees in lakhs)
	in the Balance of States and States	i Basis mikatan Ahardar aku, Tahasi	terre considered National and Alberta
Rail Freight Income Road Freight Income Handling Income Terminal Income		12,098.57 722.70 626.06 167.50	12,338.16 440.86 643.90 175.40
	Total	13,614.83	13,598.32

26 OTHER INCOME

OTHER INCOME		(Rupees in lakhs	
era karan i	ંગ્લન પ્રસ્તોપ્તિ: તેમનાગત છે, ટોલ્ટ ર	Aller and the second	
Interest income on Financial assets carried at amortised cost			
Unwinding of interest on loan to related party	-	168.45	
Unwinding of interest on Security deposit	5.76	1.28	
Other interest income			
Interest on Bank fixed deposits	21.10	29.38	
interest income on income tax refund	-	16.92	
Income on derecognistion of related party loan	-	307.69	
Other Income			
Government Grants	365.49	365.49	
Financial Guarantee Income	3.85	0.77	
Sundry Balance/ Excess provision Written Back	245.67	553.46	
Gain on sale of investment	-	5.00	
Miscellaneous Income	-	10.95	
Total	641.87	1,459.39	

27 COST OF OPERATIONS

COST OF OPERATIONS	(Rupees in lakhs)
(⁴ ,15)(04)(1).	izan minina Meninde 193, 1968: Metron Dab 2006
Rail Freight Expenses	9,434.84 10,172.63
Road Freight Expenses Handling Expenses	687.64 448.10 227.76 231.86
Terminal Expenses Other operating expenses	357.55 304.26 692.48 350.53
Total	11,400.27 11,507.38





.

a.

28 EMPLOYEE BENEFITS EXPENSE

8 EMPLOYEE BENI	EFITS EXPENSE		(Rupees in lakhs)
· · · · · · · · · · · · · · · · · · ·	n i la seven de	terret contentes. antidagis da gladis	and the second s
Contributio	ages and bonus on to provident and other funds re expenses	1,037.47 23.21 61.23	776.16 15.09 35.54
	Total	1,121.91	826.79

29 FINANCE COST

FINANCE COST			(Rupees in lakhs)
	Separation	Ann archear Beiseis an Airst	Man darim 64% Alton 45 ANR
Interest expense o	n Financial liabilities measured at amortised cost		
Interest expense o	n borrowings	6,476.99	7,451.08
Unwinding of inter	est for loan from related party (Refer Note 41)	63.70	56.88
Interest expense o	n Security Deposit	1.24	1.24
interest expense o	thers		
Interest on Delaye	d Payment of Statutory Dues	167.51	263.01
Interest on MSME	vendors	0.50	-
Other borrowing of	osts		
Guarantee Commi	ssion Expense	246.46	240.56
Interest Charges O	thers	53.33	-
Bank charges		5.84	6.17
	Total	7,015.57	8,018.94

30 DEPRECIATION AND AMORTISATION EXPENSES

DEPRECIATION AND AMORTISATION EXPENSES		(Rupees in lakhs)
Markteatta a	kran artanati 🦾 🛔 Viatnati (in Algio)) A comaint of a
Depreciation on tangible assets Amortisation on intangible assets	3,701.76 397.88	3,766.16 364.45
Total	4,099.64	4,130.61





31 OTHER EXPENSES

ER EXPENSES	Sec. 6. 4297/9	(Rupees in lakh
	eren an	Marsa a and
Repairs and maintenance		
- Building	16.73	17.
- Others	58.25	46.
Advertisement	35.62	38.
Payments to Auditors (Refer note below)	12.50	8.
Bad Debts written off	-	101.
Electricity charges	33.81	28.
Insurance	21.45	18.
Legal and professional fees	176.32	75.
Rates and taxes	15.89	7.
Rent	472.99	374.
Printing and Stationary	16.08	7.
Office Expenses	32.26	13.
Security charges	59.11	45.
Telephone and internet expenses	25.47	18.0
Travelling & conveyance expenses	185.37	46.
Vehicle Expenses	44.54	28.:
Allowance for doubtful debts and advances	24.71	
Miscellaneous expenses	22.19	18.8
Loss on sale of scrap	0.30	
Total	1,253.59	895.4

(a) Details of Payments to auditors

(a) Details of Payments to auditors		(Rupees in lakhs)
	1	Na d
As Statutory Auditor Audit Fee	12.50	8.50
Total	12.50	8.50

32 EXCEPTIONAL ITEMS

EXCEPTIONAL ITEMS				(Rupees in lakhs)
	⁶⁴ ihailteiteiteite		The debe The debe	1000 - 1000 -
Sundry Balance / exce	ss provision written back		-	(61.07)
Loss on sale/discarded	of Property, plant and equipment	1	-	182.79
Settlement of Claim	Settlement of Claim		(6,475.16)	302.54
	Total		(6,475.16)	424.26





33 Earnings per share (Basic and Diluted)

er a dep du	ja soni Spinasjenski, najveti	anaran shari t
Profit available to equity shareholders Profit/(Loss) after tax (A) (Rupees In Lakhs)	(4,159.12)	(10,745.75)
Number of equity shares Weighted average number of equity shares outstanding (Basic and Diluted) (B)	4,23,84,417	4,23,84,417
Basic & Diluted earnings per share(A/B) (Rupees)	(9.81)	(25.35)
Nominal Value of an equity share (Rupees)	10	10

34 CONTINGENT LIABILITIES & COMMITMENT

(To the extent not provided for)

I) Capital Commitments

	_						(Rupees in lakhs)
· · · · · · ·						100 (1888)	Staffe In the
的话站了这						Bigen Ste Ander	in the set of the set
		uning to be everyted	on capital account and	ant provided for (and the second second	ماري المريك المريك المريك المريك
	of contracts rema	atoug to be executed	i on capical account and	The house to f	iner	475.64	400.98
of advances paid)					1		400.50
Cash outflow expe	cted on execution	on such capital cont	racts				

(II) Contingent liabilities:

			(Rupees in lakhs)
Reaction of the second s		36976130593.220842) 31976130693.220842)	anara am
(a) Carrier Bond (Bond has been given to Principal of custom for the safeguarding duty liabilit movement from respective ports to ICD Khurja)	ty on goods	675.00	1,080.00
(b) Bank Guarantee		101.25	162.00
(c) Letter of Credit (Letter of Credit given in favour of Railways for availing e-freight facility for payment)	or haulage	100.00	100.00
(d) Claims Against the Company not acknowledged as Debt		23.39	23.39
(e) Service tax Demand		51.55	51.55
No Cesh outflow is expected in near future			

35 Certain creditors have initiated legal proceedings against the company and its directors and the company has defaulted in payment of instalments of consent terms for which the company is in process of negotiating and finalising the revised consent terms. Majority of the creditors have been settled over the past few years and some of the creditors have also shown interest and faith not only in logistics and infrastructure sector but also in the Company and are being allotted equity shares of Holding Company.

36 Operating lease commitments - Company as lessee The company has taken office on lease under non-cancellable operating lease expiring at the end of 3 years. The leases have varying terms, escalation clauses and renewal rights. On renewal, the terms of the leases are renegotiated.

The company has paid Rupees 340.65 lakhs (March 31, 2018: Rupees 278.56 lakhs)during the year towards minimum lease payment.

					(Rupees in lakhs)
in an				B. Berry and	150741 F. 2010
Commitments for minimum I	ease payments in relation	to non cancellable operat	ing leases are as		
follows					
Future Lease Payment:					
a) Within one Year				351.92	340.65
b) Later than one year but not	later than five years			699.19	1,051.11
c) Later than five years				•	
Total				1,051.11	1,391.76
Lease rents recognised as expe	inse in the year			340.65	278.56

37 EMPLOYEE BENEFIT

37.1 Disclosure pursuant to Indian Accounting Standard (IND AS) 19 - Employee Benefits

(a) Defined contribution plans

Contribution to Defined Contribution Plan, recognised as expenses for the years are as under:

		 	(Rupees in lakhs)
1 ·	en trakkoleren.	vrensensjes Mennik - He zijne	aria Mura Si 20 0
AVEDI	Employer's Contribution to Provident Fund	7.00	5.46
SURVEDI S	Employer's Contribution to Pension Scheme	15.88	12.40
IST NE	Employer's Contribution to ESIC	0.33	0.78
* (MUMBAI) +		UNREAL PROPERTY OF	
(b) Brief descriptions of the plans

The Company's defined contribution plans are Provident Fund and Employees State insurance where the Company has no further obligation beyond making the contributions. The Company's defined benefit plans include gratuity. The employees are also entitled to leave encashment as per the Company's policy.

{c } Leave Obligations

MUM

DAC

Leave Obligations		(Rupees in lakhs)
Constant Constant	 Server and	all the second second
Provisions for Leave encashment		
Current	10.61	3.69
Non-Current	48.49	16.42
1		
Total	59.10	20.11

(d) Defined benefit plan – Gratuity:

The employee's Gratuity fund is managed by the Life insurance Corporation of India. The present value of obligation is determined based on actuarial valuation using the Projected Unit Credit Method, which recognised each period of services as giving rise to additional unit of employee benefit entitlement and measures each unit sepearately to build up to final obligation.

rticulars	March 31, 2019	March 31, 2018
Actuariel assumptions	Indian Assured lives	and a firm
ortality Table		
	Mortality (2006-08) Ult	Mortality (2006-08) U
scount rate	6.95%	7.40
pected return on plan assets	6.95%	7.40
	9.00%	7.00
lary Escalation Rate	17.00%	
lithdrawal Rate	58 Years	
atirement Age		
Change in Present value of defined benefit obligations		
ability as at the beginning of the year	47.31	37.:
terest cost	3.50	2.3
urrant service cost	19.62	7.3
enefits paid	(2.95)	{0.
ctuarial (gain)/loss on obligations	24.39	0.9
ctuarial (gain)/www.on.ophgacons	41.75	
coulsition adjustment	133.62	47.3
ability as at the end of the year		
I. Change in Fair value of plan assats		
air value of plan assets as at the beginning of the year	57.35	46.
xpected return on plan assets	4.23	14.4
ctual Enterprise's Contributions		0.9
	-	(0.
enefits paid		(2.
ctuarial gain/(loss) on plan assets	61.58	57.
air value of plan assets as at the end of the year		57.
V. Actual return on plan assets		
xpected return on plan assets	4.23	14.
actuarial gain/(loss) on plan assets		(2.
ctual return on plan assets	4.23	11.
/. Liability recognised in the Balance Sheet	133.62	47.
lability as at the end of the year	61.58	57.
air value of plan assets as at the end of the year	*****	
iability/ (Asset) recognised in the Balance Sheet	72.03	(10.
VI. Percentage of each category of plan assets to total fair value of plan assets		
Insurer managed funds	100%	i 10
		1
VII. Amount recognised in the Statement of Profit and Loss	19.62	7.
Current service cost	13.02	
interest cost	(0.74)	2.
Expected return on plan assets	(0.74)	(14.)
Net actuarial (gain)/loss to be on obligation		
Expense/ (Income) recognised in Statement of Profit and Loss		(3
VIII. Amount recognised in the Other Comprehensive Income (OCI)		·
Amount recognised in OCI, beginning of the year	(7.29)	(0.
Amount recognised in Col, cogniting of the post		,
components of actuarial gain/losses on obligations to Change in financial assumptions	16.19	(1
to change in mancial assumptions	(1.82)	,
the to Change In demographic assumption	10.02	
Dire to Change in experience assumption	10.02	
Enclo Change in experience assumption Experted return on plan assets Change in Asset Ceiling	BUSA	(2
Change in Asset Ceiling	26.4	1
ocal remasurement recognised in OCI	24.39	•••
	17.10	{7.
Amount recognised in OCI, end of year		

IX. Balance Sheet reconciliation		
Opening net llability	(10.05)	(9.09)
Expenses recognised in Profit & Loss	18.88	7.16
Actual Employer Contribution	(2.94)	(0.93)
Total Remeasurement recognised in OCI	24.39	(7.19)
Acquisition adjustment	41.75	-
Closing net liability	72.03	(10.05)

(e) Salary escalation assumption has been set in discussions with the enterprise based on their estimates of overall long-term salary growth rates after taking into consideration expected earnings inflation as well as performance and seniority related increases.

37.2 Sensitivity analysis:

Particulars	Changes in assumptions	Effect on Gratuity obligation (Rupees in lakhs)
For the year ended 31st March, 2018		
Salary growth rate	+0.50%	47.66
Discount rate	+0.50% -0.50%	45.41
For the year ended 31st March, 2019		47.00
Salary growth rate	+0.1%	140.38
	-0.1%	127.32
Discount rate	+0.1%	127.26
	-0.1%	140.58

The above sensitivity analysis is based on a change in an assumption while holding all other assumptions constant. In pratice, this is unlikely to occur, and changes in some of the assumptions may be correlated. In presenting the above sensitivity analysis, the present value of defined benefit obligation has been calculated using the projected unit credit method at the end of reporting period, which is the same as that applied in calculating the defined obligation ilability recognised in the balance sheet.

These plans typically expose the Company to actuarial risks such as: longevity risk and salary risk.

- (A) Interest risk A decrease in the discount rate will increase the plan ilability.
- (5) Longevity risk The present value of the defined benefit plan liability is calculated by reference to the best estimate of the mortality of plan participants both during and after their employment. An increase in the life expectancy of the plan participants. As such, an increase the plan's liability.
- (C) Salary risk The present value of the defined plan liability is calculated by reference to the future salaries of plan participants. As such, as increase in the salary of the plan participants will increase the plan's liability.
- 37.3 The weighted average duration of the defined benefit obligation at the end of the reporting period is 5 years (31st March, 2018 6 years).
- 38 Disclosure pursuant to Indain Accounting Standard 108 Operating Segment

The company is engaged in Private Container Train Operator (PCTO) business and is holding Category-I license which allows the company to operate on Indian Rall network on pan India basis both Domestic and Exim Traffic. In the opinion of the company, the entire operations are governed by the same set of risks and returns and hence the same has been considered as representing a single primary segment.

The company provides services within India and hence does not have any operation in economic environments with different risks and returns. Hence, it is considered that the company is operating in a single geographical segment.

Customers individually contributes to more than 10% of revenue :-

There are 2 customers (March 31, 2018 - 3 customers) aggregating to Rupees 8,179.24 lakhs (March 31, 2018 Rupees 8,619.29 lakhs) constituting 60% (March 31, 2018- 64%) of Revenue.

39 DISCLOSURES UNDER MICRO, SMALL AND MEDIUM ENTERPRISES DEVELOPMENT ACT, 2006 (MSMED ACT, 2006)

To the extent, the company has received intimation from the "suppliers" regarding their status under the Micro, Small and Medium Enterprises Development Act, 2006, the details are provided as under

	Development Act, 2000, the details are provided of an add		(Rupees In lakhs)
	Lationari and an	For the year March 31, 2019	March S1, 2018
	(I) Principal amount remaining unpaid	15.77	5.24
	(ii) interest due thereon remaining unpaid	0.50	-
	(III) Interest paid by the Company in terms of Section 16 of the Micro, Small and Medium Enterprises Development Act 2006, along with the amount of the payment made to the supplier beyond the	-	-
MUMBA	(v) Interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under the Micro, Sector and Medium Enterprises Development Act, 2006.	-	-
1	(Laterest accrued and remaining unpaid (net of tax deducted at source)	0.50	
(WILIMBA	White rest accude and maining up able even in the successful grears, until such date when the interest augress above are actually paid to the small enterprise.	-	DINFR.S.
ALAND ACCOUNT	S. C.		

40 RECENT ACCOUNTING PRONOUNCEMENTS

STANDARDS ISSUED BUT NOT EFFECTIVE

On March 30, 2019, the Ministry of Corporate Affairs (MCA) has notified ind AS 116 - Leases and certain amendment to existing ind AS. These amendments shall be applicable to the Company from April 01, 2019.

A) ISSUE OF IND AS 116 - LEASES

Ind AS 116 will supersede the current standard on leases i.e. Ind AS 17- Leases. As per Ind AS 116, the lessor will have to bring to books all the noncancellable portion of leasing arrangement.

B) AMENDMENT TO EXISTING STANDARD

The MCA has also carried out amendments of the following accounting standards I. Ind AS 101- First time adoption of Indian Accounting Standards II. Ind AS 103 - Business Combinations III. Ind AS 109 - Financial Instruments IV. Ind AS 111 - Joint Arrangements V. Ind AS 12 - Income Taxes V. Ind AS 19 - Employee Benefits VI. Ind AS 19 - Employee Benefits VI. Ind AS 23 - Borrowing Costs VII. Ind AS 23 - Borrowing Costs VIII. Ind AS 28 - Investment in Associates and Joint Ventures Application of above standards are not expected to have any significant Impact on the Company's financial statements.





41 RELATED PARTY TRANSACTIONS

(i) List of related parties as per the requirements of Ind-AS 24 - Related Party Disclosures

- -	and a second sec	an a	S. SPORTANE S	าระสมกับ เราะสมกับ สายสายสายสาย สายสายสายสายสายสายสายสายสายสายสายสายสายส
1	Arshiya Limited	Holding Company	100	India
2	Arshiya Rali Siding and infrastructure Limited (till February 3, 2018)	Subsidiary	100	India
3	Arshiya Supply Chain Management Private Limited (till March 21, 2018) Arshiya Northern FTWZ Limited Arshiya Industrial & Distribution Hub Limited Arshiya Logistics Services Limited (formerly known as Laxmipati Balaji Exim Trading Limited)	Feilow Subsidiaries		India
4	Mr. Ajay S. Mittal - Director Mr. Navnik Choudhary - Director Mr. Ashish Kumar Bairagra - Independent Director Mr. Sishabh Pankaj Shah - Independent Director Mr. Siddarth Kasturia - Chief Executive Officer (w.e.f. March 26, 2019) Ms. Avani Dipakkumar Lakhani - Company Secretary (w.e.f. March 26, 2019) Mr. Vinod Jain - Chief Financial Officer (Ill March 25, 2019) Mr. Amardeep Gupta - Chief Financial Officer (w.e.f. March 26, 2019)	Key Managerial Personnel {KMP}		
5	Mrs. Archana A Mittal Mr. Ananya A Mittal	Relative of Key Managerial Personnel		

(ii) The nature and amount of transactions with the above related parties are as follows

	 Carbora US, Augustanditore - P. 		(Rupees in lakhs
	The set of	ALL SAME STRATES AND A STREET STREET	STRUCTURE AND ADDRESS
	Loans given	•	376.0
	Loan repayments**		(1,312.8
rshiya industrial & Distribution Hub Limited	Rall Freight Income	749.45	23.4
	Unwinded Interest Income on	-	67.2
	Loan to subsidiaries		
	Loans given		943.8
	Loan repayments**		(2,354.8
ushiya Northern FTWZ Limited	Unwinded interest income on	- · ·	101.3
	Loan to subsidiaries		
	Loans taken**	(788.31)	(4,831.)
	Loan repayments	2,996.44	3,751.0
	Allocation of cost and common	(419.55)	(418.)
	expenses by Holding Company*		•
Arshiya Limited	Unwinded Interest expense on	(63.70)	(56.)
	Loan from holding company		
	Financial Guarantees	(246.46)	(240.5
white Courts Chain Monoroment Driveto Heritari	Loans given	-	11.
ushiya Supply Chain Management Private Limited	Loan repayments**	•	(11.)
	Loans given		0.3
Arshiya Logistics Services Limited (formerly known as Laxmipati Balaji Exim Trading Limited)	Loan repayments	0.23	

• During the year, the Holding Company has allocated certain common cost and expenses incurred by it, to the company aggregating to Rupees 419.55 lakhs (31st March, 2018 - Rupees 418.25 lakhs) based on Holding Company's estimates of such cost and expenses attributable to the company. Hence, Employee benefit expenses (Refer Note No. 28) and certain expenses stated under other expenses (Refer Note No. 31) are presented as inclusive of such allocation of certain common costs and expenses.

** The Company has adjusted balance receivable amount of Arshiya Industrial, Distribution & Hub Limited (AIDHL) on March 31, 2019 Rupees NIL (March 31, 2018 Rupees 1,269.42 lakhs), receivable amount of Arshiya Northern FTWZ Limited (ANFTWZ) on March 31, 2019 Rupees NIL (March 31, 2018 Rupees 2,026.74 lakhs), receivable amount of Arshiya Supply Chain Management Private Limited (ASCM) on March 31, 2019 Rupees NIL (March 31, 2018 Rupees 4.45 lakhs) with Arshiya Limited.





Closing Balances	Station Mr.	(Rupees in la 3(0/4())
and the second		
Arshiya Logistics Services Limited (formerly known as Laxmipati Balaji Exim Trading Limited)	-	· · · · · ·
Arshiya Umited	304.38	2,02
Arshiya Industrial Distribution & Hub Limited	289.58	
Arshiya Limited	854.69	1,04
and the second state of the second state of the second		
Ajay S Mittal	54,120.00	51,12
Archana A Mittal	54,120.00	51,12
and a second		
Arshiya Limited	51,200.19	48,20
HARAN MARKANI AND		
Arshiya Limited	550.00	

(iv) Key managerial personnel compensation

Mr. Vinod Jain - Chief Financial Officer (till March 25, 2019)		(Rupees in lakhs)
	Lan Alto Standard	56 6
Short term employee benefits	20.49	-
Total	20.49	-
Mr. Siddarth Kasturia - Chief Executive Officer (w.e.f. March 25, 2019)		(Rupses in lakhs
	and the second	March Aller
Short term employee benefits	0.95	-
Total	0.95	
Ms. Aveni Dipakkumar Lakhani - Company Secretary (w.e.f. March 26, 2019)		(Rupees in lakhs)
Ellevender	(1) 我就是我们的。	We Western Property
Short term employee benefits	0.10	-
Total	0.10	
Mr. Amardeep Gupta - Chief Financial Officer (w.e.f. March 26, 2019)		(Rupees in lakhs)
Openant in the second	ALCONTRACTOR	Million March
Short term employee benefits	0.36	-





185

42 FAIR VALUE MEASUREMENTS

F1. ended Instruments by Cotogood 81

Financial Instruments by Category				(Rupees in lakhs)
		jarajs.	\$\$\$	NAMES IN CONTRACT
	Street of Marsh	March 41 Alexan	ADDA SALE	WERE ALL COMPANY
FINANCIAL ASSETS				
Amortised cost				
Trade Receivables	1,323.33	666.59	1,323.33	666.55
Cash and Cash Equivalents	120.24	375.35	120.24	375.3
Security Deposits	131.32	126.04	131.32	126.04
Other Bank Balances	267.30	386.16	267.30	386.16
Financial Guarantee	854.69	1,041.15	854.69	1,041.15
				-,
190	George States and States	Setter (a a a a a a a a a a a a a a a a a a a	
FINANCIAL LIABILITIES				
Amortised cost				
Borrowings	43,822.29	44,498.60	43,822.29	44,498.60
Trade Payables	1,323.21	626.12	1,323,21	626.12
Other financial liabilities	16,488.44	16,275.54	16,488.44	16,275.54
			.,	20j2/0101
. Bederi	102.111	3. AU 19.40	(3) (348) (2 ⁿ	

(II) Fair Valuation techniques used to determine fair value

The Company maintains procedures to value financial assets or financial liabilities using the best and most relevant data available. The fair values of the financial assets and liabilities are included at the amount that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date.

The following methods and assumptions were used to estimate the fair values:

- (a) The Company assessed that the fair value of cash and cash equivalent, trade receivables, trade payables, and other current financial assets and liabilities approximate their carrying amounts largely due to the short term maturities of these instruments.
- (b) The fair values for security deposits and other financial liabilities were calculated based on cash flows discounted using a current lending rate. They are classified as level 3 fair values in the Fair value hierarchy due to the inclusion of unobservable inputs including counterparty credit risk.
- (c) The fair values of non current borrowings are based on discounted cash flows using a current borrowing rate. They are classified as level 3 fair values in the fair value hierarchy due to the use of unobservable inputs, including own credit risk.

(iii) Fair value hierarchy

This section explains the judgements and estimates made in determining the fair values of the financial instruments that are recognised and measure at fair value. To provide an indication about the reliability of the inputs used in determing fair value, the company has classified its financial instruments into three levels prescribed under the accounting standard.

- (a) Level 1 Level 1 hierarchy includes financial instruments measured using quoted prices.
- (b) Level 2 The fair value of financial Instruments that are not traded in an active market (for example, traded bonds, over the counter derivatives) is determined using valuation techniques which maximise the use of observable market data and rely as little as possible on entity-specific estimates. If all significant inputs required to fair value an instrument are observable, the instrument is included in level 2.
- (c) Level 3 If one or more of the significant inputs are not based on observable market data, the instrument is included in level 3. This is ase falling isted equity shares, contingent consideration and indemnification assets included in level 3. the





43 Financial Risk Management

The Company's principal financial liabilities comprises of borrowings, trade and other payables and financial guarantees contracts. The main purpose of these financial liabilities is to manage for the Company's operations. The Company's financial assets comprises of trade and other receivables, cash and deposits that arises directly from its operations.

The company's activities expose it to variety of financial risks including credit risk, liquidity risk and market risk. The Company's risks management assessment, management and processes are established to identify and analyze the risks faced by the Company to set up appropriate risks limits and controls and to monitor such risks and compliances with the same. Risks assessment and management policies and processes are reviewed regularly to reflect changes in market conditions and the company's activities.

Credit risk	Cash and cash equivalents, trade receivables and Financial assets measured at amortised cost.	Ageing analysis	Dioussen Regular review of credit limits
Liquidity risk	Borrowings and other liabilities	Rolling cash flow forecasts	Availability of financial support from parent company
Market risk – interest rate	Long-term borrowings at variable rates	Sensitivity analysis	Unhedged

The Company's risk management is carried out by a corporate finance team under the policies approved by the Board of Directors. The Board provides written principles for overall risk management as well as policies covering specific areas, such as credit risk, interest rate risk.

(A) Credit risk

The Company is exposed to credit risk, which is risk that counterparty will default on its contractual obligation resulting in a financial loss to the Company, Credit risk arises from cash and cash equivalents as well as credit exposures to trade customers including outstanding receivables.

Trade receivables are typically unsecured and are derived from revenue earned from customers located in India. Credit risk has always been managed by the company through continuously monitoring the creditworthiness of customers to which the company grants credit terms in the normal course of business. Outstanding customer receivables are regularly monitored and any further services to major customers are approved by the senior management. Credit risk is high as only few customers' account for majority of the revenue in the year presented. On account of adoption of Ind AS 109, the company uses expected credit loss model to assess the impairment loss or gain.

(B) Liquidity risk

Uquidity risk is the risk that the Company may not be able to meet its present and future cash and collateral obligations without incurring unacceptable losses. The Company's objective is to, at all times; maintain optimum levels of liquidity to meet its cash and collateral requirements. The Company limits its liquidity risk by ensuring funds from trade receivables. The Company relies on operating cash flows and funding from holding company to meet its needs for funds.

The table below provides undiscounted cash flows towards financial liabilities into relevant maturity based on the remaining period at the balance sheet to the contractual maturity date.





Contractual maturities of financial liabilities			(Rupees in lakhs)
Alter e con		naanta ay ahaanada dahaa B ^{iya} a	and the second s
March 31, 2019			
Borrowings	11,919.62	32,369.86	
Trade payables	1,323.21		
Other financial liabilities	16,489.00		-
Total Financial Ilabilities	29,731.83	32,369.86	-
March 31, 2018			
Borrowings	11,024.26	34,150.17	-
Trade payables	626.12	-	-
Other financial liabilities	16,276.91	-	
Total Financial Ilabilities	27,927.29	34,150.17	•

(C) Market risk

Market Risk is the risk that the fair value of future cash flow of a financial instruments will fluctuate because of volatility of prices in the financial markets. Market risk can be further segregated as: 1) Foreign currency risk and 2) Interest rate risk

1) Foreign currency risk

1) Foreign currency risk is the risk that the fair value or future cash flow or an exposure will fluctuate because of changes in foreign exchange rates. However, the Company does not have any foreign currency exposure.

2) interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Company's main interest rate risk arises from long term borrowings with variable rates, which expose the Company to cash flow interest rate risk. During the 31st March, 2019 the Company's borrowings at the variable rate were mainly denominated in Rupees.

The Company's fixed rate borrowings are carried at amortised cost. They are therefore not subject to interest rate risk as defined in IND AS- 107, since neither the carrying amount nor the future cash flows will fluctuate because of a change in market interest rates.

Interest rate risk exposure

The exposure of the company's borrowing to interest rate changes at the end of the reporting period are as follows:-

	(Rupees in Jakns)
March 31, 2019	March 31, 2018
6,910.60	8,931.30

Interest sensitivity

Profit or loss is sensitive to higher/ lower interest expense from borrowings as a result of changes in interest rates.

	impact on profi	it before tax
Particulars	March 31, 2019	March 31, 2018
Interest sensitivity 50 bps increase the profit before tax by* 50 bps decrease the profit before tax by* * Holding all other variable constant	(34.55) 34.55	(44.66) 44.66





44 CAPITAL MANAGEMENT

For the company's objective when managing capital is to safeguard the company's ability to continue going concern in order to provide the return for shareholders and benefit to other stakeholders and to maintain an optional capital structure to reduce the cost of capital. Capital includes issued equity capital, share premium and all other equity reserves attributable to the equity holders of the company.

The company manages its capital structure and makes adjustments in light of changes in economic conditions and the requirements of the financial covenants. To maintain or adjust the capital structure, the company may adjust the dividend payment to shareholders, return capital to shareholders or issue new shares. The company monitors capital using a Debt equity ratio.

		(Rupees in lakhs)
	AFRAD N ZAR	- 1997 (A. 1916)
Borrowings	43,822.29	44,498.60
Other Financial Liabilities (Interest accrued)	14,990.64	14,554.19
Total Debt	58,812.93	59,052.79
Equity	4,238.44	4,238.44
Other equity	(18,197.13)	(14,073.62)
Total Equity	(13,958.69)	(9,835.18)

Notes:-

(i) Debt is defined as long term and short term borrowings including current maturities and interest.

(ii) Total equity (as shown in balance sheet) includes issued capital and all other equity.

As stated in Notes to accounts, the company is also having scheme of arrangements to reorganise the capital structure. Debt Covenants

Under the terms of Restructuring Agreement, the company is required to comply with following financial covenants:-Without prior approval of lender, the company shall not:

- (i) Loans, debenture & charge Issue or subscribe to any debentures, shares, raise any loans, deposit from public, issue equity or preference capital, change its capital structure or create any charge on its assets including its cash flow or give any guarantees.
- (II) Dividend on equity shares declare/pay dividend on equity shares unless otherwise approved by the Lender/Business Monitoring Committee and subject to the payment of recompense amount payable by the borrower to the lender in accordance with the provisions of RA.

In order to achieve this overall objective, the Capital Management, amongst other thing, aims to ensure that it meets Financial covenants attached to the Interest bearing Loans and borrowings that define Capital structure requirements, there have been breaches in the Financial covenants of Interest bearing loans and borrowing in the Current period and previous period.

The Company has not proposed any dividend in last two years in view of losses incurred.





45 Taxation

- i) in view of loss for the year, no provision for current tax has been made.
- ii) The Company has not recognised any deferred tax assets on deductible temporary differences, unused tax losses as it is not probable that the Company will have sufficient future taxable profit which can be available against the available tax losses.
- iii) Unused tax losses for which no deferred tax assets has been recognised

			(nupees in lakits)
an a	taros as sep	an a giragina da sa 200 aniatar a 110 Aniatar (110 (110 (110 (110)))	: 20 م روسه (۲۵ م مرکز) ۲۰ م ۲۰
2014-2015	14,739.87	2022-2023	6,512.68
2015-2016	-	2023-2024	5,536.76
2016-2017		2024-2025	3,192.80
2017-2018	13,093.10	2025-2026	3,717.69
2018-2019	431.07	2026-2027	3,199.58
2019-2020		2027-2028	2,104.24
Total	28,264.04		24,263.74

Unused deferred tax assets as at March 31, 2019 Rupees 9,217.76 Lakhs (March 31, 2018 - Rupees 13,290.59 Lakhs) has not been recognised, as there is no convincing evidence that sufficient taxable profits will be available against which the unadjusted tax losses will be utilised by the Company.

Details of Deferred tax assets are mentioned below:-		(Rupees in lakhs)
Frankrik		States and States
	eligiteti in theine	Andrew Constant State
Property plant equipment	4,268.76	4,084.31
Unabsorbed depreciation	(6,308.57)	(5,773.59)
Expense allowable on payments under section	(166.12)	(3,651.00)
Unabsorbed loss	 (7,348.65)	(8,397.44)
Financial Instruments	 336.82	447.13
Total Deferred Tax Assets	(9,217.76)	(13,290.59)

46 Preparation of financial statements on " Going Concern" basis

The company has incurred net loss of Rupeés 4,183.51 Lakhs during the year ended March 31, 2019 and as of that date, the company's current liabilities exceeded by its current assets by Rupees 29,449.72 lakhs. Accumulated losses have also resulted into more than net worth of the company. Some of its lenders have recalled their loans and the company is in the process of negotiating the revised payment terms with the lenders.

The Company is EBIDTA positive and with the commencement of the two dedicated freight corridors, the said company will benefit immensely. Moreover, a unique contract entered into with one of the largest global shipping lines has already started to improve the profitability. Also, the proposed merger of Fellow subsidiary with the company as mentioned in note no. 51, would make their operations not only complementary, but enhance their profitability.

In view of the focussed emphasis of the Government on logistics infrastructure sector, the proposed restructuring and considering the fact that the facilities are yet to achieve full operational potential besides the strategic locations of the facilities, the management's future outlook of its businesses is very promising. Accordingly the financials have been prepared on going concern basis, based on financial support from the Parent Company.

The management of the company is in the process of restructuring its business operations and steps are as under :

- i Focus on long term contracts with corporate clients for stable revenues.
- ii Focus on reduction of empty haulage / return business in different sectors to increase in revenue and margins.
- iii Diversified focus from only Steel industry to Cement, Agro and Tiles also so as to have a balance of product mix.
- Iv Increased focus on Khurja as a distribution hub post GST implementation





(Runnes in Jakhe)

- Government focus on Multi-modal logistic and transport services to increase the throughput of the infrastructure already created by the company;
- vi Government focus on the revamping of the Railway Boards and increasing clarity on regulatory aspects to support resolution of the regulatory issues;
- vii Two Dedicated Freight Corridor(s) (DFC) along the Western and Eastern part of India to support increase in the business volume.
- viii Government announcement of Jewar Airport and Merut Highway connecting Jewar via Khurja and thereby connecting to Yamuna Expressway(i.e. Delhi to Agra).

The above steps shall enable the Company to improve Company's Net worth and its ability to discharge its debts/liabilities in near future.

- 47 As per debt covenant, the Company are required to adhere to repayment schedule and any short payment gives Edelweiss Asset Reconstruction Company (EARC) the right to convert whole of the outstanding amount of restructured rupee loan and/or part of the default amount into fully paid up equity shares of the Company. No such notice of conversion in writing has been given by EARC and the Company continues to disclose the amount as current and non-current as per repayment schedule, in the Balance Sheet.
- 48 The company had entered into one-time settlement (OTS) with a lender during the year. The company had paid partial amount against OTS and the effect was taken as an exceptional item during the year ended March 31, 2019. However, the Company has defaulted in payment as per the terms of the OTS. As a result, the Company needs to reverse the exceptional gain recorded during the year and needs to recognise Interest on the entire liability as per the original terms. The Company is in discussion with the lender for additional time to repay. Accordingly, the company continues to reflect the outstanding amount of loan as per OTS in the financial statements. The Company has not reversed the gain, nor provided for additional interest. Had the Company reversed the gain and provided for additional interest, exceptional item would have been lower by Rs. 6,604.55 Lakhs and finance cost would have been higher by Rs. 3,500.76 Lakhs having consequential impact on total comprehensive income, liabilities and other Equity for the year ended 31st March, 2019.
- 49 The Company has procured certain capital goods under EPCG scheme at concessional rate of duty. On non fulfiliment of certain conditions, the company may become liable to pay differential custom duty along with interest thereon such procurement. The management is hopeful of completing the expected obligation within the stipulated time.
- 50 The scheme of arrangement u/s 230 to 232 is scheme of merger of the Arshiya Industrial and Distribution Hub Limited and Arshiya Transport & Handling Limited into the company, which is pending with NCLT. The scheme is conditional on various approval / sanctions and is effective thereafter; accordingly no effect of the said Scheme is given in the financial statement. The Creditors meeting of the respective companies was held on 6th May 2019. No accounting impact and disclosures is considered and necessary at this stage pending requisite regulatory approvals.
- 51 In addition to above, the Board of Directors of the Parent Company at their meeting held on 24th May, 2018, has approved a scheme to reorganize corporate structure i.e. demerger of "Domestic warehousing business" of the Parent Company into the company. This proposed scheme of arrangement is conditional upon approval of above scheme (refer note no. 50). No accounting impact and disclosures is considered and necessary at this stage pending requisite regulatory approvals.
- 52 The Company has sent request letters/ emails to various parties for confirmations of balances under borrowings, trade receivables trade payables etc, loans and advances to which only few parties have responded. Accordingly, the possible adjustment, if any, required in the financial statements will be accounted as and when the same is determinable.
- 53 Corporation Bank has filed a suit with Debt Recovery Tribunal, New Delhi, towards recovery against Arshiya Rail Infrastructure Limited, Arshiya Limited as a Corporate Guarantor and two promoter directors of the holding company as Guarantors, for Rupees 8,012.60 lakhs. The same is pending before the DRT Delhi. The matter is sub-judice.





54 The Company disaggregates revenue from contracts with customers by type of products and services, geography and timing of revenue recognition.

Effective April 1, 2018, the Company adopted Ind AS 115 "Revenue from Contracts with Customers". The effect on adoption of Ind AS 115 was insignificant.

.

.

Revenue disaggregation by type of goods and services is given note no. 25

Revenue disaggregation by geography is as follows:

		(Rupees in lakhs)
Geography	For the year ended 31 st March 2019	For the year ended 31 st March 2018
In India	13,596.16	13,560.32
Outside India	18.67	38.00

Geographical revenue is allocated based on the location of the customers.

Revenue disaggregation by timing of revenue recognition is as follows:

Geography	For the year ended 31 st March 2019	<u>(Rupees in lakhs)</u> For the year ended 31 [#] March 2018
Goods transferred at a point in time	-	-
Service transferred over time	13,614.83	13,598.32

- 55 Based on recent Supreme court judgement on Provident Fund dated February 28, 2019 there are various interpretive issues including its applicability thus prospective provision w.e.f. March 01, 2019 been considered of Rs. 0.44 lakhs.
- 56 The Company's borrowings have been assigned by bankers to an ARC/ restructured with banks under CDR Package. Certain lenders had invoked the CDR package. The ARC/ CDR lenders have charged interest/ penal interest/additional Interest amounting to Rs. 511.89 lakhs upto the year ended March 31, 2018, which was not accepted by the Company and hence is under negotiation. In light of audit qualifications in previous year as a matter of prudence, the company has recognised the said interest/ additional interest/ penal interest and has accordingly restated the finance cost and other consequential impacts in respective year. Further, during the year ended March 31, 2018 reported figure of finance cost, other Equity and Interest Accrued on borrowings was Rs. 7,507.05 lakhs, Rs. (13,561.72) lakhs and Rs. 12,671.96 lakhs respectively. Restated figures of finance cost, other equity and Interest Accrued on Borrowings are Rs. 8,018.94 lakhs, Rs. (14,073.62) lakhs and Rs. 13,183.84 lakhs respectively. Earning Per Share (EPS) also recalculated based on the restated figures.
- 57 The figures for the previous year have been re-grouped /re-arranged, wherever necessary, to correspond with the current year's classification/disclosure.

Notes to the financial statements	1-57
As per our Report of even date	
For Chaturvedi & Shah LLP Chartered Accountants Firm Registration Number 101720W/W100355	For and on behalf of Board of Directors of Arshiya Rail Infrastructure Limited
Vijay Napawaliya Partner Membership Number. 109859	Ajay S Mittal Navnit Choudhary Director Director DIN : 00226355 DIN : 00613576
Place : Mumbai Date: May 27, 2019	Amardaep Gupta Chief Financial Officer Company Secretary

C.A (C.A.A.) 2926/ MB/ 2019

IN THE NATIONAL COMPANY LAW TRIBUNAL,

MUMBAI BENCH

C.A. (CAA)/2926/MB/2019

Under Section 230 to 232 read with section 66 and section 52 and other applicable provisions of the Companies Act, 2013 In the matter of Scheme of Arrangement between Arshiya Limited bearing CIN: L93000MH1981PLC024747 (Demerged

Company/Applicant Company 1) And Arshiya Rail Infrastructure Limited bearing CIN:

U93000MH2008PLC180907 (Resulting Company/Applicant Company 2) And Their Respective Shareholders.

ARSHIYA LIMITED

)

)

302, Ceejay House, Level-3, Shiv Sagar)

Estate, F-Block, Dr. Annie Besant Road,) Worli, Mumbai 400018 Maharashtra India

)... Demerged Company/Applicant Company 1

ARSHIYA RAIL INFRASTRUCTURE)

LIMITED) 302, Ceejay House, Level-3, Shiv Sagar) Estate, F-Block, Dr. Annie Besant Road,) Worli, Mumbai 400018 Maharashtra India))... Resulting Company/Applicant Company 2

(Hereinafter collectively referred to as 'the Applicant Companies')

Order delivered on: 09.12.2019

CORAM: Hon'ble M. K. Shrawat, Member (J)

For the Applicants: Adv Gauraj Shah A/W CS Siddharth Doshi i/b. Aabid & Co. Practicing Company Secretaries

Per: M. K. Shrawat, Member (J)

ORDER

UPON the application of the Applicant Companies above named by a Company Notice of Admission **AND UPON HEARING** Advocate for the Applicant Companies **AND UPON READING** the Application along with the Notice of Admission dated 27th July, 2019, of Mr. Ajay Shankarlal Mittal, Director in the Applicant Companies, in support of Notice of Admission along with Application and Exhibits therein referred to, **IT IS ORDERED THAT:**

- The Counsel for the Applicants states that the present Scheme is a Scheme of Demerger by Demerger of "Domestic Business Undertaking" of Arshiya Limited into Arshiya Rail Infrastructure Limited and their respective shareholders under the provisions of Sections 230 to 232 read with section 66 and section 52 and other applicable provisions of the Companies Act, 2013.
- 2. The Counsel for the Applicants Companies further submits that the Applicant Company 1 is engaged in the business of developing, operating and maintaining special economic zones (SEZs)/free trade and warehouse zones (FTWZs), inland container depots (ICDs), industrial parks, logistic parks, warehouses, infrastructure or infrastructure projects and the Applicant Company 2 is engaged in the business of setting up of Rail Infrastructure/Network within India and abroad including operations/movement of Container/Goods Trains using Indian Railway Network and also to acquire, procure, obtain on lease/licence or otherwise Container Trains, Rakes, Wagons, Boggies and Create, Develop or Obtain on lease/licence basis Railway Sidings, Rail Yards, Warehouses required for the business of the Company.

3. BACKGROUND AND RATIONALE FOR THE SCHEME

Arshiya, a flagship company of Arshiya Group, is engaged in the business of developing Free Trade Warehousing Zones ('**FTWZ**') and Domestic Warehousing Areas ('**DWA**') as mentioned below to improve logistics infrastructure in India.

FTWZ Business - FTWZ at Panvel offers over 800,000 Sq. ft. of warehousing space with best in class infrastructure which is suitable for clients across industries. The facility is well connected to the National and State Highways and situated only 24 kms from the country's busiest container port and also close to the proposed International Airport in Navi Mumbai.

Domestic Business - Domestic Business comprises of Domestic Warehousing activities being carried out by Arshiya at land admeasuring 43.42 acres of land situated at Khurja (UP) and investments held in the Resulting Company pertaining to Private Freight Terminal ('**PFT**'), Rail Transportation Services ('**RAIL**') businesses being carried in the Resulting Company and Inland Container Depot ('**ICD**'), DWA business being carried in AIDHL.

Brief description of the above mentioned business :

PFT: ARIL currently under a PFT license operates Indian Railways traffic business for various customers catering to bulk goods movement and bagged cargo at sidings specifically earmarked for the same.

RAIL: ARIL holds category-I license to run container trains pan-India and is one of the largest Private Container Train Operator (PCTO's) with a rail fleet of 18 rakes and 3,200 owned containers equipped to handle a wide-range of cargo, with a pan India presence. ARIL's service is completely equipped to provide efficient movement of cargo between terminals, hubs and warehouses.

ARIL's infrastructure consists of rail siding, rail terminal equipped with three loop / blast rail lines and three non-ballast rail lines with a capacity to handle up to 20 rakes per day.

ICD: The Khurja ICD is co-located with a state-of-the-art Rail Terminal and FTWZ. ICD-Khurja is the only private ICD in the country to have exclusive connectivity with 6lane private rail siding offering regular and prompt rail connectivity through owned rakes to all the major gateway ports that service the northern region of India. ICD Khurja is located strategically with multiple road approaches from the major 4/6 lane highways providing a congestion-free movement of cargo and containers.

The Group intends to reorganize its corporate structure and integrate / consolidate its operations by housing the following businesses into two different entities / separate verticals:

FTWZ business in Arshiya Domestic business (includes DWA, ICD, Rail and PFT business) in ARIL

Though the businesses of all the Group companies complement each other, the Group believes that in the industry, there are other companies who are doing consolidated business of offering rail infrastructure, cargo/container handling services, providing ICD and Domestic warehousing services. The demand in the market for the entities providing consolidated services is higher than the entities providing individual services. In order to earn higher revenue and to cater to the needs of the market, the management intends to consolidate the rail infrastructure, transport handling business, DWA and ICD business. Hence, the management envisages the transfer of Domestic business of Arshiya into ARIL.

Accordingly, the proposed demerger of the Domestic Business (as defined hereinafter) envisaged in this Scheme ('**the Demerger**') would be in order to integrate / consolidate its DWA, RAIL, PFT & ICD business into ARIL, which would enable Arshiya to focus solely on FTWZ Business. This would provide more flexibility in terms of creating business synergies in the Resulting Company i.e. ARIL, enable cost savings, rationalizing capital requirements and optimizing utilization of valuable resources which will enhance management focus on the different businesses being housed under separate entities, thereby leading to higher operational efficiency. Further, the Scheme would be in the best interests of the shareholders, creditors and employees of Arshiya and ARIL, respectively as it would result in enhanced value for the shareholders and allow focused strategy on expansion/ operation of both the FTWZ and the Domestic Business independently. Pursuant to this Scheme all the shareholders of Arshiya will get shares in ARIL and there would be no change in the economic interest for any of the shareholders of Arshiya pre and post implementation of this Scheme.

- 4. The Counsel for the Applicant Companies submits that the Board of Directors of the Applicant Companies at their respective meetings held on 24th May, 2018 have considered and approved the proposed Composite Scheme of Arrangement for the reduction of Equity Share Capital of the Resulting Company and transfer of the "Domestic Business Undertaking" of the Demerged Company to the Resulting Company subject to necessary statutory approvals, in order to benefit the stakeholders of the Demerged Company and Resulting Company. The Board Resolutions approving the Scheme for the Applicant Company 1 and Applicant Company 2 are annexed respectively as Annexure I and II to the Company Scheme Application.
- 5. Upon the Scheme becoming effective and in consideration of the demerger including the transfer and vesting of the Demerged Undertaking in the Resulting Company, the Resulting Company shall, without any further application or deed, for every 2 (two) fully paid-up equity shares of the Demerged Company, issue and allot to each member of the Demerged Company whose name appears in the register of members of the Demerged Company as on the Record Date or to his/her heirs, executors, administrators or the successors-in-title, as the case may be, subject to the provisions of Clause 7.4 below, 1 (one) fully paid-up equity share of Rs 2 each, of the Resulting Company ('New Equity Shares').
- 6. That the meeting of the Equity Shareholders of the Applicant Company 1 be convened and held at Nehru Centre, Doctor Annie Besant Road, Worli, Mumbai, Maharashtra 400018 at 11.00 am on Monday, 13th January, 2020 for the purpose of considering and,

if thought fit, approving with or without modification(s) the proposed arrangement embodied in the Scheme of Arrangement of "Domestic Business Undertaking" of Arshiya Limited into Arshiya Rail Infrastructure Limited and their respective shareholders.

- 7. That the meeting of the Equity Shareholders of the Applicant Company 2 be convened and held at 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai 400018 at 4.00 pm on Monday, 13th January, 2020 for the purpose of considering and, if thought fit, approving with or without modification(s) the proposed arrangement embodied in the Scheme of Arrangement of "Domestic Business Undertaking" of Arshiya Limited into Arshiya Rail Infrastructure Limited and their respective shareholders.
- 8. The Counsel for the Applicant Companies submits that Applicant Companies have both the Creditors i.e. Secured Creditors and Unsecured Creditors. The Counsel for the Applicant Companies submits that as on 30th September, 2019, the total number of Secured Creditors of Demerged Company and Resulting Company are 17 (Seventeen only) amounting to Rs. 9,98,83,92,045 (Rupees Nine Hundred Ninety Eight Crore Eighty Three Lakhs Ninety Two Thousand Forty Five Only) and 15 (Fifteen only) amounting to Rs. 9,96,73,53,168 (Rupees Nine Hundred Ninety Six Crore Seventy Three Lakhs Fifty Three Thousand One Hundred Sixty Eight Only) respectively and total number of Unsecured Creditors of Demerged Company and Resulting Company are 128 (One Hundred and twenty eight only) amounting to Rs. 1,90,22,29,065/- (Rupees One Hundred Ninety Crore Twenty Two Lakhs Twenty Nine Thousand and Sixty Five Only) respectively. List where of is annexed as Annexure –A, Annexure-B, Annexure-C and Annexure-D.
- 9. That the meeting of the Secured Creditors of the Applicant Company 1 be convened and held at 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai 400018 at 11:00 am on Tuesday, 14th January, 2020 for the purpose of considering and, if thought fit, approving with or without modification(s) the proposed arrangement embodied in the Scheme of Arrangement of "Domestic Business"

Undertaking" of **Arshiya Limited** into **Arshiya Rail Infrastructure Limited** and their respective shareholders.

- 10. That the meeting of the Secured Creditors of the Applicant Company 2 be convened and held 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai 400018 at 1:00 pm on Tuesday, 14th January, 2020 for the purpose of considering and, if thought fit, approving with or without modification(s) the proposed arrangement embodied in the Scheme of Arrangement of "Domestic Business Undertaking" of Arshiya Limited into Arshiya Rail Infrastructure Limited and their respective shareholders.
- 11. That the meeting of the Unsecured Secured Creditors of the Applicant Company 1 be convened and held at 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai 400018 at 2:00 pm on Tuesday, 14th January, 2020 for the purpose of considering and, if thought fit, approving with or without modification(s) the proposed arrangement embodied in the Scheme of Arrangement of "Domestic Business Undertaking" of Arshiya Limited into Arshiya Rail Infrastructure Limited and their respective shareholders.
- 12. That the meeting of the Unsecured Secured Creditors of the Applicant Company 2 be convened and held 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai 400018 at 4:00 pm on Tuesday, 14th January, 2020 for the purpose of considering and, if thought fit, approving with or without modification(s) the proposed arrangement embodied in the Scheme of Arrangement of "Domestic Business Undertaking" of Arshiya Limited into Arshiya Rail Infrastructure Limited and their respective shareholders.
- 13. The counsel for the Applicant Companies submits that the Resulting Company being an Unlisted Public Limited Company, the shares of the same are not listed in any stock exchange. Further the equity shares of the Demerged Company are listed with BSE Limited and NSE Limited with around 9323 number of equity shareholders as on date of the Application. Further as per the requirement of SEBI Circular No. CFD/DIL3/CIR/2017/21 dated 10th March, 2017, the Demerged Company received No Objection Certificate for draft Scheme of Arrangement and Amalgamation from BSE

Limited on 12th Day of July, 2019 and from NSE Limited on 15th Day of July, 2019 the same has been disseminated by BSE Limited and NSE Limited on its website and the acknowledgement is annexed at Exhibit- 17 and Exhibit-18 to the Application.

- 14. That at least 30 days clear notice before the said Meetings of the Equity Shareholders, Secured Creditors and Unsecured Creditors of the Applicant Companies to be held as aforesaid, a notice convening the said Meeting at the place, day, date and time as aforesaid, together with a copy of the Scheme, a copy of statement disclosing all material facts as required under Section 230(3) of the Companies Act 2013 read with Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rule, 2016 and the prescribed Form of Proxy, shall be sent by Registered Post or by courier or by speed post or by hand delivery to each of the Equity Shareholders of the Applicant Companies as aforesaid at their respective registered or last known addresses or by email to the registered e-mail address of the Equity Shareholders, Secured Creditors and Unsecured Creditors as per the records of the Applicant Companies.
- **15.** That at least 30 clear days before the Meetings of the Equity Shareholders, Secured Creditors and Unsecured Creditors of the Applicant Companies to be held as aforesaid, a joint notice convening the said meetings indicating the place, date and time aforesaid and stating that copies of the Scheme and the statement required to be furnished pursuant to Section 230 (3) of the Companies Act, 2013 read with Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rule, 2016 and that the form of Proxy can be obtained free of charge at the respective Registered Offices of the Applicant Companies as aforesaid, be published once each in 'Active Times' in English and `Mumbai Lakshadeep' in Marathi, both circulated in Mumbai not less than thirty(30) days before the date fixed for the meeting.
- 16. The Applicant Companies are directed to:
 - i. Issue Notice convening meeting of the Equity shareholders, Secured Creditors and Unsecured Creditors in Form No. CAA.2 as per Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016;

- ii. Issue Statement containing all the particulars as per Section 230(3) of the Companies Act, 2013;
- Issue Form of Proxy in Form No. MGT-11 as per Rule 19 of the Companies (Management and Administration) Rules, 2014; and
- iv. Advertise the Notice convening meeting in Form No. CAA.2 as per Rule 7 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016.
- 17. That the value and number of the shares of each Equity Shareholder shall be in accordance with the books/ register of the Applicant Companies where the entries in the books/ register are disputed, the chairman of the Meeting shall determine the value for the purpose of the aforesaid meeting and his decision in that behalf would be final.
- 18. Mr. Ajay Shankarlal Mittal, Director failing whom Mr. Ashishkumar Bairagra, Directors of the Applicant Companies, shall be the Chairman for each of the aforesaid meetings to be held as aforesaid of the Applicant Companies or any adjournments thereof.
- 19. The Scrutinizer for the each of the meetings or any adjournments thereof shall be Mr. Mohammed Akram, (ACS - 9411) Companies Secretary in practice having its office at R 34, 3rd Floor, Shanti Bhawan, 198 Kalbadevi Road, Mumbai – 400002.
- **20.** The Chairman appointed for the aforesaid Meetings of the Applicant Companies to issue the notices of the Meeting of the Equity Shareholders, Secured Creditors and Unsecured Creditors as the case may be, referred to above. The said Chairman shall have all powers under the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 in relation to the conduct of the meeting(s), including for deciding procedural questions that may arise or at any adjournment thereof or any other matter including an amendment to the Scheme or resolution, if any, proposed at the meeting by any person(s).
- **21.** The quorum for the aforesaid meeting of the Equity Shareholders of the Applicant Companies shall be as prescribed under Section 103 of the Companies Act, 2013.

- **22.** That the quorum of the aforesaid meetings of the secured the unsecured creditors of the Applicant 1 and 2 shall be five present in person or through proxy.
- **23.** That the voting shall be allowed on the proposed Scheme by voting in person or by proxy. The voting by proxy or authorised representative in case of body corporate be permitted, provided that a proxy in the prescribed form / authorization duly signed by the person entitled to attend and vote at the meeting, is filed with the respective Applicant Companies at its respective Registered Offices not later than, 48 hours before the aforesaid meeting as required under Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016.
- 24. That the voting shall be allowed on the proposed Scheme in case of the Equity Shareholders, Secured Creditors and Unsecured Creditors of the Applicant Companies by voting through electronic means and Postal Ballot also.
- **25.** The Chairman to file an affidavit not less than seven days before the date fixed for the holding of the meetings of the Applicant Companies and do report this Tribunal that the direction regarding the issue of notices and advertisement have been duly complied with as per Rule 12 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016.
- 26. The Chairman to report to this Tribunal, the result of the aforesaid meeting within ten days of the conclusion of the meetings of the Applicant Companies and the said report shall be verified by his Affidavit as per Rule 14 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016.
- **27.** That the Chairman of the meetings to report to this Tribunal, the results of the aforesaid meetings within thirty days of the conclusion of the meetings.

- 28. That this bench hereby directs the Applicant Companies to issue notices for equity shareholders meeting, unsecured creditors meeting and secured creditors meeting to all respective shareholders, unsecured Creditors and secured creditors as on 30th September, 2019 by Registered Post /Speed Post / courier / Airmail / email / hand-delivery as required under section 230 (3) of the Companies Act, 2013 with a direction that they may submit their representations, if any, to the Tribunal and copy of such representations shall simultaneously be served upon the Applicant Companies respectively.
- 29. The Applicant Companies are directed to serve the notice upon the Regional Director, Western Region, Ministry of Corporate Affairs, Mumbai Maharashtra, pursuant to Section 230(5) of the Companies Act, 2013 as per Rule 8 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016. If no response is received by the Tribunal from Regional Director within 30 days of the date of receipt of the notice it will be presumed that Regional Director and/ or Central Government has no objection to the proposed Scheme as per Rule 8 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016.
- **30.** The Applicant Companies are directed to serve the notice upon the concerned Registrar of Companies, pursuant to Section 230(5) of the Companies Act, 2013 as per Rule 8 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016. If no response is received by the Tribunal from the Registrar of Companies within 30 days of the date of receipt of the notice it will be presumed that Registrar of Companies has no objection to the proposed Scheme as per Rule 8 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016.
- **31.** The Applicant Companies are directed to serve notices along with copy of scheme upon the concerned Income Tax Authority enlisted below, within whose jurisdiction respective assessments of the Applicant Companies are made pursuant to Section 230(5) of the Companies Act, 2013 as per Rule 8 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016:-

Name of the Company	PAN No.	Address
ARSHIYA LIMITED	AAACI2679A	DCIT Central Circle 6(4)
ARSHIYA RAIL	AAACI2679A	DCIT Central Circle 6(4)
INFRASTRUCTURE		
LIMITED		

If no response is received by the Tribunal from the Income Tax Authorities within 30 days of the date of receipt of the notice it will be presumed that Income Tax Authorities has no objection to the proposed Scheme as per Rule 8 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016.

32. The Demerged Company/ Applicant Company 1 to serve the notice upon the Securities and Exchange Board of India (SEBI) and BSE Limited and NSE Limited and pursuant to Section 230(5) of the Companies Act, 2013 as per Rule 8 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016. If no response is received by the Tribunal from the Securities and Exchange Board of India (SEBI) and BSE Limited and NSE Limited within 30 days of the date of receipt of the notice it will be presumed that the Securities and Exchange Board of India, BSE Limited and the NSE Limited has no objection to the proposed Scheme as per Rule 8 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016.

33. The Appointed date is 1st April, 2019

M.K. Shrawat, Member (J) Dated: 09.12.2019

Sd/-

INFORMATION DOCUMENT OF ARSHIYA RAIL INFRASTRUCTURE LIMITED IN TERMS OF SEBI CIRCULAR NO. CFD/DIL3/ CIR/2017/21 DATED MARCH 10, 2017 IN RELATION TO THE SCHEME OF ARRANGEMENT BETWEEN ARSHIYA LIMITED (HERE IN AFTER REFERRED TO AS THE "DEMERGED COMPANY" OR "AL") AND ARSHIYA RAIL INFRASTRUCTURE LIMITED (HEREIN AFTER REFERRED TO AS THE "RESULTING COMPANY" OR "AL") AND ARSHIYA RAIL INFRASTRUCTURE LIMITED (HEREIN AND CREDITORS ("SCHEME")



ARSHIYA RAIL INFRASTRUCTURE LIMITED

CIN: U93000MH2008PLC180907 Reg Off: 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018 T: +91 22 42305500/02 F: +91 22 4230 5555 E-mail: info@arshiyalimited.com /Website: www.arshiyalimited.com

PROMOTERS OF THE COMPANY

ARSHIYA LIMITED (Refer Part I of this Document i.e Promoters of the Company)

SCHEME DETAILS, LISTING AND PROCEDURE

The Board of Directors of the Company is proposing the Scheme of Arrangement between Arshiya Limited (here in after referred to as the "**Demerged Company**" or "**AL**") and Arshiya Rail Infrastructure Limited (here in after referred to as the "**Resulting Company**" or "**ARIL**" or "Company") and their Respective Shareholders and Creditors ("**Scheme**") under sections 230 to 232 read with Section 66 And Section 52 and other applicable Provisions of the Companies Act, 2013. On the Scheme becoming effective the shareholders of AL would be allotted equity shares in the Company and would become shareholders of the Company as on the determined record date. The shares of AL are listed on BSE Limited ("**BSE**"), National Stock Exchange of India Limited ("**NSE**"). The equity shares so issued pursuant to the Scheme to the shareholders of AL would be listed on stock exchanges in terms of Rule 19 of the Securities Contract (Regulation) Rules, 1957. Therefore this Information Document be read accordingly. The shareholders of ARIL may also download from the website of AL this Information Document, the Scheme as approved by the Board of Directors of AL and ARIL, the valuation report dated May 24, 2018 issued by ZADN & Associates, Chartered Accountants, the fairness opinion dated May 24, 2018 issued by Chartered Capital and Investment Limited and other relevant documents related to the Scheme.

Post sanction of the Scheme, the equity shares of the Company are proposed to be listed on BSE and NSE. For the purposes of this listing, the Designated Stock Exchange is BSE.

ELIGIBILITY CRITERIA

There being no initial public offering or rights issue, the eligibility criteria of SEBI (ICDR) Regulations, 2018, as amended ('**ICDR Regulations**') does not become applicable.

However, SEBI vide its Circular No. CFD/DIL3/CIR/2017/21 dated March 10, 2017 stated that the listed entity shall include the applicable information pertaining to unlisted entities involved in the scheme in the format prescribed for abridged prospectus as provided in Part E of Schedule VI of Securities Exchange Board of India (Issue of Capital and Disclosure Requirements) Regulations, 2018, as amended, and the same has to be annexed with the Notice or explanatory statement or proposal accompanying resolution to be sent to and passed by the shareholders while seeking approval of the scheme.

Accordingly in compliance with Regulation 37 of SEBI (Listing Obligations & Disclosure Requirements) Regulations 2015 read with SEBI Circular No. CFD/DIL3/CIR/2017/21 dated March 10, 2017 the Company has submitted the relevant information, as and where applicable to an Unlisted Company, in line with the format for Abridged Prospectus specified in Part E of Schedule VI of Securities Exchange Board of India (Issue of Capital and Disclosure Requirements) Regulations, 2018, as amended.

GENERAL RISKS

Investment in equity and equity related securities involve a degree of risk and investors should not invest any funds unless they can afford to take the risk of losing their investment. **Specific attention of the shareholders is invited to the section titled "Internal Risk Factors" on page 8 of this Information Document.**

PRICE INFORMATION OF MERCHANT BANKER

Not Applicable (Since there is no invitation to Public for Subscription by way of this Document)

COMPANY'S ABSOLUTE RESPONSIBILTY

The Company, having made all reasonable inquiries, accepts responsibility for and confirms that this Information Document contains all information with regard to the Company, which is material, that the information contained in this Information Document is true and correct in all material aspects and is not misleading in any material respect, that the opinions and intentions expressed herein are honestly held and that there are no other facts, the omission of which makes this Information Document as a whole or any of such information or the expression of any such opinions or intentions misleading in any material respect.

MERCHANT BANKER	STATUTORY AUDITORS	COMPANY SECRETARY
Chartered Capital and Investment Limited	Chaturvedi & Shah LLP	Ms. Avani Lakhani
Address :418-C, "215 ATRIUM", Andheri	Chartered Accountants	Address: 302, Ceejay House, Level-3, Shiv
Kurla Road, Andheri (East), Mumbai – 400	Address: 714-715, Tulsiani	Sagar Estate, F-Block, Dr. Annie Besant
093	Chambers, 212, Nariman Point,	Road, Worli, Mumbai - 400018
	Mumbai-400021	

	TABLE OF CONTENTS		
Sr.	Sr. Particulars Page No.		
No.			
1.	Promoters of the Company	4	
2.	Business Model/Business Overview and Strategy	4	
3.	3. Board of Directors of the Company 5		
4.	4. Objects of the Scheme 6		
5.	Audited Financial Statements 8		
6.	6. Internal Risk Factors 8		
7.	7. Summary of Outstanding Litigations, Claims and Regulatory Action 9		
8.	8. Other Regulatory And Statutory Disclosures 10		
9.	Declaration	11	

I. PROMOTER OF THE COMPANY

ARSHIYA LIMITED

Name of the Company	:	Arshiya Limited
Date of Incorporation	:	July 03, 1981
CIN	:	L93000MH1981PLC024747
Registered Office	:	302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018
Nature of Business	: The Company is engaged in the business of developing, operating and mainta special economic zones (SEZs)/free trade and warehouse zones (FTWZs), in container depots (ICDs), industrial parks, logistic parks, warehouses, infrastructur infrastructure projects.	
Names of Stock Exchanges where Equity Shares of the Company are Listed		 > BSE Limited > National Stock Exchange of India Limited

II. BUSINESS MODEL / BUSINESS OVERVIEW AND STRATEGY

Business Overview

• Arshiya Rail Infrastructure Limited is a Public Limited Company originally incorporated under the provisions of the Companies Act, 1956 on April 07, 2008 as Public Company Limited by Shares. It is a wholly owned subsidiary of Arshiya Limited. The registered office of Arshiya Rail Infrastructure Limited is located at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai- 400018, Maharashtra.

- The main objects of the Company as set out in its Memorandum of Association inter alia include:
 - 1. To carry on the business of setting up of Rail Infrastructure/Network within India and abroad including operations/ movement of Container/Goods Trains using Indian Railway Network and also to acquire, procure, obtain on lease/ licence or otherwise Container Trains, Rakes, Wagons, Boggies and Create, Develop or Obtain on lease/licence basis Railway Sidings, Rail Yards, Warehouses required for the business of the Company.
 - 2. To carry on the business predominantly in the Northern Region of India, of container freight stations which, inter-alia includes setting up of bonded warehousing infrastructure and services, facilities for customs examination, EDI, empty container yard for storage of shipping containers, repairs and refurbishment of containers, truck, cargo and material handling equipments, transportation, non-bonded warehousing, IT & ITES infrastructure and services, warehousing, cold storage and other cargo related activities.
 - 3. To carry on the business of Transport & Handling of Containers/ Goods/ Network within India and abroad including operations/movement of Container/Cargo/Goods Trains using India Railway Network and also to acquire, procure, obtain on lease/licence or otherwise Container Trains, Rakes, Wagons, Boggies and Create, Develop or obtain on lease/licence basis Railway Sidings, Rail Yards, Warehouses required for the business of the Company.
- The Company is engaged in the business of setting up of Rail Infrastructure/Network within India and abroad including operations/movement of Container/Goods Trains using Indian Railway Network and also to acquire, procure, obtain on lease/licence or otherwise Container Trains, Rakes, Wagons, Boggies and Create, Develop or Obtain on lease/licence basis Railway Sidings, Rail Yards, Warehouses required for the business of the Company.

Business Strategy

Sharpening focus on railway business combined with Inland Container Depot (ICD) and Domestic Warehousing Business to provide excellent logistic facility. The company's ICD at Khurja is co-located with a state-of-the-art Rail Terminal. ICD-Khurja is the only private ICD in the country to have exclusive connectivity with 6-lane private rail siding offering regular and prompt rail connectivity through owned rakes to all the major gateway ports that service the northern region of India. ICD Khurja is located strategically with multiple road approaches from the major 4/6 lane highways providing a congestion-free movement of cargo and containers.

Sr. No.	Name of the Director	Designation (Independent / Whole time / Executive / Nominee)	Experience including current/past position held in other firms
1.	Mr. Ajay Shankarlal Mittal	Director	Mr. Mittal is the key driving force behind Arshiya's growth as an integrated supply chain management and logistics infrastructure solutions company. With over three decades of experience, Mr. Mittal has successfully scaled Arshiya Limited by developing Free Trade Warehousing Zones (FTWZs), Rail & Rail Infrastructure, Industrial & Domestic Hub, and Transport & Handling to its unified business portfolio.
2.	Mr. Navnit Jugal Kishore Choudhary	Director	Mr. Navnit Choudhary, a Chartered Accountant, looks after the Commercial and Taxation aspects of the company. He is associated with the Arshiya group since last 10 years. During his 20 years of career, he was associated with many organisations like KK Birla group. RPG Group, Future Group.
3.	Mr. Ashishkumar Bairagra	Independent Director	Mr. Ashishkumar Bairagra has extensive experience in handling internal audits, statutory audits, management audits, tax advisory and business advisory assignments. He is a Partner of M. L. Bhuwania & Co LLP, Chartered Accountants, which is an independent member of Geneva Group International (GGI).
4.	Mr. Rishabh Pankaj Shah	Independent Director	Mr. Rishabh Shah is a practicing legal counsel and a legal consultant who advises on several areas of Civil Law. He has over 20 years of experience representing major corporations as legal counsel.

III. BOARD OF DIRECTORS OF THE COMPANY

IV. OBJECTS OF THE SCHEME

(A) General Objects of the Scheme

The Board of Directors of the Company is proposing the Scheme of Arrangement between Arshiya Limited (here in after referred to as the "**Demerged Company**" or "**AL**") and Arshiya Rail Infrastructure Limited (here in after referred to as the "**Resulting Company**" or "**ARIL**" or "**Company**") and their Respective Shareholders and Creditors ("**Scheme**") under sections 230 to 232 read with Section 66 And Section 52 and other applicable Provisions of the Companies Act, 2013.

The proposed demerger of the Domestic Business Undertaking of Arshiya Limited into Arshiya Rail Infrastructure Limited as envisaged in the Scheme would be:

- To integrate / consolidate its DWA, RAIL, PFT & ICD business into ARIL, which would enable Arshiya to focus solely on FTWZ Business ;
- To provide more flexibility in terms of creating business synergies in the Resulting Company i.e. ARIL, enable cost savings, rationalizing capital requirements and optimizing utilization of valuable resources which will enhance management focus on the different businesses being housed under separate entities, thereby leading to higher operational efficiency;
- To enhance value for the shareholders and allow focused strategy on expansion/ operation of both the FTWZ and the Domestic Business independently;
- To achieve and fulfill their objectives more efficiently and economically and the same is also in the interest of all the stakeholders.

On the Scheme becoming effective the shareholders of AL would be allotted equity shares in the Company and would become shareholders of the Company as on the determined record date. The shares of AL are listed on BSE Limited ("BSE"), National Stock Exchange of India Limited ("NSE"). The equity shares so issued pursuant to the Scheme to the shareholders of AL would be listed on stock exchanges in terms of Rule 19 of the Securities Contract (Regulation) Rules, 1957. Post sanction of the Scheme, the equity shares of the Company are proposed to be listed on BSE and NSE.

- (B) Means of financing: Not Applicable
- (C) Schedule of Deployment of Issue Proceeds: Not Applicable
- (D) Name of Monitoring Agency: Not Applicable
- (E) Terms of Issuance of convertible security: Not Applicable
- (F) Shareholding Pattern as on December 10, 2019

Sr. no.	Particulars	Pre – Issue Number of equity shares	% Holding of Pre Issue
1.	Promoter & Promoter Group*	4,23,84,417	100.00
2.	Public	0	0.00
	Total	4,23,84,417	100.00

*Note: Promoter and Promoter Group Shareholding include shares being held jointly or through nominee.

(G) Number/amount of equity shares proposed to be sold by Selling Shareholders, if any – Not Applicable

V. AUDITED FINANCIALS STATEMENTS OF THE COMPANY

					(RS. IN LUKNS)
PARTICULARS	F.Y 2018-19*	F.Y 2017-18*	F.Y 2016-17*	F.Y 2015-16	F.Y 2014-15
Total income from operations (net)	13,614.83	13,598.32	15,539.26	19,758.55	23,823.90
Net Profit / (Loss) before tax and extraordinary items	(4,159.12)	(10,745.75)	(8,275.22)	(6,819.48)	(10,328.81)
Net Profit / (Loss) after tax and extraordinary items	(4,159.12)	(10,745.75)	(9,233.34)	(8,330.30)	(10,409.61)
Equity Share Capital	4,238.44	4,238.44	4,238.44	4,088.92	4,088.92
Reserves and Surplus / Other Equity	(18,197.13)	(14,073.62)	(3,330.44)	(9,573.12)	(1,066.62)
Net worth	(13,958.69)	(9,835.18)	908.00	(5484.20)	3,022.29
Basic earnings per share (Rs.)	(9.81)	(25.35)	(22.58)	(20.37)	(25.46)
Diluted earnings per share (Rs.)	(9.81)	(25.35)	(22.58)	(20.37)	(25.46)
Return on net worth (%)	(29.80)	(109.26)	(1016.89)	(151.90)	(344.43)
Net asset value per share (Rs.)	(32.93)	(23.20)	2.14	(13.41)	7.39

*Figures are based on the statutory financial statements prepared in accordance with the Ind AS as prescribed u/s 133 of the Companies Act, 2013.

VI. INTERNAL RISK FACTORS

The below mentioned risks are the top 6 risk factors:

- 1. The Resulting Company is party to certain litigations which if determined against it could adversely affects its financial position and business operation.
- 2. The Resulting Company is required to obtain certain approvals, licenses and permits in the ordinary course of business and is required to comply with certain rules and regulations to operate the business and failure to obtain any such approvals, licenses and permits or to comply with the rules and regulations in a timely manner may adversely affect the operations.
- 3. The success of the Resulting Company is largely dependent upon the knowledge and experience of its Promoters and Key Management Personnel. Any loss of Key Management Personnel could adversely affect the business, operation and financial conditions of the Resulting Company.
- 4. The Resulting Company is exposed to foreign currency exchange rate fluctuations which may impact its cash flow and operating and financial results.
- 5. If the Resulting Company is unable to identify expansion opportunities or if the Resulting Company experiences delays or other problems in implementing such projects, the Company's growth, financial condition, cash flows and results of operations may be adversely affected.
- 6. The Resulting Company has certain contingent liabilities which if materializes may adversely affect the financial position of the Resulting Company.

VII. SUMMARY OF OUTSTANDING LITIGATIONS, CLAIMS AND REGULATORY ACTION

A. Total number of outstanding litigations against the Company and amount involved

ARIL is involved in 3 litigations against them. Total Amount involved in all the litigations, wherever quantifiable is Rs. 204.10 Crores.

(De in Labbe)

B. Brief Details of top 5 material outstanding litigation against the Company and amount involved

Sr. No	Particulars	Litigation Filed by	Current Status	Amount Involved (Rs. In Crores)
1.	Default in Repayment of loans	Corporation Bank	The next date of the matter is scheduled dated 19.12.2019 in National Company Law Tribunal, Mumbai.	81.24
2.	Non-Payment of service charges and Dishonour of Cheques for the services availed by the Company.		The next date of the matter is scheduled dated 18.12.2019 in National Company Law Tribunal, Mumbai.	5.86
3.	Default in Repayment of loans	Bank of India	The next date of the matter is scheduled dated 13.01.2020 in Debt Recovery Tribunal, Delhi.	117

C. Regulatory Action, if any - disciplinary action taken by SEBI or stock exchanges against the Promoters / Group companies in last 5 financial years including outstanding action, if any

Nil

D. Brief details of outstanding criminal proceedings against Promoters

Nil

VIII. OTHER REGULATORY AND STATUTORY DISCLOSURES

Authority for the Scheme:

The Scheme has been approved by the Board of Directors of the Company in their meeting held on May 24, 2018. The same is subject to the approval from the SEBI, Shareholders, Stock Exchanges, National Company Law Tribunal, Mumbai Branch, Regional Director, Registrar of Companies, Official Liquidator and such other regulatory authorities, as may be applicable.

• Expert opinion obtained, if any:

Valuation Report and Fairness Opinion

• Documents for Inspection:

- 1. Memorandum and Articles of Association of the Company
- 2. Audited Financial Statement for the last five Financial Years
- 3. Shareholding Pattern of the Company as on December 10, 2019
- 4. Draft Scheme of Amalgamation
- 5. Valuation Report issued by ZADN & Associates, Chartered Accountants
- 6. Fairness Opinion on Valuation Report issued by Chartered Capital and Investment Limited, SEBI registered Category Merchant Banker, having SEBI Permanent Registration No. INM000004018
- 7. Networth Certificate Pre and Post Scheme
- <u>Time and Place of Inspection of Documents:</u>

Copies of the above mentioned Documents may be inspected at the Registered Office of our Company situated at 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai - 400018 between 10.00 a.m. and 2.00 p.m. from Monday to Friday, except public holidays from the date of this Information Document until the Listing Approval.

IX. DECLARATION

We hereby declare that all relevant provisions of the Companies Act, 1956, the Companies Act, 2013 and the guidelines/ regulations issued by the Government of India or the guidelines/regulations issued by the Securities and Exchange Board of India, established under Section 3 of the Securities and Exchange Board of India Act 1992, as the case may be, have been complied with and no statement made in this disclosure document is contrary to the provisions of the Companies Act, 1956, the Companies Act, 2013, the Securities and Exchange Board of India Act, 1992, or rules made or guidelines or regulations issued there under, as the case may be. We further certify that all statements in this disclosure document are true and correct.

For Arshiya Limited

Sd/-Ajay Shankarlal Mittal Managing Director DIN: 00226355

Date: 10th December, 2019Place: Mumbai

Following additional documents are required to be submitted for Demerger cases wherein a division of a listed company is hived off into an unlisted company or where listed company is getting merged with an unlisted company:

1. Clarification as to what will be listing status of the Resulting/Transferee Company/ies.

The New Equity Shares to be issued in terms of Scheme of Arrangement shall be listed and / or admitted to trading on the BSE Limited and National Stock Exchange of India Limited where the equity shares of the Demerged Company are listed.

Also Attached as an Annexure - XIII-A

2. Details of Assets and Liabilities of the Demerged division that are being transferred.

		(In Crores)
De	merged division	FY 2017-18
Ass	sets:	
a.	Land	74.99
b.	Trade Receivable:	00.24
	Sundry Debtors	
	Total Assets	75.23
Lia	bilities:	
a.	GST Liabilities	00.04
b.	Inter Division Loans	74.99
с.	Profit & Loss retained earnings	00.20
	Total Liabilities	75.23

Also Attached as an Annexure - XIII-B

- 3. Confirmation from the Managing Director/ Company Secretary, that:
 - a) There will be no change in Share Capital of the resulting/transferee company till the listing of the equity shares of the company on BSE Limited.
 - b) The shares allotted by the resulting company pursuant to the Scheme shall remain frozen in the depositories system till listing/trading permission is given by the designated stock exchange.

Attached as an Annexure - XIII-C

- 4. Confirmation by the Managing Director/ Company Secretary of the resulting/transferee company on the letter head of resulting company that:
 - a) Equity shares issued by the company pursuant to the scheme of amalgamation/ arrangement shall be listed on the BSE Limited, subject to SEBI granting relaxation from applicability under Rule 19(2) (b) of the Securities Contract (Regulation) Rules, 1957.
 - b) The company shall comply with all the provisions contained in SEBI circular no. CFD/DIL3/CIR/2017/21 dated March 10, 2017.
 - c) The company shall also fulfill the Exchange's criteria for listing and shall also comply with Rules, Byelaws, and Regulations of the Exchange and other applicable statutory requirements.

Attached as an Annexure - XIII-D

5. Percentage of Net Worth of the company, that is being transferred in the form of demerged undertaking and percentage wise contribution of the Demerged division to the total turnover and income of the company in the last two years as per the following format:

(Rs.	in	crores)
------	----	---------

Particulars	Financial	Net worth	% to total	Turnover	% to total	Profit after	% to total
	Year					Tax	
Demerged	2017-18	75.19	6.41%	0.20	0.23%	0.20	0.38%
division	2016-17	-	0.00%	-	0.00%	-	0.00%
FTWZ	2017-18	1,097.71	93.59%	85.22	99.77%	51.79	99.62%
divisions	2016-17	(102.50)	100.00%	75.81	100.00%	(105.35)	100.00%
Total	2017-18	1,172.91	100.00%	85.42	100.00%	51.99	100.00%
	2016-17	(102.50)	100.00%	75.81	100.00%	(105.35)	100.00%

FOR ARSHIYA RAIL INFRASTRUCTURE LIMITED

Navnit Choudhary **Director** DIN: 00613576

Date: 4th February, 2019

Pre De-merger Shareholding Pattern of Resulting Company (Arshiya Rail Infrastructure Limited)

Sr. No.	Name of the Shareholders	Address	Number of Shares	% of Holding	Amount per share (In Rs.)
1	Arshiya Limited	302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. A. B. Road, Worli, Mumbai- 400 018	4,23,83,817	100.00	10
2	Mr. Ajay S Mittal (Nominee of Arshiya Limited)	Mittal Bhavan, 62-A ,Pedder Road, Mumbai-400 026	100	-	10
3	Mrs. Archana A Mittal (Nominee of Arshiya Limited)	Mittal Bhavan, 62-A ,Pedder Road, Mumbai-400 026	100	-	10
4	Arshiya Limited jointly with Mr. Navnit Choudhary	302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. A. B. Road, Worli, Mumbai- 400 018	100	-	10
5	Arshiya Limited jointly with Mr. Navin Saraf	302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. A. B. Road, Worli, Mumbai- 400 018	100	-	10
6	Arshiya Limited jointly with Mrs. Joyce Collaco	302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. A. B. Road, Worli, Mumbai- 400 018	100	-	10
7	Arshiya Limited jointly with Mrs. Mini Suresh	302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. A. B. Road, Worli, Mumbai- 400 018	100	-	10
	Total		4,23,84,417	100.00	

For ARSHIYA RAIL INFRASTRUCTURE LIMITED

Navnit Choudhary Director DIN: 00613576 Post De-merger Shareholding Pattern of Resulting Company (Arshiya Rail Infrastructure Limited)

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	Category of	No.s of Share	No. of fully	No.s of Partly	No. of shares	Total nos. shares	Share holding	Nur	nber of V ield in ea	Number of Votting Rights held in each class of		No. of Shares	Share holding,	Number of Locked in	er of d in	Number of Shares	r of ss	Number of equity
Share equity Depo (1)+(1) moof Anares Total Anares Total Anares		holders (III)	paid up equity	paid- up	unde rlying	held (VII)=	as a % of total		secu (I	rities X)		Under lying	as a % assuming	shar (XII	es)	pledged or otherwise	d or ⁄ise	shares held
			Share	equity	Depo	(V)+(VI)	no. of	Ž	o of Votin	ß	Total	Out stating	full			encumbered	ered	in
			neld (IV)		sitory Receipts	(IV)+	shares (calculated	5	Rights	Ē	as a % of	conver tible	conversion	No.	Asa %	(XIII) No.) Asa %	dema terialized
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$					UN		as ner	LIASS	Llass	lotal	-ta	securities	convertible	(a)	of	(a)	of	form
1371 x y x y x y x y $(Y11)$ $(Y11)$ $(Y11)$ Asa % of Asa Asa Asa Asa Asa	_			2			SCRR	Equity	Uthers		B+C)	finchiding	securities	6	total	C	tota	(XIV)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$							1957)	x	y		6	Warrants)	as a %		charee		charac	
Asa % of (A+B+C2) Asa % of (A+B+C2)												(cruminant)	of diluted		held		held	
							As a % of					, ,	share		(q)		(q)	
							(A+B+C2)						capital)					
(3458113) (0) (3458113) (21.58) (3458113) (21.58) (34.8412) <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>=(IX)</td><td></td><td></td><td></td><td></td><td></td></td<>													=(IX)					
(3458113 0 (3458113 51.58 (53458113 51.58 (53458113 51.58 (3478+72) (As a%of As a %%of As a %%of <													(X)+(IIA)					
2 63458113 0 63458113 51 63458113 51.58 63458113 51.58 1000000 7 9590 59564220 0 59564220 48.42 59564220 48.42 1218187 1 0 9590 59564220 0 59564220 48.42 59564220 48.42 1218187 1 1 0 0 0 0 0 0 0 95964220 48.42 59564220 48.42 1218187 1													As a % of					
59564220 0 59564220 48.42 59564220 48.42 1218187 1218187 0 0 0 59564220 48.42 59564220 48.42 1218187 1218187 0 0 0 0 0 0 0 0.00 48.42 1218187 0 0 0 0 0 0 0 0 0.00 48.42 1218187 0 0 0 0 0 0 0 0 0.00 48.42 1218187 0 0 0 0 0 0 0 0 0.00 0.00 0.00 0	s s		63458113	0	0	63458113	51.58	63458113	0	63458113	51.58	00.0	51.58	1000000	15.76	63458113	100.00	63458113
5956420 0 59564220 48.42 59564220 48.42 0.00 48.42 1218187 0 0 0 0 59564220 48.42 59564220 48.42 1218187 1 0 0 0 0 0 0 0 48.42 1218187 1 0																		
5956420 0 59564220 4842 59564220 4842 1218187 1218187 0 0 0 0 95664220 4842 59564220 4842 1218187 0 0 0 0 0 0 0 4842 1218187 0																		
$ \begin{bmatrix} 1 \\ 12 \\ 30 \\ 20 \\ 30 \\ 30 \\ 30 \\ 30 \\ 30 \\ 3$		9590	59564220	0	0	59564220	48.42	59564220	0	59564220	48.42	0.00	48.42	1218187	2.05	NA	NA	59389483
123022333 0 0 0 0 0 0 0 0 0 0 0 0 1		0	0	0	0	0	0	0	0	0		00.00	0.00	0		NA	NA	
$ \begin{bmatrix} 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0$																		
0 0																		
0 0		0	0	0	0	0	NA	0	0	0		0.00	0.00	0		NA	NA	
0 0 0 0 0.00 0.00 123022333 0 123022333 100 123022333 100 11218187																		
0 0 0 0 0.00 0.00 0.00 10.00																		
123022333 0 0 123022333 100.00 100.00 100.00 102.00		0	0	0	0	0	0	0	0	0		00.00	0.00	0		NA	NA	0
123022333 0 123022333 100 123022333 0 1230200 100.00 <td></td>																		
123022333 0 0 123022333 100 123022333 0 123022333 100.00 0.00 100.00 11218187																		
123022333 0 0 123022333 100 123022333 0 123022333 0 123022333 100.00 10218187																		
		9592	123022333	0	0	123022333	100	123022333	0	123022333	100.00	0.00	100.00	11218187	9.12	9.12 63458113		51.58 122847595

Table I - SUMMARY STATEMENT HOLDING OF SPECIFIED SECURITIES
Ь
ER GROU
< GR
TEF
0M0
PRO
QN
RA
OTER AND P
Σ
E PR
F THE PRO
OF
ERN
ATTE
G P/
EHOLDING PA
HOL
LRE
SHA
ING
0MI
HS.
ENT
TATEMENT SHOWING SHAR
c n
le II - S
ble
Ta

ß
LDI
OH
ARE
SH/
LIC
UB
HEF
FT
IO N
G PATTERN OF THE PUBLIC SHARI
ATT
G P
DIN
IOL
REF
SHAREHOLDIN
NGS
MI
SHO
E I
ME
ATEMEN'
L
- III
able
Ľ

			<u> </u>	0	0	0	0	4	0	0	0	0	4		0	-		0	4	ъ
Number of equity shares held in	demat erialized form	(VIX)						661494	1822500	1387500			2496494					5242284.50	9095704	1488135
r of ss 1 or ise	ered)	As a % of total shares held (b)		NA	NA	NA	NA	NA	NA	NA	NA	NA	NA		NA	NA		NA	NA	NA
Number of Shares pledged or otherwise	encumbered (XIII)	No. (Not appli cable) (a)		NA	NA	NA	NA	NA	NA	NA	NA	NA	NA		NA	NA		NA	NA	NA
er of 1 shares 1)		As a % of total shares held (b)		0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00		0.00	0.00		0.06	0.93	0.00
Number of Locked in shares (XII)		No. (a)		0	0	0	0	0	0	0	0	0	0		0	0		73313	1144874	0
Total Share holding, as a % assuming	full conversion of convertible securities (as a percentage of diluted share capital) (XI)=(VII)+(X) As a % of (A+B+C2)		0.01	0.00	0.00	0.00	0.54	1.48	1.13	0.00	0.00	2.03		0.00	00.0		4.26	7.39	1.21
No. of Shares Underlying Outstating	convertible securities (including	Warrants) (X)		0	0	0	0	0	0	0	0	0	0		0	0		0	0	0
s	Total as a % of	70 01 Total Voting Rights		0.01	0.00	0.00	0.00	0.54	1.48	1.13	0.00	0.00	2.03		0.00	0.00		4.26	7.39	1.21
Number of Voting Rights held in each class of securities (1X)	20	Total		12500	0	0	0	661494	1822500	1387500	0	0	2496494		0	0		5242285	9095704	1488135
umber of held in ea secu	No of Voting Rights	Class Others y		0	0	0	0	0	0	0	0	0	0		0	0		0	0	0
Z	z	Class Equity x		12500	0	0	0	661494	1822500	1387500	0	0	2496494		0	0		5242285	9095704	1488135
Share holding % calculated	as per SCRR,1957) As a % of	(A+B+C2) (VIII)"		0.01	0.00	0.00	0.00	0.54	1.48	1.13	0.00	0.00	2.03		0.00	0.00		4.26	7.39	1.21
Total nos. shares held (VII)=	+(\1)+(\1)			12500	0	0	0	661494	1822500	1387500	0	0	2496494		0	0		5242285	9095704	1488135
No.s of shares underl ying	Depos itory Receipts	(IV)		0	0	0	0	0	0	0	0	0	0		0	0		0	0	0
Partly paid-up equity Share	held (V)			0	0	0	0	0	0	0	0	0	0		0	0		0	0	0
No. of fully paid up equity	Share held (IV)	, ,		12500	0	0	0	661494	1822500	1387500	0	0	2496494		0	0		5242285	9095704	1488135
No.s of Share holders (III)				-	0	0	0	1	2	1	0	0	4		0	0		9174	46	1
PAN (II)										AADCI6523Q										AAAPP9781B
Category & Name of the shareholders (1)			Institutions	Mutual Funds	Venture Capital Funds	Alternate Investment Funds	Foreign Venture Capital Investors	Foreign Portfolio Investors	Financial Institution/ Banks	IDFC BANK LIMITED	Insurance Companies	Provident Funds/ Pension Funds	SUB TOTAL (B)(1)	Central / State government(s)	Central Government/ State Government(s)/ President of India	SUB TOTAL (B)(2)	Non-institutions	Individuals - i. Individual shareholders holding nominal share capital up to Rs. 2 lakhs.	INDIVIDUAL - ii. Individual shareholders holding nominal share capital in excess of Rs. 2 lakhs.	NAISHADH JAWAHAR PALEJA
			1	(a)	(q)	(c)	(p)	(e)	(J)		(g)	(µ)		2	(a)		3	(a.1)	(a.2)	

Number of equity shares held in	demat erialized form	(XIX)	0	0	0	40017275	4867500	3914215	1250476	2008954	2079638	4967701	2188095	1717160	1676195	6429215	711476
r of s lor ise	ered	As a % of total shares held (b)	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Number of Shares pledged or otherwise	encumbered (XIII)	No. (Not appli cable) (a)	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
oer of n shares [1]		As a % of total shares held (b)	0.00	00.0	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Number of Locked in shares (XII)		No. (a)	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Total Share holding, as a % assuming	full conversion of convertible securities (as a percentage of diluted share capital) (XI)=(VII)+(X) As a % of (A+B+C2)	0.00	0.00	2.03	32.53	3.96	3.18	1.02	1.63	1.69	4.04	1.78	1.40	1.36	5.23	0.58
No. of Shares Underlying Outstating	convertible securities (including	Warrants) (X)	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
	Total as a % of	Total Voting Rights	0.00	00.0	0.00	32.53	3.96	3.18	1.02	1.63	1.69	4.04	1.78	1.40	1.36	5.23	0.58
Number of Voting Rights held in each class of securities (IX)		Total	0	0	0	40017275	4867500	3914215	1250476	2008954	2079638	4967701	2188095	1717160	1676195	6429215	711476
umber of V held in ea secu (I	No of Voting Rights	Class Others y	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Ñ	Ž	Class Equity x	0	0	0	40017275	4867500	3914215	1250476	2008954	2079638	4967701	2188095	1717160	1676195	6429215	711476
Share holding % calculated	as per SCRR,1957) As a % of	(A+B+C2) (VIII)"	0.00	00.0	0.00	32.53	3.96	3.18	1.02	1.63	1.69	4.04	1.78	1.40	1.36	5.23	0.58
Total nos. shares held (VII)=	+(v) +(vi)		0	0	0	40017275	4867500	3914215	1250476	2008954	2079638	4967701	2188095	1717160	1676195	6429215	711476
No.s of shares underl ying	Depos itory Receipts	(IA)	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Partly paid-up equity Share	held (V)		0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
No. of fully paid up equity	Share held (IV)		0	0	0	40017275	4867500	3914215	1250476	2008954	2079638	4967701	2188095	1717160	1676195	6429215	711476
No.s of Share holders (III)			0	0	0	164	1	1	1	1	1	1	1	1	1		47
PAN (II)							AABCE8997N	AAATE7452N	AAATE6688G	AAATE6841P	AAATE6490L	AAATE7680C	AAATE6570B	AAATE6849F	AAATE6341A	AAATE7922P	
Category & Name of the shareholders (1)			NBFCs registered with RBI	Employee Trusts	Overseas Depositories (holding DRs) (balancing figure)	Any Other(BODIES CORPORATE)	ECAP EQUITIES LIMITED	EDELWEISS ASSET RECONSTRUCTION COMPANY LIMITED	EDELWEISS ASSET RECONSTRUCTION COMPANY LTD	Any Other(CLEARING MEMBER)							
			q	(C)	(q)	(e)											(e)

	Category & Name of the shareholders (1)	PAN (II)	No.s of Share holders (III)	No. of fully paid up equity	Partly paid-up equity Share	No.s of shares underl ying	Total nos. shares held (VII)=	Share holding % calculated	Ź	umber of V held in ea secui	Number of Voting Rights held in each class of securities (1X)		No. of Shares Underlying Outstating	Total Share holding, as a % assuming	Number Locked in s (XII)	Number of Locked in shares (XII)	Number of Shares pledged or otherwise	r of s l or ise	Number of equity shares held in
				Share held (IV)	held (v)	Depos itory Receints	+(N) +(VI)	as per SCRR,1957) As a % of	Ż	No of Voting Rights		Total as a	convertible securities fincluding	full conversion of convertible securities (encumbered (XIII)	ered)	demat erialized form
_						(Iv)		(A+B+C2) (VIII)"	Class Equity x	Class Others y	Total	% of Total Voting Rights	Warrants) (X)	as a percentage of diluted share capital) (XI)=(VII)+(X) As a % of (A+B+C2)	No. (a)	As a % of total shares held (b)	No. (Not appli cable) (a)	As a % of total shares held (b)	(XIX)
(e)	Any Other(DIRECTORS RELATIVES)		2	3000	0	0	3000	0.00	3000	0	3000	0.00	0	5.23	0	0.00	NA	NA	3000
(e)	Any Other(EMPLOYEE)		8	8776	0	0	8776	0.01	8776	0	8776	0.01	0	0.01	0	0.00	NA	NA	8776
(e)	Any Other(FOREIGN NATIONALS)		4	150000	0	0	150000	0.12	150000	0	150000	0.12	0	0.12	0	0.00	NA	NA	25000
(e)	Any Other(NON RESIDENT INDIANS (NON REPAT))		41	327957	0	0	327957	0.27	327957	0	327957	0.27	0	0.27	0	0.00	NA	NA	327957
(e)	Any Other(NON RESIDENT INDIANS (REPAT))		97	377694	0	0	377694	0.31	377694	0	377694	0.31	0	5.23	0	0.00	NA	NA	327957
(e)	Any Other(OVERSEAS BODIES CORPORATES)		ŝ	1133560	0	0	1133560	0.92	1133560	0	1133560	0.92	0	5.23	0	0.00	NA	NA	1133560
	SUB TOTAL (B)(3)		9586	57067726	0	0	57067726	46.39	57067726	0	57067726	46.39	0	46.39	1218187	0.99	NA	NA	56892989
	Total Public Shareholding (B) = (B)(1)+(B)(2)+(B) (3)		9590	59564220	0	0	59564220	48.42	59564220	0	59564220	48.42	0	48.42	1218187	66.0	NA	NA	59389483
Detail	Details of the shareholders acting as persons in Concert including their Shareholding (No. and %): 0	ing as persons in Co.	ncert inclu	tding their Sh	areholding (No. and %)	0 ::												
"Detai	"Details of Shares which remain unclaimed may be given hear along with details such as number of shareholders, outstanding shares held in demat/unclaimed suspense account, voting rights which are frozen etc.	n unclaimed may be	given hea	r along with d	etails such.	as number (of shareholder	s, outstanding	shares held in	demat/un	claimed susp	ense account	t, voting rights	which are frozen e	itc.				
Note: (1) PA	Note: (1) PAN would not be displayed on website of Stock Exchange(s).	1 on website of Stoc	k Exchange	9(S).															
(2) TF Institu	(2) The above format needs to be disclosed along with the name of following persons: Institutions/Non Institutions holding more than 1% of trial number of shares."	be disclosed along v	with the na ‰ of total r	time of followi.	ng persons: rec"														
22011	·· ·····	TUNING TIME TIME	10 01 01	in the the the the	1100														

- NON PUBLIC SHAREHOLDER	
TERN OF THE NON PROMOTER -	
SHAREHOLDING PAT	
Table IV - STATEMENT SHOWING	

Cate Nam shar (I)	Category & Name of the shareholders (I)	PAN (II)	No. of Share holders (III)	No. of fully paid up		No.s of shares under lying	Total no. shares held (VII)=	Share holding % calculated	un Piq	Number of Votting Rights held in each class of securities (IX)	tting Rig h class of ities)		No. of Shares Underlying Outstating	Total Shareholding, as a % assuming	Num Lock sha (X	Number of Locked in shares (XII)	Number of Shares pledged or otherwise	oer of res ed or wise	Number of equity shares held
				equity Share	Share held	Deposi tory	+(^) ()	as per SCRR,1957)	Ň	No of Voting Rights		1	convertible securities	full conversion of convertible			encumbered (XIII)	bered II)	in dema
				held (IV)	S	Receipts (VI)	(IV)	As a % of (A+B+C2) (VIII)		0		% of (A+B +C)	(including Warrants) (X)	securities(as a % of diluted share	No. (a)	As a % of total	No. (Not applicable)	As a % of total shares held	terialized form (XIV)
									Class Equity x	Class Others y	Total			capital) (XI)=(VII)+(X) As a % of (A+B+C2)		shares held (b)	(a)	(Not applicable) (b)	
1 Custod Holder	Custodian/DR Holder																		
I			0	0	0	0	0	0.0000	0	0	0 0	0.0000 0		0.0000	0	0.0000	NA	NA	0
"Toti	"Total Non		0	0	0	0	0	0.0000	0	0	0 0	0.0000 0	_	0.0000	0	0.0000	NA	NA	0
Pror	Promoter-																		
Non Shar	Non Public Shareholding																		
(C)=(C) (1)+(C)	(C)=(C) (1)+(C)(2)"																		
"Note																			
(1) PAN	(1) PAN would not be displayed on website of Stock Exchange(s).	be displa	iyed on we	sbsite of 5	Stock Exch	ange(s).													

The above format needs to disclose bame of all holders holding more than 1% of total number of shares.
 W.r.t. the information pertaining to Depository Receipts, the same may be disclosed in the respective columns to the extent information available."

Table V - STATEMENT SHOWING DETAILS OF SIGNIFICANT BENEFICIAL OWNERS

Sr. No	Details of th	Sr. No Details of the significant beneficial owner (I)	eneficial owner	Details	ils of the registered owner (II)	red owner	Details of ho	lding/ exercis whe	ercise of right of the SBO in th whether direct or indirect *: (III) Whether by virtue of:	SBO in the repo direct *: e of:	Details of holding/ exercise of right of the SBO in the reporting company, whether direct or indirect *: (III) Whether by virtue of:	Date of creation / acquisition of significant beneficial interest* (IV)
	Name	PAN/ Passport No. in case of a foreign national	Nationality	Name	PAN/ Passport No. in case of a foreign national	Nationality	Shares %	Voting rights %	Rights on distributable dividend or any other distribution	Rights on Exercise of istributable control ividend or any other istribution	Exercise of significant influence	

THSPACE LET INTERNOOD ALL BURNES



CIN: L93000MH1981PLC024747 Registered Office: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-block, Dr. Annie Besant Road, Worli, Mumbai-400018

FORM NO. MGT-11 (PROXY FORM)

[Pursuant to section 105(6) of the Companies Act, 2013 and rule 19(3) of the Companies (Management and Administration) Rules, 2014]

Meeting of all the Secured Creditors (including Debenture Holders) of Arshiya Limited ("the Company")

(Convened Pursuant to The Order Dated 9th Day Of December, 2019 Passed By The Hon'ble National Company Law Tribunal, Mumbai Bench)

Time & Date: 11:00 a.m.(1100 Hrs), Tuesday, 14th January, 2020

Venue: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-block, Dr. Annie Besant Road, Worli, Mumbai-400018

Nai	me of the Secured Creditor (s):
Reg	gistered Address:
E-n	nail Id:
I/W	e, being the Secured Creditor (s) for Rs of the above-named company, hereby appoint
1.	Name:
	Email ID:
	Address:
	Signature:
2.	Name:
	Email ID:
	Address:
	Signature:Or failing him/her
3.	Name:
	Email ID:
	Address:
	Signature:

as my/our proxy to attend and vote, in case of a poll, for me/us and on my/our behalf at the said Meeting of the Company, to be held on Tuesday, the 14th January, 2020 at 11.00 a.m. (1100 Hrs) at Hall of Culture (Nehru Centre), Dr. Annie Besant Road, Worli, Mumbai- 400 018 and at any adjournment thereof in respect of resolutions are indicated below:

Item No.	Resolutions	Outstanding Amount owed	I assent to the resolution	I dissent from the resolution
SPECIA	L BUSINESS:			
1	Approval of Composite Scheme of Arrangement Between Arshiya Limited And Arshiya Rail Infrastructure Limited And Their Respective Shareholders And Creditors.			

**This is only optional Please put a $\sqrt{}$ in the appropriate column against the resolutions indicated in the Box. Alternatively you may mention the Outstanding amount owed to you in the appropriate column in respect of which you would like your proxy to vote, if you leave all the columns blank against any or all the resolutions your proxy will be entitled to vote in the manner as he/she thinks appropriate.

Signed ______ this day of _____2020

Signature of Secured Creditor: _____

Signature of Proxy holder(s): _____

- 1) This form of proxy in order to be effective should be duly completed and deposited at the Registered Office of the Company not less than 48 hours before the commencement of the said meeting A proxy need not be a Secured Creditor of the company.
- 2) In case the appointer is a body corporate the proxy form should be signed under its seal or be signed by an office or an attorney duly authorized by it and an authenticated copy of such authorization should be attached to the proxy form.



CIN: L93000MH1981PLC024747 Registered Office: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-block, Dr. Annie Besant Road, Worli, Mumbai-400018

FORM NO. MGT-11 (PROXY FORM)

[Pursuant to section 105(6) of the Companies Act, 2013 and rule 19(3) of the Companies (Management and Administration) Rules, 2014]

Meeting of all the Unsecured Creditors of Arshiya Limited ("the Company")

(Convened Pursuant to The Order Dated 9th Day Of December, 2019 Passed By The Hon'ble National Company Law Tribunal, Mumbai Bench)

Time & Date: 02:00 a.m.(1400 Hrs), Tuesday, 14th January, 2020

Venue: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-block, Dr. Annie Besant Road, Worli, Mumbai-400018

Na	ne of the Unsecured Creditor (s):
Re	istered Address:
E-r	nail Id:
I/W	e, being the Unsecured Creditor (s) for Rs of the above-named company, hereby appoint
1.	Name:
	Email ID:
	Address:
	Signature: , Or failing him/he
2.	Name:
	Email ID:
	Address:
	Signature: Or failing him/her
3.	Name:
	Email ID:
	Address:
	Signature:

as my/our proxy to attend and vote, in case of a poll, for me/us and on my/our behalf at the said Meeting of the Company, to be held on Tuesday, the 14th January, 2020 at 02.00 p.m. (1400 Hrs) at Hall of Culture (Nehru Centre), Dr. Annie Besant Road, Worli, Mumbai- 400 018 and at any adjournment thereof in respect of resolutions are indicated below:

Item No.	Resolutions	Outstanding Amount owed	I assent to the resolution	I dissent from the resolution	
SPECIAL BUSINESS:					
1	Approval of Composite Scheme of Arrangement Between Arshiya Limited And Arshiya Rail Infrastructure Limited And Their Respective Shareholders And Creditors.	I			

**This is only optional Please put a $\sqrt{}$ in the appropriate column against the resolutions indicated in the Box. Alternatively you may mention the Outstanding amount owed to you in the appropriate column in respect of which you would like your proxy to vote, if you leave all the columns blank against any or all the resolutions your proxy will be entitled to vote in the manner as he/she thinks appropriate.

Signed ______ this day of _____2020

Signature of Unsecured Creditor:

Signature of Proxy holder(s): _____

- 1) This form of proxy in order to be effective should be duly completed and deposited at the Registered Office of the Company not less than 48 hours before the commencement of the said meeting A proxy need not be a Unsecured Creditor of the company.
- 2) In case the appointer is a body corporate the proxy form should be signed under its seal or be signed by an office or an attorney duly authorized by it and an authenticated copy of such authorization should be attached to the proxy form.



CIN: L93000MH1981PLC024747

Reg. Off: 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400018 T: +91 22 42305500/02 F: +91 22 4230 5555 E-mail: info@arshivalimited.com I Website: www.arshivalimited.com

Attendance Slip for Meeting of all the Secured Creditors (including Debenture holders) of Arshiya Limited ("the Company").

(Convened Pursuant to The Order Dated 9th Day Of December, 2019 Passed By The Hon'ble National Company Law Tribunal, Mumbai Bench)

I, a Secured Creditor/ proxy / authorised representative for a such Creditor of the Company, hereby record my presence at the said Meeting of the Company on Tuesday, the 14th January, 2020 at 11:00 a.m.(1100 Hrs) at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai- 400018, Maharashtra.

Name and Address of the Secured Creditor	:		
If Secured Creditor(s) please sign here	If Proxy, please mention name and sign here Name of Proxy Signature		

- (1) Secured Creditors / Proxy holders as the case may be are requested to produce the attendance slip duly signed at the Meeting entrance.
- (2) Said creditors holding slip in physical form, are requested to advise change in their address, if any, to the Registered address of the Company as mentioned above.



CIN: L93000MH1981PLC024747

Reg. Off: 302, Ceejay House, Level-3, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400018 T: +91 22 42305500/02 F: +91 22 4230 5555 E-mail: info@arshivalimited.com I Website: www.arshivalimited.com

Attendance Slip for Meeting of all the Unsecured Creditors of Arshiya Limited ("the Company").

(Convened Pursuant to The Order Dated 9th Day Of December, 2019 Passed By The Hon'ble National Company Law Tribunal, Mumbai Bench)

I, a Unsecured Creditor/ proxy / authorised representative for a such Creditor of the Company, hereby record my presence at the said Meeting of the Company on Tuesday, the 14th January, 2020 at 02:00 p.m.(1400 Hrs) at 302, Level 3, Ceejay House, F- Block, Shiv Sagar Estate, Dr. Annie Besant Road, Worli, Mumbai- 400018, Maharashtra.

Name and Address of the Unsecured Creditor :								
If Unsecured Creditor(s) please sign here	If Proxy, please mention name and sign here Name of Proxy	e Signature						

- (1) Unsecured Creditors / Proxy holders as the case may be are requested to produce the attendance slip duly signed at the Meeting entrance.
- (2) Said creditors holding slip in physical form, are requested to advise change in their address, if any, to the Registered address of the Company as mentioned above.

CIN: L93000MH1981PLC024747

Regd Off: - 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400 018 Tel: 022 4230 5500 Fax: 022 4230 5555 E-mail: info@arshiyalimited.com Website: www.arshiyalimited.com

Meeting of all the Secured (including Debenture holders) /Unsecured Creditors of Arshiya Limited ("the Company") (Convened Pursuant to The Order Dated 9th Day Of December, 2019 Passed By The Hon'ble National Company Law Tribunal, Mumbai Bench)

> **Time & Date for Secured Creditors:** 11:00 a.m (1100 Hrs)., Tuesday, 14th January, 2020 T**ime & Date for Unsecured Creditors:** 02:00 p.m (1400 Hrs)., Tuesday, 14th January, 2020

> Venue: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400 018.

BALLOT PAPER

[Pursuant to section 109(5) of the Companies Act, 2013 and rule 21(1) (c) of the Companies (Management and Administration) Rules, 2014]

Name of the Company: Arshiya Limited

Registered office: 302, Level-3, Ceejay House, Shiv Sagar Estate, F-Block, Dr. Annie Besant Road, Worli, Mumbai- 400 018

S1.	Particulars	Details						
No.								
1.	Name of the Secured/unsecured Creditor (In block letters)							
2.	Postal address							
3.	Class of Creditor (Please tick appropriate box)							
		Secured	Unsecured					
I hereby exercise my vote in respect of Special resolutions enumerated below by recording my assent or dissent to the said resolution in the following manner:								
Item No.	Resolutions	Outstanding Amount owed	I assent to the resolution	I dissent from the resolution				
SPECI	SPECIAL BUSINESS:							
1	Approval of Composite Scheme of Arrangement Between Arshiya Limited and Arshiya Rail Infrastructure Limited and Their Respective Shareholders And Creditors.							

Date:

Place:

(Signature of the Secured/Unsecured Creditor)

